

Diane Roy Vice President, Regulatory Affairs

Gas Regulatory Affairs Correspondence Email: gas.regulatory.affairs@fortisbc.com

Electric Regulatory Affairs Correspondence Email: <u>electricity.regulatory.affairs@fortisbc.com</u> FortisBC 16705 Fraser Highway Surrey, B.C. V4N 0E8 Tel: (604) 576-7349 Cell: (604) 908-2790 Fax: (604) 576-7074 Email: <u>diane.roy@fortisbc.com</u> www.fortisbc.com

June 9, 2017

British Columbia Utilities Commission Suite 410, 900 Howe Street Vancouver, B.C. V6Z 2N3

Attention: Mr. Patrick Wruck, Commission Secretary and Manager, Regulatory Support

Dear Mr. Wruck:

Re: FortisBC Energy Inc. (FEI)

Project No. 3698899

2016 Rate Design Application (the Application)

Response to the British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1

On December 19, 2016, FEI filed the Application referenced above. In accordance with Commission Order G-30-17 setting out the Regulatory Timetable for the review of the Application, FEI respectfully submits the attached response to BCUC IR No. 1.

If further information is required, please contact the undersigned.

Sincerely,

FORTISBC ENERGY INC.

Original signed:

Diane Roy

Attachments

cc (email only): Registered Parties



C ^{TN}	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 1

1	Tab	Table of Contents				
2	Α.	CHAI	PTER 2 – APPROVALS SOUGHT	4		
3		1.0	Reference: APPROVALS SOUGHT			
4	В.	CHAI	PTER 4 – STAKEHOLDER ENGAGEMENT	5		
5		2.0	Reference: RESIDENTIAL CUSTOMER RESEARCH SURVEY	5		
6	C.	CHAI	PTER 5 – RATE DESIGN PRINCIPLES	10		
7		3.0	Reference: RATE DESIGN PRINCIPLES	10		
8		4.0	Reference: RATE DESIGN PRINCIPLES	12		
9	D.	CHAI	PTER 5 – GOVERNMENT POLICY	15		
10		5.0	Reference: GOVERNMENT POLICY	15		
11		6.0	Reference: GOVERNMENT POLICY	19		
12	Е.	CHAI	PTER 6 – FEI COST OF SERVICE STUDY	22		
13		7.0	Reference: FEI COST OF SERVICE STUDY	22		
14		8.0	Reference: FEI COST OF SERVICE STUDY	33		
15		9.0	Reference: FEI COST OF SERVICE STUDY	35		
16		10.0	Reference: FEI COST OF SERVICE STUDY	43		
17		11.0	Reference: FEI COST OF SERVICE STUDY	46		
18		12.0	Reference: FEI COST OF SERVICE STUDY	51		
19		13.0	Reference: FEI COST OF SERVICE STUDY	53		
20		14.0	Reference: FEI COST OF SERVICE STUDY	55		
21		15.0	Reference: FEI COST OF SERVICE STUDY	61		
22	F.	CHAI	PTER 7 – RATE DESIGN FOR RESIDENTIAL CUSTOMERS	69		
23		16.0	Reference: RATE DESIGN FOR RESIDENTIAL CUSTOMERS	69		
24		17.0	Reference: RATE DESIGN FOR RESIDENTIAL CUSTOMERS	71		
25		18.0	Reference: RATE DESIGN FOR RESIDENTIAL CUSTOMERS	77		
26		19.0	Reference: RATE DESIGN FOR RESIDENTIAL CUSTOMERS	83		
27		20.0	Reference: RATE DESIGN FOR RESIDENTIAL CUSTOMERS	85		
28	G.	CHAI	PTER 8 – RATE DESIGN FOR COMMERCIAL CUSTOMERS	91		
29		21.0	Reference: RATE DESIGN FOR COMMERCIAL CUSTOMERS	91		
30		22.0	Reference: RATE DESIGN FOR COMMERCIAL CUSTOMERS	96		
31		23.0	Reference: RATE DESIGN FOR COMMERCIAL CUSTOMERS	99		
32		24.0	Reference: RATE DESIGN FOR COMMERCIAL CUSTOMERS	103		
33	Н.	CHAI	PTER 9 – RATE DESIGN FOR INDUSTRIAL CUSTOMERS	116		
34		25.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS			



 I.

	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
ISPC	2016 Rate Design Application (the Application)	June 9, 2017
IO BC	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 2
26.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	131
27.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	133
28.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	142
29.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	144
30.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	147
31.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	151
32.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	158
33.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	167
34.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	169
35.0	Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS	177
36.0	Reference: FEI COST OF SERVICE STUDY	180
CHAP	TER 11 – FEI GENERAL TERMS AND CONDITIONS AND RATE	
SCHE	DULES FOR SERVICE	184
37.0	Reference: RATE DESIGN FOR RESIDENTIAL CUSTOMERS	

17 18		39.0	Reference: FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES FOR SERVICE	189
19 20		40.0	Reference: FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES FOR SERVICE	193
21 22		41.0	Reference: FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES FOR SERVICE	195
23	J.	CHAF	PTER 12 – FEI FINAL COST OF SERVICE RESULTS AND REBALANCING	197
24		42.0	Reference: FEI FINAL COST OF SERVICE RESULTS AND REBALANCING	197
25	FO	RT NE	LSON SERVICE AREA	204
26	Κ.	CHA	PTER 13 – STAKEHOLDER ENGAGEMENT	204
27		43.0	Reference: STAKEHOLDER ENGAGEMENT	204
28 29	L.	CHAF METH	PTER 13 – FORT NELSON COST OF SERVICE ALLOCATION IODOLOGY	207
30		44.0	Reference: Fort Nelson Cost of Service Allocation Methodology	207

38.0 Reference: FEI GENERAL TERMS AND CONDITIONS AND RATE

31		45.0	Reference: FORT NELSON COST OF SERVICE ALLOCATION	
32			METHODOLOGY	211
33	М.	CHA	PTER 13 – RESIDENTIAL RATE DESIGN FOR FORT NELSON	215
34		46.0	Reference: RESIDENTIAL RATE DESIGN FOR FORT NELSON	215
35		47.0	Reference: RESIDENTIAL RATE DESIGN FOR FORT NELSON	217



		FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date:
RTIS BC [™]		Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 3
N. C	НАР	TER 13 – COMMERCIAL RATE DESIGN FOR FORT NELSON	219
4	8.0	Reference: COMMERCIAL RATE DESIGN FOR FORT NELSON	

2		48.0	Reference: COMMERCIAL RATE DESIGN FOR FORT NELSON	219
3		49.0	Reference: COMMERCIAL RATE DESIGN FOR FORT NELSON	220
4		50.0	Reference: COMMERCIAL RATE DESIGN FOR FORT NELSON	222
5	0.	CHAF	PTER 13 – INDUSTRIAL RATE DESIGN FOR FORT NELSON	224
6		51.0	Reference: INDUSTRIAL RATE DESIGN FOR FORT NELSON	224
7 8	Ρ.	CHAP REB/	PTER 13 – FORT NELSON FINAL COST OF SERVICE RESULTS AND ALANCING	229
9 10		52.0	Reference: FORT NELSON FINAL COST OF SERVICE RESULTS AND REBALANCING	229
11 12		53.0	Reference: FORT NELSON FINAL COST OF SERVICE RESULTS AND REBALANCING	232
13		TRAN	ISPORTATION SERVICE REVIEW	234
14	Q.	CHAF	PTER 10 – TRANSPORTATION SERVICE REVIEW	234
15		54.0	Reference: TRANSPORTATION SERVICE REVIEW	234
16		55.0	Reference: TRANSPORTATION SERVICE REVIEW	237
17		56.0	Reference: TRANSPORTATION SERVICE REVIEW	243
18		57.0	Reference: TRANSPORTATION SERVICE REVIEW	249
19		58.0	Reference: TRANSPORTATION SERVICE REVIEW	251
				054
20		59.0	Reference: TRANSPORTATION SERVICE REVIEW	254
20 21		59.0 60.0	Reference: TRANSPORTATION SERVICE REVIEW Reference: TRANSPORTATION SERVICE REVIEW	254 258
20 21 22		59.0 60.0 61.0	Reference: TRANSPORTATION SERVICE REVIEW Reference: TRANSPORTATION SERVICE REVIEW Reference: TRANSPORTATION SERVICE REVIEW	254 258 266



1 A. CHAPTER 2 – APPROVALS SOUGHT

2	1.0	Referen	ce: APPROVALS SOUGHT
3			Exhibit B-1, Section 2.3, pp. 2-5 to 2-6
4			Implementation
5		On page	2-5 of Exhibit B-1, FEI states:
6 7 8 9		F 2 a	El is seeking to implement its proposed rate design changes effective June 1, 018. In order to provide adequate time to prepare for the implementation of pproved changes, including billing system changes and notification to ustomers of the changes. EEL requests a Commission decision early in 2018
10 11		FEI then 2-6.	provides reasons for targeting a June 1, 2018 effective date on pages 2-5 and
12 13 14		1.1 F s	Please explain if an August 1, 2018 effective date for implementation could atisfy the reasons provided for FEI's target effective date.
15	<u>Resp</u>	onse:	
16 17	Augus could	st 1, 2018 work and	as an effective date for implementation of the proposed rate design changes satisfy the reasons provided above. As mentioned in the Application, the

implementation date is a target date and is dependent on the Commission's ability to issue a decision early in 2018. In addition to the timing of a Commission decision, the effective date to implement rate design changes may also need to consider the timing of other Commission decisions, such as quarterly gas cost filings, which may also impact customer rates. Implementation timing objectives would be to allow for clear customer communication while minimizing potential confusion from multiple changes occurring at the same time.

- 24
- 25
- 26
 27 1.2 Please state the latest date that the Commission could issue a decision on the
 28 FEI 2016 Rate Design Application that would allow FEI to implement the
 29 proposals by June 1, 2018.
- 31 **Response:**

FEI requests the Commission issue a decision by March 1, 2018 at the latest to allow adequate time for FEI to implement its rate design proposals by June 1, 2018.

34



-	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 5

1	В.	СНАР	TER 4	- STAKEHOLDER ENGAGEMENT
2	2.0	Refer	ence:	RESIDENTIAL CUSTOMER RESEARCH SURVEY
3 4				Exhibit B-1, Section 4.6.1, pp. 4-6 to 4-7; Section 7.4.5, p. 7-16; Appendix 4-5
5				FEI survey methodology and scope
6		On pa	ges 4-6	and 4-7 of Exhibit B-1, FEI states:
7 8 9		•	The s sampl Interio	urvey of FEI's customers outside Fort Nelson used a total recommended e size of 750 (250 for each of Metro Vancouver, Vancouver Island and the r). This resulted in 753 final surveys in these regions;
10 11		•	The f reside	inal data set was weighted geographically to accurately reflect FEI's ntial customer base across the province;
12		On pa	ge 7-16	of Exhibit B-1, FEI states:
13 14 15 16 17 18			FEI be desigr are all easy f and cu flat rat	elieves that its existing flat rate structure provides the best balance of rate a considerations for residential customers FEI's residential customers ready familiar with this rate structure, flat rates are simple to administer and o understand and provide more stability in terms of both utility revenues ustomers' rates. The customer research survey results also show that the e structure is preferred by the majority of residential customers
19 20 21 22		2.1	Please weigh Kelow	e provide a table/chart/map showing the locations of the geographically ted final data set for the FEI survey (i.e. percent from Vancouver, Victoria, na, etc.).
23	Resp	onse:		
24	Sentis	s contrib	uted to	the following response.

The final dataset was weighted to reflect the distribution of FEI customers by region – Lower
Mainland, Interior, and Vancouver Island. The dataset is not weighted by town/city in these
regions.

28 The table below provides the weighted distribution of the sample by region:

Region/City	Weighted Percentage (N=753)
Lower Mainland	62%
Interior	27%
Vancouver Island	11%



- 1 2
- 2.2 Please elaborate on how FEI chose its sample of FEI's customers outside Fort Nelson.
- 3 4
- 5 **Response:**
- 6 Sentis provides the following response.

7 Sentis used a third-party research panel provider to invite a random sample of B.C. residents to8 this survey. Each B.C. resident was screened for the following:

- 9 They must be aged 18 or older;
- They and no member of their immediate family or in their household were employed in
 the following sectors:
- 12 o Utility Company;
- 13 o Natural Gas Company or Gas Marketer;
- 14 o Electricity Company;
- 15 o Market Research Company;
- 16 o Newspaper, Radio, or TV Network; and
- 17 o Utility Regulatory Body;
- 18 They must receive a natural gas bill from FortisBC; and
- They must be the person in the household who is responsible for, or who shares responsibility for, making payment decisions for energy bills.
- 21
 22
 23
 24
 25
 26
 2.2.1 Please confirm, or otherwise explain, that the sample was a random sample.
 28
 29 Response:
- 30 Confirmed. It was a random sample of B.C. residents that are FEI natural gas customers.



ти	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 7

1 2	
3 4 5 6 7 8	2.3 Please confirm, or explain otherwise, that this survey was only conducted in English and could only be completed on-line (i.e. the customer did not have the option to complete a survey over the phone or by mail). <u>Response:</u>
9 10	Confirmed. The survey was conducted in English only and could be completed only online There was no option to complete the survey by phone or mail.
11 12	
13 14 15 16	2.3.1 Please discuss the advantages and disadvantages of online surveys. Response:
17	Sentis provides the following response.
18	The advantages and disadvantages of online surveys are identified below.
19	Advantages:
20	Less intrusive for respondents;
21 22	 Can usually be completed faster by the respondent than a telephone survey or paper survey;
23	Has a faster turnaround time for data collection;
24	 Allows for visuals to be shown or audio to be played, if needed, as part of the survey;
25	Costs less than telephone, mail or in-person surveying;
26 27 28	 More accurate since respondents are entering their responses directly into the system more traditional survey methods require data entry by research personnel, meaning human error can be a factor;
29 30	 Can typically receive more honest and candid responses from respondents, since they do not have to give their responses to an interviewer; and
31 32	 Easier to pre-screen respondents to ensure that only those that meet the target profile are surveyed.



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 8

1 **Disadvantages:**

- Certain populations are less likely to have internet access and to respond to online questionnaires;
- Not as suitable for very complex questions that are more qualitative in nature and may
 need more explanation; and
- A lack of a trained interviewer to clarify and probe can possibly lead to less complete
 data on open-ended questions.
- 8
- 9
- Ũ
- 10
- 11 2.4 Please explain why the FEI and Fort Nelson surveys were conducted in the 12 summer months rather than in the winter months when customers have a greater 13 interest in their bills.
- 14

15 Response:

16 The timing of the survey was dictated by when FEI was preparing its Application. FEI was

17 required to file the Application before the end of 2016. FEI was not in a position to conduct the

18 survey in the winter of 2015/16.

A number of the survey questions are related to customers' knowledge and understanding of
 the current rate structure and bill components of their natural gas bill. These areas would not be
 affected by seasonality.

The remaining questions on principles important to customers when designing a rate structure, and which rate structure is preferred, could potentially be affected by seasonality. However, if that is the case, then no one season would be an ideal time to conduct the study. One could argue that a more measured response can be collected from customers during summer months, since they are not dealing with above-average natural gas bills which could inflate or exaggerate the influences that they consider important when it comes to rate design.

- 28
- 29

30 31

- 2.5 Please explain if FEI considers the FEI customer survey to be representative of the customers across FEI's system.
- 32 33

34 **Response:**

35 Yes. FEI considers the survey to be representative of the customers across FEI's system,

36 subject to the limitations of all quantitative research. All research is subject to margins of error.



TH	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 9

1 2	For this study, at the 95 percent level of confidence, the main sample of 753 has a margin of error of +/-4 percent and the Fort Nelson sample of 65 has a margin of error or +/-12 percent.
3 4	
5 6 7 8 9	2.6 With regard to specific proposals in the Application, when does FEI take into consideration the FEI customer research survey results?Response:
10 11	FEI considered the results of the residential customer research survey, along with other rate design considerations, in conducting a full review of FEI's rate design for residential customers.
12	The survey results are supportive of FEI's proposal to maintain the current flat rate structure.
13 14 15 16 17 18 19	FEI's proposal to limit the Basic Charge increase to 5 percent is also informed by the survey results. Respondents gave similar ratings to rate stability, fairness, efficiency and government policy, indicating that they prefer a balanced approach, and that the improvements from rate design in aligning with the cost causation principle should consider bill impacts as well as government energy policies.
20 21 22 23 24	2.6.1 Please explain how much weight is given to the customer research survey results when determining rate design proposals. Response:
25 26 27 28 29 30 31 32	FEI does not apply any particular weighting to customer research survey results or any other individual rate design consideration. As explained in Section 1.2 of the Application, rate design is a complex balancing process as it frequently requires the application of multiple, and sometimes conflicting, principles and the consideration of viewpoints from various stakeholders. The residential customer research survey results indicate that residential customers prefer rate design proposals that are easy to understand (ease of understanding principle) and that strike a balance among other competing rate design principles. This is what FEI's rate design proposals strive to achieve.



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 10

1 C. CHAPTER 5 – RATE DESIGN PRINCIPLES

2	3.0	Refer	ence:	RATE DESIGN PRINCIPLES
3				Exhibit B-1, Section 7.8.1, p. 7-22;
4 5				BC Hydro 2007 Rate Design Application Phase I Order G-130-07 and Decision dated October 26, 2007, p. 69
6				Rate impact considerations
7 8 9		On pa bill im custor	age 7-22 pact to mers."	2 of Exhibit B-1, FEI states: "Any rate design proposal should consider the customers and should be implemented in a way that avoids rate shock to
10		The C	Commis	sion, on page 69 of its October 2007 Decision on BC Hydro's 2007 Rate
11		Desig	n Applic	ation Phase 1, stated:
12			With r	egard to acceptable level of bill impact, BC Hydro has endeavored to limit
13			the co	mbined annual impact of rebalancing and restructuring on any individual
14 15			custor	ner bill to no more than ten percent, exclusive of any changes arising from
16			circum	increases. This is not a fulle that is interfued to be binding in every
17			impac	ts to exceed 10 percent per annum where the absolute dollar value of the
18			increa	se is very small.
19		3.1	Please	e explain the guidelines which FEI uses to consider bill impact to customers
20			and ra	ite shock.
21				
22	Resp	onse:		
23	FEI a	grees \	with the	guidelines regarding rate design-related rate impacts expressed in the
24	quote	in the	pream	ble from the Commission Decision on BC Hydro's 2007 Rate Design

- 25 Application Phase 1.
- 26 FEI finds the definition of rate shock used by the South Dakota Supreme Court to be useful:¹
- 27 "Rate Shock" is a term used to describe "the effect on utility customers when a
- utility implements a significantly increased rate immediately or in a relatively short
 time span."

What constitutes rate shock must be assessed in the circumstances of each case. In its
 Decision on BC Hydro's 1992 Rate Design Application, the Commission stated the following
 with respect to what constitutes rate shock:

¹ US West Communications, Inc. v. AT&T Communications of the Midwest, Inc., Sprint Communications, Company, L.P., MCI Telecommunications Corporation, Telecommunications Action Group and Dakota Telecommunications Group [2000 SD 140].



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 11

1 As indicated by the evidence, whether a particular increase constitutes rate 2 shock depends on the overall rate environment and the circumstances of the 3 particular customer (T. 175-178). It is the Commission's responsibility to assess 4 these circumstances and determine when rate shock may be properly said to 5 have occurred.

6 In its Decision on BC Hydro's Residential Inclining Block Application (Order No. G-124-08), p. 7 105, the Commission similarly stated that it should not adopt a "one size fits all" approach to 8 rate shock, but "should evaluate each application on its own merits".

9

- 10 11 12 3.1.1 Does FEI consider that, in setting a maximum bill impact guideline, the 13 focus of the percentage change should be based on the total customer 14 bill, each component (commodity rate, delivery rate, fixed basic charge), 15 or a combination of charges? Please explain. 16
- 17 Response:

18 FEI believes the appropriate point of reference for the rate design bill impact guideline is the 19 total customer bill. The percentage changes in individual line items on the bill are of limited 20 value since they do not express the full bill impact experienced by customers from the change. 21 Further, some rate design changes are done in combinations, such as a shifting of cost 22 recovery between the fixed and volumetric charges. In those situations, the impact of changes 23 in individual line items are offset or partly offset by rate design changes affecting other line 24 items.

25 FEI may analyze the bill impact for individual rate design proposals, but as a guideline in setting 26 the maximum bill impact, FEI has considered the combined annual impact of rebalancing as

27 well as the individual rate design proposals.



S DO™	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
S BC.	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 12

1	4.0	Reference:	RATE DESIGN PRINCIPLES
2			Exhibit B-1, Section 5.3, p. 5-2;
3			BC Hydro Residential Inclining Block (RIB) Rate Re-Pricing
4 5			Application, Order G-45-11 with Reasons for Decision dated March 14, 2011, Appendix A, p. 5;
6			Application for Approval of Rates between BC Hydro and FortisBC
7			Inc. with regards to Rate Schedule 3808, Tariff Supplement No. 3 –
8 Q			No. 2 to Rate Schedule 3817 (BC Hydro RS 3808). Order G-60-14 and
10			Decision dated May 6, 2014, pp. 32, 35;
11 12			FortisBC Inc. 2009 Rate Design and Cost of Service Analysis, Order G-156-10 and Decision dated October 19, 2010, p. 7;
13			BC Gas 1996 Rate Design Application, Volume 1, Exhibit 2A, Tab 3,
14			p. 6;
15			FBC Application for Stepped and Stand-By Rates for Transmission
16			Voltage Customers, Order G-67-14 and Decision dated May 26, 2014,
17			рр. 15, 17–18
18			Evaluation framework
19		FEI describes	s its rate design principles on page 5-2 of Exhibit B-1.
20		On pa	age 5 of the BC Hydro RIB Rate Re-Pricing Application Reasons for
21		Decisi	ion (Order G-45-11), the Commission lists eight Bonbright Principles. These
22		princip	bles were also included on page 32 of the Commission's 2014 Decision on
23		BC Hy	ydro RS 3808 and on page / of the Commission's decision on the FortisBC
2 4 25		Princi	nle 3 speaks to efficiency and includes in brackets "consideration of social
26		issues	s including environmental and energy policy."
27		BC Gas state	d in Exhibit 2A, tab 3, page 6 of the 1996 Rate Design Application (RDA):
28		The p	purpose of the rate design review is to help determine if cost burdens are
29		prope	rly borne by each class, if rates reflect the proper economic signals, if rates
30		will pr	ovide stability both for the customer and for the utility, if the rates promote
31		simpli	city and administrative ease and allow for the recovery of the revenue
32		requir	ement.
33		The Commiss	sion stated on pages 15, 17 and 18 of the 2014 FBC Industrial stepped and
34		standby rate	decision (Order G-67-14):
35		the	Panel notes that any change in rate design naturally results in some initial
36		increa	se in rate instability. As such, the Panel does not see the need to change
37		an exi	isting rate designs unless there is a clear need to do so

BC [™]	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 13

1 The Panel considers that before making any changes to previously approved rate 2 design, the Panel should be satisfied that greater efficiencies or cost savings 3 would accrue to the benefit of ratepayers overall, or that the existing rate is now 4 outside of fairness norms from a cost causation perspective. The Panel should 5 also be satisfied before making any changes to previously approved rate design 6 that the magnitude of the changes to the affected parties are acceptable and that 7 benefits in the broad public interest would result.

- The Commission stated on page 35 of the RS 3808 Decision (Order G-60-14):
- 9 ... the Commission Panel also acknowledges that existing rates are, by
 10 necessary implication, not unjust, unreasonable, unduly discriminatory or unduly
 11 preferential if they have already been approved by the Commission.
- 12 Because the 1993 PPA was approved by the Commission as fair, the 13 Commission Panel will only evaluate fairness where there is clear evidence that 14 changes in circumstances require the previous fairness determination to be 15 revisited.
- 164.1Is FEI supportive of using the G-45-11 definition of the efficiency principle17(including the brackets) for rate evaluation purposes? If no, please explain why18not.
- 19

8

20 **Response:**

FEI is generally supportive of using the G-45-11 definition of the efficiency principle; however, FEI notes that for a natural gas distributor such as FEI, environmental and energy policies are not necessarily aligned with efficient use of the distributor's system and therefore should be considered separately as is done by FEI.

For electric utilities with generation assets, the marginal cost of new generation and other marginal costs of providing service are usually higher than the embedded costs, meaning that it is more economically efficient to conserve energy than to build new generation capacity to serve the new load. For FEI however, the marginal delivery cost of an additional unit of consumption is less than the average cost and, therefore, an increase in consumption does not reduce FEI's economic efficiency. For more information regarding FEI's marginal delivery cost, please refer to EES Consulting study provided in Appendix 4-4 of the Application.

- 32
- 33

- 4.2 Does FEI agree with the purpose of the rate design review as articulated by BC
 Gas in the 1996 RDA extract above? If no, please explain why not.
- 37



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 14

1 Response:

Yes. More specifically, the main objective of rate design is to strike a balance among all of the above mentioned considerations. The quote from the 1996 RDA is, in effect, a summary that directly or indirectly encompasses the eight Bonbright principles that have been articulated in this Application (Page 5-2) and, as noted in the preamble, other rate design applications before the Commission.

- 7
- 8
- 9

13

104.3Does FEI agree with the conditions to be satisfied before making any changes to11previously approved rate design as articulated in the G-67-14 and G-60-1412extracts above? If no, please explain why not.

14 **Response:**

15 Yes. FEI generally agrees that the conditions articulated in the extracts from Orders G-67-14 16 and G-60-14 should be considered before making rate design changes. FEI observes. 17 however, that both of the decisions cited deal with much narrower rate design issues, such as 18 one rate class or one contract. In this Application, FEI is addressing a wide scope of rate 19 design issues, including minor ones, such as housekeeping changes to its General Terms & 20 Deciding whether the conditions cited in the two quotes are met involves a Conditions. 21 considerable amount of judgment, and should take into account the magnitude of the change 22 and the applicable rate design considerations in each case.



 FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 15

1 D. **CHAPTER 5 – GOVERNMENT POLICY** 2 5.0 Reference: **GOVERNMENT POLICY** 3 Exhibit B-1, Section 5.4, p. 5-3, Section 7.5.1, p. 7-17, Section 8.3.5, 4 p. 8-14 5 **Government policy environment** 6 FEI states on page 5-3 of Exhibit B-1: 7 One of the major developments since FEI's rate design proceeding in 2001 is the 8 implementation of the provincial government's climate action and energy policies. 9 The overall thrust of these policies for FEI is twofold: (i) to promote energy 10 efficiency and conservation through demand side and tax measures to curb GHG 11 emissions; and (ii) to promote the role of natural gas in the transportation sector. 12 FEI states on page 7-17 of Exhibit B-1: 13 By Order G-141-09, the Commission approved FEI's 2010-2011 NSA. As part of 14 the 2010-2011 NSA, and in alignment with government's energy conservation 15 policies, the monthly Basic Charge was fixed at 2009 levels and all annual margin increases since 2009 have been allocated to variable volumetric charges. 16 17 On page 8-14 of Exhibit B-1, FEI states that one factor which mitigates "making 18 significant changes to the Basic Charge" is "Government energy efficiency and conservation policies discourages higher fixed charges." 19 20 5.1 Does FEI consider that the government's focus on energy conservation has 21 increased, decreased or stayed the same since the (i) 2001 and (ii) 2010-2011 22 rate design proceedings? Please explain. 23 24 Response:

FEI assumes that the question refers to the 2001 rate design and 2010-2011 revenue requirement proceedings.

27 The provincial government's focus on energy conservation and curbing GHG emissions has 28 been trending upward both between 2001 and 2009 and between 2010-2011 to today. The 29 difference is that while the majority of policies developed between 2001 and 2009 affecting 30 natural gas utilities were focused on decreasing energy use in water heating and space heating 31 applications through demand-side management, tax measures and improving energy efficiency 32 in buildings, the policies developed since 2010 have expanded the focus to cover the 33 transportation sector as well as other industrial and commercial sectors. This is explained 34 further below.



1 The evolution of government policies between 2001 and 2009:

As explained in Section 5.4.1 of the Application, the main development in energy and climate change policy in this period relates to the 2007 BC Energy Plan. This was followed by government announcement in February 2008 to introduce the BC Carbon Tax. To implement the policy items outlined in the 2007 BC Energy Plan and the carbon tax, the provincial government passed legislation in the spring of 2008, including the following:

- 7 Greenhouse Gas Reduction Targets Act,
- 8 Utilities Commission Amendment Act, 2008;
- 9 Greenhouse Gas Reduction (Renewable and Low Carbon Fuel Requirements) Act,
- 10 Greenhouse Gas Reduction (Emission Standards) Statutes Amendment Act, 2008; and
- 11 Carbon Tax Act.

The cumulative and individual impacts of these pieces of legislation on the cost of natural gas for FEI's customers have been significant. In addition, these Acts, along with other exogenous factors such as the increasing share of multi-family dwellings in new housing starts, have led to a decline in market share of natural gas in space and water heating applications and declining use per customer in the residential sector.

17 *The evolution of government policies from 2010 to the Present:*

- 18 Since 2010, government energy and climate change related policies have evolved in two ways:
- Changes in existing regulations through amendments: For example, and as explained in Section 5.4.1 of the Application, the *Utilities Commission Amendment Act, 2008* introduced amendments to the *Utilities Commission Act* (UCA) that provided authority for the *Demand-Side Measures Regulation* (enacted in November 2008). In July 2014, the provincial government modified the *Demand-Side Measures Regulation* through B.C. Reg. 141/2014. This amendment raised the low income program eligibility threshold and added a deemed list of eligible low income customers.
- 26 Introduction of new regulations or policies: This includes the 2010 Clean Energy Act 27 (CEA) and the subsequent enactment of Greenhouse Gas Reduction (Clean Energy) 28 Regulation (GGRR) in 2012. The GGRR was the first legislation which recognized the 29 role of natural gas as a cost-effective means of reducing GHG emissions in the 30 transportation sector. The GGRR has been amended several times since its initial 31 inception to expand the eligibility criteria and/or increase related programs' expenditure 32 caps. More recently, in August 2016, the GGRR was amended (B.C. Reg. 214/2016) to 33 expand the eligibility criteria for incentives and to introduce two new prescribed 34 undertakings: one for incentives to support the adoption of natural gas for remote power 35 generation; and a second for LNG storage and infrastructure to enhance the LNG



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 17

1 distribution network to serve LNG customers. (For more information regarding the CEA 2 and GGRR please refer to the Section 5.4.2 of the Application).

3 Another example of developments since the 2010-2011 revenue requirement proceeding relates 4 to the BC Climate Leadership Plan published in August of 2016 as well as recent tax measures 5 announced in government's 2017 budget to gradually phase out the PST on electrical power for 6 commercial and industrial sectors which will decrease the price competitiveness of natural gas 7 in these sectors. The BC Climate Leadership Plan in particular noted the role of electrification in 8 achieving BC's climate change policy goals, and included initiatives in several areas such as 9 transportation and upstream natural gas production and processing that are expected to result 10 in low-carbon electrification.

11 In March 2017, subsequent to the filing of the Rate Design Application, the BC government 12 made further amendments to the GGRR by Orders-in-Council Nos. 101/2017 and 161/2017. 13 OIC 161/2017 increases the amount that public utilities may spend in providing incentives and 14 building infrastructure pertaining to NGT initiatives, with a particular focus on the marine sector. 15 OIC 101/2017 adds a new section to the GGRR allowing public utilities to pursue low carbon 16 electrification initiatives for the purpose of reducing greenhouse gas emissions by replacing 17 higher emitting energy sources with electricity. OIC 101/2017 present risks for FEI since there is 18 the potential for natural gas to be the fuel displaced by some of the low carbon electrification 19 initiatives.

- 20
- 21
- 22

- 23 24 25

26

27

5.2 Does FEI consider that government's current energy conservation policies suggest that the Basic Charge should remain fixed, with no annual changes and only be updated periodically? Please explain your response.

28 **Response:**

29 The government's energy and climate policies provide high-level guidance and do not prescribe 30 whether the Basic Charge should increase or not. Nevertheless, keeping the Basic Charge fixed 31 with periodic updates in the context of rate design proceedings, and flowing general rate 32 increases to the Delivery Charge is more aligned with government policies than flowing general 33 rate increases to both the Basic Charge and Delivery Charge. The former approach increases 34 the volumetric price signals and provides customers who want to invest in demand-side 35 measures with more certainty that the potential savings will pay for the investment they have 36 made.

- 37
- 38



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 18

3

- 5.3 Please explain what FEI considers to be a "significant change" to the Basic Charge?
- 4 5 **Response:**

6 Significant is a situation-specific concept, such that it cannot be quantified with a defined 7 threshold or a formula. Significant can mean different things to different customer groups or to 8 individual customers in the same customer group. FEI uses tools such as bill impact analysis 9 and experience-based judgment, along with other rate design considerations (i.e., to establish 10 an economic crossover point between two rate schedules), to decide how much the Basic 11 Charge for a specific customer class should change (if any). For instance, in the case of the 12 residential rate class, recovery of 100 percent of fixed delivery costs with fixed charges would 13 be significant enough to discourage some customers from engaging in energy efficiency 14 measures and, therefore, as explained in Section 7.5.2 of the Application, a complete alignment 15 between fixed costs and fixed charges is not desirable from an energy conservation 16 perspective.

- 17
 18
 19
 20 5.3.1 Does FEI consider a "significant change" to the Basic Charge to be different for different customer groups?
 23 <u>Response:</u>
 24 Yes. Please refer to the response to BCUC-FEI IR 1.5.3.
- 25



TN	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
	2016 Rate Design Application (the Application)	June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 19

1	6.0	Reference:	GOVERNMENT POLICY
2			Exhibit B-1, Section 7.5.2, pp. 7-18 to 7-19;
3 4			BC Hydro RS 3808, Order G-60-14 and Decision dated May 6, 2014, p. 55;
5 6 7			FBC Application for Stepped and Stand-By Rates for Transmission Voltage Customers, Order G-67-14 and Decision dated May 26, 2014, p. 13
8			Conservation
9		FEI states on	pages 7-18 and 7-19 of Exhibit B-1:
10 11 12 13 14 15 16 17 18 19 20 21		As me was the theory charge This b scena effort of nate volume custor consu level c	entioned above, alignment with government's energy conservation policy he basis for the 2009 decision to hold the Basic Charge constant. The suggests that excessively high fixed charges (relative to volumetric es) can lead to consumption behaviours that result in excessive usage. ehaviour, sometimes described by economists as a "buffet effect", refers to rios in which customers strive to consume more than desired levels in an to justify the break-even costs of a high fixed charge. For the specific case ural gas utilities, excessively high fixed charges, and correspondingly lower etric charges, may affect customers' behaviour through decreased ner participation in energy saving activities rather than a direct increase in mption. That is, the customer may lose the incentive to achieve the desired of energy savings.
22 23 24 25 26		In ligh Charg engag fixed o efficier	t of government's energy policy considerations, any increase in the Basic e should be done in a manner that does not discourage customers' ement in energy saving initiatives. As such, a complete alignment between costs and fixed charges is not desirable from an energy conservation and ncy perspective.
27		The Commiss	ion stated on page 55 of the BC Hydro RS 3808 Decision (Order G-60-14):
28 29 30 31		Efficie consu operat British	ncy benefits can be described as promotion of: (i) efficient customer mption and investment decisions, (ii) efficient utility investment and ional decisions and (iii) innovation. The Panel also considers any effect on Columbia social issues, including environmental and energy policy.
32 33		The Commiss decision (Ord	ion stated on page 13 of the 2014 FBC Industrial stepped and standby rate er G-67-13):
34 35 36		There need f custor	fore, the Panel agrees with BCPSO that the key question in determining if a for the Stepped Rate exists is whether the Stepped Rate promotes efficient ner behaviour rather than merely results in less electricity consumption.



2

3

4

5

FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 20

6.1 Does FEI consider that rate design evaluation requires consideration of whether the proposal promotes efficient customer behaviour, rather than merely whether it results in less gas consumption or whether the rate signals long-run marginal costs at the margin? Please explain.

6 Response:

Promoting efficient use of the products and services provided by the utility is one of the goals of
rate design. This is articulated in rate design Principle 3 on page 5-2 of the Application (price
signals that encourage efficient use and discourage inefficient use).

10 The preamble to this question refers to government policy considerations in FEI's rate design 11 and, in particular, in relation to FEI's Basic Charge. Government energy and climate policies are 12 predominately concerned with curbing GHG emissions. For the building sector, this mainly 13 translates to "less natural gas consumption" in space and water heating (other than renewable 14 natural gas consumption which is encouraged) through energy efficiency and conservation measures. Government policy does not concern itself with FEI's load factor (for a natural gas 15 16 distributor, load factor is a measure of efficiency) or FEI's marginal delivery cost. Government 17 policy considerations are only one item among others, and FEI strives to strike a balance 18 among competing rate design considerations.

For instance, in Table 7-2 of the Application FEI evaluated different rate structure options based on major rate design principles including economic efficiency. This evaluation indicates that some rate structure options such as seasonal rates may provide better price signals from an economic efficiency perspective but considering all rate design principles, including customer bill impact, ease of understanding and administration, customer acceptance and government policy, the flat rate structure provides a better balance and is the preferred option.

Regarding marginal cost based rate signals, FEI asked its consultant, EES Consulting, to conduct a study of FEI's marginal delivery cost. The results of this study are provided in Appendix 4-4 to the Application and indicate that FEI's marginal delivery cost is lower than its historical embedded costs. This suggests that rate structures such as inclining block rates which charge more dollars per unit of consumption after a certain consumption threshold is passed are not appropriate for FEI as they send the wrong price signals.

- 31
- 32

- 346.2Does FEI consider that its rate design evaluation requires consideration of how35any change would affect customer decisions to: connect/disconnect to the gas36line; make an investment in gas consuming equipment; decide how much gas to37consume and when to consume it? Please explain.
- 38



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 21

1 Response:

2 FEI's connection and disconnection policies are discussed in Section 12 of its General Terms 3 and Conditions (GT&C). FEI's new customer connection policy was recently reviewed by the 4 Commission in FEI's 2015 System Extension application and decision (Order G-147-16, dated 5 September 16, 2016) and therefore was excluded from the scope of this proceeding. The main 6 extension test considers various factors, such as expected gas consumption over time as well 7 as type and number of gas consuming appliances expected to be installed, to determine 8 whether or not an extension can proceed without a contribution in aid of construction from 9 customers wishing to connect to FEI's distribution system. A similar test exists for the service 10 line cost allowance. Proposals to increase the share of fixed costs recovered by the Basic 11 Charge can improve the results of extension test as it will slightly improve the reliability of the 12 revenue stream forecast in the discounted cash flow models.

13 FEI's rate design evaluations did consider the impact of its proposals on customer behavior. For 14 instance, FEI's proposal to limit the revenue-neutral increase to the Basic Charge for RS 1 to five percent was informed by the potential impact of much higher fixed charges on low-use 15 16 customers, as well as its effect on customers' decisions in terms of investing in energy efficiency 17 measures. Regarding the timing of consumption, FEI did consider seasonal rates for residential 18 customers along with other potential rate structures, but decided that the existing flat rate 19 structure provides the most balanced outcome. The timing of consumption was also considered 20 for the rate design of some commercial and industrial rate classes. For instance, for RS 5/25, 21 the mechanism to set contract demand for the upcoming year favors customers whose peak 22 consumption occurs in the non-peak months (April 1st to October 31st) by halving the demand 23 charge they would otherwise pay if their peak was in the winter months. Another example is RS 24 4, which is a seasonal service (firm in the summer and interruptible in the winter).

- 25
- 26
- 27
- 21
- 28 29

30

6.2.1 Does FEI consider that these considerations should not be affected by whether FEI does or does not supply the commodity? Please explain.

31 **Response:**

- FEI agrees generally that the considerations identified in response to BCUC-FEI IR 1.6.2 should not be affected by whether FEI does or does not supply the commodity.
- 34



RTIS BC"		FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017				
		Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 22				
E.	СНА	PTER 6 – FEI COST OF SERVICE STUDY					
7.0	Refe	rence: FEI COST OF SERVICE STUDY					
		Exhibit B-1, Section 6.2.1.2, p. 6-4;; Section 6.3.5.4, p Appendix 6-5, pp. 1-2; Appendix 6-6, p. 3;	p. 6-18 to 6-19;				
		New Brunswick Energy and Utilities Board (NBEUB) matter of a Review of a Cost of Service Study filed by New Brunswick LP, December 21, 2010 (2010 Enbridg Decision), p. 7	Decision in the [,] Enbridge Gas je COSA				
		http://142.166.3.251/Documents/Decisions/NG/E/2010 20EGNB%20Cost%20of%20Service%20Decision%20-	http://142.166.3.251/Documents/Decisions/NG/E/2010%2012%2021% 20EGNB%20Cost%20of%20Service%20Decision%20-%20E.pdf				
		Minimum System Study and sizing of distribution pipe standards					
	On p	On page 6-4 of Exhibit B-1, FEI states:					
The resu distributio related.		The result of the MSS [Minimum System Study] determines distribution mains costs that are customer related versus costs related.	the proportion of that are demand				
	On p	age 6-19 of Exhibit B-1, FEI states:					
		The MSS results allocate 30% of the distribution system costs related component and 70% to the demand-related component.	to the customer-				
	On p	age 1 of Exhibit B-1, Appendix 6-5, FEI states:					
		To estimate the value of mains required from a customer co demand component FEI follows the steps outlined below:	nnection vs. the				
		4. Value FEI's mains at the minimum standard size and mater	rial (60mm PE)…				
	On p	age 3 of Exhibit B-1, Appendix 6-6, FEI states:					
		Effective Nov 3, 2008 (per IB 2008-43 Elimination of 88 m restricted use of 42 mm PE Pipe) 88 mm PE is no longer beinstallations and 42 mm PE will be restricted to single services w Where these 88 mm and 42 mm material would have been self the next larger pipe size, 114 mm and 60 mm respectively [Emphasis added]	Im PE pipe and ng used for new vithout branches. <u>ected in the past</u> , must be used.				
	On p	age 7 of the 2010 Enbridge COSA Decision, the NBEUB states:					
		A specific issue in this cost of service analysis is whether to use 1¼-inch mains in the calculation of the minimum system. Dr. O to use 2-inch mains in the calculation. His position, based on	e 2-inch mains or vercast proposes discussions with				

		FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
FORTISBC		Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 23
1		EGNB engineers, is that this is the minimum sized main current	ly being installed.
2 3		He does acknowledge that 1 ¹ / ₄ -inch pipe continues to be used areas like cul-de-sacs but he does not truly consider these to be	to service certain mains.
4 5		For the purposes of the minimum system study, the Board o use a 1¼-inch main as the minimum.	rders that EGNB
6 7 8	7.1	Please provide document "IB 2008-43 Elimination of 88 m restricted use of 42 mm PE Pipe".	m PE pipe and
9	Response:		
10	Please refer	to Attachment 7.1 for the requested document.	
11			
12 13			
14 15 16 17 18 19	7.2	Please provide schedules of the mains installed in each of 2015 same format as Table 1 in Exhibit B-1, Appendix 6-5, page provide the percentage of mains that are equal to or less than year.	5 and 2016 in the 2. Please also 1 60mm for each
20	Response:		
21 22 23	The followin The percent of mains len	g two tables show the Distribution Mains addition by pipe size for age of mains length for 60mm pipe and smaller is shown, as well a gth for the sizes of pipe that are smaller than 60mm (i.e., excluding	[•] 2015 and 2016. Is the percentage 60mm pipe).

In 2015, approximately 4.5 percent of the mains installed were 42mm or smaller. The dominant
pipe size installed was the 60mm pipe at 108.75 kilometers or 66.1 percent (108.75 / 164.51) of
the total.

In 2016, approximately 5.3 percent of the mains installed were 42mm or smaller, and since
there were no mains at 48mm, the same 5.3 percent of mains installed were smaller than
60mm. The dominant pipe size installed in 2016 was 60mm pipe, at 65.57 kilometers, or 68.6
percent (65.57 / 95.63) of the total.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 24

2015 Combined PE and Steel Mains					
					% of Length
	Diam	neter		Cumulative	60mm or
Line No.	Inches	mm	Length (m.)	Length (m.)	less
1	0.6	15	763	763	
2	0.8	21	6	769	
3	1	26	5,825	6,593	
4	1.25	33	-	6,593	
5	1.7	42	766	7,360	4.5%
6	1.9	48	-	7,360	
7	2.4	60	108,753	116,113	70.6%
8	2.9	73	-	116,113	
9	3.5	88	159	116,272	
10	4	101	-	116,272	
11	4.5	114	26,276	142,547	
12	6.6	168	17,519	160,066	
13	8.6	219	3,722	163,788	
14	10.7	273	10	163,798	
15	12.7	323	708	164,506	
16	16	406	-	164,506	
17	18	457	-	164,506	
18	20	508	-	164,506	
19	24	609	-	164,506	
20	30	762	-	164,506	
21	36	914	-	164,506	
22	42	1066	-	164,506	
23	Total		164,506		



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
	Julie 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 25

2016 Combined PE and Steel Mains								
					% of Length			
	Diame	ter		Cumulative	60mm or			
Line No.	Inches	mm	Length (m.)	Length (m.)	less			
1	0.6	15	144	144				
2	0.8	21	-	144				
3	1	26	4,365	4,509				
4	1.25	33	-	4,509				
5	1.7	42	561	5,070				
6	1.9	48	-	5,070	5.3%			
7	2.4	60	65,570	70,640	73.9%			
8	2.9	73	-	70,640				
9	3.5	88	93	70,733				
10	4	101	-	70,733				
11	4.5	114	15,001	85,735				
12	6.6	168	8,196	93,930				
13	8.6	219	1,702	95,632				
14	10.7	273	-	95,632				
15	12.7	323	-	95,632				
16	16	406	-	95,632				
17	18	457	-	95,632				
18	20	508	-	95,632				
19	24	609	-	95,632				
20	30	762	-	95,632				
21	36	914	-	95,632				
22	42	1066	-	95,632				
23	Total		95.632					

7.2.1 If FEI installed 42 mm mains in 2015 or 2016, please explain why 42 mm is not the minimum main size used in the Minimum System Study. In your response, please take into consideration the determination on page 7 of the 2010 Enbridge COSA Decision.

10 Response:

Since 2003, FEI's standard has been to connect customers to a new main that is at least a 60mm size and it is by exception only that a smaller main would be used. The use of 42mm

13 pipe is infrequent and is only used for repairs to existing 42mm pipe, or for single services with



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 26

no branches. Approximately 766 meters of 42mm pipe were installed in 2015 and 561 meters in
2016. In contrast, approximately 108,753 meters of 60mm pipe was installed in 2015 and
65,570 meters in 2016. FEI uses the 60mm size pipe almost exclusively as the minimum sized
pipe for new mains installation (refer to the response to BCUC-FEI IR 1.7.2).

5 It is difficult to draw any implications for FEI's Minimum System Study from the 2010 Enbridge 6 COSA Decision quoted in the preamble:

- As discussed in the quoted decision, Enbridge in that case had not completed a
 Minimum System Study, and the New Brunswick Energy Utilities Board had to make its
 determination "[i]n the absence of any detailed study of EGNB's own system".²
- The New Brunswick Energy Utilities Board provides no rationale in its decision for why it concludes that a system with 1¼-inch mains better calculates the appropriate minimum system for Enbridge's minimum system study, other than noting the use of 1¼-inch pipe on Enbridge's system. In the absence of any clear rationale for the Board's determination, it is difficult to determine if the decision is applicable to FEI's Minimum System Study.
- It is unclear what the rate impacts of using 1¼-inch mains for the minimum system were
 for Enbridge's customers, or how this interacted with any other rate design issues or
 considerations in that case.

For FEI's Minimum System Study, the use of 60 mm polyethylene pipe is appropriate as it has been FEI's minimum standard size and material since 2003, and is used more widely in the system than 42 mm pipe.

22 23 24 7.3 25 Please reproduce the following tables, to the best of your ability, to show the impact of using a minimum main size of 42mm in the Minimum System Study: 26 27 Table 1: Minimum System Results for All Mains (Exhibit B-1, Appendix 6-5, p. i. 28 2) 29 ii. Table 6-16: Delivery Cost of Service Allocation to Rate Schedules (Exhibit B-30 1, p. 6-27) 31 iii. Table 12-2: COSA R:C and M:C Results after Rate Design Proposals (Exhibit 32 B-1, p. 12-5) 33 iv. Table 12-4: FEI Rate Proposal Summary (Exhibit B-1, p. 12-8)

² 2010 Enbridge COSA Decision, page 7. Online: http://142.166.3.251/Documents/Decisions/NG/E/2010%201;

http://142.166.3.251/Documents/Decisions/NG/E/2010%2012%2021%20EGNB%20Cost%20of%20Service%20Decision%20-%20E.pdf.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 27

2 Response:

3 FEI considers that the Minimum System Study using 60 mm pipe is the correct approach as it is

4 FEI's minimum standard in most cases when installing mains for the distribution system. As

5 identified in response to BCUC-FEI IR 1.7.2, the number of 42 mm mains installed in 2015 and

6 2016 was small compared to FEI's standard of 60 mm.

Nevertheless, FEI has updated the Minimum System Study to use 42mm as the minimum as requested. FEI calculated the average replacement cost based on the 42 mm minimum system at \$89.53 per meter. This amount is higher than the \$55.68 per meter for the 60 mm that was used in the Application. Material (pipe) cost is only a small portion of the cost of installing a main; since the 42 mm is used in circumstances where existing 42 mm needs replacement, which are smaller scale jobs, the cost per meter is higher. When using the 42 mm as the minimum, distribution mains are split 40 percent Customer and 60 percent Demand as can be

14 found in Table 1 below.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 28

Table 1: Minimum System Results for All Mains

COMBIN	ED STEEL & P	LASTIC MA	INS						
	Diame	eter						Mi	nimum Size Cost
				Un	it Cost / Length			(A	l Pipe Valued at
Line No.	Inches	mm	Length in Meters	(\$/m)		Weighted Cost			60mm PE)
	(1)	(2)	(3)		(4)		(5)		(6)
1	0.6	15	201,739	\$	92.53	\$	18,667,526	\$	18,061,558
2	0.8	21	38,914	\$	240.94	\$	9,375,869	\$	3,483,990
3	1.0	26	1,491,415	\$	196.47	\$	293,012,603	\$	133,525,605
4	1.3	33	17,750	\$	240.56	\$	4,269,924	\$	1,589,152
5	1.7	42	8,176,149	\$	130.49	\$	1,066,929,326	\$	732,006,502
6	1.9	48	41,693	\$	241.46	\$	10,067,387	\$	3,732,769
7	2.4	60	9,344,973	\$	148.83	\$	1,390,781,394	\$	836,650,682
8	0.6	15.0	0			\$	-	\$	-
9	0.8	21.0	200	\$	241.73	\$	48,329	\$	17,900
10	1.0	26.0	2,303	\$	241.73	\$	556,680	\$	206,178
11	1.3	33.0	2	\$	241.73	\$	555	\$	206
12	1.7	42.0	9,481	\$	241.73	\$	2,291,804	\$	848,816
13	1.9	48.0	0			\$	-	\$	-
14	2.4	60.0	48,205	\$	241.73	\$	11,652,632	\$	4,315,790
15	2.9	73	585	\$	274.33	\$	160,579	\$	52,406
16	3.5	88	1,629,167	\$	167.72	\$	273,236,425	\$	145,858,462
17	4.0	101	592	\$	275.56	\$	163,058	\$	52,978
18	4.5	114	2,714,754	\$	208.66	\$	566,447,291	\$	243,050,523
19	6.6	168	1,190,799	\$	449.10	\$	534,788,514	\$	106,611,623
20	8.6	219	292,284	\$	1,876.21	\$	548,386,780	\$	26,168,078
21	10.7	273	49,070	\$	2,274.10	\$	111,590,603	\$	4,393,228
22	12.7	323	125,597	\$	2,274.19	\$	285,631,012	\$	11,244,615
23	16.0	406	33,359	\$	2,274.22	\$	75,866,002	\$	2,986,621
24	18.0	457	1,947	\$	2,274.22	\$	4,428,391	\$	174,333
25	20.0	508	57,658	\$	6,171.01	\$	355,805,428	\$	5,162,051
26	24.0	609	1,466	\$	6,171.01	\$	9,045,949	\$	131,239
27	30.0	762	11,779	\$	6,171.01	\$	72,687,404	\$	1,054,554
28	36.0	914	0			\$	-	\$	-
29	42.0	1066	0			\$	-	\$	-
32	Т	OTAL	25,481,880			\$	5,645,891,466	\$	2,281,379,858
33									
34	Customer Re	elated Con	nponent L	Line 32, Column (6) / Line 32, Column (5)					<u>40%</u>
35	Demand Rel	ated Com	ponent		1 -	Lin	e 34, Column (6)		<u>60%</u>

2

The PLCC is the peak load carrying capacity embedded in the minimum system. When using a 4 A2 mm minimum system, a new PLCC must be calculated. When the minimum system is 5 reduced from 60 mm to 42 mm, the PLCC adjustment is reduced from 0.205 to 0.113.



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 29

- 1 The combination of the increased customer component of distribution mains and the reduction
- 2 of the PLCC results in an increase in the allocation of costs to RS 1 and a decrease in the costs
- 3 allocated to most of FEI's other rate schedules as shown in Table 2 (updated Table 6-16) below.

Table 2: Delivery Cost of Service Allocation to Rate Schedules

Rate Schedule	\$(000)	Percentage of total
1	524,102	66.9%
2	125,158	16.0%
3/23	89,052	11.4%
4	51	0.0%
5/25	32,573	4.2%
6	147	0.0%
7/27	1,544	0.2%
22	807	0.1%
22A	6,815	0.9%
22B	2,598	0.3%
Total	782,847	100.0%

5

6 Two of FEI's rate proposals are affected by the reallocation of costs as described above: RS 1 7 and RS 22. With less costs allocated to the proposed new RS 22 Firm, a larger downward 8 adjustment to RS 22 customers revenue is required when setting the rates at a 100 percent R:C 9 ratio. The adjustment decreases the proposed rates from the RS 22 Firm and increases the RS 10 1 revenue adjustment, ultimately increasing the RS 1 volumetric rate as can be seen in Table 3

11 (updated Table 12-2) and Table 4 (updated Table 12-4) below.



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
-	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 30

Table 3: COSA R:C and M:C Results after Rate Design Proposals

Rate Schedule	Initial COSA		Revenue Shift (\$000)	Approximate Annual Bill Change	COSA after Rate Design Proposals	
	R:C	M:C			R:C	M:C
Rate Schedule 1	95.6%	03.1%	2 610 0	0.3%	94 9%	02.1%
Residential Service	33.070	35.170	2,013.0	0.070	54.570	52.170
Rate Schedule 2	101 20/	102 50/	(1 174 1)	0.5%	104 10/	107 70/
Small Commercial Service	101.5%	102.5%	(1,1/4.1)	-0.5%	104.1%	107.7%
Rate Schedule 3/23						
Large Commercial Sales and	101.6%	103.3%	1,174.1	0.6%	106.8%	114.7%
Transportation Service						
Rate Schedule 5/25						
General Firm Sales and	104.9%	112.2%	45.2	0.0%	109.3%	124.7%
Transportation Service						
Rate Schedule 6/6P	121 20/	150 1%			133 304	163.0%
Natural Gas Vehicle Service	131.270	159.170			133.270	105.970
Rate Schedule 22A						
Transportation Service (Closed)	109.5%	109.8%			113.2%	113.5%
Inland Service Area						
Rate Schedule 22B						
Transportation Service (Closed)	99.7%	99.7%			103.2%	103.2%
Columbia Service Area						
Rate Schedule 22						
Large Volume Transportation	1425.5%	1864.4%	(2,586.8)	-11.7%	100.0%	100.0%
Service						

Rate Schedule (rates not set using allocated costs)	Initial COSA		Revenue Shift (\$000)	Approximate Annual Bill Change	COSA after Rate Design Proposals	
	R:C	M:C	(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	.	R:C	M:C
Rate Schedule 4	1 4 7 4 9/	550.0%	12.2	1.09/	150.0%	572.0%
Seasonal Firm Gas Service	147.470	550.9%	13.5	1.970	150.0%	575.0%
Rate Schedule 7/27						
General Interruptible Sales and Transportation Service	139.6%	712.3%	(90.7)	-0.3%	139.2%	711.6%

2

3

Table 4: FEI Rate Proposal Summary

Rate Schedule	Estimated COSA-Based 2018 Rates	Proposed Rate Changes	Estimated 2018 Rates After Proposed Changes
RS 1 – Residential			
Basic Charge (daily)	\$0.3890	\$0.0195	\$0.4085
Delivery Charge (\$/GJ)	\$4.821	(\$0.050)	\$4.771



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 31

Rate Schedule	Estimated COSA-Based 2018 Rates	Proposed Rate Changes	Estimated 2018 Rates After Proposed Changes
RS 2 – Small Commercial			
Basic Charge (daily)	\$0.8161	\$0.1324	\$0.9485
Delivery Charge (\$/GJ)	3.850	(\$0.186)	3.664
RS 3/RS 23 – Large Commercial			
Basic Charge (daily)	\$4.3538	\$0.4357	\$4.7895
Delivery Charge (\$/GJ)	\$3.189	\$0.001	\$3.190
RS 4			
Basic Charge (Monthly)	\$439	Nil	\$439
Delivery Charge (\$/GJ) Off Peak	\$1.278	\$0.114	\$1.392
Delivery Charge (\$/GJ) Extended Period	\$2.183	(\$0.018)	\$2.165
RS 5/RS 25			
Basic Charge (Monthly)	\$587.00	Nil	\$587.00
Delivery Charge (\$/GJ)	\$0.887	Nil	\$0.887
Demand Charge (\$/Month/GJ)	\$21.596	\$3.00	\$24.596
RS 6/RS 26			
Basic Charge (Monthly)	\$61	Nil	\$61
Delivery Charge (\$/GJ)	\$4.873	(\$1.318)	\$3.555
RS 7/RS 27			
Basic Charge (Monthly)	\$880.00	Nil	\$880.00
Delivery Charge (\$/GJ)	\$1.455	(\$0.012)	\$1.443
RS 22			
Basic Charge (Monthly)	\$3,664.00	Nil	\$3.664.00
Firm Demand Charge (\$/Month/GJ)	n/a		\$22.478
Firm MTQ (\$/GJ)	n/a		\$0.150
Interruptible MTQ (\$/GJ)	\$1.060	(\$0.171)	\$0.889

7.4 Please complete the following table to summarize the impact of changing the minimum main size in the minimum system study on the allocation of distribution system costs between demand and customer components.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 32

	Minimum System Study Results (2001 to 2016)			Alternate Scenario	
	BC Gas 2001 Rate	2012 COSA ⁽¹⁾	2016 FEI Rate	2016 FEI Rate	
	Design Application		Design Application	Design Application	
Minimum Main Size			60 mm	42 mm	
Customer Related Component			30%		
Demand Related Component			70%		
Notes					

(1) Please use the minimum main size that was used in the development of the 2012 COSA that was used to inform the FEU 2012 Common Rates Amalgamation and Rate Design Application

1 2

3 **Response**:

- 4 Please refer to the table below.
- 5 Since FEI installs very small amounts of 42 mm pipe (refer to the response to BCUC-FEI IR
- 6 1.7.2), the pricing per unit length results were higher than when using the 60 mm pipe. This
- 7 leads to a higher cost minimum system and, consequently, a larger Customer Related
- 8 Component.

	Minimum Syster	Alternate Scenario		
	BC Gas 2001 Rate Design Application	2012 COSA (1)	2016 FEI Rate Design Application	2016 FEI Rate Design Application
Minimum Main Size	42 mm	60 mm	60 mm	42 mm
Customer Related Component	26%	39%	30%	40%
Demand Related Component	74%	61%	70%	60%

- 9 Note:
- 10 ¹ Response to BCUC IR 1.135.11 from the FEU 2012 Common Rates Amalgamation and Rate
- 11 Design Application.
- 12



BC™	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 33

1	8.0	Refere	ence:	FEI COST OF SERVICE STUDY	
2				Exhibit B-1, Section 3.2.1.2, p. 3-5	
3				Storage and Transport Costs	
4	On page 3-5 of Exhibit B-1, FEI states:				
5 6 7 8			Storag contrac sales c [Em	e and transport costs are <u>primarily</u> incurred as a result of resources ated by FEI to facilitate the flow of gas on FEI's system so that the load of sustomers can be served and the system stays in balance on a daily basis phasis added]	
9 10 11 12		8.1	When if there discuss	FEI says storage and transport costs are "primarily incurred," please state are reasons why FEI incurs storage and transport costs other than those sed in the preamble above.	
13	<u>Resp</u>	onse:			
14 15 16 17 18 19 20 21	FEI clarifies that storage and transportation resources are fixed costs under the Annual Contracting Plan and are secured for the primary purpose of serving the load for core sales customers under Rate Schedules 1 through 7. A secondary benefit of these resources is that they are used to balance the system as a whole, including for customers served under the transportation model. A third benefit of holding these resources is to support FEI as the supplier of last resort. Should an operational upset occur, such as a reduction in pressure on FEI's transmission line or the interconnecting pipeline, these resources would be called upon to maintain system pressure and integrity.				
22 23					
24 25 26 27	<u>Resp</u>	onse:	8.1.1	Please elaborate as to what these reasons are.	
28	Pleas	e refer t	o the res	sponse to BCUC-FEI IR 1.8.1.	
29 30					
31 32 33 34		8.2	Please impact	explain if transportation customers balancing practices have had any on the storage and transport costs for the resources contracted.	



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 34

1 Response:

Storage and transportation resources are fixed costs under the Annual Contracting Plan and are secured for the primary purpose of serving the load for core sales customers under Rate Schedules 1 through 7. The balancing practices of transportation customers have had no impact on these fixed costs. Transportation customers benefit from these resources when they are used to balance the system as a whole; however, FEI does not collect any fees from transportation customers for this balancing service to compensate for the cost to hold these resources.

9 Please also refer to BCOAPO-FEI IR 1.10.1b) for a discussion of the potential cost reduction to

10 midstream resources if the proposed daily balancing and tolerances are implemented.



TN	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
	2016 Rate Design Application (the Application)	June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 35

1	9.0	Reference:	FEI COST OF SERVICE STUDY
2			Exhibit B-1, Section 6.3.1.1., p. 6-6; Section 6.3.2, Table 6-5, p. 6-10
3			to 6-12; Section 6.3.4.1, p. 6-13
4			Tilbury Expansion Project costs
5		On page 6-1	1 of Exhibit B-1, FEI states:
6		The ⁻	Filbury Expansion Project is an expansion to FEI's existing LNG facility
7		locate	d in Delta. The Project includes additional liquefaction of 35 TJ/Day and a 1
8		BCF	LNG storage tank to serve growing LNG demand. The cost recovery of
9		exper	iditures associated with the Tilbury Expansion Project was authorized by
10		Direct	ion No. 5 to the Commission as amended (OIC No. 557/2013 and OIC No.
11		749/2	014) The Tilbury Expansion Project is estimated to cost \$400 million
12		plus o	levelopment costs and AFUDC FEI's general approach for known and
13		meas	urable changes has been to include in its COSA model the annual cost of
14		servio	e for 2018 for the CTS projects and the annual cost of service for the first
15		year	of operations for LMIPSU. For the Tilbury Expansion Project, which is the
16		only p	project that has associated revenues, FEI adopted a different approach. As
17		descr	ibed below, FEI used a ten-year levelized margin approach in the COSA
18		mode	I to more accurately reflect the ongoing impact of this project on customers.
19		On page 6-10) of Exhibit B-1, FEI provides Table 6-5 which presents the expected project
~~		• • • • • • •	

- 19 On page 6-10 of Exhibit B-1, FEI provides Table 6-5 which presents the expected project 20 in-service dates and COSA costs. Table 6-5 shows that in the COSA study, the Tilbury 21 Expansion Project had a mid-year rate base addition of \$399 million and a cost of 22 service of \$7 million.
- 239.1Please prepare a schedule that reconciles the 2016 FEI approved revenue24requirement of \$1,237.5 million in Table 6-1 on page 6-6 to the 2016 delivery25cost of service functionalization of \$782.847 million in Table 6-8 on page 6-13.
- 26

27 Response:

28 Please find the requested reconciliation in the table below:

Particulars	\$ million \$ million	n Reference
2016 FEI Revenue Requirement	1,237.	5 Table 6-1, Page 6-6
Cost of Gas	(477.	7) Table 6-1, Page 6-6
Bio-Methane Cost of Service in Bio-Methane Variance Account	(1.	5)
Lower Mainland lintermediate Pressure System Upgrade Project	25.	1 Table 6-5, Page 6-10
Coastal Transmission System	13.	8 Table 6-5, Page 6-10
Tilbury Expansion		
Cost of Service	46.8	
RS 46 Delivery Revenue	(39.7) 7.	0 Table 6-5, Page 6-10
BC Hydro IG Delivery Revenue	(15.	7)
Joint Venture Delivery Revenue	(4.	5)
Bypass Delivery Revenue	(1.	1)
Functionalized Cost of Service	782.	8 Table 6-8, Page 6-13
FORTIS BC^{**}

5

6

7

FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 36

- 1 2 3 4 9.2 Please provide a table which shows (i) the cost of service and (i
 - 9.2 Please provide a table which shows (i) the cost of service and (ii) RS 46 revenues for the Tilbury Expansion Project by year for the each of the ten years included in the levelized Tilbury Expansion Project cost of service.

8 **Response:**

9 The following paragraph describes the rationale for using a 10 year levelized approach for the 10 Tilbury Expansion Project.

11 The Tilbury Expansion Project, which has both incremental volumes (revenues) and costs, is 12 unlike the Lower Mainland Intermediate Pressure System Upgrade Projects and the Coastal 13 Transmission System Project, which have costs but do not have incremental volumes 14 associated with them. For the Tilbury Expansion Project, the incremental volumes are not all 15 realized at the time that the full costs of the Tilbury Expansion Project are included in rate base. 16 Reflecting only the first year of incremental revenues would not be representative of the longer 17 term impact that the Tilbury Expansion Project will have on the revenue requirement. As such, 18 and as described in Section 6.3.2.3, FEI used a 10-year levelized approach for inclusion of 19 costs and revenues for the Tilbury Expansion Project.

\$000 Year 1 Year 2 Year 3 Year 4 Year 5 Year 6 Year 7 Year 8 Year 9 Year 10 0&M 3,570 3,928 4,331 4,823 5,722 7,072 7,213 7,357 7,504 7,654 Depreciation 3,096 3,130 3,161 3,188 3,211 3,341 3,360 3,409 3,429 3,450 Structures & Improvements 2,211 2,236 2,258 2,277 2,294 2,400 2,435 2,450 2,464 Gas Holders - Storage 2,387 **Compressor Equipment** 9,434 9,540 9,634 9,717 9,787 10,183 10,241 10,391 10,451 10,513 Negative Salvage Provision 1,423 1,423 1,423 1,423 1,423 1,423 1,423 1,423 1,423 1,423 Income Tax (751) (174)438 929 1,396 1,761 1,116 521 (131) (750) LNG Tax Credit (425)(842) (962) (1,342) (1,494) (2,252) (2,350) (2,421) (2,465) (2,510) LNG Tax 406 0 39 63 186 380 393 419 432 165 Earned Return 28,425 27,682 26,895 26,062 25,268 24,432 23,514 22,590 21,636 20,798 Total Cost of Service (i) 46,984 46,963 47,241 47,241 47,793 48,727 47,311 46,112 44,716 43,474 23,770 32,204 34,893 RS46 Delivery Revenue (ii) 11,220 21,463 51,278 52,300 53,343 54,406 55,490

20 The requested table is provided below:

22

21

FEI notes that the LNG Tax and LNG Tax Credit included in the cost of service for the Tilbury Expansion Project have been enacted but have not been proclaimed in force by the Lieutenant

25 Governor in Council for BC.

26



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 37

9.3 Please use calculations to explain what the cost of service included in the COSA would be for the Tilbury Expansion project if FEI used its general approach for known and measurable changes and included the estimated annual cost of service and revenue for only 2018 in the COSA.

7 <u>Response:</u>

8 If FEI had used the forecast 2018 costs and RS 46 revenues in the analysis, the total cost of

- 9 service to be included would be \$46,963 thousand and the RS 46 delivery revenue would be
- 10 \$21,463 thousand. Details are included in the table below.

\$000	Year 2
0&M	3,928
Depreciation	
Structures & Improvements	3,130
Gas Holders - Storage	2,236
Compressor Equipment	9,540
Negative Salvage Provision	1,423
Income Tax	(174)
LNG Tax Credit	(842)
LNG Tax	39
Earned Return	27,682
Total Cost of Service (i)	46,963
RS46 Delivery Revenue (ii)	21,463
(i) less (ii)	25,500

- 9.3.1 Please reproduce the following tables to show the impact of using the estimated 2018 cost of service and revenue associated with the Tilbury Expansion project in the COSA:
 - i. Table 6-8: Delivery Cost of Service Functionalization Summary (Exhibit B-1, p. 6-13)
 - ii. Table 6-16: Delivery Cost of Service Allocation to Rate Schedules (Exhibit B-1, p. 6-27)
 - Table 12-2: COSA R:C and M:C Results after Rate Design Proposals (Exhibit B-1, p. 12-5)



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 38

- 2 Please refer to the adjusted Table 6-8, Table 6-16 and Table 12-2 below.
- 3 The predominant change to the COSA from using 2018 Tilbury Expansion forecast costs and
- 4 revenues³ in the COSA is that the known and measurable changes are increased by \$15,383
- 5 thousand. This increase is mainly due to the difference between the forecast revenues in 2018
- 6 compared to the ten-year levelized forecast revenues.
- 7 The results are minor changes to most of the R:C ratios, with only RS 22 Interruptible showing a
- 8 significant change. RS 22 Interruptible attracts very few costs because it does not contribute to
- 9 system peak demand, yet the net cost of Tilbury Expansion is allocated to all non-bypass
- 10 customers based on margin. The table below shows the minor changes to the R:C ratios from
- 11 treating Tilbury like the other two projects.

Rate Schedule	Change in R:C Ratio			
RS 1	+0.2%			
RS 2	-0.2%			
RS 3/23	-0.3%			
RS 5/25	-0.3%			
RS 6/6P	+0.9%			
RS 22A	+2.0%			
RS 22B	+1.8%			
RS 22	0.0%			
50 /	4.40/			
RS 4	+1.1%			
RS 7/27	+0.9%			

12 13

Table 6-8 (Adjusted): Delivery Cost of Service Functionalization Summary

Function	\$ thousands	Percentage of total
Gas Supply Operations	2,016	0.3
Tilbury LNG Storage	50,701	6.4
Mt. Hayes LNG Storage	7,577	0.9
Transmission	172,135	21.6
Distribution	463,423	58.1
Marketing	50,153	6.3
Customer Accounting	52,226	6.4
Total	798,230	100.0

³ Treatment of Tilbury Expansion is in the same manner as the LMIPSU and CTS projects in the COSA. FEI has included the project costs and revenues as forecast for 2018.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 39

Table 6-16 (Adjusted): Delivery Cost of Service Allocation to Rate Schedules

Rate Schedule	\$ thousands	Percentage of total
1	519,186	65.0%
2	133,067	16.7%
3/23	97,838	12.3%
4	51 0.0%	
5/25	36,138	4.5%
6	152	0.0%
7/27	1,542 0.2%	
22	808 0.1%	
22A	6,840	0.9%
22B	2,608	0.3%
Total	798,230	100.0%



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 40

Table 12-2 (Adjusted): COSA R:C and M:C Results after Rate Design Proposals

Rate Schedule	Initial COSA		A Revenue Approximate Shift Annual Bill (\$000) Change		COSA after Rate Design Proposals	
	R:C	M:C		Ū	R:C	M:C
Rate Schedule 1	05.7%	03.3%	757 /	0.1%	96.6%	04 7%
Residential Service	95.770	93.370	757.4	0.176	90.0%	94.770
Rate Schedule 2	101 10/	102.00/	(1 174 1)	0.5%	102.09/	102 69/
Small Commercial Service	101.1%	102.0%	(1,174.1)	-0.5%	102.0%	103.0%
Rate Schedule 3/23						
Large Commercial Sales and	101.3%	102.6%	1,174.1	0.6%	103.3%	106.7%
Transportation Service						
Rate Schedule 5/25						
General Firm Sales and	104.6%	111.2%	45.2	0.0%	106.0%	114.8%
Transportation Service						
Rate Schedule 6/6P	132.0%	160.3%			132.6%	161.6%
Natural Gas Vehicle Service	152.070	100.570			152.070	101.070
Rate Schedule 22A						
Transportation Service (Closed)	111.4%	111.7%			115.0%	115.4%
Inland Service Area						
Rate Schedule 22B						
Transportation Service (Closed)	101.4%	101.4%			104.9%	104.9%
Columbia Service Area						
Rate Schedule 22						
Large Volume Transportation	1451.3%	1898.1%	(725.2)	-3.3%	100.0%	100.0%
Service						

Rate Schedule (rates not set using allocated costs)	Initial COSA		Revenue Shift (\$000)	Approximate Annual Bill Change	COSA after Prop	Rate Design oosals
	R:C	M:C	(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		R:C	M:C
Rate Schedule 4	140 50/	560.0%	10.0	1.0%	151 20/	E99 69/
Seasonal Firm Gas Service	140.3%	500.9%	13.3	1.9%	151.5%	500.0%
Rate Schedule 7/27						
General Interruptible Sales and Transportation Service	140.5%	725.4%	(90.7)	-0.3%	140.2%	727.1%

- 9.4 If the cost for the Tilbury Expansion Project exceeded \$400 million (excluding development costs and AFUDC) and was instead \$450 million, please discuss how these additional costs (\$50 million) would be treated in the COSA and recovered from customers.



11

N	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 41

1 Response:

- 2 The COSA treatment will be the same for all Tilbury Expansion Project costs as described in 3 Section 6.3.2.3. Please also refer to the response to BCUC-FEI IR 1.9.5.
- 6
 7 9.5 If the cost for the Tilbury Expansion Project exceeded \$400 million (excluding development costs and AFUDC) and was instead \$450 million, please discuss if it is within the Commission's jurisdiction to determine the rates and customers from whom these additional costs (\$50 million) would be recovered.
- 12 **Response**:
- The B.C. Government recently passed amendments to Direction No. 5 by Order in Council No.
 162/2017 dated March 21, 2017, which included an increase in the spending limit for the Tilbury
- 15 Phase 1A expansion from \$400 million to \$425 million. At this time, FEI expects the capital 16 costs to be within the \$425 million spending limit.
- Direction No. 5 determines how and from whom the costs of the Tilbury Expansion Project are
 recovered. Specifically, under section 4(c) of Direction No. 5, the capital costs are recoverable
 from "applicable customers", which is a defined term. Section 4 of Special Direction 5 states in
- 20 part:
- 4 (2) In setting rates under the Act for FortisBC Energy Inc., the commission must
 do <u>all</u> of the following:
- 23
- 24 (c) include in the calculation of rates for applicable customers
- 25(i) the annual revenues from the sale of LNG from phase 1A26facilities and phase 1B facilities,
- 27 (ii) the annual operating costs of phase 1A facilities and phase 1B28 facilities, and
- 29(iii) the capital costs, construction carrying costs, sustaining capital30costs, decommissioning and salvaging costs and feasibility and31development costs respecting phase 1A facilities32facilities;

Direction No. 5 defines "applicable customers" as being synonymous with FEI's non-bypasscustomers:



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 42

- "applicable customers" means customers of a utility other than customers
 receiving service
- 3 (a) under a fixed rate,

4 (b) in the Fort Nelson service area of the utility, unless the Fort Nelson 5 service area no longer has a distinct rate base, or

6 (c) under the transportation rate schedule;

7 Non-bypass customers also obtain the benefit of the revenues from the Tilbury Expansion8 Project.



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 43

1	10.0	Refere	ence: I	FEI COST O	F SERVI	CE STU	DY				
2			E	Exhibit B-1,	Section	6.3.4.3,	p. 6-14;	Section	6.3.4.4, p.	6-16	
3			٦	Tilbury Expa	ansion P	roject c	osts				
4		On pag	ge 6-14 c	of Exhibit B-1	, FEI sta	tes:					
5 6 7 8 9 10			As discu is incluc Tilbury I Expansi (within custome	ussed in Sec ded in the L Expansion de ion Costs are the function ers.	tion 6.3. NG Stor oes not t e directly) and t	2.3 of th age fund follow th allocate he net	e Applic ction. Ho at of the ed to RS difference	ation, the owever, the existing 46 and c ce is allo	Tilbury Ex ne allocations storage ploffset with l pocated to	pansion project on approach fo ant. The Tilbury RS 46 revenues all non-bypass	rt r y s s
11 12		On pa functio	age 6-14 on from Ti	of Exhibit I ilbury LNG S	B-1, FEI torage."	states	"Mt. Ha	iyes LNG	Storage	has a separate	е
13 14 15 16		10.1	Please (function which is	confirm, or o in the above a separate o	therwise e quote, category	explain FEI is re from the	, that wh eferring e Mt. Hay	nen FEI re to the Tilk yes Storae	efers to the oury LNG ge function	e "LNG Storage Storage functior	n
17	<u>Respo</u>	onse:									
18	Confir	med.									
19 20											
21 22 23 24		On pa provide preser	ge 6-16 c e a small nts the for	of Exhibit B-1 amount of L recast RS 46	I, FEI sta NG for th demand	ates that ne NGT i that wil	"In the market." I be supp	near term In Table plied from	, Mt. Haye 6-10 (extra Mt. Hayes	s is expected to icted below) FE	o I
				Table 6-10:	RS 46 Der	nand Fore	ecast Serv	ved by Mt.	Hayes (TJ)		
25					2016 20	2017 100	2018 100	2019 100			
26 27		In Tab that fo	ole 6-6 or rms the b	n page 6-12 basis for the t	of Exhib en year	it B-1, F levelizec	El provi I revenu	ides the f e includeo	orecast of d in the CC	RS 46 demand SA model.	d
28 29 30 31		10.2	Please o RS 46 includec Expansi	confirm that demand for d in the ten y ion Cost with	the RS ∠ ecast_in ear level iin the T	46 dema Table (ized RS ilbury LN	nd supp 5-6 and 46 reve NG Stora	lied by M the reve nue that i age function	t. Hayes is nue from s used to c on before t	included in the this demand is offset the Tilbury the difference is	e s y s

- 32 allocated to all non-bypass customers.
- 33



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 44

- 2 Not confirmed. The RS 46 demand supplied by Mt. Hayes is not included in Table 6-6. Table 6-
- 3 6 was provided to inform the reader of the RS 46 demand forecast to be produced by the Tilbury
- 4 Expansion. FEI does confirm that the RS 46 demand supplied by Mt. Hayes is included in the 5 revenue that is used to offset Tilbury Expansion costs within the Tilbury Storage function.
- 5 revenue that is used to onset Tlibury Expansion costs within the Tlibury Storage function
- In addition to the volume included in Table 6-6, an RS 46 demand of 668.7 TJ⁴ is embedded in FEI's 2016 test year. Included in the 668.7 TJ is the 20 TJ shown in Table 6-10 from the preamble. Within the COSA, both the levelized demand from RS 46 from Table 6-6 plus the 668.7 TJ demand embedded in FEI's test year are used to offset the Tilbury Expansion Cost within the Tilbury LNG Storage function before the difference is allocated to all non-bypass
- 11 customers. A revised and expanded Table 6-6 is included below.

	TJ	Year 1	Year 2	Year 3	Year 4	Year 5	Year 6	Year 7	Year 8	Year 9	Year 10	Levelized and included in COSA
	Tilbury Expansion	2,956	5,545	6,021	7,998	8,496	12,242	12,242	12,242	12,242	12,242	8,726
	RS 46 demand included in Test Year (includes the 20 TJ demand served by Mt. Hayes in Table 6-10)								669			
	Total RS 46	demano	d include	ed in CO	SA mode	el						9,395
12 13												
14 15 16 17 18	Response	10 e:	0.2.1	lf not o Table 6	confirm -6 with	ed, plea the RS	ase exp 46 dem	lain and and sup	d provid plied by	e an ur Mt. Hay	odated v es incluc	ersion of led.
19	Please refer to the response to BCUC-FEI IR 1.10.2.											
20 21												
22 23 24 25 26	10 <u>Respons</u> e	.3 Pl ex e:	ease p presse	orovide d in TJ.	the n	naximur	m annu	ial lique	efaction	capacit	y at M	t. Hayes
27	Mt. Hayes	has a	liquefac	ction ca	pacity o	of 8.2 T.	J per da	у.				

⁴ Commission Order G-193-15, dated December 11, 2015, Compliance Filing, Schedule 18.



TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 45

- FEI's standard procedure is to limit liquefaction activities to the non-winter months, from April to October, for a maximum of 214 days. This means a theoretical 1,755 TJ total volume of liquefaction in a year, but plant maintenance activities will decrease this amount to some extent.
- 4
 5
 6
 7
 10.3.1 What is the maximum amount of Mt. Hayes liquefaction capacity that could be used to supply RS 46 demand without impacting the allocation of costs to Mt. Hayes LNG Storage function within the COSA model and/or to the Storage and Transport (Midstream) Costs component of the Gas Cost Allocation model.
- 13 **Response:**

The maximum liquefaction capacity that could be used to supply RS 46 customers is ninety days in the summer. Ninety days of liquefaction capacity at Mt. Hayes is equal to 729 TJ of LNG that could be produced for the NGT market before impacting the allocation of costs in the

17 COSA.



אז	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
-	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 46

1	11.0	Reference:	FEI COST OF SERVICE STUDY
2			Exhibit B-1, Section 6.3.4.4, pp. 6-14 to 6-16;
3 4 5			Commission Letter L-20-16 dated August 4, 2016, appendix A; Non- confidential FEI 2016/2017 Annual Contracting Plan – Executive Summary, p. E-5:
6 7			FortisBC Energy Utility 2014 Long Term Resource Plan Application, Exhibit B-2, BCUC IR 1.58.1
8			LNG facility cost allocation
9		On page 6-14	4 of Exhibit B-1, FEI states:
10 11 12 13 14 15 16 17 18 19		The e to sup requir suppo activit has a has a transr serve provio	existing Tilbury LNG Storage facility serves as a needle peaking resource pport the CTS's [Coastal Transmission System] ability to meet customer rements on extreme cold days. The Tilbury LNG Storage facility also orts transmission and distribution operations during maintenance and repair ties, emergency outages and supply constraints Mt. Hayes LNG Storage separate function from Tilbury LNG Storage The Mt. Hayes LNG facility a dual purpose of serving as (1) a gas supply storage facility and (2) a mission facility which provides additional transmission system capacity to customers in the same fashion that pipeline looping and compression de such capacity.
20 21 22 23		11.1 Does opera supply	the Mt. Hayes LNG facility also support transmission and distribution tions during maintenance and repair activities, emergency outages and y constraints?

Yes. Mt. Hayes is included in FEI's resource portfolio as a peaking gas supply resource for RS 1-7 customers. It also serves as a replacement for transmission capacity by providing additional system capacity on Vancouver Island to serve customers in the same fashion that pipeline looping and compression could have provided such capacity. Furthermore, Mt. Hayes supports transmission and distribution operations during maintenance and repair activities, emergency outages and supply constraints.

- 31
- 32
- 33
- 34
- On page E-5 of the non-confidential FEI 2016/2017 Annual Contracting Plan Executive
 Summary that is Appendix A to Commission Letter L-20-16, FEI provides the following
 table summarizing the Midstream gas supply portfolio:



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 47

Peak Day Portfolio (TJ/d)	2016/17 Portfolio-	2015/16	
	Planned	Portfolio	
Fort Nelson Division	5	5	
Alberta Baseload Supply (CCRA gas & Mktrs)	83	84	
Station 2 Baseload Supply (CCRA gas & Mktrs)	248	251	
Total Commodity Supply	331	335	
Seasonal Supply	175	195	
Seasonal Storage	196	197	
Market Area Storage	210	210	
Peaking Supply	-	-	
Spot Supply	46	24	
Mt. Hayes LNG	163	163	
Tilbury LNG	163	163	
Industrial Curtailment	28	28	
Total Midstream Supply	981	980	
Total Resources (TJ/d)	1,317	1,320	
Peak Day Demand (TJ/d)	1,317	1,320	

11.2 Please confirm that for the 2016/17 annual contracting plan, the Tilbury LNG and Mt. Hayes LNG resources respectively each constituted 12.4 percent of the total Midstream portfolio of resources required to meet the peak day demand.

Response:

The Tilbury LNG and Mt. Hayes LNG resources both constituted 16.6 percent of the total
Midstream portfolio of resources required to meet the 2016/17 peak day demand, calculated as
follows:

- 10 Tilbury LNG: 163 / 981 = 16.6%
- 11 Mt. Hayes LNG: 163 / 981 = 16.6%

In terms of the total peak day requirements (i.e., Commodity and Midstream) for the 2016/17
 Annual Contracting Plan, each of Tilbury LNG and Mt. Hayes facility constitute 12.4 percent.

- 1711.2.1Does FEI anticipate any significant changes to the role that the Tilbury18LNG and Mt. Hayes LNG resources play in the Midstream portfolio in19the future? If so, please elaborate.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 48

2 FEI does not anticipate any significant changes to the role that the existing Tilbury LNG and Mt. 3 Hayes LNG resources play in the Midstream portfolio. Both will continue to provide peaking 4 supply to Rate Schedules 1 to 7 customers as well as a small amount of LNG supply to RS 46 5 customers, if needed. As RS 46 sales increase in the future, it is possible that the production of 6 LNG at the existing Tilbury and Mt. Hayes facilities will need to increase (on a temporary basis) 7 requiring those facilities to operate more often than in the past. Increased RS 46 sales are 8 expected to be served by additional LNG capacity at Tilbury, such as the current Phase 1A 9 expansion nearing completion. Depending on how the growth of RS 46 and other LNG sales 10 develop, FEI may be able to replace some of its downstream storage or alternative resources in 11 the Midstream gas supply portfolio with storage, liquefaction and vaporization at Tilbury, if 12 available.

- 13
- 14

15 16

17

18

19

11.2.2 Please describe the similarities and the differences in the roles of the Tilbury LNG facility and Mt. Hayes LNG facility, respectively, as a gas supply storage facilities in the Midstream portfolio.

20 Response:

21 The Tilbury LNG facility and Mt. Hayes LNG facility provide similar roles as gas supply storage 22 facilities in the Midstream portfolio. Both are used to provide peaking supply for Rate Schedules 23 1 to 7 customers, and gas supply during emergency conditions. Although both facilities have the 24 same withdrawal or vaporization capability of 163 TJ per day, Mt. Hayes provides approximately 25 10 days of supply while Tilbury provides approximately 4 days of supply, if withdrawn at the 26 maximum daily capability. Given that Mt. Hayes holds more inventory, gas is usually sent out 27 from Mt. Hayes before Tilbury, which tends to be reserved for the peak days with the highest 28 demands.

- 29
- 30
- 31
- 3211.3Please populate the following table to illustrate the extent to which the Mt. Hayes33LNG facility and the Tilbury LNG facility were used to supply the Midstream34portfolio requirements over the past five contract years.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission)	Page 40

Information Request (IR) No. 1

Page 49

Contract Voor	Mt. Hayes LN	IG Facility	Tilbury LNG Facility	
Contract real	No. of Days	Total TJs	No. of Days	Total TJs
Nov 1, 2011 to Oct 31, 2012				
Nov 1, 2012 to Oct 31, 2013				
Nov 1, 2013 to Oct 31, 2014				
Nov 1, 2014 to Oct 31, 2015				
Nov 1, 2015 to Oct 31, 2016				
100 1, 2010 10 000 01, 2010				

1

2

3 **Response:**

- 4 The following table provides the gas sent out from the Mt. Hayes and Tilbury LNG facilities to
- 5 supply the Midstream portfolio requirements over the past five contract years, plus the 2016/17
- 6 contract year up to May 2017.

Contract Voor	Mt. Hayes LNG Facility		Tilbury LNG Facility	
Contract Year	No. Days	Total TJs	No. Days	Total TJs
Nov 1, 2011 to Oct 31, 2012	5	143	2	38
Nov 1, 2012 to Oct 31, 2013	2	30	4	29
Nov 1, 2013 to Oct 31, 2014	12	539	18	148
Nov 1, 2014 to Oct 31, 2015	10	360	4	42
Nov 1, 2015 to Oct 31, 2016	2	43	4	25
Nov 1, 2016 to Oct 31, 2017	13	360	1	13

- 7
- 8 9

10

11

12 13 In the FortisBC Energy Utilities 2014 Long Term Resource Plan (FEU 2014 LTRP) proceeding, in response to BCUC IR 1.58.1 regarding FEU's consideration of the option of increasing either the number of days duration and/or peak day quantities sourced from the existing Tilbury and Mt. Hayes LNG storage facilities for the purpose of meeting the peak design day portfolio load requirements, FEU responded:

14 The FEU also evaluate opportunities on an on-going basis within its own 15 operating region to improve infrastructure leading to better diversity and reliability 16 within the portfolio over the long term. For example, FEI is currently planning to 17 expand the liquefaction and storage capacity at the Tilbury site, primarily to meet 18 the growing market for LNG applications. This may provide an opportunity for the 19 FEU to source additional on-system storage resources, in particular if additional 20 vaporization facilities can be incorporated into the expanded facility. The addition 21 of vaporization to the facility and ability to liquefy at a greater rate than the 22 original peak shaving Tilbury facility could allow FEI to utilize this resource as a 23 market area storage resource during cold weather events. FEI could potentially 24 replace expiring Mist and NWP transportation contracts in the future or replace 25 incremental resources that may be required to meet growing load requirements.



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 50

- 1The FEU will continue to assess this potential opportunity as part of the annual2contracting process.
- 11.4 If FEI were to add vaporization to the expanded Tilbury LNG facility in the future
 so it could be used to replace market area storage contracts such as the Mist
 and NWP transportation contracts in the Midstream portfolio, please describe
 how FEI would propose to address any associated cost allocation issues. For
 instance, would FEI anticipate it would propose an allocation of some portion of
 the Tilbury LNG Storage function to the Storage and Transport portion of the gas
 Allocation model?
- 10

If vaporization equipment was added to the expanded Tilbury LNG facility, it could be used to replace some portion of market area storage contracts. If this situation were to arise in the future, FEI would propose an allocation method to the Commission for its review and approval. The following comments represent FEI's current thinking about a potential allocation methodology, which are speculative at this time, and subject to change based on the facts and considerations relevant at the time FEI were to make its application to the Commission.

As discussed in Section 6.3.2.3 of the Application, the cost allocation under the Tilbury Expansion's current configuration of having no vaporization functionality is that the Tilbury Expansion costs are directly allocated to RS 46 and offset with RS 46 revenues, with the net difference being allocated to all non-bypass customers.

If vaporization equipment were added to the expanded Tilbury LNG facility, it would become a multi-purpose facility and be able to serve as (1) a RS 46 LNG demand facility, and (2) a gas supply storage facility. FEI expects that the allocation of costs with respect to the RS 46 LNG demand function would continue, as described above; however, FEI expects that there would also be an allocation of costs to FEI's midstream portfolio for the gas supply storage function.

27 A potential basis or benchmark for the allocation of costs to the midstream portfolio in this 28 situation would be the estimated cost of avoided third party storage and transportation gas 29 supply resources that would otherwise need to be contracted if available on behalf of the core 30 market. FEI expects that the regulatory treatment to administer such a charge would be to 31 credit or reduce the total delivery costs associated with FEI's operations with the appropriate 32 amount, and debit or charge FEI's midstream portfolio with the equivalent amount (this follows 33 the cost allocation described in Section 6.3.4.4). FEI expects that there would also be an 34 allocation of the variable vaporization costs and a portion of the variable liquefaction costs to the 35 midstream portfolio.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 51

1	12.0	Refer	ence: FEI COST OF SERVICE STUDY
2			Exhibit A2-2, Section 2, p. 5
3			Frequency of COSA studies
4		On pa	ge 5 of Exhibit A2-2, Elenchus states:
5 6 7 8 9			Cost of service allocation studies are conducted periodically by utilities to compare the costs attributable to the various customer classes with the revenues being collected from the customer classes. The frequency with which COSA studies are updated varies across jurisdictions and is typically linked to the rate setting process. Updates are typically expected at least every five years.
10 11 12 13		Elencl five ye ATCO genera	hus further explains that Union Gas Limited conducts a cost of service study every ears, Enbridge Gas Distribution conducts a cost of service study every year, and 9 Gas' most recent cost of service studies are from 2008/2009 and 2011/2012 al rate applications.
14 15 16 17	Resp	12.1 onse:	Please state the most recent time when FEI included a full and up-to-date COSA study for (i) FEI; and (ii) Fort Nelson in a regulatory application.
18 19	The n Nelso	nost reo n in a re	cent time when FEI included a full and up-to-date COSA study for FEI and Fort egulatory application was in 2012.
20 21			
22 23 24 25 26 27		12.2	Please provide a detailed explanation of the costs, time and effort in person- hours to prepare a full and up-to-date COSA study for each of FEI and Fort Nelson (excluding rate design and rebalancing) for regulatory review. Please explain if any external resources are required.
28	<u>Resp</u>	onse:	
29 30 31	The re time t chang	esource hat has jes to th	s required to prepare a COSA can vary depending on factors such as the length of passed since the previous COSA was undertaken, whether there are significant ne underlying costs or the introduction of known and measureable changes, the

- number of supporting studies that need to be done or updated and the extent to which external
 consultants are engaged. FEI will respond to this question in consideration of the resources
 required to complete the current COSAs for FEI and Fort Nelson.
- Internal resources have been utilized extensively in the preparation of the COSA studies for FEI
 and Fort Nelson with this Application. Although FEI has not tracked the labour hours associated
- 37 with the two COSAs, FEI estimates 2,000 hours in total for FEI and 900 hours in total for Fort



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 52

Nelson. In total, the internal fully-loaded labour cost is estimated in the range of \$275 thousand, split 70 percent to FEI and 30 percent to Fort Nelson (FEI notes, however, that Fort Nelson will receive 0.00244 percent of FEI's labour costs through the shared services allocation and not a separate allocation for the internal costs of the COSA). In addition to internal labour, FEI has incurred \$100 thousand of external consultant costs to review and provide an expert opinion on the COSA and supporting studies for FEI and \$5 thousand for Fort Nelson to date.

- 7 8
- 9
- 1012.3Please explain the benefits and disadvantages of performing a full COSA study11every five years as opposed to less frequently, for example every 10 years or12longer.
- 13

14 <u>Response:</u>

Updating a COSA study every five years as opposed to every 10 years has the potential benefit of highlighting any changes among the rate groups that may require further examination and alerting the Commission and the utility of any rate design matters that should be considered. The disadvantage is the time and cost involved with more frequent reviews. For the internal resources, staff with the specific skillset to undertake rate design activities may not be available due to conflicts with other major applications that are underway, and for the external resources, there will be incremental costs involved.

A 10-year interval between full COSA studies would have the potential to leave important rate design issues unknown and unaddressed for an extended period of time.

FEI's predecessor companies from the 1980s through to 2001 have completed COSA studies every 3 to 7 years. FEI is of the opinion that a COSA study that is completed every 4 to 6 years is a reasonable time period to consider if there are issues that need to be raised in a regulatory proceeding, but that significant changes in FEI's business may require more frequent examination of specific limited scope issues. These issues could be raised by FEI, by the Commission or by interveners.



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
*	2016 Rate Design Application (the Application)	June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 53

1 13.0 Reference: FEI COST OF SERVICE STUDY

2 3

4

5

6

7

Exhibit B-1, Section 6.5.2, Table 6-19, p. 6-36; Section 3.2.2, Table 3-2, p. 3-6

Additional data

Table 6-19 in Exhibit B-1 presents the revenue to cost (R:C) and margin to cost (M:C) ratio results for rate schedules that are not set using cost of service allocations. These include RS 4, RS 7, RS 27 and RS 22.

- 8 13.1 Please provide an updated version of Table 6-19 which includes the revenue to 9 cost and margin to cost ratios from the COSA results and before rate design 10 proposals and rebalancing for (i) RS 26; (ii) RS 46; (iii) the aggregate of 11 customers with bypass agreements; and (iv) non-bypass contract customers.
- 12

13 Response:

FEI does not have any customers taking service under RS 26 so there is no revenue and no allocated costs to produce an R:C ratio. RS 46 revenues and directly assigned costs are used to produce the ratios in the table below. Bypass customers and non-bypass contract customers attract costs based on their contribution to the cost allocators in the COSA.

18 An updated Table 6-19 is provided below.

Rate Schedule / Customer	R:C	M:C
Rate Schedule 4	147 9%	548 0%
Seasonal Firm Gas Service	147.070	540.970
Rate Schedule 7/27	140 10/	700 00/
General Interruptible Sales and Transportation Service	140.170	120.070
Rate Schedule 22	1102 20/	1477 60/
Large Volume Transportation Service	1193.3%	1477.3%
(ii) Rate Schedule 46	00.0%	94.00/
Liquefied Natural Gas	90.0%	04.9%
(iii) Customers with Bypass Agreements	14.6%	14.6%
(iv) Non-Bypass Contract Customers	99.7%	99.7%

- 19
- 20
- 21
- 22
- 2313.2Table 3-2 on page 3-6 of Exhibit B-1 shows that RS 50 has no customers. Please24state if FEI could obtain customers for RS 50 within the next five years and25explain what their expected energy demand would be on FEI's system.
- 26



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 54

2 Yes, FEI may have RS 50 customers join the system within the next five years, although no 3 project proponents have made final investment decisions at this time. For example, the 4 Woodfibre LNG project is still under active development and would become an RS 50 customer 5 if that project goes ahead. The expected energy demand on FEI's system for the Woodfibre 6 LNG project would be up to 110 PJ / year⁵. There are other potential RS 50 customers in the 7 Lower Mainland, including LNG projects of similar size to Woodfibre LNG, but FEI does not 8 have certainty on whether they will proceed, and if they do, whether they will commence 9 operations within the five-year period mentioned in the question.

⁵ Woodfibre LNG's website identifies that the Export License is for 2.1 million tonnes of LNG per year which is approximately 110 PJ per year.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 55

1	14.0	Refere	ence:	FEI COST OF SERVICE STUDY
2				Exhibit B-1, Section 3.3.2, p. 3-10
3				Historical R:C ratio range of reasonableness
4		On pa	ge 3-10	of Exhibit B-1, FEI states:
5 6 7 8 9			In Apr consid BC Ga R:C to as a g	ril 1993, BC Gas filed the Phase B Rate Design Application, which ered the allocation of all other utility costs, other than gas supply costs as determined the allocated cost of service of customer rate schedules with cost ratios and proposed a range of 90% to 110% on this ratio to be used uideline for setting rates.
10 11 12 13		14.1	Please in the of thes	e explain the reasons behind the proposed R:C ratio range of 90% to 110% BC Gas 1993 Phase B Rate Design Application and discuss whether each se reasons remain applicable for the FEI 2016 Rate Design Application.
14	<u>Respo</u>	onse:		
15 16 17	The C ratios followi	ommiss in the E ng quot	sion's ao 3C Gas te from j	cceptance of a 90 percent to 110 percent range of reasonableness for R:C 1993 Phase B Rate Design proceeding relied on previous precedent. The page 11 of the 1993 Phase B decision (Order G-101-93) illustrates this:
18 19		The B peak	CGUL F	DC study used three different methods of allocating capacity costs: sibility, non-coincident peak, and average and excess demand.

20 These three methods were identified by the Applicant, Intervenor and 21 Commission staff expert witnesses as being the most commonly used in the gas 22 industry in North America. All three methods indicated that BCGUL's current 23 rates are less than the allocated historical costs for residential customers in all 24 Divisions, although the revenue to cost ratios for Inland residential customers 25 were within 10 percent of the theoretical ideal of a one-to-one correspondence 26 between costs and revenues. In previous decisions the Commission has 27 accepted a 10 percent band as reasonable. Similarly, all three methods indicated 28 that Lower Mainland industrial customers were contributing revenues in excess 29 of the costs allocated to them. (emphasis added)

FEI believes that a 90 percent to 110 percent range of reasonableness for R:C ratios in rate
 design represented an established practice for the Commission at the time.

An example of a recent precedent (at that time) accepting the 90 percent to 110 percent range
of reasonableness is found in a 1991 reconsideration decision involving Ocelot Chemicals Inc.
and Pacific Northern Gas. The following quote is taken from page 38 of the Decision attached to
Order G-23-91 (dated February 27, 1991):



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 56

- Based on the above, the Commission finds the following:
- 2 1. The Commission accepts that firm rates should move as rapidly as 3 possible towards costs, modified by the zone of reasonableness whereby, 4 in the absence of compelling evidence to the contrary, a revenue to cost 5 ratio of 90 to 110 percent shall be seen as revenue cost equality. 6 Accordingly, the Commission orders that residential rates increase by five 7 percent and commercial rates by three percent per annum for three years 8 commencing in accord with the implementation date described in Section 9 6.2. (emphasis added)

10 The Commission made further findings with respect to COSA studies and range of 11 reasonableness at pp. 28-29:

- A cost of service study is a guide to determine whether the revenues generated by the rates charged to a particular class of customer are sufficient to cover the cost of serving that class of customer. As such, cost of service studies should reflect costs only. Other considerations, while important in determining fair, just and reasonable rates, should be included following a review of the cost of service study results.
- Given the above, the results of cost of service studies should be seen as a tool to
 be used in the setting of fair, just and reasonable rates. They are not, in and of
 themselves, fair, just and reasonable rates.
- 21 The Commission is also cognizant of the considerable reliance upon judgement 22 involved in the undertaking of a cost of service study. Although judgement is 23 required in lesser amounts to determine the specific component of the total cost 24 of service and functionalization of costs, significant judgement is required to 25 classify costs between capacity, commodity and customer components. Even 26 greater judgement is required in determining the appropriate method to allocate 27 these costs amongst rate classes. For example, compressor costs have been 28 allocated 100 percent to capacity even though annual usage contributes to a 29 decreased service life. Similarly, different classes of customers impose different 30 levels of risk on the utility, but quantifying the appropriate cost differential is not 31 attempted in these studies. Finally, there are benefits attributable to serving 32 certain classes of customers but these, too, have not been included as an offset 33 against costs within the study as they are not easily quantified.
- Therefore, even as a tool for indicating the level of costs attributable to serving a particular class of customer, cost of service studies must be viewed as an indicator only, of the sufficiency or insufficiency of rates to cover a particular set of costs. Given the imprecision inherent in cost of service studies in general, and in particular the studies in issue, the Commission believes that as long as revenues from a particular class of service and costs allocated to that class of



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 57

service do not differ by more than 10 percent, there is no compelling evidence to
 determine that the cost of service results indicate rate restructuring is required.

The Commission Decision in the Columbia Natural Gas Rate Design and Fording Coal Ltd.
Complaint May 17, 1988, pp. 52-53 found:

- 5 The Commission is not, <u>however</u>, <u>convinced of the wisdom of embracing a</u> 6 <u>targetted (sic) revenue to cost ratio for any class</u>. This imputes an absolute 7 <u>standard of correctness regarding revenue to cost ratios which is misplaced for</u> 8 <u>several reasons</u>:
- 9
 1. Cost, especially in regard to joint use facilities, is not a precise concept.
 10
 11 Using either the dd or the ps allocation methodology leads, for example,
 11 to a Fording net revenue to cost ratio range of some 35 percent (92
 12 percent to 127 percent). Thus the notion of "true cost" in this situation is
 13 somewhat misleading.
- 14
 2. Even for a given allocation methodology (dd or ps), there can be considerable variation in determining costs due to judgment and the various other refinements that can be used. During the Inland rate design hearing, the evidence indicated that costs could be out by as much as 10 percent. (See Inland Rate Design Hearing transcripts pp. 794, 6584, 6593-9).
- 203. Depreciation can be vulnerable to considerations other than physical life21of property.
- 224. Revenues can vary significantly from year to year, especially for industrial23gas use which is dependent on volatile international or national economic24conditions.
- 5. The revenue to cost ratio also varies depending on whether or not the
 FACOS is prorated to revenues or revenues are prorated to the FACOS.
- 6. Historically, the gross revenue to cost ratio was used for comparative purposes. With the advent of direct gas purchasing, bypass and the segmented gas pricing policy, gross revenue to cost ratios have largely lost their meaning. Yet, net revenue to cost ratios are unstable in the sense that the biggest share of total costs is due to gas costs. The difference is margin or non-gas costs which result in exaggerated divergence of the revenue to cost ratio from unity. (emphasis added)

Although the Commission's reasons for acceptance of the 90 percent to 110 percent range of reasonableness in the 1993 Phase B Rate Design Decision were not directly stated, FEI believes it is reasonable to expect that the findings on range of reasonableness in then-current decisions, such as the 1988 Columbia Natural Gas – Fording Coal decision and the 1991 PNG-



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 58

- 1 Ocelot Chemicals decision (both cited above), were in view. The Commission's findings in
- 2 those decisions continue to be applicable today. Please refer also to the discussion on the
- 3 range of reasonableness in Section 6.5.1 of the Application.
- In addition to the references cited above, FEI's review of the Commission Decisions with respect to the range of reasonableness for natural gas utilities indicate that R:C ratios within a range of 90 percent to 110 percent have been consistently found acceptable. The relevant excerpts from these decisions are reviewed below.
- 8 The Centra Gas Fort St. John Inc. 1996 and 1997 Revenue Requirements and Rate Design
 9 Application Phase II Rate Design Decision also used threshold limits, at p. 3:
- 10 The Commission recognizes that judgment is involved in undertaking a cost of 11 service study. Considerable judgment is involved not only in classifying costs into 12 capacity, commodity and customer-related components, but also in determining 13 the appropriate method of allocating these charges among different rate classes. 14 In recognition of these inherent difficulties, the band of reasonableness for rate 15 restructuring adopted by the Commission to date, is the commonly accepted 16 band of plus or minus 10 percent around the ideal 1.0 benchmark ratio. In this 17 Decision the Commission applies these threshold limits with the commodity cost of gas excluded from consideration. 18
- 19 In that case some rates were still outside the band, but the Commission found them acceptable,20 at page 12:
- The Commission anticipates that the resulting revenue to cost ratio will still be well above the band of reasonableness of 0.90 to 1.10 adopted by the Commission. As a move towards this objective the Commission is satisfied that the proposed rates are justified.
- In the Centra Gas British Columbia Inc. 2002 Rate Design Application, the Commission
 indicated R:C ratios should tend toward the range of reasonableness, and also accepted a
 departure from the typical range on the basis of the immaturity of the utility, at pp. 40-41:
- For a financially healthy and mature utility, the Commission would expect the range of revenue to cost ratios across customer classes to tend toward 0.9 to 1.1, all other objectives being satisfied. The Commission finds that in the circumstances of an immature utility it would be unreasonable to limit revenue to cost ratios within a narrow range and thereby limit the consideration of other circumstances in the design of rates which meet the public interest. The Commission views Centra as an immature utility under its current circumstances.
- 35 The range of reasonableness of 90 to 110 percent was also used in:
- The FEI (then BC Gas) 1996 Rate Design, approved by the Commission in Order G-98-96 dated October 7, 1996.



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
M	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 59

1 2	•	The F 116-01	EI (then BC Gas) 2001 Rate Design, approved by the Commission in Order G-I dated November 7, 2001.
3 4 5	•	The F Comm Comm	El Common Rates, Amalgamation and Rate Design Application, decided by the ission in Order G-26-13 dated February 25, 2013 and reconsidered by the ission in Order G-21-14 dated February 26, 2014.
6 7			
8 9 10 11		14.2	Please explain whether the quality of FEI's customer data, load data and costing data has improved since the 1993 Phase B Rate Design Application.

13 Although the customer data has improved since the 1993 Phase B Rate Design Application, in 14 particular for General Firm Service, it has not sufficiently improved for the vast majority of FEI 15 customers, i.e., Residential and Commercial Sales service, to warrant changing the range of 16 reasonableness. General Firm Service (Rate Schedules 5 and 25) only accounts for 17 approximately 775 customers, whereas Residential and Commercial Sales service (Rate 18 Schedules 1, 2 and 3) represent approximately 995 thousand customers of the approximately 19 998 thousand total customers. In terms of 2017 annual forecast volumes, General Firm Service 20 accounts for approximately 15,840 TJ, whereas Residential and Commercial Sales service 21 accounts for approximately 121,480 TJ of the total forecast of 214,640 TJ⁶.

22 For Residential, Small Commercial and Large Commercial Sales Service, which make up the 23 majority of the customer demand, the available data is from monthly customer meter reads, 24 which occur in multiple cycles throughout the month. This is an improvement from 1993 when 25 these customers' meters were typically read every second month. However, even with these 26 improvements the necessary data to know what actual customer consumption is during peak 27 conditions is not available. As such, the load factors of individual customers, and even the 28 residential and commercial classes as a whole, continue to be estimates, meaning there is still a 29 measure of uncertainty in the demand allocators in the COSA.

Also, at the time of the 2001 Rate Design, while customers in the other Industrial rate schedules had demand meters and daily measurement data available, a large number of RS 5 customers' volume data were still based on monthly meter reads. This has now changed to all RS 5 measurement readings being available on a daily basis. This is also an improvement on the customer load data which allows for considering alternate methods of determining Daily Demand coupled with setting the Demand Charge to apply to General Firm Service customers.

The Company has made investments in tracking costing data when it switched its accounting and management systems to SAP, several years after the 1993 Rate Design, which tracks costs

⁶ FEI's Compliance Filing dated December 12, 2016, Appendix A, Section 11, Schedules 17 and 19



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 60

- 1 on an activity basis. The activities cover an array of capital and operating activities, including
- 2 those related to LNG assets and operations, Transmission assets and operations, and
- 3 Distribution assets and operations.



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
•	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 61

1	15.0	Reference	e: FEI COST OF SERVICE STUDY
2 3			Exhibit B-1, Section 6.5, p. 6-31; Section 6.5.2, p. 6-34; Appendix 6-4, Schedule 1;
4			Exhibit A2-2, Section 6.1.1, p. 28
5			Use of the R:C ratio versus the M:C ratio range of reasonableness
6		On page	6-31 of Exhibit B-1, FEI states:
7 8 9 10 11 12		T re th in ra so	the R:C ratios show whether the rates charged to each rate schedule adequately cover their allocated cost of service. For FEI's transportation rate schedules that have companion sales rate schedules (RS 23, RS 25 and RS 27) FEI aputes a cost of gas so that when the R:C ratios are calculated the final R:C to is on the same basis (delivery margin plus cost of gas) as for the sales rate chedules
13		On page	6-34 of Exhibit B-1, FEI states:
14 15 16 17 18		T cc va ei fr	ne margin to cost ratio is calculated by dividing the total delivery margin ollected from a rate schedule which includes Basic Charge, demand charge, olumetric Delivery Charge and administrative charge revenues, by the allocated mbedded delivery costs. Gas and storage and transport costs are excluded om both the numerator and denominator when calculating the M:C ratios.
19		On page 28 of Exhibit A2-2, Elenchus states:	
20 21 22 23 24		T w ra R R	ne definition of R:C and M:C ratios implies that the calculated R:C ratio range ould always be less than the calculated M:C ratio range. Specifically, the M:C atio would be less than the calculated R:C ratio for the same rate schedule if the :C ratio is less than 1.00 and the M:C ratio would be greater than the calculated :C ratio for the same rate schedule if the R:C ratio is greater than 1.
25 26		15.1 P al	lease explain, with calculations, how FEI imputes the cost of gas in order to be ole to calculate the R:C ratio for RS 23, RS 25 and RS 27.

27 28 **B**eener

28 **Response:**

FEI uses the cost of gas from the companion Sales rate schedule and applies it to the companion Transport rate schedule. The companion rate schedule's cost of gas includes the appropriate commodity and storage and transport costs, which is different for all three sales rate schedules.

The companion Sales and Transport Rate Schedules are RS 3 and RS 23, RS 5 and RS 25,
and RS 7 and RS 27.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 62

- 1 As an example, the imputed cost of gas for RS 23 is derived by using the RS 3 cost of gas,
- 2 divided by RS 3 volume and multiplied by RS 23 volume, as shown in the calculations in the
- 3 following table.

Line No.	Particulars	Amount	Reference
1	RS 3 Cost of Gas (\$000)	67,784	Appendix 6-2, Schedule 17, Line 6, Column 3
2	RS 3 Energy Volume Sold (TJ)	18,121.3	Appendix 6-2, Schedule 18, Line 6, Column 3
3	RS 23 Energy Volume Sold (TJ)	8,968.8	Appendix 6-2, Schedule 18, Line 7, Column 3
4	RS 23 Imputed Cost of Gas (\$000)	33,548	Line 1 / Line 2 x Line 3

- 4
- 5
- 6
- 7
- 8

10

- 15.2 Please explain, with calculations, how FEI determined (i) the cost of gas; and (ii) the R:C ratio for each of RS 22A and RS 22B large volume transportation service customers as presented in Table 6-18 on page 6-35 of Exhibit B-1.
- 11

12 Response:

13 The R:C ratios are calculated in the same manner as all other R:C ratios in the COSA using 14 revenue divided by allocated costs, where revenue equals delivery margin plus cost of gas, and 15 allocated costs equals allocated delivery costs plus cost of gas.

16 The cost of gas for RS 22A and 22B is equal to these rate schedules allocation of unaccounted 17 for (UAF) gas from FEI's test year revenue requirement. UAF refers to gas that is not 18 specifically accounted for in gas energy balance of receipts, deliveries, and operations use and 19 is associated with both the transmission and distribution system.

20 Since FEI moves gas across its delivery system for both Sales and Transport customers, both 21 Sales and Transport customers receive an allocation of UAF gas and that allocation is included 22 in the COSA for calculating the R:C ratio for RS 22A and RS 22B. Given that the UAF gas cost 23 is small, the R:C ratios are nearly equal to the M:C ratios for RS 22A and 22B.

- 24
- 25
- 26
- 27 15.3 Please take into consideration Schedule 1 (Fully Distributed Cost of Service 28 Allocation with R:C and M:C ratios) in Appendix 6-4 of Exhibit B-1 and explain 29 whether FEI is in agreement with Elenchus' statement shown in the preamble.
- 30



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 63

Based on FEI's explanation of R:C and M:C ratios and Elenchus' statement on

1 Response:

- FEI agrees with Elenchus' statement in the preamble. Since the same cost of gas amount is added to both the numerator and denominator in the R:C ratio for each rate schedule, Elenchus'
- 4 statement is a mathematical certainty.
- 5
- 6
- Ũ
- 7
- 8 9
- 10
- 10
- R:C and M:C ratios in the preamble, please explain if FEI considers that an R:C ratio range of 0.90 to 1.10 (+/- 10%) is equivalent to a M:C ratio range of the same 0.90 to 1.10 (+/- 10%).
- 11 same 0 12

15.4

13 <u>Response:</u>

No. R:C and M:C ratios are not mathematically equivalent, and an R:C ratio range of 0.90 to
1.10 (+/- 10 percent) is not equivalent to a M:C ratio range of the 0.90 to 1.10 (+/- 10 percent).
As described in Section 6.5.1 of the Application, FEI considers 0.90 – 1.10 to be the appropriate
range when examining R:C ratios. As described by Elenchus in the preamble, the equivalent
M:C ratio range would be wider than 0.90 – 1.10.

19

20

21

2215.5Based on FEI's explanation of R:C and M:C ratios and Elenchus' statement on23R:C and M:C ratios in the preamble, please explain if an R:C ratio range should24be greater than a M:C ratio range in order to be applied in an equivalent manner25during rate design.

26

27 Response:

To provide an equivalent basis for determining the need for rebalancing, the M:C ratio range should be wider than the R:C ratio range, not the reverse as the question suggests. As Elenchus' indicates, for any given R:C ratio, the M:C ratio will be further away from 100%, which means that a wider range of reasonableness should be applied to the M:C ratio.

FEI has included 2 tables below that demonstrate that for any R:C ratio not equal to 100 percent, the M:C ratio moves farther from 100 percent. For both of the following tables, the only input that is different is the Delivery Margin on Line 1. The R:C ratio includes cost of gas in both the revenue and cost side of the equation, because the cost of gas is a flow through. Customers pay exactly FEI's cost of gas so that revenue equals the cost. The M:C ratio excludes the cost of gas in both the numerator and denominator of the ratio calculation.



- 1 The first table calculates the R:C and M:C ratios when Delivery Margin is greater than the
- 2 allocated cost.

Line	Particulars	R:C	M:C	Description / Reference
1	Delivery Margin	750	750	Delivery Margin based on volume and existing rates
2	Cost of Gas	300		Cost of Gas based on volume and market based gas costs
3	Total Revenue	1050	750	Line 1 + Line 2
4				
5	Allocated Delivery Cost	700	700	Allocated Delivery Cost of Service from COSA Study
6	Cost of Gas	300		Line 2
7	Total Cost	1000	700	Line 5 + Line 6
8				
9	Ratio	105.0%	107.1%	Line 3 / Line 7

4 As can be seen on line 9, the equivalent M:C is further from 100 percent than the R:C.

5 The second table calculates the R:C and M:C ratios when Delivery Margin is less than the

6 allocated cost.

Line	Particulars	R:C	M:C	Description / Reference
1	Delivery Margin	650	650	Delivery Margin based on volume and existing rates
2	Cost of Gas	300		Cost of Gas based on volume and market based gas costs
3	Revenue	950	650	Line 1 + Line 2
4				
5	Allocated Delivery Cost of	700	700	Allocated Delivery Cost of Service from COSA Study
6	Cost of Gas	300		Line 2
7	Total	1000	700	Line 5 + Line 6
8				
9	Ratio	95.0%	92.9%	Line 3 / Line 7

- 8 As can be seen on line 9, the equivalent M:C is further from 100 percent than the R:C.
- 9 When the R:C is 100 percent, the M:C is also 100 percent.
- 10

7

- 11
- 12
- 1315.6Please complete the following table by placing the calculated figures in the cells14highlighted in yellow to show the impact of the inclusion of different magnitudes15of the cost of gas (gas and storage and transport costs) when calculating the R:C16ratio from M:C ratios of 90%. As per Schedule 1, Total Gas Cost Revenue (Row175) should be equal to Total Cost of Gas (Row 6).



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission)	Page 65

	Column 1		Column 2		Column 3		Column 4		Column 5		Column 6	(Column 7
		\square	RS 1 ⁽¹⁾	Sc	enario A (2)	Sce	enario B ⁽³⁾	Sc	enario C ⁽⁴⁾	Sc	enario D ⁽⁵⁾	Sce	enario E ⁽⁶⁾
			(000's)		(000's)		(000's)		(000's)		(000's)		(000's)
Row 1	Total Delivery Revenue Margin	\$	475,312	\$	450,000	\$	450,000	\$	450,000	\$	450,000	\$	450,000
Row 2	Allocated Cost of Service	\$	510,655	\$	500,000	\$	500,000	\$	500,000	\$	500,000	\$	500,000
Row 3	Margin to Cost (M:C) Ratio		93.1%		90.0%		90.0%		90.0%		90.0%		90.0%
Row 4													
Row 5	Total Gas Cost Revenue	\$	287,646	\$	450,000								
Row 6	Total Cost of Gas	\$	287,646	\$	450,000								
Row 7													
Row 8	Total Revenue (Row 1 + Row 5)	\$	762,958	\$	900,000								
Row 9	Total Cost of Service (Row 2 + Row 6)	\$	798,301	\$	950,000								
Row 10													
Row 11	Revenue to Cost (R:C) Ratio		95.6%		94.7%			1					
	Notes									-			
	(1) RS 1 figures taken from Exhibit B-1,	Appe	endix 6-4, Sche	dul	e 1								
	(2) Scenario A: Delivery Revenue Marg	in (50)%), Gas Cost I	Reve	enue (50%)								
	(3) Scenario B: Delivery Revenue Marg	in (60	%), Gas Cost F	Reve	enues (40%)								
	(4) Scenario C: Delivery Revenue Marg	in (70)%), Gas Cost F	Reve	enues (30%)								
	(5) Scenario D: Delivery Revenue Marg	in (40)%), Gas Cost I	Reve	enues (60%)								
	(6) Scenario E: Delivery Revenue Margi	in (30	%), Gas Cost F	Reve	enues (70%)								

3 Response:

4 As demonstrated in the following table, as the proportion of the Delivery Margin to Total

5 Revenue or Cost of Gas Revenue to Total Revenue changes by increments of 10 percent, while

6 holding the Delivery Revenue Margin (Row 1) and the Allocated Cost of Service (Row 2)

7 constant, the Revenue to Cost Ratio changes by 1 percent with each 10 percent change, from a

8 low of 93 percent (Scenario C) to a high of 97 percent (Scenario E).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 66

	Column 1	Column 2		Column 3	Cc	olumn 4	Column 5		Column 6		Column 7	
		RS 1 ⁽¹⁾	s	cenario A ⁽²⁾	Scer	nario B ⁽³⁾	Scenario C ⁽⁴⁾		Scenario D ⁽⁵⁾		Sce	enario E ⁽⁶⁾
		\$000's		(\$000's)	(\$	5000's)	(\$000's)		(\$000's)			(\$000's)
Row 1	Total Delivery Revenue Margin	\$ 475,312	2 \$	450,000	\$	450,000	\$	450,000	\$	450,000	\$	450,000
Row 2	Allocated Cost of Service	\$ 510,655	; \$	500,000	\$	500,000	\$	500,000	\$	500,000	\$	500,000
Row 3	Margin to Cost Ratio (M:C) Ratio	93.19	6	90.0%		90.0%		90.0%		90.0%		90.0%
Row 4												
Row 5	Total Gas Cost Revenue	\$ 287,646	; \$	450,000	\$	300,000	\$	192,857	\$	675,000	\$	1,050,000
Row 6	Total Cost of Gas	\$ 287,646	; \$	450,000	\$	300,000	\$	192,857	\$	675,000	\$	1,050,000
Row 7												
Row 8	Total Revenue (Row 1 + Row 5)	\$ 762,958	\$	900,000	\$	750,000	\$	642,857	\$	1,125,000	\$	1,500,000
Row 9	Total Cost of Service (Row 2 + Row 6)	\$ 798,301	. \$	950,000	\$	800,000	\$	692,857	\$	1,175,000	\$	1,550,000
Row 10												
Row 11	Revenue to Cost (R:C) Ratio	95.69	6	94.7%		93.8%		92.8%		95.7%		96.8%
	Notes											
	(1) RS 1 figures taken from Exhibit B-1,	RS 1 figures taken from Exhibit B-1, Appendix 6-4, Schedule 1										
	(2) Scenario A: Delivery Revenue Marg	in (50%), G	ost Revenue									
	(3) Scenario B: Delivery Revenue Marg	in (60%), Ga	as Co	ost Revenue (40%)							
	(4) Scenario C: Delivery Revenue Marg	in (70%), G	as Co	ost Revenue (30%)							
	(5) Scenario D: Delivery Revenue Marg	in (40%), G	as Co	ost Revenue	(60%)							
	(6) Scenario E: Delivery Revenue Marg	in (30%), Ga	is Co	ost Revenue (70%)							

15.7 Please complete the following table by placing the calculated figures in the cells highlighted in yellow to show the impact of the inclusion of different magnitudes of the cost of gas (gas and storage and transport costs) when calculating the R:C ratio from M:C ratios of 110%. As per Schedule 1, Total Gas Cost Revenue (Row 5) should be equal to Total Cost of Gas (Row 6).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission)	Page 67

	Column 1		Column 2		Column 3		Column 4		Column 5	Column 6			Column 7
			RS 2 ⁽¹⁾ (000's)	Sc	enario A ⁽²⁾ (000's)	Sc	enario B ⁽³⁾ (000's)	Sc	enario C ⁽⁴⁾ (000's)	Sc	enario D ⁽⁵⁾ (000's)	So	enario E ⁽⁶⁾ (000's)
Row 1	Total Delivery Revenue Margin	\$	133,094	\$	143,000	\$	550,000	\$	550,000	\$	550,000	\$	550,000
Row 2	Allocated Cost of Service	\$	129,862	\$	130,000	\$	500,000	\$	500,000	\$	500,000	\$	500,000
Row 3	Margin to Cost (M:C) Ratio		102.5%		110.0%		110.0%		110.0%		110.0%		110.0%
Row 4													
Row 5	Total Gas Cost Revenue	\$	111,133	\$	143,000								
Row 6	Total Cost of Gas	\$	111,133	\$	143,000								
Row 7													
Row 8	Total Revenue (Row 1 + Row 5)	\$	244,227	\$	286,000								
Row 9	Total Cost of Service (Row 2 + Row 6)	\$	240,995	\$	273,000								
Row 10													
Row 11	Revenue to Cost (R:C) Ratio		101.3%		104.8%								
	Notes			_						-			
	(1) RS 2 figures taken from Exhibit B-1	. Appe	endix 6-4, Sche	dule	e 1								
	(2) Scenario A: Delivery Revenue Mar	gin (50	%), Gas Cost R	ever	nue (50%)								
	(3) Scenario B: Delivery Revenue Mar	zin (60	%), Gas Cost R	ever	nues (40%)								
	(4) Scenario C: Delivery Revenue Marg	zin (70	%), Gas Cost R	ever	nues (30%)								
	(5) Scenario D: Delivery Revenue Mar	gin (40)%), Gas Cost F	eve	nues (60%)								
	(6) Scenario E: Delivery Revenue Mar	zin (30	%), Gas Cost R	ever	nues (70%)								

3 Response:

Similar to the results in response to BCUC-FEI IR 1.15.6, the change in the revenue to cost ratio is 1 percent with each 10 percent increment in the proportion of delivery margin revenue to total revenue - the lowest being Scenario E at 103 percent and the highest being Scenario C at 107 percent. Although the Delivery Margin and Allocated Cost of Service (Row 2) change significantly from Scenario A to Scenario B, as long as the Delivery Margin share of the total revenue remains unchanged, the change in the Revenue to Cost Ratio is only 1 percent with

10 each 10 percent change in the share of total revenue (i.e., 104.8 percent to 105.8 percent).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 68

	Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7
		RS 2 ⁽¹⁾	Scenario A ⁽²⁾	Scenario B ⁽³⁾	Scenario C ⁽⁴⁾	Scenario D ⁽⁵⁾	Scenario E ⁽⁶⁾
		\$000's	(\$000's)	(\$000's)	(\$000's)	(\$000's)	(\$000's)
Row 1	Total Delivery Revenue Margin	\$ 133,094	\$ 143,000	\$ 550,000	\$ 550,000	\$ 550,000	\$ 550,000
Row 2	Allocated Cost of Service	\$ 129,862	\$ 130,000	\$ 500,000	\$ 500,000	\$ 500,000	\$ 500,000
Row 3	Margin to Cost Ratio (M:C) Ratio	102.5%	110.0%	110.0%	110.0%	110.0%	110.0%
Row 4							
Row 5	Total Gas Cost Revenue	\$ 111,133	\$ 143,000	\$ 366,667	\$ 235,714	\$ 825,000	\$ 1,283,333
Row 6	Total Cost of Gas	\$ 111,133	\$ 143,000	\$ 366,667	\$ 235,714	\$ 825,000	\$ 1,283,333
Row 7							
Row 8	Total Revenue (Row 1 + Row 5)	\$ 244,227	\$ 286,000	\$ 916,667	\$ 785,714	\$ 1,375,000	\$ 1,833,333
Row 9	Total Cost of Service (Row 2 + Row 6)	\$ 240,995	\$ 273,000	\$ 866,667	\$ 735,714	\$ 1,325,000	\$ 1,783,333
Row 10							
Row 11	Revenue to Cost (R:C) Ratio	101.3%	104.8%	105.8%	106.8%	103.8%	102.8%
	Notes						
	(1) RS 1 figures taken from Exhibit B-1,	Appendix 6-					
	(2) Scenario A: Delivery Revenue Marg	gin (50%), Gas	Cost Revenue	(50%)			
	(3) Scenario B: Delivery Revenue Marg	in (60%), Gas	Cost Revenue	(40%)			
	(4) Scenario C: Delivery Revenue Marg	in (70%), Gas	Cost Revenue				
	(5) Scenario D: Delivery Revenue Marg	gin (40%), Gas	Cost Revenue	(60%)			
	(6) Scenario E: Delivery Revenue Marg	in (30%), Gas	Cost Revenue	(70%)			



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 69

1	F.	CHAPTER 7	- RATE DESIGN FOR RESIDENTIAL CUSTOMERS
2	16.0	Reference:	RATE DESIGN FOR RESIDENTIAL CUSTOMERS
3			Exhibit B-1, Section 7.2.2, p. 7-4;
4 5			BC Hydro 2015 Rate Design Application, Exhibit B-23, BCUC IR 2.174.2;
6 7			FEU 2014 LTRP, Exhibit B-1, pp. 20, 21, Order G-189-14 and Decision dated December 3, 2014, p. 38
8			Cost competitiveness
9 10		FEI states on use for space	page 7-4 of the Application that there is an increasing share of electricity heating for residential customers and domestic water heating.
11		BC Hydro sta	ted in response to BCUC IR 2.174.2 in the 2015 Rate Design Application:
12 13 14 15 16 17 18 19 20 21		There compe indivic descri gas is on cur up fro higher advan influer	are a number of factors to consider when comparing the cost etitiveness between natural gas and electric space and water heating and lual consumers may put more or less weight on certain factors. As bed in the preamble to this IR the FEU 2014 LTRP identified that natural less expensive for space and water heating on an energy cost basis based rent market pricing, however, natural gas systems can carry a much higher nt cost than electric baseboards. BC Hydro agrees with FEU's view that the upfront capital cost of natural gas end-use applications erodes the cost tage of natural gas compared to electricity and plays an important role in noing customer energy choice for space and water heating.
22		The Commiss	sion's decision on the FEU 2014 LTRP states on page 38:
23 24 25 26		Consis Comm compe should	stent with the Commission's determination in Order G-120-11, the hission Panel finds that (i) the FEU's objective of maintaining the etitiveness of natural gas with other energy sources is inappropriate and a not be included in a future PRMP
27 28 29		In the 2014 Hydro Step 1 cost differenc	FEU LTRP Application, FEU provided a comparison of gas rates to BC and Step 2 electricity rates (Figure 2-5, p. 20), and a comparison of the e for space and water heating – natural gas vs. electricity (p. 21, Table 2-1).
30 31 32 33		16.1 Please Applic and (ii	e update Figure 2-5 and Table 2-1 provided in the 2014 FEU LTRP ation using BC Hydro's current rates and: (i) FEI's existing residential rate,) FEI's proposed residential rate.

Please refer to the figure and table below, which update Figure 2-5 and Table 2-1 provided in
 the 2014 FEU LTRP Application with BC Hydro's current Rate Schedule 1101 Residential rates



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 70

- 1 effective April 1, 2017, and FEI's current RS 1 Residential rates effective January 1, 2017. FEI
- 2 has also updated the capital and maintenance costs in Table 2-1 with more recent information
- 3 that was filed in its Application for Common Equity Component and Return on Equity.
- 4 As set out in the Application, FEI is not requesting Commission approval of Residential rates
- 5 effective June 1, 2018 (except for approval of a Basic Charge per Day of \$0.4085 per day), but
- 6 rather a specific amount of change to the Delivery Charge per GJ that will be in place at the time
- the rates are implemented. Therefore, FEI is not able to include specific proposed June 1, 2018
- 8 rates in the figure and table below.

9 FEI Residential Annual Natural Gas Rates Compared with BC Hydro Residential Electric Rates



10

11

Capital Cost Difference for Space and Water Heating – Natural Gas vs. Electricity^{7,8}

	Space Heating	Water Heating
Capital costs for natural gas	\$9,000	\$2,000
Capital costs for electricity	\$4,435	\$1 ,000
Upfront capital cost premium for natural gas compared to electricity	\$4,565	\$1,000
Annual difference in capital costs ¹	\$422	\$116
Annual maintenance costs	\$100	\$0
Total annual difference in capital and maintenance costs	\$522	\$116
Energy consumption per year (GJ)	38	22
Difference in cost between natural gas and electricity over measurable life (\$/GJ)	\$13.84	\$5.25

¹ Represents the difference in capital costs per year, assuming a stream of equal annual payments with an interest rate of 6% and a measureable life of 18 years for a space heating furnace and 13 years for a hot water tank.

⁷ Analysis based on Tables C-6 and C-7 of Appendix C of the FEI Application for its Common Equity Component and Return on Equity for 2016.

⁸ Assumptions based on the new construction of a home in the Lower Mainland (Medium Size Dwelling at approximately 3,000 square feet).



FortisBC Energy Inc. (FEI or the Company) Submission Date: 2016 Rate Design Application (the Application) June 9, 2017 Response to British Columbia Utilities Commission (BCUC or the Commission) Page 71 Information Request (IR) No. 1

1	17.0	Refe	rence: F		IGN FOR R	ESIDENTI	AL CUSTO	MERS				
2			E	Exhibit B-1	, Section 7	.2.3, pp. 7-	5 to 7-6					
3			F	Residentia	consumpt	ion patter	n					
		•										
4		On pa	age 7-5 of Exhibit B-1, FEI states:									
5 6 7			As shown in Figure 7-5 below, the 2015 residential annual consumption distribution forms a bell curve. There is a slight skew to the right relative to the mean annual consumption which is estimated at 81 GJ/year excluding outliers.									
8 9 10		17.1	Please consum	provide t ption.	he standa	rd deviatio	on for the	e 2015 re	sidential a	annual		
11	Respo	onse:										
12 13	The standard deviation for the 2015 residential annual consumption per customer as depicted in Figure 7-5, which excludes outliers, is approximately 44 GJ/year.											
14 15												
16 17 18 19 20		17.2	Please breakdo normaliz in the ta	complete t wn of the i zed consum ble and the	he following number of r nption from notes belo	g table, to esidential 2011 to 20 w the table	the best c customers f 15. Please	of your abil for different use the incl	lity, to prov levels of a rements pro	vide a annual ovided		
			Number of FEI residential customers ⁽¹⁾ with an annual normalized consumption of:						Total Residential			
	Yea	Year	0 - 10 GJ	11 - 20 GJ	21 - 30 GJ		231 - 240 GJ	240 - 250 GJ	> 250 GJ	Customers		
	20:	15								886,169		
	20:	14								873,661		
	20:	13								863,189		

Notes:

2012

2011

(1) Use increments of 10 GJ (2) Actual residential customer totals taken from Exhibit B-2, Appendix A2, Section 2, p. 2 of the FEI Annual Review for 2017 Rates proceeding

21 22

23 **Response:**

The requested data is provided below. Note that the table is split into three sections to 24 accommodate all of the consumption ranges. 25

854,050

860,403
FORTIS BO	С
-----------	---

Response to British Columbia Utilities Commission (BCUC or the Commission) Page 72	BC [™]	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Information Request (IR) No. 1		Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 72

Year	0-10	10-20	20-30	30-40	40-50	50-60	60-70	70-80
2015	26,037	32,321	40,947	52,592	66,074	79,363	88,191	89,693
2014	26,368	30,991	37,739	49,266	62,545	76,296	86,084	89,247
2013	25,615	32,164	39,343	50,472	63,469	77,422	87,066	89,216
2012	25,111	31,161	36,618	46,474	57,730	71,428	81,255	85,952
2011	25,001	30,508	35,373	44,797	56,492	69,365	80,131	84,228

80-90	90-100	100-11	10	110-	120	120	0-130	13	30-140	140-15	0	150-160	160-170
82,720	71,000) 58,0)47	45	5,227		35,090		26,486	20,3	99	15,499	12,110
83,537	72,313	3 58,9	920	46	6,691		35,787		27,237	20,9	59	15,940	12,194
82,214	70,516	5 56,8	333	44	l,375		34,110		25,852	19,7	98	14,959	11,563
82,648	72,834	4 60,6	573	47	7,955		37,214		27,952	21,3	28	16,278	12,301
81,295	72,440	0 60,4	105	48	3,443		37,841		28,876	22,1	27	16,901	13,108
170-180	180-190	190-200	20	0-210	210-	-220	220-2	30	230-240	240-2	50	>250	Total

170-180	180-190	190-200	200-210	210-220	220-230	230-240	240-250	>250	Total
9,347	7,464	5,786	4,472	3,469	2,666	2,102	1,674	304	879,080
9,488	7,657	5,776	4,442	3,562	2,633	2,153	1,638	23	869,486
8,923	7,045	5,227	4,190	3,175	2,465	1,878	886	-	858,776
9,650	7,351	5,712	4,455	3,452	2,608	2,006	1,494	-	851,640
10,059	7,608	6,131	4,770	3,569	2,784	2,186	1,711	355	846,504

Note that the customer totals do not match those in the preamble but do reconcile to the
numbers included in Figure 7-5. As noted on page 7-5 of the Application, Figure 7-5 excludes
outliers, defined as those data points beyond the 99 percentile.

- 17.2.1 If FEI is unable to respond to the previous question, please explain why and explain the cost, time and effort in person-hours that is required to respond to this information request.

Response:

15 FEI was able to respond to the question. Please refer to the response to BCUC-FEI IR 1.17.2.

- 19 17.2.2 Please provide copies of Figure 7-5 showing the:
- i. 2014 Residential Normalized Consumption Distribution
- 22 ii. 2013 Residential Normalized Consumption Distribution



2

	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
C	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 73

- iii. 2012 Residential Normalized Consumption Distribution
 - iv. 2011 Residential Normalized Consumption Distribution

3 4 <u>Response:</u>

5 The requested histograms are provided below.





TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 74







FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 75



- 17.3 Please describe and compare the following attributes for (i) customers consuming equal to or less than 20 GJ/year; (ii) customers consuming from 70 to 90 GJ/year; and (iii) customers consuming greater than 140 GJ/year:
 - end uses (furnaces, boilers, domestic water heaters, and fireplaces, range and BBQ);
- 10 ii. typical load factor;
 - iii. price elasticity estimate as it relates to the variable charge; and
 - iv. any significant non-energy benefits (for example, environmental for BBQ use, customer satisfaction for cooking etc.)

Response:

16 FEI provides the following answers:



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 76

- 1 (i) End uses (furnaces, boilers, domestic water heaters, and fireplaces, ranges and BBQs): 2 FEI does not have any record of the actual appliances used by its existing customers. 3 Nevertheless, the results of the 2012 Residential End-Use survey (REUS) can be used 4 to provide a general response to this question. The 2012 REUS indicates that the 5 majority of customers consuming less than 20 GJ/year use electricity as their primary 6 heating fuel source, with electric baseboards as the most used heating method. Almost 7 half of these customers use natural gas fireplaces as their secondary space heating 8 method. Similarly, electricity is the main fuel source for water heating in this group. 9 Additionally, approximately 15 percent have natural gas ranges and 30 percent of these 10 customers have natural gas BBQs.
- (ii) The REUS shows that natural gas is the primary source of space and water heating for customers consuming between 70 and 90 GJ/year. The natural gas furnace is the primary space heating method for this group. Approximately 20% of customers in this group have natural gas ranges and natural gas BBQs.
- (iii) The results for customers consuming more than 140 GJ/year is similar to the customers consuming between 70 and 90 GJ/year, as the majority of customers in this group use natural gas as their primary space and water heating fuel source. Similarly, natural gas furnaces are the primary space heating method. More than 25 percent of customers in this group have natural gas ranges and approximately 30 percent use natural gas BBQs as well.
- 21 (iv) Typical load factor: Figure 7-8 of the Application provides a scatter plot for the estimated 22 load factor for RS 1 customers and their respective annual consumption. This figure 23 demonstrates that customers consuming less than 20 GJ/year can have both high and 24 low load factors indicating that there is no "typical" load factor associated with this group 25 of residential customers. The customers consuming between 70 to 90 GJ per year also 26 have a broad range of load factors with the majority of these customers having load 27 factors between 20 to 40 percent. For customers consuming more than 140 GJ, the 28 range of estimated load factors is slightly smaller and typically around 25 to 35 percent.
- (v) Price elasticity estimate as it relates to the variable charge: The requested information is
 not available. The price elasticity studies are typically at a higher level of aggregation
 (residential, commercial and industrial) and do not separate customers based on
 consumption.
- (vi) Any significant non-energy benefit (for example, environmental for BBQ use, customer
 satisfaction for cooking): There is no clear relationship between non-energy benefits of
 natural gas, such as customer preference and satisfaction for cooking with natural gas,
 and the individual customer's consumption level.
- 37



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Ċ	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 77

18.0	Refere	ence:	RATE DESIGN FOR RESIDENTIAL CUSTOMERS
			Exhibit B-1, Appendix 7-1, p. 1;
			BC Gas Utility Ltd. Phase B Rate Design Application, Order G-101-93 and Decision dated October 25, 1993, pp. 24–25
			Low Use Residential Customers
	Appen dated	idix 7-1 July 16,	of Exhibit B-1 contains the 2012 FEU Residential End-Use Study (REUS) 2014. Page 1 of 2012 FEU REUS states:
		Use ra declinii and 4 ^o somew	ates (weather normalized gas consumption per-household) have been ng across FEU's regions since 1999. Use rates are down 24% since 1999 % since the last REUS (2008). The decline since 2008 is understated what due to a change in the use rate calculation method for 2012.
	Page	1 of the	2012 REUS then provides reasons that result in the declining use rates.
	18.1	Please regions	explain if FEI considers that the trend of declining use rates across FEU's s has continued since the 2012 FEU REUS.
Resp	onse:		
The tr 2012. norma graph despit indica contin	rend of Pleas alized U indicate re occas te that i ue.	declining se refer PC". Fig es that l sional y n the m	g residential use rates across FEI's service territory has continued since to Figure 7-6 of the Application entitled "FEI's historical residential gure 7-6 provides the residential UPC rates from 2006 until 2015. This JPC has decreased from 87.6 GJ in 2012 to 84.4 GJ in 2015. In addition, ear over year UPC increases, FEI's long-term resource plan forecasts edium and long-term, the declining residential use per customer trend will
	18.0 Respondent The tr 2012. normal graph despit indical contin	18.0 Reference Appendated Page 1 18.1 Response: The trend of 2012. Please normalized U graph indicate dated u despite occase indicate that i continue.	 18.0 Reference: Appendix 7-1 dated July 16, Use raddeclining and 49 somew Page 1 of the 18.1 Please regions Response: The trend of declining 2012. Please refer normalized UPC". Fig graph indicates that U despite occasional y indicate that in the m continue.

- 25
- 20
- 26 27
- 28The Commission stated in the BC Gas Utility Ltd. Phase B Rate Design Application29Decision (Order G-101-93), pp. 24, 25:
- When trying to meet the objective of aligning marginal rates with LRIC, variables that can be adjusted are (1) the basic charge and (2) the intra-marginal rate (this would be the summer rate in a seasonal rate design). Thus, changes to the basic charge may be required, simply to ensure that the Utility recovers its costs, and these changes may require decreases rather than increases, even though the FDC studies indicate that the customer related costs significantly exceed the current basic charge.



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
C	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 78

1 2		The Commission is therefore unwilling at this time to accept the full increase in the basic charge proposed by BCGUL
3 4	18.2	Does FEI consider that increasing the residential fixed charge could result in low- use residential customers leaving FEI's system or being reluctant to connect to
5		FEI's system? Please explain your response.

7 **Response:**

6

8 In general, residential customers are known to have low elasticity of demand, meaning that their 9 demand for natural gas does not significantly change with changes in price levels. Therefore, it 10 is unlikely that a small increase of 5 percent in Basic Charge along with a corresponding 11 decrease in volumetric charge will lead to a material decrease in number of customers. Previous 12 increases in FEI's Basic Charge, such as the 15 percent increase in the Basic Charge from the 13 2001 rate design decision, did not lead to a material decrease in the number of customers. 14 Nevertheless, if the magnitude of increase in fixed charges is significant, low-use customers 15 such as those with convenience load (for instance, customers who use natural gas only for 16 fireplaces, BBQs or dryers) may decide to leave the system.

With regard to new customers, developers and builders are the primary decision-makers for attaching to the natural gas distribution network and small increases to fixed charges are unlikely to have a negative impact on their decisions. An increase to the Basic Charge could potentially have a positive impact on new connections, by increasing the economic viability of main extensions under the Main Extension test (MX test) as discussed in the response to BCUC-FEI IR 1.6.2.

- 23
- 24
- 25
- 2618.3Please explain, with calculations, if the (i) existing rate design; and (ii) the27proposed rate design results in low-use customers paying less than their28customer-related costs according to the COSA?
- 29

30 **Response:**

The table below provides the monthly bill amount as well as the percentage of customer attributed cost recovered from the monthly average bill amount for various low consumption levels under the two requested scenarios: (i) existing rate design (no change in Basic Charge) and (ii) proposed rate design (i.e., a 5 percent increase in the Basic Charge and offsetting decrease in the volumetric charge).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:	
2016 Rate Design Application (the Application)	June 9, 2017	
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 79	

Annual	Average	Existing rate design (no change in basic charge)			Proposed rate design		
Consumption (GJ)	Monthly Consumption (GJ)	Mor ar	ithly bill nount	% of customer related cost recovery	Mor a	nthly bill mount	% of customer related cost recovery
0	0	\$	11.8	44%	\$	12.43	46%
6	0.5	\$	14.3	53%	\$	14.81	55%
12	1	\$	16.7	62%	\$	17.18	63%
18	1.5	\$	19.1	70%	\$	19.55	72%
24	2	\$	21.5	79%	\$	21.93	81%
30	2.5	\$	23.9	88%	\$	24.30	90%
36	3.0	\$	26.3	97%	\$	26.67	98%

2 As can be seen in the table above, in the extreme case of a customer with zero annual use, the 3 monthly bill amount will only recover 44 percent and 46 percent of customer-related costs for 4 the existing and proposed rate designs, respectively. With increases in consumption levels, the 5 percentage of recovery gradually increases. For instance, at 24 GJ annual consumption, the 6 average monthly bill amounts will recover 79 percent and 81 percent of average customer-7 related costs for the existing and proposed rate designs, respectively. At annual consumptions 8 above 36 GJ, the percentage of recovery will be close to 100 percent. This analysis shows that 9 the proposed rate design with a one-time revenue-neutral 5 percent increase in the Basic 10 Charge improves the cost recovery by approximately 2 percent.

- 11
- 12
- 13
- 14 15 16
- 18.3.1 Please calculate the annual residential consumption in GJ that is required for FEI to recover customer-related costs as determined by the COSA study through:
- i. The existing rate design; and
- 18 19

- ii. The proposed rate design.
- 20 Response:
- This response also addresses BCUC-FEI IR 1.18.3.2, which requests the same information but for the total of both the customer-related and demand-related costs.
- 23 The minimum annual consumption required can be calculated based on the following formula:



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 80

Minimum annual consumption = (Annual average cost⁹ - Daily Basic Charge*365.25) /
 Volumetric charge

3 Based on the above formula and the average customer-related and total fixed costs (sum of

4 demand and customer-related costs) provided in Table 7-5 of the Application, the annual

5 consumption thresholds required to recover these costs under the two requested scenarios are

6 as follows:

Description		sting rate design	Proposed rate design	
Daily Basic Charge	\$	0.3890	\$	0.4085
Volumetric Charge	\$	4.832	\$	4.746
Annual consumption to recover customer-related costs (GJ)		38		37
Annual consumption to recover sum of demand and customer- related costs (GJ)		80		80

7

8 As can be seen from the table, at the current Basic Charge level of \$0.389 per day, customers

9 consuming less than approximately 38 GJ per year do not pay all of their allocated customer-

10 related costs. If the Basic Charge is increased by 5% and the volumetric charge decreased by

11 an offsetting amount, the minimum annual consumption under which the customer does not pay

12 the allocated customer-related costs decreases to approximately 37 GJ per year.

13 Similarly, the minimum annual consumption required to recover total fixed costs in both 14 scenarios is close to FEI's average use at approximately 80 GJ. The approximately equal 15 consumption level calculated in both scenarios reflects the fact that for an average use 16 customer the proposed increase in the Basic Charge is revenue neutral.

- 17
- 18

- 2018.3.2Please calculate the annual residential consumption in GJ that is21required for FEI to recover both customer-related costs and demand-22related costs as determined by the COSA study through:
- 23

- i. The existing rate design; and
- ii. The proposed rate design.
- 24 25
- 26 **Response:**
- 27 Please refer to the response to BCUC-FEI IR 1.18.3.1.

⁹ Customer-related or sum of demand and customer related costs.

FORTIS BC

FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 81

- 1 2
- 2

4

5

6

7

18.4 If feasible, using the FEI COSA study, please unbundle the residential customer class into two segments: (i) consuming at or below 20 GJ/year, and (ii) consuming above 20 GJ/year. Please provide the R:C and M:C ratios of these two customer segments, and comment on whether FEI's analysis indicates that low-use residential customers could be set up as a separate rate class.

8 9

10 Response:

11 To respond to this question, in the COSA model FEI separated customers in RS 1 into two 12 segments, RS 1 > 20 GJ per year (RS 1A) and RS 1 <= 20 GJ per year (RS 1B). FEI assumed 13 that the customer weighting factors for both meters and services and administration and billing 14 are the same for RS 1A and RS 1B. This is a reasonable assumption but FEI recognizes that 15 the customer weighting factor for meters and services for RS 1B could be slightly less because 16 of the nature of the location of these customers in multi-family dwellings, which can have 17 multiple meters on one service line.

The R:C and M:C ratios for the two separated residential groups can be found in the followingtable.

Rate Schedule	R:C	M:C
RS 1A (Consumption > 20 GJ/yr)	96.6%	94.6%
RS 1B (Consumption <= 20 GJ/yr)	61.6%	57.0%

20

As can be seen in the table above, the R:C and M:C ratios for the RS 1B group are lower than that of RS 1A. The Basic Charge for the residential group collects approximately 45 percent of the customer and demand related costs; consequently, the balance of these costs must be recovered through the volumetric charge. With the lower volume customers consuming less than the average there is a smaller volume over which to collect the balance of the allocated costs resulting in lower R:C and M:C ratio.

While this exercise identifies that low consumers within a rate schedule pay less than their allocated cost, it is also true that customers consuming more than the average are paying more than their allocated costs. This is true for all rate schedules.

30 In the response to BCOAPO-FEI IR 1.7.4, FEI discusses the nature of customers and costs and

31 contends that every customer ultimately has a different cost to serve. When considering how to

32 resolve this perceived intra-class inequity other issues must be considered such as government

33 policy, administrative burden and customer impact.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 82

1 One way to solve much of the intra-class inequity might be to increase the Basic Charge to 2 collect more or nearly all of the customer related charges, but this would be contrary to 3 government policy as described in Section 5.4 of the Application regarding energy conservation

4 as it reduces the conservation price signal to consumers.

5 Another option could be to create a separate rate schedule with a different rate structure for customers consuming between 0 and 20 GJ annually. There are approximately 62,000 RS 1 6 7 customers that fall into this consumption range and approximately 824,000 customers that 8 consume more than 20 GJ annually. In this scenario, the customers from these two groups that 9 fall near the 20 GJ threshold would have to be reviewed annually (as is done with FEI's small 10 and large commercial customers) so that if the customer's consumption changed they could be 11 moved to the correct rate schedule. This process would be administratively costly and could 12 result in rate instability for customers and reduce customer satisfaction. In addition, creating a 13 separate rate schedule for low consumption customers would increase the rates and the annual 14 bills for these customers. The increase in rates may cause some of these customers to leave 15 the system, which would in turn increase rates for all non-bypass customers.

Finally, the under recovery from this group of customers costs all other RS 1 customers approximately \$0.19 per year or 0.02 percent on the annual bill. Considering this small cross subsidization, the similar characteristics of FEI's residential customers, government policy, rate stability and the administrative burden and the costs associated with separating residential customers into two classes, maintaining one residential rate class with the rates as proposed in the Application is a balanced solution.

22
23
24
25
18.4.1 If not feasible, please explain the costs, time and effort in person-hours that will be required in order to provide the information requested.
27
28 Response:
29 Please refer to the response to BCUC-FEI IR 1.18.4.
30



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 83

1	19.0	Refere	nce: RATE DESIGN FOR RESIDENTIAL CUSTOMERS
2			Exhibit B-1, Section 7.5.2, p. 7-18, Sections 7.8 and 7.8.1, p. 7-22;
3			2017 RIB Rate Report, pp. 6, 8, 9;
4 5			BC Hydro 2015 Rate Design Application, Order G-5-17 and Decision dated January 20, 2017, p. 14
6			Proposal and problem definition
7 8 9 10		FEI cor of Exhi energy constar	npares the existing residential rate to the proposed residential rate on page 7-22 bit B-1. FEI states on page 7-18 of Exhibit B-1: "alignment with government's conservation policy was the basis for the 2009 decision to hold the Basic Charge nt."
11		The Co	mmission states in the 2017 RIB Rate Report on page 6:
12 13 14 15 16 17 18			The RIB rates are conservation rates; that is, their purpose is to conserve energy or promote energy efficiency by providing a higher incentive, in the form of a higher rate for electricity purchased in the second tier, for higher-use customers to reduce consumption. Since it is not the purpose of the RIB rates to benefit any customers at the expense of other customers, this supports the Commission's view based on the R/C ratios that there is no undue discrimination in the RIB rate.
19		The Co	mmission states in the 2017 Decision on BC Hydro's 2015 RDA:
20 21 22 23 24 25			the Panel notes that one of the reasons BC Hydro maintains the RIB rate as status quo is because the rate structure appears to be achieving its overall objective of encouraging conservation through customer response to higher marginal prices at the Step 2 energy rate. In the Panel's view, in assessing the rate design proposals for the various customer classes it is important to consider the efficiency criterion in balance with other principles.
26 27 28 29 30 31 32	_	19.1	Please clearly identify the problem with the existing recovery of residential costs through fixed vs. variable charges that this Application is intended to address. If the problem relates to an efficiency concern, please identify the specific concern (i.e. how is it negatively affecting customer investment/consumption decisions), and if the problem relates to a fairness concern, please provide evidence that the existing level of cost recovery is outside of fairness norms.

33 Response:

The fixed vs. variable revenue recovery concern is mainly an intra-class fairness issue, which must be balanced with competing rate design issues or principles, such as cost recovery, bill impacts, revenue stability, or having rate structures that align with energy conservation objectives or other government policy objectives. As discussed below, FEI's proposed one-time



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 84

1 adjustment to the residential Basic Charge and offsetting decrease to the Delivery Charge 2 strikes an appropriate balance between these competing considerations.

3 The main objective in rate design is to improve the balance among various competing rate 4 design considerations. With the passage of time, this balance may be shifted towards one rate 5 design principle at the expense of others and therefore there might be a need to introduce 6 measures that can improve the situation. The 5 percent increase in the Basic Charge and 7 corresponding decrease in the volumetric Delivery Charge proposed by FEI for the residential 8 rate class is in line with this objective. The evidence provided in Section 7.5.1 of the Application 9 indicates that there is a need to improve the intra-rate schedule fairness within the residential 10 rate class so that the balance among various rate design considerations is improved. This will 11 also improve the balance of interests between low-use residential customers and medium and

12 high use customers.

13 As shown in Figure 7-9 of the Application, during the last eight years and compared to low use 14 customers, medium and high use customers have been bearing a greater share of delivery 15 margin increases, which has led to the intra-rate schedule imbalance. (The analysis shows that 16 within the 2009 to 2016 period, the delivery margin for customers with 25 GJ, 85 GJ, and 145 17 GJ annual consumption has increased by 16 percent, 30 percent, and 36 percent, respectively). 18 This is while all residential customers receive the same safe and reliable service irrespective of their consumption level. A one-time 5 percent increase in the Basic Charge and corresponding 19 20 decrease in the Delivery Charge will help to improve the imbalance in intra-rate schedule 21 fairness, and will not have any material impact on other rate design considerations such as rate 22 impact or government energy policy.

23 As explained on page 7-17 of the Application, the Commission previously approved increases in 24 FEI's Residential Basic Charge that were higher than the 5 percent increase proposed by FEI. 25 For example, as part of the 1996 NSA, the monthly Basic Charge was increased by 26 approximately 11 percent from \$6.32 to \$7.00. In the 2001 NSA, the monthly Basic Charge was 27 increased by 15 percent from \$8.66 to \$10.00. These increases were deemed to be fair by the 28 The proposed 5 percent increase in the Basic Charge and corresponding Commission. 29 decrease in the Delivery Charge is similarly fair and reasonable.

- 30 31 32 33 Please identify the specific changes in circumstances since 2009 which 19.1.1 34 have resulted in the creation of each identified problem. 35 36 **Response:** 37 Please refer to the response to BCUC-FEI IR 1.19.1.
- 38



SBC [™] -	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 85

1	20.0 Ref	erence:	RATE DESIGN FOR RESIDENTIAL CUSTOMERS
2 3			Exhibit B-1, Section 7.5.1, pp. 7-17 to 7-18, Section 7.8.1, pp. 7-23 to 7-25
4			Fixed and variable Cost Recovery
5	On	page 7-17	of Exhibit B-1, FEI states:
6 7 9 10 11 12		In the about and <u>misali</u> issue appro part o	e current residential rate structure, the current basic charge recovers 44% of the customer costs and only about 27% of the total of customer demand costs allocated to the residential rate schedule <u>The</u> <u>gnment between fixed costs and the Basic Charge has been a re-occurring</u> <u>in FEI's rate design proceedings</u> . The Commission has previously ved increases in the share of fixed costs recovered by fixed charges. As f the 1996 NSA, the monthly Basic Charge was increased by approximately rom \$6.32 to \$7.00 In the 2001 NSA, the monthly Basic Charge was again
13 14		increa	ised by an additional 15% from \$8.66 to \$10.00. [Emphasis added]
15 16	On whic	page 7-2: ch:	2 of Exhibit B-1, FEI recommends a rate design for residential customers
17 18 19 20 21		•	Improves the alignment between the fixed costs allocated to the residential rate schedule and the fixed charges recovered from residential customers by a one-time 5% increase to Basic Charge and corresponding decrease in the volumetric Delivery Charge. [Emphasis Added]
22	On	page 7-23	3 of Exhibit B-1, Section 7.8.1, FEI states:
23 24 25 26		Impler in the decrea GJ.	menting the proposed 5% increase in Basic Charge results in an increase daily Basic Charge from \$0.3890 to \$0.4085 per day and a corresponding ase in the volumetric Delivery Charge from the \$4.832 per GJ to \$4.746 per
27 28 29 30 31	20.2	Please the to is reco	e confirm, or otherwise explain, that 56% of the customer costs and 73% of tal of customer and demand costs allocated to the residential rate schedule overed through the variable charge(s).
30	Confirmed	•	
52	Commed.		
33 34			
35			



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 86

20.2 Please provide, for each year since 2009, the percentage of total residential delivery costs recovered through (i) fixed charges and (ii) variable charges.

3

4 <u>Response:</u>

- 5 FEI cannot provide the requested information for all years as it would require the creation of a
- 6 COSA study for each year. FEI did prepare a COSA study in 2012 using 2013 test year data,7 and can provide the requested percentages for that year.
- In its 2012 Common Rates, Amalgamation and Rate Design Application, FEI provided a COSA study that was an amalgamated COSA including Mainland (FEI), Vancouver Island (FEVI), Whistler (FEW) and Fort Nelson (FN). At that time, all four of these areas were separate utilities with separate rate structures. To provide the information requested in this question, FEI has utilized that COSA, but assumed that the Mainland rate structure was in place for this amalgamated entity (FEI + FEVI + FEW + FN) and has used this assumption to determine the following results.
- In the 2013 amalgamated COSA (which included Fort Nelson), the Basic Charge recovered about 42 percent of the customer-related costs, and about 28 percent of the total of customer and demand-related costs allocated to the residential rate schedule. The balance of the total allocated costs would be recovered through the volumetric charge. These percentages are reasonably close to the results of the current COSA.
- Another point of comparison available relates to the 2001-2002 period. In 2001 and before the Commission's decision to increase the share of fixed cost recovery in the Basic Charge, the Basic Charge recovered approximately 43 percent and 24 percent of customer-related and total customer-related and demand-related costs, respectively. After the Commission's decision to increase the Basic Charge by 15 percent, these percentages improved to approximately 50 percent and 27 percent, respectively.
- 26
- 27
- 28
- 29 20.3 Please explain if FEI's proposal for "a one-time 5% increase to Basic Charge and 30 corresponding decrease in the volumetric Delivery Charge" will result in a 31 misalignment between fixed costs and the Basic Charge over time as the 32 volumetric Delivery Charge is changed annually.
- 33

34 **Response:**

The proposed 5 percent increase to the Basic Charge and offsetting decrease in volumetric charge does not result in misalignment, but rather decreases the misalignment. However, it is correct that over time, as the delivery margin increases and the Basic Charge is held constant,



FortisBC Energy Inc. (FEI or the Company)	Submission Date:	
2016 Rate Design Application (the Application)	June 9, 2017	
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 87	

- the impact of the proposed improvement in alignment will gradually diminish. For this reason, it
 is important to review and potentially adjust the recovery of fixed costs from time to time.
- 3
 4
 5
 6 20.4 Please provide the increase in the daily Basic Charge and the corresponding decrease in the volumetric Delivery Charge from increasing the Basic Charge by 10 and 15 percent.

10 Response:

11 The requested information is provided in the table below.

	Title	COSA after Rebalancing	10% Increase in Basic Charge and offsetting Decrease in Delivery Charge	15% Increase in Basic Charge and offsetting Decrease in Delivery Charge
	Daily Basic Charge (\$/day)	0.3890	0.4279	0.4474
	Delivery Charge (\$/GJ)	4.832	4.661	4.575
12				
13				
14				

- 1520.4.1In the same format as Table 7-9 in Exhibit B-1, page 7-25, please16provide the bill impact of increasing the Basic Charge by 10 percent and1715 percent.
- 18
- 19 Response:
- 20 The following table provides the bill impact of increasing the Basic Charge by 10 percent.

	Annual Bill impact due to the 10% increase in Basic Charge				
Annual Consumption	Dollar Amount	Percentage of Total Bill			
0 GJ	\$14.0	10.0%			
40-45 GJ	\$7.0	1.4%			
60-65 GJ	\$4.0	0.5%			
80-85 GJ	\$0.0	0.0%			
100-105 GJ	\$(3.0)	-0.3%			
120-125 GJ	\$(7.0)	-0.6%			

21

22 The following table provides the bill impact of increasing the Basic Charge by 15 percent.



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 88

	Annual Bill impact due to the 15% increase in Basic Charge				
Annual Consumption	Dollar Amount	Percentage of Total Bill			
0 GJ	\$21.0	15.0%			
40-45 GJ	\$10.0	2.1%			
60-65 GJ	\$5.0	0.8%			
80-85 GJ	\$0.0	0.0%			
100-105 GJ	\$(5.0)	-0.5%			
120-125 GJ	\$(10.0)	-0.8%			

2

- 3
- 4 5

6

7

20.5 For the proposed FEI residential, please complete the table below in 5 GJ increments for the 0–30 GJ range and 10 GJ increments for the 31–140 GJ range. Also include fully a functional electronic spreadsheet for the data in the table.

	Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7	Column 8	Column 9
	Annual	Annual Revenue	Annual Fixed Cost		Difference as a %	Annual Revenue	Annual Variable		Difference as a %
	Annual	from Proposed	based on COSA	Difference	of Annual Fixed	from Proposed	Cost based on	Difference	of Annual
	consumption	Basic Charge	Results		Cost	Variable Charge	COSA Results		Variable Cost
		(a)	(b)	(c) = (a) - (b)	(c) / (b)	(d)	(e)	(f) = (d) - (e)	(f) / (e)
Row 1	0 - 5 GJ								
Row 2	6 - 10 GJ								
Row 3	11 - 15 GJ								
Row 4	16 - 20 GJ								
Row 5	21 - 25 GJ								
Row 6	26 - 30 GJ								
Row 7	31 - 40 GJ								
Row 8	41 - 50 GJ								
Row 9	51 - 60 GJ								
	131 - 140 GJ								

8

9

10 Response:

- 11 For clarity FEI has renamed the columns from the table provided in the following way:
- Column 3 renamed to Annual Customer Related Cost based on COSA Results;
- Column 5 renamed to Difference as a percent of Annual Customer Related Costs;
- Column 6 renamed to Annual Revenue from Proposed Volumetric Charge;
- Column 7 renamed to Total Annual Cost based on COSA Results to be recovered through Volumetric Charge; and



FortisBC Energy Inc. (FEI or the Company)	Submission Date:	
2016 Rate Design Application (the Application)	June 9, 2017	
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 89	

Column 8 renamed to Difference as a percent of costs to be recovered through
 Volumetric Charge.

It is important to note that both customer-related and demand-related costs are predominantly fixed. Of the total delivery costs, there are very few costs that are variable with consumption. Because FEI's costs are predominantly fixed each customer within a rate schedule is responsible for the same amount, and for this response FEI is describing this as the annual revenue responsibility of each customer. FEI assumes that the annual revenue from proposed Basic Charge (column 2) plus the annual revenue from proposed volumetric charge (column 6) sums to the annual revenue responsibility of each customer.

10 The annual revenue responsibility of each customer is calculated in the following manner. The 11 total RS 1 COSA allocated costs equal \$504,452 thousand¹⁰ multiplied by 94.4 percent M:C 12 ratio¹¹ equals \$476,203 thousand. This is the total annual revenue responsibility for all 13 customers in RS 1 and when divided by 886,652 RS 1 customers¹² the annual revenue 14 responsibility for each customer equals \$537.

Column 2 is populated using the Proposed Daily Basic Charge of \$0.4085 (as provided in Table
7-7) * 365.25.

To populate column 3 FEI has used the customer-related costs from the COSA. The customer
related costs of \$305,518 thousand¹³ multiplied by 94.4 percent M:C ratio¹⁴ equals \$288,409
thousand divided by 886,652 customers¹⁵ equals a customer related cost of \$325¹⁶ per
customer.

Column 6 is populated using the Proposed Delivery Charge of \$4.746/GJ (as provided in Table
 7-7) multiplied by consumption from column 1.

Column 7 is the annual revenue responsibility per customer of \$537 less recoveries from the proposed Basic Charge of \$149 from column 2 which equals \$388. This column represents the costs that need to be recovered through FEI's volumetric Delivery Charge.

26 The requested table is provided below.

¹⁰ Appendix 12, Schedule 4, Line 36, Rate 1.

¹¹ Appendix 12, Schedule 1, Line 32, Rate 1.

¹² Appendix 12, Schedule 7, Line 6, Rate 1.

¹³ Appendix 12, Schedule 4, Line 39, Rate 1.

¹⁴ Appendix 12, Schedule 1, Line 32, Rate 1.

¹⁵ Appendix 12, Schedule 7, Line 6, Rate 1.

¹⁶ Alternate calculation: \$27.10 per month (from Table 7-5) multiplied by 12 months.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 90

	Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7	Column 8	Column 9
	Annual Consumption (GJ)	Annual Revenue from Proposed Basic Charge	Annual Customer Related Cost based on COSA Results	Difference	Difference as a % of Annual Customer Related Costs	Annual Revenue from Proposed Volumetric Charge	Total Annual Cost based on COSA Results to be recovered through Volumetric Charge	Difference	Difference as a % of costs to be recovered through Volumetric Charge
		(a)	(b)	(c) = (a) - (b)	(c) / (b)	(d)	(e) = 537 - (a)	(f) = (d) - (e)	(f) / (e)
Row 1	5	149	325	(176)	-54%	24	388	(364)	-94%
Row 2	10	149	325	(176)	-54%	47	388	(340)	-88%
Row 3	15	149	325	(176)	-54%	71	388	(317)	-82%
Row 4	20	149	325	(176)	-54%	95	388	(293)	-76%
Row 5	25	149	325	(176)	-54%	119	388	(269)	-69%
Row 6	30	149	325	(176)	-54%	142	388	(245)	-63%
Row 7	40	149	325	(176)	-54%	190	388	(198)	-51%
Row 8	50	149	325	(176)	-54%	237	388	(151)	-39%
Row 9	60	149	325	(176)	-54%	285	388	(103)	-27%
Row 10	70	149	325	(176)	-54%	332	388	(56)	-14%
Row 11	80	149	325	(176)	-54%	380	388	(8)	-2%
Row 12	90	149	325	(176)	-54%	427	388	39	10%
Row 13	100	149	325	(176)	-54%	475	388	87	22%
Row 14	110	149	325	(176)	-54%	522	388	134	35%
Row 15	120	149	325	(176)	-54%	570	388	182	47%
Row 16	130	149	325	(176)	-54%	617	388	229	59%
Row 17	140	149	325	(176)	-54%	664	388	277	71%



4

5

	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
N	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 91

1 G. CHAPTER 8 – RATE DESIGN FOR COMMERCIAL CUSTOMERS

2 21.0 Reference: RATE DESIGN FOR COMMERCIAL CUSTOMERS

Exhibit B-1, Section 8.3, p. 8-11, 8-12; FEI Rate Schedule 2¹⁷; FEI Rate Schedule 3¹⁸

Economic crossover point

FEI states on page 8-11 of Exhibit B-1 that "The economic crossover point between RS
2 and RS 3 is the annual volume at which a customer would have the same annual total
cost whether served under either RS 2 or RS 3."

- 9 Table 8-3 on page 8-12 of Exhibit B-1 shows the economic crossover volume for RS 2 10 and RS 3 is 1,457 GJ.
- 11 The current rates for RS 2 and RS 3 are shown in FEI's gas tariff.

12 21.1 Please explain when the economic crossover point was last re-aligned, and what 13 the crossover point was set at.

14

15 **Response:**

16 The economic crossover point was last re-aligned in the 2001 Rate Design Application. At that 17 time, the threshold, or crossover point, between Small Commercial and Large Commercial was 18 2,000 GJ/year. The threshold at 2,000 GJ/year has been the same since it was originally set in 19 the 1993 Phase B Rate Design Application.

- 20
- 21
- 22
- 23 21.2 Please explain the methodology and inputs to calculate the economic crossover
 24 point when it was last re-aligned with the customer segmentation threshold
 25 between RS 2 and RS 3.
- 26

27 Response:

In FEI's 2001 Rate Design, a COSA study was performed which at that time indicated that there was no need for rebalancing, i.e., no revenue shift required between the various rate classes. However, FEI noted that the rates in place at that time resulted in a crossover at 1,280 GJ between the Commercial rate schedules, and that adjustments to the Basic Charge and Delivery Charge were necessary for Rate Schedules 2 and 3/23 to achieve an economic crossover at 2,000 GJ. Since no rebalancing was required between classes, a requirement was that the changes made to the rates for RS 2 would also need to yield the same total revenue for

¹⁷ <u>https://www.fortisbc.com/About/RegulatoryAffairs/GasUtility/NatGasTariffs/Documents/RateSchedule_2.pdf.</u>

¹⁸ https://www.fortisbc.com/About/RegulatoryAffairs/GasUtility/NatGasTariffs/Documents/RateSchedule 3.pdf.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 92

1 RS 2 and similarly the changes to the rates for RS 3/23 would need to result in the same total 2 revenue for RS 3/23. Inputs needed to complete this calculation were the Basic Charge 3 (\$/Month), Delivery Charge (\$/GJ), Cost of Gas (\$/GJ), the number of customers and annual 4 volumes for each of the rate schedules.

5 The objective of the RS 2 and RS 3/23 changes was that the revised rates for RS 2 and RS 3 6 should result in the same annual customer bill at a consumption level of 2,000 GJ and achieve 7 the same total revenues in each of RS 2 and RS 3/23. Achieving this required changes to the 8 Basic Charge and Delivery Charge of both rate schedules.

- 9
- 10
- 11

15

- Please explain whether the methodology to calculate the economic crossover
 point as presented in Table 8-3 differs from those described above. If yes, please
 elaborate and explain the rationale for any differences.
- 16 **Response:**

17 The methodology used to determine the economic crossover point is the same in the current 18 Application as it was in FEI's 2001 RDA, except for one variation. In 2001 the total revenues 19 from RS 2 and RS 3 were held unchanged. The changes in the Basic Charge and Delivery 20 Charge moved the economic crossover to 2,000 GJ.

In the 2016 Rate Design Application the rates have been changed so that the economic crossover is at 2,000 GJ (same methodological result as in 2001), but the rates also result in an equal offsetting revenue shift between RS 2 and RS 3/23 (this is different from 2001). The reason for the revenue shift was to limit the maximum bill impact any customer would experience to a maximum of 10 percent. As can be seen in Exhibit B-1, Table 12-2, Page 12-5, an approximately \$1.2 million revenue shift occurs between RS 2 and RS 3/23.

Also in the current Application, the rates used in Table 8-3 to calculate the crossover are the estimated COSA-Based rates (refer to Table 12-4 on Page 12-8 for the Basic Charge and Delivery Charge), i.e., the 2016 approved rates <u>plus</u> known and measureable changes (discussed in Section 6 of the Application). FEI views the inclusion of known and measurable changes as an appropriate refinement to the COSA study, rather than a change to the RS 2 / RS 3 economic crossover methodology.

If the statement in the preamble, "The current rates for RS 2 and RS 3 are shown in FEI's gas
tariff", is implying that current RS 2 and RS 3 rates are used in the economic crossover
calculation in the Application that would be incorrect. Please refer to the response to BCUC-FEI
IR 1.21.5 where the crossover is calculated using the current tariff rates.

- 37
- 38



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 93

4

2 3 21.4 Please explain whether the rates used to calculate the economic crossover point as shown in Table 8-3 includes all rate riders. If not, why not?

5 Response:

6 The economic crossover rates as shown in Table 8-3 do not include rate riders. FEI has not 7 included rate riders in the crossover calculation in its Rate Design Applications for the following 8 reasons.

- 9 1. Rate rider recoveries are not part of the current period utility revenues from customers, 10 but are recoveries or refunds of previous period deferred charges.
- 11 2. Rate riders related are generally temporary in duration, usually in place for a pre-12 determined period of time. For example, the amalgamation-related rate riders are only 13 in place to phase in the transition to common rates over a three-year period, and are 14 also set at different levels for different service areas.
- 3. An individual rate rider can vary between recoveries and refunds depending on whether 15 16 there were over or under-recoveries of the relevant costs or revenues in the prior period.
- 17
- 18

- 19 20 21.5 Please replicate Table 8-3 using the current rates as referenced in the preamble, 21 and reference the corresponding itemized rates contained in the Table of 22 Charges in RS 2 and RS 3.
- 23

24 **Response:**

25 Based on approved current rates (excluding rate riders) effective April 1, 2017 applicable to 26 Rate Schedules 2 and 3, the economic crossover point is at 1,716 GJ of annual consumption. 27 Note that the Delivery Charge shown below is lower than those used in Table 8-3 because the 28 original table includes known and measureable changes. Also, the Cost of Gas is lower than 29 what was included in the 2016 Annual Review.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission)	Page 94

Rate Components	RS 2	RS 3	Difference
1 Basic Charge (per day)	\$ 0.8161	\$ 4.3538	
2 Times number of days	365.25	365.25	
3 = Basic Charge Revenue	\$ 298.08	\$1,590.23	\$1,292.14
4 Delivery Charge (\$/GJ)	\$ 3.523	\$ 2.939	
5 Plus Cost of Gas (\$/GJ)	\$ 3.070	\$ 2.901	
6 = Total Variable Cost (\$/GJ)	\$ 6.593	\$ 5.840	\$ 0.753
7 Economic Crossover Point			
(Line 3 / Line 6)			1,716 GJ

3 The cost of gas for RS 2 is the sum of Storage and Transport Charge of \$1.020 / GJ plus 4 Commodity Cost Recovery Charge of \$2.050 / GJ.

5 The cost of gas for RS 3 is the sum of Storage and Transport Charge of \$0.851 / GJ plus 6 Commodity Cost Recovery Charge of \$2.050 / GJ.

- 7
- 8

- 9
- 10 21.6 Please explain what would be the threshold difference between the economic 11 crossover point and the customer consumption threshold to require a re-12 alignment.
- 13

14 Response:

15 FEI recommends not specifying a threshold difference for when a realignment would be 16 automatically triggered. The reasons for this are:

- 17 1. A review of revenue to cost ratios and rates is to be undertaken by FEI approximately 18 every five years, and any necessary revenue rebalancing and changes to rates, 19 including the realignment of the crossover point, can be done at that time.
- 20 2. Changes in rates will cause some misalignment. Gas cost changes can either increase 21 or decrease the misalignment, causing the economic crossover to be greater or less 22 than 2,000 GJ. As shown in the illustration below the \$0.90/GJ - RS 2 / \$0.84/GJ - RS 3 gas cost decrease only results in a \$0.06/GJ gap at 2,000 GJ, but it increases the 23 24 economic crossover above 2,000 GJ by approximately 180 GJ. These changes can 25 then also reverse over a shorter time frame.

26 Regardless, if a trigger threshold difference were to be established, it should be large enough to 27 leave a persistent price signal if left unaddressed. For that reason, FEI suggests that it be set at



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 95

1 a plus or minus 500 GJ difference between the economic crossover point and the RS 2 - RS 3 2 consumption threshold. Where the economic crossover has moved to 500 GJ above or below 3 the specified consumption threshold, the percentage difference at 2,000 GJ between RS 2 and 4 RS 3 annual bills is still not large, but 500 GJ of annual load difference would be outside the 5 year-to-year swings in consumption that might be expected to occur for customers with annual 6 consumption near the 2,000 GJ level. A plus-or-minus 500 GJ movement in the economic 7 crossover would also be outside the range of fluctuations in the economic crossover caused by 8 gas cost changes.

9 The following illustration provides an example of how the economic threshold can change, using 10 the change in cost of gas from the amounts included in the COSA rates to the cost of gas rates 11 that the Commission has approved in 2017. The economic threshold moves from 2,004 GJ to 12 2,182 GJ and at 2,000 GJ the annual bill difference is \$117. What this illustrates is that a rate 13 change difference between RS 2 and RS 3 of approximately \$0.06/GJ can cause the economic

14 threshold to move by approximately 180 GJ. But the annual bill variance at 2,000 GJ is only

15 \$117, or less than 1 percent.

	P	Proposed Rates With Average Gas					P	roposed Ra	ates as 4	s With Cur	rent 7	Cost of
		RS 2	11 2	RS3	Dif	ference	_	RS 2	as r	RS3	., Dif	ference
Rate Components												
Basic Charge \$ / Day	\$	0.9485	\$	4.7895			\$	0.9485	\$	4.7895		
Times number of days		365.25		365.25				365.25		365.25		
= Basic Charge Revenue	\$	346.44	\$	1,749.36	\$1	,402.93	\$	346.44	\$	1,749.36	\$1	L,402.93
Delivery Charge (\$ / GJ)	\$	3.664	\$	3.190			\$	3.664	\$	3.190		
Plus Cost of Gas (\$/GJ)	\$	3.967	\$	3.741			\$	3.070	\$	2.901		
= Total Variable Cost (\$/GJ)	\$	7.631	\$	6.931	\$	0.700	\$	6.734	\$	6.091	\$	0.643
						2.004						2 4 0 2
Economic Crossover						2,004						2,182
Annual Bill at 2,000 GJ	\$1	5,608.44	\$:	15,611.36	\$	2.93	\$	13,814.44	\$	13,931.36	\$	116.93



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
BC	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 96

1	22.0	Refer	ence:	RATE DESIGN FOR COMMERCIAL CUSTOMERS
2				Exhibit B-1, Section 8.5, p. 8-16;
3 4				BC Gas 2001 Rate Design Application Order G-116-01, Appendix 1, p. 2
5				Economic crossover point alignment
6 7		The C Applic	commiss ation (a	sion stated in Order G-116-01 on the BC Gas Utility Ltd. 2001 Rate Design pproval of the negotiated settlement, Appendix 1, p. 2):
8 9 10 11			In orde (Rate approa propos	er to achieve an economic breakpoint between Small Commercial Service Schedule 2) and Large Commercial Service (Rate Schedules 3/23) that aches 2,000 GJ per year, Rate Schedules 2 and 3/23 will be revised as sed by BC Gas in the Application under Tab 6, page 3.
12		On pa	ige 8-16	of Exhibit B-1, FEI explains that:
13 14 15 16 17 18			This n energy and bi year-to move instabi	nisalignment gives an incentive to customers on RS 2 to consume more y so they can move above the 2,000 GJ threshold to achieve a lower rate II. The misalignment might also cause rate instability for customers whose p-year fluctuations in annual demand may occasionally cause them to back and forth between these rate schedules. This can also cause revenue ility for the utility.
19 20 21 22	Boon	22.1	With refactors crosso	eference to the various components of RS 2 and RS 3, please explain the s that contributed from an alignment to a misalignment of the economic over point with the customer segmentation threshold.
23	Resp	onse:		
24 25	The fatter the ch	actors tl nanges i	hat wou in the Co	Id have contributed to the misalignment over the past 15 years would be ost of Gas, Basic Charge, and Delivery Charge.
26 27 28	Since costs 1.21.6	2001 th such as 5, gas c	he avera s around osts (co	age cost of gas has decreased significantly, albeit with periods of high gas 3 2008. As demonstrated in the example in the response to BCUC-FEI IR mbined commodity and midstream) declines at different rates between RS

29 2 and RS 3/23 can cause the economic crossover to move away from 2,000 GJ.

30 General increases from revenue requirements were applied to the Basic Charge and Delivery 31 Charge in equal percentage until 2010. Since 2010, the recovery of increased revenue 32 requirements has been flowed through only on the Delivery Charge. The practice since 2010 33 keeps the difference static between the RS 2 and RS 3 Basic Charges, but the difference in the 34 RS 2 and RS 3 volumetric Delivery Charges tends to increase over time. The result is that, other 35 components such as gas costs being equal, the economic crossover will decrease gradually.



SC™	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 97

1 2			
3 4 5 6 7	22.2 <u>Response:</u>	Please e RS 3 wa	explain whether the customer segmentation threshold between RS 2 and s set at any other value than 2,000 GJ in the past.
8 9	Since the init between the t	ial approv wo rate so	al of RS 2 and RS 3 in the 1993 Phase B Rate Design the threshold chedules has been at 2,000 GJ/year.
10 11			
12 13 14 15 16 17	Response:	22.2.1	If yes, please explain the circumstance that supported a modification of the customer segmentation threshold to the existing 2,000 GJ threshold in favour of relying solely on rate changes.
18	Please refer t	o the resp	onse to BCUC-FEI IR 1.22.2.
19 20			
21 22 23 24 25 26	22.3	Please e addresse incentive instability	explain how the proposed rate changes to the delivery and basic charge es each of the issues identified regarding the misalignment, namely (i) to consume more, (ii) rate instability for the customer, and (iii) revenue y for the utility.
27	<u>Response:</u>		
28	Below is the r	response f	or each of the listed categories:
29 30 31 32 33 34 35 36	i) Incent custor This o propo econo achiev less u 3.	tive to Cor mers woul could ince sals, whic mic cross ve a lower nder RS 2	nsume More – Figure 8-11 shows that for consumption above 1,500 GJ, d pay a lower effective rate under RS 3 than they would under RS 2. nt customers to move to RS 3 by consuming more gas. Under FEI's h are shown in Figure 8-12, the customer segmentation threshold and sover are matched so that the incentive to increase consumption to rate is eliminated. For all volumes up to 2,000 GJ, a customer will pay and, for all volumes above 2,000 GJ, a customer will pay less under RS

FORTIS BC

FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 98

- 1 ii) Customer Rate Instability – The proposed rates improve rate stability for customers. For 2 those customers consuming at or near 2,000 GJ per year, the average cost for a 3 customer will be the same (i.e., at 2,000 GJ) or close to being the same under either 4 RS2 or RS 3. For customers whose normal annual consumption is close to 2,000 GJ, 5 the current rates have a gap of approximately 25 cents per GJ; this is decreased by the 6 proposed rates. For example, at consumption of 1,900 GJ, based on the proposed rates 7 for RS 2 and 3 the difference in the average cost per GJ is only 3.2 cents per GJ, and at 8 consumption of 2,100 GJ, the difference in the average cost per GJ is only 3.8 cents per 9 GJ.
- iii) Utility Revenue Instability The increased Basic Charges for both RS 2 and RS 3/23
 improves the utility revenue stability because less of the revenue is dependent on
 consumption. With the proposed rate realignment, the utility will experience improved
 revenue stability as customers are incented to receive service under the correct service
 offering of Small Commercial Service or Large Commercial Service and FEI will be
 better able to predict the revenues to be received under each of those rate schedules.
- 16
- 17
- . .
- 18
- 1922.4Please explain what would be the consequence of an economic crossover point20that is higher than the customer segmentation threshold.
- 21

22 <u>Response:</u>

If the economic crossover point was greater than the customer segmentation threshold, customers whose normal expected consumption was greater than the customer segmentation point but less than the economic crossover point would be incented to be reclassified as Small Commercial and / or attempt to reduce their consumption to below the customer segmentation threshold.



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
BC™	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 99

123.0Reference:RATE DESIGN FOR COMMERCIAL CUSTOMERS2Exhibit B-1, pp. 8-1, 8-14, 8-21

Proposed rate change

4 FEI states on page 8-1 of Exhibit B-1:

5 FEI proposes to increase the Basic Charge and to reduce the Delivery Charges 6 of RS 2, RS 3 and RS 23 to eliminate the customer bill differential for customers 7 whose annual consumption is close to the 2,000 GJ threshold.

8 In table 8-3 on page 8-21 of Exhibit B-1, FEI presents the proposed changes to the 9 commercial rates:

Rate Schedule	COSA ¹³⁸ Based Rate	Proposed Rate	Proposed Change
RS 2 – Small Commercial			
Basic Charge (daily)	\$0.8161	\$0.9485	\$0.1324 or 16.2%
Delivery Charge (\$/GJ)	\$3.850	\$3.664	\$-0.186 or -4.8%
RS 3/23 – Large Commercial			
Basic Charge (daily)	\$4.3538	\$4.7895	\$0.4357 or 10.0%
Delivery Charge (\$/GJ)	\$3.188	\$3.189	\$0.001 or 0.03%

Table 8-3: Proposed Changes to Commercial Rates

- 11 On page 8-14 of Exhibit B-1, FEI states
 - ... FEI's commercial customer related costs are reasonably well recovered by the Basic Charge
 - Government energy efficiency and conservation policies discourages higher fixed charges
 - Increasing the Basic Charge would result in bill impacts and rate instability for commercial customers
- 18Based on these competing principles and considerations, FEI believes that the19basic charges provide a reasonable recovery of FEI's commercial customer20allocated fixed costs.
- 21 23.1 Please provide the basic charge as a percentage of total bill for each of the 22 average RS 2 and RS 3 customers.
- 23

10

12

13

14

15

16



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 100

1 Response:

- 2 Line 8 of the following table shows the Basic Charge as a percentage of the total customer bill
- 3 for an average RS 2 customer and for an average RS 3 customer using the COSA Based Rate
- 4 and the Proposed Rate for each rate schedule.

		R	S 2 - Small	Con	nmercial	F	RS 3 - Large	Cor	nmercial
		CO	SA Based	Р	roposed	C	OSA Based	P	Proposed
Line		_	Rate		Rate		Rate		Rate
1	Average # of Days		365.25		365.25		365.25		365.25
2	Use / Customer (GJ)		332.6		332.6		3,587		3,587
3	Basic Charge	\$	0.8161	\$	0.9485	\$	4.3538	\$	4.7895
4	Delivery Charge	\$	3.850	\$	3.664	\$	3.188	\$	3.189
5	Cost of Gas	\$	3.967	\$	3.967	\$	3.741	\$	3.741
6	Annual Basic Charge	\$	298.08	\$	346.44	\$	1,590.23	\$	1,749.36
7	Total Bill	\$	2,898.01	\$	2,884.51	\$	26,444.55	\$	26,607.27
	Basic Charge as a % of								
8	Total Bill		10%		12%		6%		7%

- 5
- 6
- 7
- 8
- 9
- 10
- 23.2 Please elaborate on which competing principles support FEI's proposed changes to the basic and delivery charges for RS 2 and RS 3.
- 11 12

13 Response:

- 14 It is FEI's view, the proposed rates for Small and Large Commercial customers are in alignment15 with the eight Bonbright principles (Exhibit B-1, Page 5-2).
- Principle 1: Recovering the Cost of Service the proposed rates will continue to recover the
 cost of service.
- Principle 2: Fair apportionment of costs among customers the increase in the Basic
 Charges moves the Company to having appropriate cost recovery in rates.
- 20 **Principle 3**: Price signals that encourage efficient use and discourage inefficient use the 21 rate structure will encourage customers to focus on efficient consumption as there will not



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 101

- be a gap in the average cost at and around 2,000 GJ where it would encourage customers
 to consume more gas just to have a lower total bill (economic crossover consideration).
- Principle 4: Customer understanding and acceptance and Principle 5: Practical and cost effective to implement no changes are being recommended as the same rate structures
 are being proposed.
- 6 Principle 6: Rate stability and Principle 7: Revenue stability please refer to the response
 7 to BCUC-FEI IR 1.22.3.

8 Principle 8: Avoidance of undue discrimination - will be improved as the interclass equity
9 will be enhanced as customers who consume approximately 2,000 GJ will have,
10 approximately, the same cost.

- 11
- 12

14

15

16

- 13
- 23.2.1 In consideration of the competing principles, please explain FEI's process and analysis in determining the appropriate basic and delivery charges.
- 1718 <u>Response:</u>

19 On page 8-14 of the Application, preceding the quoted points in the preamble, FEI stated that: 20 "The rate design principle to fairly apportion costs would suggest that FEI move the Basic 21 Charge upwards to be in closer alignment with FEI's customer costs". At this point in the 22 Commercial Rate Design discussion (i.e. on page 8-14), the analysis and assessment of what 23 needs to be done is incomplete. Following page 8-14, FEI provides an assessment of Rate 24 Design Options for changing the consumption threshold between Small and Large Commercial 25 or adjusting the Basic Charges and Delivery Charges of RS 2 and RS 3. Part of the assessment 26 took into consideration the impact on customers. The decision made by FEI was to propose to 27 keep the threshold at 2,000 GJ/year and adjust the charges.

- 28 FEI's analysis in determining the appropriate basic and Delivery Charges were the following:
- At the segmentation threshold of 2,000 GJ per year a customer that is in either RS 2 or
 RS 3 would have the same annual bill.
- For annual volumes less than 2,000 GJ the lower annual cost should be achieved under
 RS 2 rates.
- 33 3. For annual volumes greater than 2,000 GJ the lower annual cost should be achieved34 under RS 3 rates.



111	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 102

- 4. The rates applied to the number of customers in each rate schedule and the volumes in
 each rate schedule will generate the same total revenue from existing rates for each rate
 schedule.
- 4 5. Minimize to the extent possible the bill impact for each customer.

5 In FEI's view, the proposed rate changes are not major, but rather adjustments to rates to 6 balance complex rate design objectives.

- 7
- 8
- C
- 9 10
- 23.3 Please explain whether FEI considered other rate adjustment options that can realign the economic crossover point at 2.000 GJ. If not, why not?
- 11
- 12

13 **Response:**

14 FEI did not consider any other rate structure options to realign the economic crossover at 2,000 15 GJ. In an effort to minimize bill impacts for RS 2 and RS 3/23 customers, FEI did try different 16 Basic Charge and volumetric charge combinations to reset the economic crossover volume to 17 2,000 GJ per year. FEI used the Excel Solver function to derive the final proposed rates for RS 18 2 and RS 3/23 and used the constraints functionality in Excel Solver. The constraints (factors) 19 that were used when solving for the 2,000 GJ economic crossover point, in priority order, 20 included: minimize the revenue shift between small and large commercial rate schedules, 21 eliminate any revenue shifts from commercial to other rate schedules, set maximum annual bill 22 impact to any one customer to 10 percent and minimize the bill impact to customers consuming 23 at the 2,000 GJ per year level.

24
25
26
27
23.3.1 If yes, please provide the comparative analysis of various rate changes considered by FEI, clearly explain the factors considered when comparing various rate change combinations/magnitudes, and explain why the proposed option is superior than other options considered.
31
32 Response:

33 Please refer to the response to BCUC-FEI IR 1.23.3.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 103

1 24.0 **Reference: RATE DESIGN FOR COMMERCIAL CUSTOMERS**

2

3

Exhibit B-1, pp. 8-4, 8-8, 8-9, 8-22, 8-23

Impact of proposed rate change

4 FEI shows in Figure 8-1 on page 8-4 of Exhibit B-1 the commercial customer market 5 segments. Figure 8-6 on page 8-8 and Figure 8-7 on page 8-9 show the small and large 6 commercial customer bill frequency, respectively.

7 FEI presents the bill impact analysis of its proposed rate adjustment for RS 2 customers in figure 8-13 and for RS 3 customers in figure 8-14. 8

9 24.1 Please replicate Figure 8-13 and 8-14 in terms of actual bill change in dollar 10 terms (\$/year).

11

12 **Response:**

13 The first graph below is the bill impacts for RS 2 (Figure 8-13 using \$ / Year) and the second 14

- graph is the bill impact for RS 3 / 23 (Figure 8-14 using \$ / Year).
- 15

Figure 8-13 (adjusted): RS 2 Customer Bill Impacts (\$ / Year)





	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 104



Figure 8-14 (adjusted): RS 3/23 Customer Bill Impacts (\$ / Year)

24.1.1 In consideration of the magnitude of bill change for RS 2 and RS 3 customers, please comment on whether the proposed rate changes will result in a change in consumer behavior.

10 Response:

2

3 4

5 6

7

8

9

For RS 3/23 customers whose bill impact is between zero and one percent, FEI does not
believe the changes will have an impact on consumer behavior.

Although for RS 2 customers in general FEI does not believe that the rate proposal will result in a noticeable change in consumer behavior, it is more difficult to assess what impact the change will have, as larger volume customers in this Rate Schedule will experience a 2 percent decrease in their bill, and the smallest volume users (less than 40 GJ per year) will have a 10 percent increase from the combined effect of the increased Basic Charge and Delivery Charge. For the smallest volume users who will experience a 10 percent increase, this will equate to an annual increase of \$45 per year or an increase of less than \$4 per month.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 105

- 24.2 Please present the bill frequency of small commercial customers and large
 - commercial customers similar to that presented in Figure 8-6 and Figure 8-7 for each customer market segments presented in Figure 8-1.

Response:

- Below are the requested histograms for small and large commercial accounts by those market
- segments presented in Figure 8-1 that FEI has available data for from its billing system.

Small Commercial





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 106

Table 2: College and Universities



Table 3: Schools





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 107

Table 4: Accommodation



Table 5: Hospital




FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 108

Table 6: Retail - Non-Food



Table 7: Logistics and Warehouses





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 109

Table 8: Food Service



Table 9: Apartments







Table 10: Office



Large Commercial



Table 11: Retail Food





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 111

Table 12: College and Universities



Table 13: Schools Schools Rate 3/23





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 112

Table 14: Accommodation



Table 15: Hospital





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 113

Table 16: Retail - Non Food



Table 17: Logistics and Warehouses





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 114

Table 18: Food Service



Table 19: Apartments







Table 20: Office





M	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 116

1 H. CHAPTER 9 – RATE DESIGN FOR INDUSTRIAL CUSTOMERS

2	25.0	Reference:	RATE DESIGN FOR INDUSTRIAL CUSTOMERS
3 4			Exhibit B-1, Section 2.1, p. 2-2; Section 6.3.1.5, pp. 6-8 to 6-9; Section 9.8.1, pp. 9-37 to 9-38
5			Tariff supplements – bypass agreements and contract customers
6		On page 2-2	of Exhibit B-1, FEI states:
7 8 9 10		FEI h tariff s the Co supple	as a number of tariff supplements, including bypass agreements. These supplements are negotiated agreements and are approved separately by ommission and, as such, FEI is not proposing any changes to existing tariff ements in this Application.

11 On page 6-9 of Exhibit B-1, FEI presents Table 6-4 as follows:

Table 6-4: Information on Bypass Customers⁶⁴

	RS 22	RS 22A	RS 25	Other	Total
Customers (#)	2	4	4	1	11
2016 Forecast Volume (TJ)	8,	396	851	375	9,622
2016 Forecast Revenue (\$000s)	846		435	44	1,325

12

13 Footnote 64 on page 6-9 of Exhibit B-1 states:

- FEI has included Teck Coal (Byron Creek) with bypass customers [in Table 6-4]
 in its Revenue Requirements. The contract is a Pipeline Agreement which
 specifies how the 'Actual Annual Service Charge' is determined. The annual
 service charge is not affected by Commission approved rate changes. As such, it
 is similar to FEI's bypass contracts.
- 1925.1Please confirm, or otherwise explain, that Teck Coal (Byron Creek) is not a20customer with which FEI has a bypass agreement.
- 21

22 Response:

- FEI confirms that Teck Coal (Byron Creek) is not a customer with which FEI has a bypass agreement. The agreement with Teck Coal is a Pipeline Agreement that was approved by the Commission effective January 1, 2003 by Order G-36-03. The charges to Teck Coal are based on the contract and are not subject to general rate changes that would be part of an annual review or revenue requirement application.
- 28
- 29
- 30



2

3

4

5

FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 117

25.1.1 If confirmed, please provide an updated version of Table 6-4 which represents only customers with which FEI has bypass agreements and also explain any differences between the table in the response and Table 6-4 in Exhibit B-1.

6 Response:

- 7 An updated version of Table 6-4 that excludes Teck Coal (Byron Creek) is provided below with
- 8 the additional rows requested in BCUC-FEI IR 1.25.2: (i) 2016 Forecast Peak Demand; (ii) 2016
- 9 Forecast Load Factor; and (iii) Costs determined by the COSA Study.
- 10 The 2016 Forecast Peak Day Demand is the Daily Transportation Quantity (DTQ's) of each of
- 11 the Bypass customers. FEI does not forecast Load Factors for the Bypass customers and in the
- 12 COSA Study there is no allocation of costs to Bypass customers, but the revenues from the
- 13 Bypass customers have been allocated to all non-bypass customers as credits to their allocated
- 14 cost of service in proportion to all non-bypass customers Delivery Margin allocated cost.

Table 6-4 (Updated): Information on Bypass Customers (excludes Teck Coal - Byron Creek)

	RS 22	RS 22A	RS 25	Total
Customer (#)	2	4	4	10
2016 Forecast Volume (TJ)	8,396		851	9,247
2016 Forecast Renenue (\$000's)	\$846		\$435	\$1,281
i) 2016 Forecast Peak Day Demand (GJ)	36,608		5,931	42,539
ii) 2016 Forecast Load Factor	Not Applicable			
iii) COSA Allocated Costs	Not Applicable			

- 15
- 16
- 17
- 18
- 19 20

- 25.2 Please add additional rows to Table 6-4 to provide the following additional information for FEI's Bypass customers only (i) 2016 Forecast Peak Demand; (ii) 2016 Forecast Load Factor; and (iii) Costs determined by the COSA Study.
- 22
- 23 Response:
- 24 Please refer to the response to BCUC-FEI IR 1.25.1.1.
- 25 26
- 27
 28 25.3 Please provide a list of the industrial sector makeup for (i) bypass customers;
 29 and (ii) non-bypass contract customers. Please provide a pie chart along with



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 118

- your responses that breakdown the industries by throughput for the most recent
 year of actual data (in a manner similar to Figure 9-1 on page 9-3 of Exhibit B-1).
- 3 4

Response:

- 5 (i) If FEI does not include Teck Coal (Byron Creek) with the Bypass Customers, there are then
- 6 10 bypass customers. The 10 bypass customers can be broken down into three segments
- 7 as follows:

Industrial Sector	# Customers in Sector
Pulp & Paper	4
Wood Products	5
Oil & Gas	1
Total	10

8

9 The following chart shows the bypass customers broken down into sector by 2016 10 throughput.



- 11 12
- (ii) To answer this question FEI has assumed that "non-bypass contract customers" in the
 question is referring to the nine RS 22A customers and the five RS 22B customers. These
 14 non-bypass RS 22A & 22B customers can be broken down into six segments as follows:



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 119

Industrial Sector	# Customers in Sector
Pulp & Paper	4
Mining - Coal	4
Mining - Metal	2
Chemical	1
Wood Products	2
Manufacturing	1
Total	14

- 2 The following chart shows the non-bypass contract customers broken down into sector by
- 3 2016 throughput.



- At the time that FEI developed the COSA Study, Teck included on their website that mining at Coal Mountain Operations will be completed by the end of 2017. Consequently, FEI eliminated their revenue of \$44 thousand from the COSA model. The costs to serve Byron creek remain in
- 15 the cost of the utility and are allocated to all other non-bypass customers.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 120

- 1 2 3
 - 25.5 Please explain if there are any other customers that are similar to FEI's bypass customers and was included in the data provided in Table 6-4 above.

7 <u>Response:</u>

8 No other customers that are not bypass customers, besides Teck Coal (Byron Creek), were 9 included in Table 6-4.

10

4

5

6

- 11
- 12 13

24

On page 9-37 of Exhibit B-1, FEI provides Table 9-22 which states that the total number
of Rate Schedule 22A customers is 9 and the total number of Rate Schedule 22B
customers is 5. On page 9-38 FEI describes the history of Rate Schedules 22A and 22B
and quotes the following from the 1993 Phase B Rate Design Decision:

- ...considering that most of these interior customers had either individually
 negotiated rates (Inland bypass customers) or a uniquely linked rate design
 (Columbia customers) and few if any were likely to be requiring load increases,
 closed rates were argued to be appropriate.
- 22 25.6 Please explain the key differences between a bypass customer and a non-23 bypass contract customer.

25 **Response:**

The key difference between a bypass customer and a non-bypass customer is that the nonbypass customer will pay the applicable Commission-approved rate schedule charges as endorsed in the Company's tariff. In contrast, bypass customers have negotiated a contract and rates that are different and lower than the standard approved applicable rate schedules.

The original bypass agreements arose because certain customers that were close to the upstream pipeline were seeking to build their own infrastructure and avoid purchase of service from the utility (i.e., bypass the utility). Rather than have the customer leave the utility service, bypass rates were negotiated. The negotiated bypass rates are long term contracts based upon the costs as if the customer left the system and built, owned and operated a pipeline from the upstream pipeline to their facility. The negotiated rates are approved by the Commission.

The principle for bypass rates is to recognize the possibility of a bypass and seek to retain the customer on the system to avoid stranding infrastructure and the complete loss of delivery



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 121

revenue while avoiding the need for construction of a physical bypass pipeline. The concept of bypass rates was supported by government policy to urge the industrial customer and the utility to negotiate a competitive transportation agreement and to avoid building unnecessary infrastructure, while helping the utility maintain service to the customer.

- 5
 6
 7
 8 25.6.1 For the five RS 22A customers that are not bypass customers, please describe the nature of the individually negotiated contracts and the associated rates for these customers.
- 11

12 **Response:**

The five RS 22A customers that are not bypass customers do not have individually negotiated contracts. The applicable associated rates are those found in the Company's tariff for Rate Schedule 22A – Transportation Service (Closed) Inland Service Area. The rates are comprised of the following:

- monthly Basic Charge;
- 18 firm Delivery Charges;
- 19 interruptible Delivery Charge;
- unauthorized overrun charges;
- balancing service charges;
- backstopping gas charge;
- replacement gas charge; and
- monthly administration charge.
- 25
- 26
- 27 28
- 29
- 25.6.2 Please describe the nature of the individually negotiated contracts and the associated rates for the five RS 22B customers. Please confirm that none of the RS 22B customers are considered to be bypass customers.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 122

1 **Response:**

2 The five RS 22B customers do not have individually negotiated contracts. The applicable 3 associated rates are those found in the Company's tariff for Rate Schedule 22B -4 Transportation Service (Closed) Columbia Service Area. In the RS 22B Table of Charges 5 Elkview Coal Corporation rates are listed separately from the rates that are applicable to all 6 other RS 22B customers. The rates are comprised of the following:

- 7 monthly Basic Charge; •
- 8 firm Delivery Charges;
- 9 interruptible Delivery Charge;
- unauthorized overrun charges; 10
- 11 backstopping gas charge; and
- 12 monthly administration charge.
- 13

14 Four of the five RS 22B customers that are coal mines do have Commission approved RS 22B 15 tariff supplements with respect to the applicable RS 22B interruptible charges. The standard RS 16 22B interruptible charges are at a premium to firm service and the tariff supplements do not 17 allow the four customers to decrease their firm contract demand but reduce the interruptible 18 charges to be aligned with the RS 22B firm Delivery Charges. The customers requested such a 19 tariff supplement so that if market conditions presented themselves the tariff supplements 20 would, if interruptible pipeline capacity is available, remove any impediment to those customers 21 who are mines of using more natural gas instead of coal, while maintaining the spirit of the rate 22 schedule.

- 23 There are no RS 22B bypass customers.
- 24
- 25
- 26

31

- 27 25.7 Please provide a list describing the end-uses for (i) bypass customers; and (ii) 28 non-bypass contract customers. Please provide a pie chart along with your 29 responses that breaks down the end-uses by throughput for the most recent year 30 of actual data (in a manner similar to Figure 9-2 on page 9-4 of Exhibit B-1).
- 32 Response:

33 The following two pie charts provide a breakdown of the end-uses for the industry sectors as a 34 whole that represent (i) 10 bypass customers and (ii) 14 RS 22A and RS 22B non-bypass 35 contract customers. The source of the data is FEI's 2015 Conservation Potential Review (CPR) 36 using a 2014 base year which is consistent with Figure 9-2 on page 9-4 of Exhibit B-1.



(i) The 10 bypass customers are in the Pulp & Paper, Wood Products and Oil & Gas industry sectors. The following pie chart provides the end-use breakdown for those industry sectors:



(ii) The 14 non-bypass customers within RS 22A and RS 22B are Pulp & Paper, Mining-Coal, Mining-Metal, Chemical, Wood Products and Manufacturing industry sectors. The following pie chart provides the end-use breakdown for those industry sectors:





2

3

TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 124

25.8 Please explain the process for initiating a change in the terms or rates for bypass agreements.

4 <u>Response:</u>

5 To initiate a change to the terms or rates for bypass agreements, FEI would first need to have 6 reason to believe, or direction from the Commission to the effect, that the existing bypass 7 agreements were no longer just and reasonable. Second, any change to the terms or rates for 8 any individual bypass agreement, through either an amendment or new agreement, would then 9 need to be negotiated between FEI and each bypass customer, and filed with the Commission 10 for approval. FEI would expect any new or amended agreement to commence following the end 11 of the term of the existing agreement.

Given the different circumstances of each bypass customer, FEI's view is that negotiation of individual bypass agreements, subject to Commission approval, remains the best process. As stated by the Commission in the Inland Natural Gas Co.'s Rate Design Decision, Order No. G-80-87, dated December 11, 1987:

Given this geographic variation, a bypass postage stamp rate would either have to be set low enough to keep the lowest cost bypass customers on the system, and thereby reduce the contributions of other higher cost bypass customers, or, risk losing those customers. Given this fact, and considering the time already spent in negotiations between the potential large bypass customers and Inland, the Commission concludes the negotiation process is most effective method for fixing individual rates to potential bypass customers.

23 At this time, it is unclear on what basis, or according to what principles, FEI would seek to 24 renegotiate the bypass agreements, unless approached by the customer regarding changes to 25 the agreement or a need for different service. The terms of the existing bypass agreements 26 were negotiated and approved in accordance with the principles for a reasonably competitive 27 bypass rate set out by Commissioner Millard in the 1987 BCUC Report and Recommendations 28 to the Lieutenant-Governor in Council in the Matter of Applications for Energy Project 29 Certificates, which were endorsed by the Commission in the Inland Natural Gas Co.'s Rate 30 Design Decision, Order G-80-87, dated December 11, 1987. The charges in each bypass rate 31 are set based on the costs that the customer would have incurred to construct, own and operate 32 its own bypass pipeline. The bypass agreements rates include provisions to address changing 33 circumstance, including:

- changes in costs that would have been incurred by the customer as result of increases
 in gas volume, increases in capacity or additional facilities that would have been
 required had the customer constructed and operated a bypass pipeline; and
- increases in FEI's costs, calculated as changes in the operating and other costs that the
 bypass customer would have incurred for the operation of a bypass pipeline.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 125

1 The bypass agreements also contain provisions dealing with extension of the term of the 2 agreements. At this time, FEI believes it continues to be reasonable to extend the agreements 3 according to their terms.

4
5
6
7 25.8.1 Please describe FEI's rights with regard to termination of current bypass agreements.
9
10 Response:

FEI has individually negotiated bypass agreements with each customer and those agreements may have differing clauses, but generally FEI is permitted to terminate the current bypass agreements only if the Commission disallows FEI the recovery from other customers of any revenue shortfalls resulting from the negotiated rates under the bypass agreement. Please refer to the response to BCUC-FEI IR 1.25.10 for copies of the Commission approved bypass tariff supplements.

- 17 18 19
 - 25.8.2 Please describe the bypass customers' rights to termination of current bypass agreements.

23 **Response:**

20

21

22

24 FEI has individually negotiated bypass agreements with each customer and those agreements 25 may have differing clauses regarding the customers' rights of termination. Generally, some 26 bypass agreements allow for termination by the customer on one year's notice, while other 27 agreements allow for the customer to terminate the bypass agreement if the BCUC sets rates at 28 a level in excess of the negotiated rates set out in the agreement. Some agreements also allow 29 for termination by the customer under other circumstances, such as a permanent cessation of 30 Please refer to Attachment 25.10, provided in the response to BCUC-FEI IR operations. 31 1.25.10 for copies of the Commission approved bypass tariff supplements.

32 33		
34		
35	25.8.3	If any of the bypass customers have had their bypass contracts
36		renegotiated to accommodate increases in load, please describe the
37		process that led to termination of the original bypass contract,



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
,	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 126

3

renegotiation of a new bypass agreement and Commission approval of a new bypass agreement.

4 Response:

5 A portion of the response to this question is being filed confidentially with the Commission, 6 pursuant to Section 18 of the Commission's Rules of Practice and Procedure regarding 7 confidential documents, established by Order G-1-16. FEI requests that the response be kept 8 confidential as it contains confidential customer information for which FEI does not have the 9 authority or permission to disclose. Given the private and commercially sensitive nature of the 10 information, FEI submits that only the Commission should have access to the unredacted 11 confidential version of this IR response.

12 A redacted version of the response has been provided in the publicly available response

Two of the ten bypass customers have increased their load which has resulted in either a new bypass agreement being entered into (Husky Energy Marketing Inc.) or an amended and restated bypass agreement (Dunkley Lumber Limited).

16 Husky Energy's bypass agreement under RS 22A was terminated when Husky and FEI 17 negotiated a new bypass agreement due to Husky wanting to increase the delivery pressure to 18 425 PSIG. This change was significant enough that under the 1993 Phase B Rate Design 19 Decision the new agreement would have to be completed under RS 22 rather than RS 22A as 20 the original agreement had been. This was a significant change to the model that the original 21 bypass rate was based on. The new negotiated bypass agreement was submitted to the 22 Commission and was approved by the Commission (refer to the response to BCUC-FEI IR 23 1.25.9 for more information).

In the case of Dunkley Lumber served under RS 25, there were two increases in load that
resulted in amending the bypass agreement as a result of increased capital and operating costs.
The amended bypass agreement was submitted to the Commission for each revision and was
approved by the Commission. The table below provides the original agreement annual volumes
and DTQ and subsequent amended volumes for Dunkley Lumber.





FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 127

3

25.9 Please complete the following table summarizing each of FEI's current bypass agreements. Please add rows as necessary and provide the annual demand using the most recent year of actual data.

		FEI Bypass Agreements					
	Name of Customer	Commission Order approving Agreement	Effective date of Agreement	Current Term of Agreement	Tariff Supplement No.	Rate Schedule	Annual Demand (TJ)
Row 1							
Row 2							
Row 3							

4 5

6 Response:

7 A portion of the response to this question is being filed confidentially with the Commission, 8 pursuant to Section 18 of the Commission's Rules of Practice and Procedure regarding 9 confidential documents, established by Order G-1-16. FEI requests that the response be kept 10 confidential as it contains confidential customer information for which FEI does not have the 11 authority or permission to disclose. Given the private and commercially sensitive nature of the 12 information, FEI submits that only the Commission should have access to the unredacted 13 confidential version of this IR response.

A redacted version of the response has been provided in the publicly available response 14

15 The following table provides the requested information. The last column Annual Demand (TJ) is 16 the energy volumes for calendar 2016. Agreements that originated prior to the 1993 Phase B 17 Rate Design Decision were amended and restated effective November 1, 1993. The Husky 18 Energy original bypass agreement was dated November 1, 1987 and was amended and 19 restated effective November 1, 1993 under RS 22A. The amended and restated agreement 20 from November 1, 1993 was terminated and the current bypass agreement was entered into

FEI Bypass Agreements						
Name of Customer	Commission Order approving Agreement	Effective Date of Agreement	Current Term of Agreement	Tariff Supplement No.	Rate Schedule	-
Dunkley Lumber	G-35-04	Nov. 1, '04	11/01/04 — 11/01/18	E-2	RS 25	
Tolko Industries		Nov. 1, '93	11/01/93 – 11/01/17	E-5	RS 25	
Tolko Industries		Nov. 1, '93	11/01/93 — 11/01/17	E-6	RS 25	
West Fraser Mills		Nov. 1, '93	11/01/93 – 11/01/20	E-8	RS 25	
West Fraser	G-68-98	Nov. 1, '96	11/01/96 –	G-10	RS 22	

21 effective February 1, 2006.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 128

FEI Bypass Agreements						
Name of Customer	Commission Order approving Agreement	Effective Date of Agreement	Current Term of Agreement	Tariff Supplement No.	Rate Schedule	•
Mills (West Pine)			11/01/17 (autorenewal ¹⁹)			
Husky Energy	G-82-05	Feb. 1, '06	02/01/06 – 11/01/17 (autorenewal ²⁰)	G-20	RS 22	
Canadian Forest Products (Prince George Pulp & Paper)	G-33-03	Nov. 1, '93	11/01/93 – 11/01/17	G-5	RS 22A	
West Fraser Mills	G-33-03	Nov. 1, '93	11/01/93 – 11/01/17	G-6	RS 22A	
Canadian Forest Products (Northwood Pulp & Timber)	G-33-03	Nov. 1, '93	11/01/93 – 11/01/17	G-7	RS 22A	
Cariboo Pulp & Paper	G-33-03	Nov. 1, '93	11/01/93 – 11/01/17	G-8	RS 22A	

¹

3

4

5

6 7

8 **Response**:

25.9.1

data.

9 The response to this question is being filed confidentially with the Commission, pursuant to 10 Section 18 of the Commission's Rules of Practice and Procedure regarding confidential 11 documents, established by Order G-1-16. FEI requests that the response be kept confidential 12 as it contains confidential customer information for which FEI does not have the authority or 13 permission to disclose. Given the private and commercially sensitive nature of the information, 14 FEI submits that only the Commission should have access to the confidential version of this IR 15 response.

Please provide a bar chart showing the Annual Throughput on the y-

axis for each bypass customer, using the most recent year of actual

¹⁹ Evergreen clause of 1 year unless 6 months' notice prior to the end of the contract year then in effect.

²⁰ Evergreen clause of 1-year extension unless 12 months' notice prior to the end of the current termination date.



тм	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 129



Please provide a bar chart showing the Annual Throughput on the y-

axis for each non-bypass contract customer, using the most recent year

3

3 4

5

6

7

8 **Response:**

25.9.2

of actual data.

9 The response to this question is being filed confidentially with the Commission, pursuant to 10 Section 18 of the Commission's Rules of Practice and Procedure regarding confidential 11 documents, established by Order G-1-16. FEI requests that the response be kept confidential 12 as it contains confidential customer information for which FEI does not have the authority or 13 permission to disclose. Given the private and commercially sensitive nature of the information, 14 FEI submits that only the Commission should have access to the confidential version of this IR 15 response.

16



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 130

25.10	For each of FEI's current bypass customers, please provide a copy of the Commission approved tariff supplement.

78 <u>Response:</u>

9 Please refer to Attachment 25.10 for copies of the ten current bypass tariff supplements.



οτιο			FortisBC 2016 Rate	Energy Inc. (FEI Design Application	or the Compar on (the Applica	יע) tion)		Submission Date: June 9, 2017
R115 BC [™]		Resp	onse to British Colum	bia Utilities Comr formation Reques	mission (BCUC t (IR) No. 1	or the Commiss	sion)	Page 131
26.0 Refe		rence:	RATE DESIGI	n for Indu	ISTRIAL C	USTOMER	S	
			Exhibit B-1, S	Section 9.3,	Table 9-2,	p. 9-7		
			Industrial cus	stomer data				
	26.1	Please that is	e state the perc represented by	centage of th RS 5 Gener	ne total 20 ral Firm Sa	16 sales cu les.	stomer (demand forecast
Resp	onse:							
2016	Foreca	ast Sales	Volumes for RS	6 5 are 1.8 p	ercent of th	e total fored	cast sale	s volumes.
	26.2	Please foreca	e state the perce ist that is repres	entage of the ented by RS	e total 2016 25 Genera	6 transportat al Firm Tran	tion cust sportatio	omer throughput on.

Response:

2016 Forecast Transportation Service Volumes for RS 25 are 17.7 percent of the total non-bypass transportation service volumes (including BC Hydro IG and Vancouver Island Joint Venture volumes, but excluding Bypass customers and Teck Coal (Byron Creek) volumes). If the Bypass customers and Teck Coal (Byron Creek) volumes are included in the total, the RS 25 share becomes 15.7 percent.

- 26.3 Please provide an updated version of Table 9-2 by splitting the row with "RS 22 / 22A / 22B – Large Volume Transportation" into 3 rows: (i) RS 22; (ii) RS 22A; and (iii) RS22B.
- Response:
- An Updated Table 9-2 is provided below, with Large Volume Transportation RS 22, RS 22A and
- RS 22B number of customers and volumes shown separately.



-	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 132

Table 9-2 (updated): Industrial Customer Data²¹

Rate Schedule	2016 Average Number of Customers	2016 Demand Forecast (PJ)	Percentage of Industrial Total
RS 4 – Seasonal	18	0.1	0.1%
RS 5 – General Firm Sales	230	2.2	3.1%
RS 25 – General Firm Transportation	566	13.5	19.4%
RS 7 – General Interruptible Sales	5	0.2	0.3%
RS 27 – General Interruptible Transportation	108	6.5	9.3%
Large Volume Transportation			
RS 22	26	13.2	
22A	9	9.0	
22B	<u>5</u>	<u>5.3</u>	
Total Large Volume Transportation	40	27.6	39.6%
Large Industrial Contract	2	19.7	28.3%
Industrial Total	984	69.7	100.0%

2

²¹ 2016 Forecast Customers and Energy from the compliance filing for the Annual Review for 2016 Rates (Order G-193-15).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 133

1	27.0	Reference:	RATE DESIGN FOR INDUSTRIAL CUSTOMERS
2			Exhibit B-1, Section 9.1, p. 9-1
3 4			General Firm Service (RS 5) and General Firm Transportation Service (RS 25)
5		On page 9-1	of Exhibit B-1, FEI states:
6 7 9 10 11 12 13 14		FEI's factor has a load formu peak not fu FEI is 1.25 te	General Firm Service (RS 5 and RS 25) is designed to serve high load and process customers with efficient utilization of the system. RS 5/RS 25 Demand Charge designed to provide lower average rates to these higher factor customers. The Demand Charge includes a peak day demand la with a 1.25 multiplier to estimate the peak day demand from the average monthly demand. Based on peak daily consumption information that was Ily available when the RS 5/RS 25 demand charge was originally designed, proposing to update the multiplier in the peak day demand formula from to 1.1.
15		On page 9-9	of Exhibit B-1, FEI states:
 16 17 18 19 20 21 22 23 24 25 26 27 28 		For pr custor Dema estima shows peak Dema estima custor averag requir	urposes of calculating the Demand Charge, RS 5 and RS 25 estimate a mer's peak day demand (referred to in the rate schedules as the "Daily nd") through a formulaic calculation that includes a 1.25 multiplier to ate peak Daily Demand from peak monthly demand FEI's analysis that the current method of using a multiplier of 1.25 is over-estimating the day demand FEI considered various options for calculating the Daily nd FEI is proposing to maintain the formula to determine the Daily nd, but to update the multiplier from 1.25 to 1.10 to more accurately ate the RS 5/RS 25 average consumption during the 5 coldest days in the mers' respective region for the past 5 years compared to their peak monthly ge consumption The change in method to calculate the Daily Demand
28 29 30 31		price incent than sched	signals so that only customers with greater than 40% load factor have an ive to take service under RS 5/RS 25. Customers with a load factor less 40% should be taking service under FEI's Large Commercial rate ules.

32

3435 <u>Response:</u>

27.1

FEI does not know the number of years that the 1.25 multiplier has been resulting in an overestimation of customers' peak demand. However, FEI has undertaken the same analysis that yielded the 2015 results for the years 2011 to 2014 and 2016 and in each of the years the

overestimation of peak day demand for RS 5 and RS 25 customers.

Please state the number of years that the 1.25 multiplier has been resulting in an

39 current formula results in a higher demand in almost all cases. For customers whose load factor



,	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 134

- 1 was greater than 50 percent, the overestimation is approximately 35 percent to 43 percent
- 2 (current method result divided by modified formula result minus one times 100) for the latest
- 3 three years. The comparative results of the different methods to estimate daily demand as
- 4 shown in Table 9-9 is provided below for the years 2011 through 2016.

Average Daily Demand (GJ) Per Customer (Combined Totals for RS 5 and RS 25 Customers)

2011	Method 1	Method 2	Method 3	Method 4	Method 5	Method 6	
	Current	Current	FEI System	Average	Average	Modified	
	Formula for	Formula	Maximum	Consumption	Consumption	Formula	
	Daily	Updated	Day Send	on Coldest 3	on Coldest 5	with 5 Day	
	Demand	Multiplier	Out	Days	Days	Average	
< 40% Load Factor	129	107	178	195	172	172	
40% to <45% Load Factor	145	124	111	128	150	150	
45% to <50% Load Factor	112	127	102	92	87	[,] 87	
>50% Load Factor	91	84	84	79	77	77	
All Customers	99	87	93	86	84	84	

⁵

Average Daily Demand (GJ) Per Customer (Combined Totals for RS 5 and RS 25 Customers)

2012	Method 1	Method 2	Method 3	Method 4	Method 5	Method 6	
	Current	Current	FEI System	Average	Average	Modified	
	Formula for	Formula	Maximum	Consumption	Consumption	Formula	
	Daily	Updated	Day Send	on Coldest 3	on Coldest 5	with 5 Day	
	Demand	Multiplier	Out	Days	Days	Average	
< 40% Load Factor	164	208	160	164	170	170	
40% to <45% Load Factor	118	129	105	109	119	119	
45% to <50% Load Factor	93	111	94	87	90	90	
>50% Load Factor	94	83	85	87	82	82	
All Customers	99	87	100	100	94	94	

6

Average Daily Demand (GJ) Per Customer (Combined Totals for RS 5 and RS 25 Customers)

2013	Method 1	Method 1 Method 2		Method 4	Method 5	Method 6	
	Current	Current	FEI System	Average	Average	Modified	
	Formula for	Formula	Maximum	Consumption	Consumption	Formula	
	Daily	Updated	Day Send	on Coldest 3	on Coldest 5	with 5 Day	
	Demand	Multiplier	Out	Days	Days	Average	
< 40% Load Factor	171	195	157	151	165	176	
40% to <45% Load Factor	104	127	103	104	106	113	
45% to <50% Load Factor	83	98	98	81	85	91	
>50% Load Factor	101	83	81	78	81	93	
All Customers	101	89	93	86	89	103	



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 135

Average Daily Demand (GJ) Per Customer (Combined Totals for RS 5 and RS 25 Customers)

2014	Method 1	Method 2	Method 3	Method 4	Method 5	Method 6	
	Current	Current	FEI System	Average	Average	Modified	
	Formula for	Formula	Maximum	Consumption	Consumption	Formula	
	Daily	Updated	Day Send	on Coldest 3	on Coldest 5	with 5 Day	
	Demand	Multiplier	Out	Days	Days	Average	
< 40% Load Factor	131	97	150	125	118	118	
40% to <45% Load Factor	99	128	110	103	109	109	
45% to <50% Load Factor	71	91	75	92	76	76	
>50% Load Factor	115	85	94	83	85	85	
All Customers	101	89	100	90	88	89	

Average Daily Demand (GJ) Per Customer (Combined Totals for RS 5 and RS 25 Customers)

2015	Method 1	Method 2	Method 3	Method 4	Method 5	Method 6	
	Current	Current	FEI System	Average	Average	Modified	
	Formula for	Formula	Maximum	Consumption	Consumption	Formula	
	Daily	Updated	Day Send	on Coldest 3	on Coldest 5	with 5 Day	
	Demand	Multiplier	Out	Days	Days	Average	
< 40% Load Factor	174	149	160	150	159	152	
40% to <45% Load Factor	93	169	89	97	109	109	
45% to <50% Load Factor	73	87	82	77	72	72	
>50% Load Factor	105	84	25	71	72	75	
All Customers	100	88	82	77	77	80	

2

1

Average Daily Demand (GJ) Per Customer (Combined Totals for RS 5 and RS 25 Customers)

2016	Method 1	Method 2	Method 3	Method 4	Method 5	Method 6
	Current	Current	FEI System	Average	Average	Modified
	Formula for	Formula	Maximum	Consumption	Consumption	Formula
	Daily	Updated	Day Send	on Coldest 3	on Coldest 5	with 5 Day
	Demand	Multiplier	Out	Days	Days	Average
< 40% Load Factor	120	142	134	134	134	132
40% to <45% Load Factor	83	79	99	85	88	88
45% to <50% Load Factor	79	74	83	78	76	76
>50% Load Factor	140	101	95	96	97	98
All Customers	110	97	101	97	98	98

3

- 4 5
- Ŭ
- 6
- 7 8

9

10

27.2 Please explain the underlying reason(s) why the 1.25 multiplier is now overestimating peak day demand for RS 5 and RS 25 customers. For example:

 i. did inadequate/inaccurate data lead to the development of an inappropriate multiplier initially?



- 11	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
-	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 136

- ii. has the load factor/typical daily demand/typical peak day demand of RS 5/RS 25 customers changed overtime?
- 3 4
- iii. any other reasons?

5 **Response:**

6 FEI believes the primary reason was that detailed data was not available at the time the 7 multiplier was initially developed. However, the number of customers taking service under RS 5 8 and 25 has tripled since the 1.25 multiplier was adopted, so changes in demand profiles with 9 this customer growth may provide some explanation of why the 1.25 multiplier is too high.

10 The 1.25 multiplier was derived and approved as part of the 1996 Rate Design Application. It 11 was agreed to as part of the Negotiated Settlement which the Commission subsequently 12 approved through Order G-98-96, dated, October 7, 1996. At that time FEI had approximately 13 250 customers in the General Firm Sales and Transportation Service customers. A number of 14 those customers did not have daily measurement of consumption data and an estimation of 15 peak day demand was calculated as part of a workshop in 1996, which resulted in a factor of 16 1.25 that represented estimated peak day consumption to highest average day consumption of 17 any winter month.

18 Subsequently, all or almost all General Firm customers have metering that provides daily 19 measurement and so it is possible to calculate a more accurate multiplier that takes into 20 consideration the actual daily consumption measurement of all customers, which was not 21 possible in 1996.

- 22
- 23
- 24
- 25
- 26 27
- 27.3 Please explain the issue(s) FEI or customers may experience due to an overestimation of the peak demand based on the multiplier of 1.25. Please use calculations to explain the impact to customer bills and FEI's operations.
- 28

29 Response:

The first issue FEI is addressing through the rate design for General Firm Service customers is to adopt a method that is a better estimation of customers' peak demand for billing purposes, i.e., the Daily Demand. The Daily Demand needs to be an appropriate billing determinant for customers' firm service use of the delivery system. This is particularly true for the majority of customers whose load factor is greater than 50 percent for which the 1.25 multiplier exceeds customers' peak demand (refer to response to BCUC-FEI IR 1.27.1).

The second issue is to have a Demand Charge rate that provides a price signal to attract customers whose loads are sufficiently large enough when coupled with their load profile for it to



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 137

1 make economic sense to be served under RS 5/25 (refer to the response BCUC-FEI 1.31.2 for

2 the economic crossover volumes at various load factors).

3 Total revenue from the RS 5/25 is not an issue as the R:C ratio is within the range of 4 reasonableness. Please refer to Exhibit B-1, Table 12-2, Page 12-5, Initial COSA result and 5 COSA after Rate Design Proposals.

6 The important point is that the two changes FEI has proposed will better align determination of 7 customer peak for billing purposes and rates to incent customers with sufficient load and load 8 factor to take service under the appropriate rate schedule, i.e., General Firm Service or Large 9 Commercial Service. FEI's goal is to get the correct pricing signal for the service being provided 10 and to appropriately recover the cost of service.

11 The first two rows in the following table provides the average customer bill impact using the

12 2015 Average Daily Demand for all customers using the 1.25 and 1.10 multipliers, respectively,

13 times the COSA Based Demand Charge. The last row shows the Average Daily Demand using

14 the proposed 1.10 multiplier times the Proposed Demand Charge.

2015 Average Daily Demand	Demand Charge \$ / GJ	Annual Demand Charge Revenue
100	\$21.596	\$25,915
88	\$21.596	\$22,805
88	\$24.596	\$25,974

15

16 Using a more accurate multiplier for determining daily demand to be applied to the demand 17 charge will have no effect on FEI operations.

18

19

20

- 2127.4Please calculate the difference to the bills of moving from a 1.25 multiplier to a221.1 multiplier for the five largest RS 5/RS 25 customers based on 2015 actual23consumption and load factor.
- 2425 **Response:**

The following table shows, for the five largest RS 5/25 customers, the actual annual volumes, the daily demand using the 1.25 multiplier and using the 1.10 multiplier, the annual demand charges for each, and the difference. For the fifth largest customer the variance is approximately

29 \$19 thousand and for the largest customer it is approximately \$52 thousand.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 138

		Current Method		Proposed Me	ethod	Annual Demand Charges				
		Daily		Daily						
		Demand		Demand						
	Actual	From 1.25		From 1.1						
	Annual	Multiplier	Load	Multiplier	Load	Demand	From 1.25	From 1.10		
	Volume (GJ)	(GJ)	Factor	(GJ)	Factor	Charge	Multiplier	Multiplier	Dif	ference
Customer 1	384,477	1,666	63%	1,466	72%	\$ 21.596	\$ 431,747	\$ 379,938	\$	51,810
Customer 2	293,681	1,369	59%	1,205	67%	\$ 21.596	\$ 354,779	\$ 312,206	\$	42,573
Customer 3	214,147	1,035	57%	911	64%	\$ 21.596	\$ 268,222	\$ 236,036	\$	32,187
Customer 4	200,759	982	56%	864	64%	\$ 21.596	\$ 254,487	\$ 223,949	\$	30,538
Customer 5	176,803	619	78%	545	89%	\$ 21.596	\$ 160,415	\$ 141,165	\$	19,250

The annual demand charge is calculated by multiplying the daily demand (GJ) x 12 x Demand Charge. Please note that the results in the table are reflective of the impact of the proposed change in the multiplier only, and do not include the mitigating effect of the proposed higher demand charge for RS 5/25. If both proposed changes are considered (i.e., a 1.10 multiplier and the higher proposed demand charge), the annual demand charges for these customers will be almost the same as under the current formula.

- 8
- 9
- 10
- 10 11
- 12

27.5 Please provide an updated version of Table 9-4 on page 9-11 of Exhibit B-1 to include (i) existing RS 5 and RS 25 charges; and (ii) FEI's proposed RS 5 and RS 25 charges.

13 14

15 **Response:**

16 The following table provides the existing charges, effective January 1, 2017 for Rate Schedules 17 5 and 25 and then provides the proposed charges for these two rate schedules. The increases 18 in the Demand Charge and Delivery Charge from the existing rates to the proposed rates also 19 include the effect of known and measurable changes as described in Exhibit B-1, Section 6.3.2 20 of the Application.

> **RS 5 RS 25** Existing Charges effective January 1, 2017 Basic Charge \$ / Month \$587.00 \$587.00 Demand Charge \$ / Month / GJ of Daily Demand \$20.077 \$20.077 Delivery Charge \$ / GJ \$0.825 \$0.825 Administration Charge \$ / Month N/A \$78.00 Multiplier for Daily Demand 1.25 1.25



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 139

	RS 5	RS 25
Proposed Charges ¹⁾		
Basic Charge \$ / Month	\$587.00	\$587.00
Demand Charge \$ / Month / GJ of Daily Demand	\$24.596	\$24.596
Delivery Charge \$ / GJ	\$0.887	\$0.887
Administration Charge \$ / Month ²	N/A	\$39.00
Multiplier for Daily Demand	1.10	1.10
s:	1.10	1.10
e Exhibit B-1, Section 12.4, page 12-9 for proposed ch	arges.	

- 27.6 Please provide in table form, for each of RS 5 and RS 25, the percentage of costs recovered through all fixed charges using (i) existing charges; and (ii) FEI's proposed charges.
- 12 Response:

From the following table the percentage of fixed revenue to total Delivery Margin Revenue for
RS 5 at existing rates and proposed rates is 72 percent and 71 percent (Line 15), respectively.
For RS 25 the corresponding percentage for both existing rates and proposed rates is
approximately 65 percent (Line 15).

17 The source data for number of customers and annual volumes is from the 2015 actual billed 18 data. Daily Demand using the 1.25 Multiplier is the summation of the daily demands for each 19 customer using the current methodology to derive Daily Demand. Daily Demand using the 1.10 20 Multiplier is derived using the results for 1.25 Multiplier and dividing it by 1.25 times 1.1.



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date:
	00110 0, 2011
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 140

Line		Existing Rates ¹⁾		Rates ¹⁾ Proposed		d Rates
No.	Particulars	RS 5	RS 25		RS 5	RS 25
1	Number of Customers	239	535		239	535
2	Annual Volume (TJ)	2,280	12,751		2,280	12,751
2	Daily Demand Using 1.25 Multiplier (GJ)	12,784	64,537			
5	Daily Demand Using 1.10 Multiplier (GJ)				11,250	56,792
4	Monthly Charges					
5	Basic Charge \$ / Month	\$ 587.00	\$ 587.00		\$ 587.00	\$ 587.00
6	Administration Charge \$ / Month	N / A	\$ 78.00		N / A	\$ 39.00
7	Demand Charge \$ / Month / GJ of Daily					
,	Demand	\$ 20.077	\$ 20.077		\$ 24.596	\$ 24.596
8	Delivery Charge \$ / GJ	\$ 0.825	\$ 0.825		\$ 0.887	\$ 0.887
9	Revenues (\$000's)					
10	Total Monthly Charge Revenue ²⁾	\$ 1,684	\$ 4,269		\$ 1,684	\$ 4,019
11	Demand Charge Revenue ³⁾	3,080	15,548		3,320	16,762
12	Total Fixed Charges Revenues	4,763	19,818		5,004	20,781
13	Delivery Charge Revenue ⁴⁾	1,881	10,519		2,022	11,310
14	Total Delivery Margin Revenue	\$ 6,644	\$ 30,337		\$ 7,026	\$ 32,091
15	% of Fixed Charge Revenue to Total					
13	Delivery Margin Revenue ⁵⁾	71.7%	65.3%		71.2%	64.8%

- 1) Refer to the response to BCUC-FEI IR 1.27.5 for the existing charges and proposed charges.
- 2) Total Monthly Charge Revenue is equal to Line 1 x 12 months x Line 5 / 1,000 for RS 5 whereas for RS 25 it is the sum of Lines 5 and 6.
- 3) Demand Charge Revenue is equal to Line 3 x 12 months x Line 7 / 1,000.
- 4) Delivery Charge Revenue is equal to Line 2 x Line 8.
- Percentage of Fixed Charge Revenue to Total Delivery Margin Revenue is equal to Line 12 / Line 14.



TN .	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 141

- 1
- 2 3

27.6.1 Please discuss any considerations FEI made when determining the RS 5 and RS 25 proposals, regarding the percentage of costs recovered through fixed charges.

5 **Response:**

In the process of reviewing the charges applicable to RS 5 and RS 25, FEI did not consider it
desirable or necessary to change the Basic Monthly Charge. As shown in the response to
BCUC-FEI 1.27.6, the fixed charges for these rate schedules recover approximately 65 percent
to 70 percent of the delivery cost of service. By way of a separate assessment, FEI is proposing
to decrease the monthly Administration Charge applicable to RS 25.

11 In addition to already recovering a large portion of the overall costs through fixed charges, the 12 rationale for not changing the Basic Monthly Charge is that the revenue to cost ratio and margin

13 to cost ratio for RS 5 and RS 25 are reasonable and that the change to the determination of

14 Daily Demand and Demand Charge is to incent smaller volume customers (generally less than

15 10,000 GJ per year of demand) and less than 40 percent Load Factor, to be served under Large

16 Commercial Service (RS 3 or RS 23).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 142

1 28.0 Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS

2

3

4

5

6

7

8

Exhibit B-1, Section 9.5.5.1, Table 9-11, p. 9-20; Appendix 9-2, p. 1

RS 5 and RS 25 daily demand new multiplier calculation

- 28.1 On page 9-20 of Exhibit B-1, FEI states that Appendix 9-2 contains a detailed description of the method for deriving the RS 5 and RS 25 daily demand multiplier. Please provide a functional Excel spreadsheet showing the data and calculation of the multiplier (1.02) for 2015, as seen in Table 9-11. Please arrange the spreadsheet data and calculations in a way that follows the sequence of steps outlined in Appendix 9-2.
- 9 10

11 Response:

The functional Excel spreadsheet in Attachment 28.1 provided in response to this question is being filed confidentially with the Commission, pursuant to Section 18 of the Commission's Rules of Practice and Procedure regarding confidential documents, established by Order G-1-16. FEI requests that the response be kept confidential as it contains confidential customer information for which FEI does not have the authority or permission to disclose. Given the private and commercially sensitive nature of the information, FEI submits that only the Commission should have access to the confidential spreadsheet.

19

20

27

21
22 28.2 Appendix 9-2 shows that data from 2011 to 2015 was used to calculate the new multiplier for RS 5 and RS 25. Please explain the effort required to update the multiplier for RS 5 and RS 25 Daily Demand on an annual basis beginning with data from 2012 to 2016. Please include a discussion regarding costs, time and effort in person-hours.

28 **Response:**

FEI estimates based on managers' experience in performing the calculations that the incremental activity to calculate the year's multiplier from all General Firm customers is 3 days FTE equivalent. A significant portion of the time is reviewing the raw data for each customer to ensure the details do not include customers who are no longer in Rate Schedules 5 or 25, and that there are no anomalies or missing data. FEI expects that there would be no, or minimal, incremental costs associated with this effort.

- 35
- 36
- 37



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 143

- 28.3 Please explain the benefits and disadvantages of updating the multiplier on an annual or biennial basis using recent years of actuals, as was done for this Application.
- 3 4

2

5 **Response:**

6 The benefit of updating the multiplier on an annual or biennial basis would be that there would 7 be assurance that the multiplier remains accurate. However, updating the multiplier on an 8 annual or biennial basis using recent years' actuals is not necessary, and would lead to 9 changes in other RS 5/25 charges and in other rate schedules, increasing rate instability and 10 decreasing customer understanding:

- Given that the population size of the RS 5/25 rate class is more established now as compared to when the original multiplier was established, FEI does not expect there will be a need to change the multiplier from year to year.
- The multiplier should not be looked at in isolation. Updating the multiplier may require changes to the RS 5/25 demand and Delivery Charges to maintain the proper price signals.
- Updating the multiplier or other components of the RS 5/25 charges would impact the rates in other rate classes, especially RS 7, RS 27 and RS 4 which are derived from RS 5/25.
- Frequent changes to the RS 5/25 charges, and other industrial rates, will increase rate instability and decrease customer understanding of the multiplier billing determinants.

For these reasons, reviewing the multiplier as part of a general rate design application, rather than more frequently, is more appropriate.


FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 144

1 **RATE DESIGN FOR INDUSTRIAL CUSTOMERS** 29.0 **Reference:**

2 3

4

- Exhibit B-1, Section 9.5.5, p. 9-16; Table 9-8, p. 9-17; Table 9-9, p. 9-18; Section 9.4, Table 9-3, pp. 9-8 to 9-9
 - RS 5 and RS 25 peak day demand estimate options and evaluation
- 5 On page 9-17 of Exhibit B-1, FEI provides Table 9-8 which shows the number of customers by load factor segment with the combined total for RS 5 and RS 25. 6
- 7 On page 9-18 of Exhibit B-1, FEI provides Table 9-9 which shows the average daily 8 demand by load factor segment with the combined total for RS 5 and RS 25.
- 9 On page 9-17 of Exhibit B-1, FEI states:

10 [F]or approximately 450 of the 774 customers the current method yields an 11 average Daily Demand that is 46% higher than the average consumption on the 12 five coldest days (105 GJ / 72 GJ - 1).

13 29.1 Please confirm, or correct where necessary, the figures in the following table and that the highlighted section of the table refers to the quotation from 9-17 of 14 15 Exhibit B-1 in the preamble.

	Column 1	Column 2	Column 3	Column 4	Column 5	Column 6
	N		customers	Average Daily Demand (GJ) per Customer		
	Load Easter Segment	Method 1 ⁽¹⁾	Method 4 (1)	Method 1 ⁽²⁾	Method 4 ⁽²⁾	
	(RS 5/25 Combined)	Current Formula for Daily Demand	Average Consumption on 5 Coldest Days	Current Formula for Daily Demand	Average Consumption on 5 Coldest Days	Percentage difference ⁽³⁾
Row 1	Customers with Zero Demand	1	4			-
Row 2	< 40% Load Factor	55	33	174	159	9%
Row 3	40% to < 45% Load Factor	75	43	93	109	-15%
Row 4	45% to < 50% Load Factor	196	87	73	72	1%
Row 5	> 50% Load Factor	447	607	105	72	46%
Row 6	All customers	774	774	100	77	30%
	Notes					
	(1) Data from Exhibit B-1, Table 9-	8, p. 9-17				
	(2) Data from Exhibit B-1, Table 9-9, p. 9-18					

- (3) This column shows the percentage difference between (i) the average daily demand determined using the current formula for daily demand; and (ii) the average consumption on the 5 coldest days using 2011 - 2015 historical data
- 16 17

18 Response:

19 FEI confirms the numbers in the table of this IR are consistent with the numbers contained in 20 Exhibit B-1, Tables 9-8 and 9-9. FEI also confirms the quote from page 9-17 is related to the

21

results highlighted in Row 5. Similar percentage differences relative to Table 9-9 can be

22 achieved by comparing the average daily demand from the Current Formula (Method 1) to the

23 average Daily Demand from Method 4 using the average on the coldest 3 days or the Modified

24 Formula (Method 5). When choosing a method to calculate Daily Demand, it is important not to

25 select a method that can result in an undesired anomalous consequence such as customers



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 145

having a zero demand. The purpose of selecting the method for determining Daily Demand is
not to estimate the peak day demand of General Firm customers, but rather to generate a billing
determinant that can be applied to the demand charge for the firm service use of the delivery
capacity of FEI's system. The results presented in Table 9-8 are supportive of using Method 2 –
Current Formula Updated Multiplier (1.10) or Method 5 – Modified Formula with 5 Day Average,

6 both of which have the least number of customers with a zero demand (i.e., 1 customer).

- 7
- 8 9

10 On page 9-16 of Exhibit B-1, FEI explained that they considered five options for 11 estimating peak demand: (i) Status Quo/Current Formula; (ii) Current Formula with 12 Updated Multiplier; (iii) FEI System Maximum Day Send Out; (iv) Average Consumption 13 on 3 or 5 Coldest Days in Region; and (v) Modified Formula.

- 1429.2FEI conducted a review of industrial rates offered by Canadian natural gas15utilities and summarized the results in Table 9-3 on pages 9-8 and 9-9 of Exhibit16B-1. For each of the Canadian natural gas utilities that use a demand charge for17industrial customers, please state which of FEI's five options listed in the18preamble was used to estimate peak day demand or elsewhere in the19determination of their demand charge.
- 20

21 Response:

Based on FEI's review it would appear that none of the other utilities use any of the five options employed by FEI in its analysis, and that there is diversity in what is done at each utility. For instance, the eastern utilities use cubic meter volume and contract demand for the billing determinant whereas Atco Gas and AltaGas use energy (gigajoules).

Union Gas Rate 100 and Enbridge Rate 110 use contracted demand in cubic meters. Union
Gas Rate 2 makes a distinction for contract demand less than and greater than 70,000 cubic
meters.

The Atco Gas and AltaGas billing determinant is expressed in energy (gigajoules), as is FEI's, versus the eastern Canadian utilities that use cubic volume (cubic meters). Another similar feature that AtcoGas and AltaGas employ is to divide the summer volumes by two, which FEI also does to determine the Daily Demand for the same April 1 to October 31 period. (For Atco Gas, if the customer is **only** taking service during the summer period then the amount of gas is **not** divided by 2 in determining the billing determinant).

The following table shows how the billing determinant is derived for service provided by the five utilities other than FEI that have demand charges – Atco Gas, Alta Gas, Union Gas, Enbridge and Gazifere.



4	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 146

Utility	Billing Determinant
Atco Gas and Pipelines Ltd. – North: High Use Delivery Service	Determination of Billing Demand: The Billing Demand for each billing period shall be the greatest amount of gas in GJ delivered in any Gas Day (i.e., 8:00 am to 8:00 am) during the current and preceding eleven billing periods provided that the greatest amount of gas delivered in any Gas Day in the summer period shall be divided by 2. Provided that for a Customer who elects to take service only during the summer period, the Billing Demand for each billing period shall be the greatest amount of gas in GJ in any Gas Day in that billing period. In the first contract year, the Company shall estimate the Billing Demand from information provided by the Customer.
Alta Gas: Optional Demand General Service – Rate No. 3	 The billing demand shall be the greater of: i) 100 GJ or ii) The Contract Demand or iii) The greatest amount of gas in GJ in any consecutive 24-hour period during the current and preceding eleven billing periods provided that the greatest amount of delivered in any 24 consecutive hours in the summer period (April 1 to October 31) shall be divided by 2.
Union Gas: Rate 2 Medium Volume Firm Service & Rate 100: Large Volume High Load Factor Firm Service	Contracted Daily Demand in cubic meters
Enbridge: Rate 110 – Large Volume Load Factor Service	Contract Demand in cubic meters
Gazifere: Rate 5 – Large Volume Firm Service	Minimum annual volume: Subscribed Volume (m ³) x number of days in year x load factor provided for in the contract (Minimum load factor is 50%)



SBC™	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 147

1	30.0 Refer	ence: RATE DESIGN FOR INDUSTRIAL CUSTOMERS
2 3		Exhibit B-1, Section 9.4, Table 9-3, pp. 9-8 to 9-9; Section 9.5.3.3, pp. 9-12; Section 9.5.3.5, p. 9-14
4		Minimum load factor eligibility criterion for RS 5 and RS 25
5	On pa	age 9-12 of Exhibit B-1, FEI states:
6 7 8 9 10		FEI reviewed firm industrial rates offered by natural gas utilities in other jurisdictions. Based on this review, a demand charge with a volumetric delivery charge rate design is used by 6 out of 10 Canadian utilities as shown in Table 9- 3. That is, six of the ten utilities surveyed used some form of demand charge. Also, three utilities required a minimum load factor to qualify for the rate.
11	On pa	age 9-14 of Exhibit B-1, FEI states:
12 13 14 15 16		RS 5 and RS 25 are designed for customers with higher load factors of 40% or above. The Demand Charge in RS 5 and RS 25 results in these higher load factor customers receiving a lower average cost. Customers with load factors lower than 40% should generally be taking service under Large Commercial Service RS 3/RS 23, where the average load factor is approximately 37%.
17 18 19	30.1	Please confirm, or otherwise explain, that when FEI is excluded from Table 9-3 it is five of nine utilities surveyed that used some form of demand charge.
20	<u>Response:</u>	
21	Confirmed.	
22 23		
24 25 26 27 28 29	30.2 Response:	Please discuss the benefits and disadvantages of using a minimum load factor eligibility criterion for RS 5 and RS 25 in a manner similar to Union Gas, Enbridge Gas or Gazifere.
30	EEL consider	s that the preferable option is to design the rate so that it is "self-policing" and
31	allows custor	ners to choose the service they would like or need on a prospective basis based on

FEI considers that the preterable option is to design the rate so that it is "self-policing", and allows customers to choose the service they would like or need on a prospective basis based on the customer's economics and business needs. Rates should be designed so that customers can choose the appropriate service they need based on how the billing determinants, Daily Demand and Annual Demand, are derived, coupled with the price(s) for the Demand Charge and Delivery Charge. If the proper price signals are in place, as proposed, then customers without a sufficient load factor and / or annual load will not choose to take service under RS 5 or



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 148

- 1 RS 25. FEI believes that the RS 5 and RS 25 have been working in this way since 1996, and
 2 sees no need to impose a minimum load factor.
- 3 The benefits of using a minimum load factor include:
- Customers have an incentive to maintain a higher load factor (or manage peak demand use) to qualify for the higher load factor rate.
- 6 The disadvantages of using a minimum load factor include:
- The load factor threshold is somewhat arbitrary and customers that fall just under the
 threshold are penalized by being grouped along with low load factor customers.
- 9 Customers with load factors less than the minimum load factor, but with sufficient annual volume would be harmed if forced to take service under a different service offering that had higher annual charges.
- Customers can be incented to 'flare' gas in off-peak period in order to achieve the minimum load factor to compensate for significant restart from a production downturn in an off-peak period for equipment maintenance or other customer economic/business reasons.
- Load factors can change from year to year, which may require customers to be moved to
 different rates from year to year leading to increased administrative burden and rate
 instability.
- The addition of a load factor threshold would have significant impacts on some customers.
- Since the inception of a demand charge for RS 5 and 25 in 1996, the general assessment of FEI is that the service is working the way it was intended and what is required at this time is to make adjustments to how the Daily Demand is determined and to adjust the Demand Charge.
- 24
- 25
- 25

- 2730.3Please explain any differences in circumstances that would prevent FEI from28using a minimum load factor eligibility criterion for RS 5 and RS 25 in a manner29similar to Union Gas, Enbridge Gas or Gazifere.
- 31 **Response:**

Of the three other utilities mentioned in the question FEI does not believe that the Union Gas
 service offering is a relevant comparator in this case. The Union Gas Large Volume High Load
 Factor Firm service offering with a minimum volume of 3,825 GJ per day would be the



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 149

equivalent of FEI's Large Transportation Service RS 22. This is not comparable to FEI's RS
 5/25 service.

FEI is unaware of any circumstances related to types of customers that would prevent FEI from using a minimum load factor eligibility criterion. Implementing a minimum load factor eligibility criterion could require the collection of metered daily demand amounts and ongoing monitoring of whether each customer would still qualify based upon the minimum load factor criterion. The review of customer accounts to see if customers continue to meet the criterion and if the customer(s) should be transferred to another rate schedule would create unnecessary additional work for both FEI and the customer.

10		
11		
12		
13	30.4	If FEI were to implement a minimum load factor eligibility criterion for RS 5 and
14		RS 25:
15		i. Please explain what the desired minimum load factor would be; and
16		ii. Please explain how FEI could implement the minimum load factor and
17		ensure customers are being placed in the appropriate rate class and being
18		billed accordingly.
19		

20 Response:

FEI does not recommend a minimum load factor eligibility criterion. The rest of the response is based on adopting a load factor eligibility criterion, contrary to FEI's recommendation.

Load Factor is a derived value of average consumption divided by peak consumption; for FEI, it is average day consumption divided by peak consumption. The derivation of the load factor is not as important as the derivation or definition of peak consumption. In response to the two questions posed above:

i) In FEI's judgment, the minimum Load Factor should be 40 percent; the class average is
 anticipated to be approximately 50 percent to 55 percent.

ii) FEI would review customers' historical daily demands and consequent load factors to
 see if the customer should be moved to an alternate rate schedule. The review would
 also consider the forecast demand and expected load factor as well.

Whether or not a minimum eligibility criterion is adopted, what is most important is the determination of the appropriate Daily Demand and the Demand Charge. A proper determination of Daily Demand with the Demand Charge should be 'self-policing' to incent customers on a prospective basis to take service under the most economic rate schedule. To ensure these firm customers have an appropriate billing determinant, FEI recommends using Method 2 or Method 5. With an appropriate determination of Daily Demand (or Peak), a



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 150

- 1 customer's Load Factor can be derived. By adopting Method 2 or Method 5, all customers would
- 2 fairly contribute to the recovery of the rate schedules' allocated cost of service.

3 With a minimum load factor requirement, similar to the Commercial customers, annual reviews

- 4 of customers' consumption and load factor would need to be done to identify customers that
- 5 should consider switching to another rate schedule.



אז	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 151

1	31.0	Refer	ence:	RATE DESIGN FOR INDUSTRIAL CUSTOMERS
2				Exhibit B-1, Section 9.5.4, p. 9-15; Section 9.5.6, Table 9-14, p. 9-23
3 4				Economic crossover volume between RS 3/RS 23 and RS 5/RS 25 at proposed rates
5		On pa	ige 9-′	15 of Exhibit B-1, FEI states:
6 7 8 9 10			The rates 40% relat RS 5	economic crossover volumes at the 2016 COSA rates show that the existing s provide sufficient incentive for customers whose load factor is less than to receive service under RS 3/RS 23, rather than RS 5/RS 25. There are ively few customers whose annual volumes would be high enough to make 5/RS 25 economic at a load factor lower than 40%.
11 12 13		Table econo is 19,8	9-14 omic ci 374 G	on page 9-23 of Exhibit B-1 shows that at 40 percent load factor, the rossover volume between RS 3/RS 23 and RS 5/RS 25 at the proposed rates J.
14		31.1	Plea	se state:
15 16			i.t	he number of RS 3/RS 23 customers currently above 19,874 GJ per year and the combined amount of throughput they represent;
17 18			ii. t a	he number of RS 3/RS 23 customers currently below 19,874 GJ per year and the amount of throughput they represent;
19 20			iii. t a	he number of RS 5/RS 25 customers currently above 19,874 GJ per year and the combined amount of throughput they represent; and
21 22 23			iv. t a	he number of RS 5/RS 25 customers currently below 19,874 GJ per year and the amount of throughput they represent.
24	<u>Respo</u>	onse:		
25	The r	equeste	ed info	ormation is provided in the table below. The number of customers and

In requested information is provided in the table below. The number of customers and volumes are from the 2016 billed data. For RS 3/23, the volumes have been weather normalized. For RS 5/25 the number of customers and volumes do not include the four bypass customers and do not include customers that are new or left these rate schedules during the year, as these customers would not have a full 12 months of data.



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
TN	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 152

Customer Count	Under 19,874 GJs	Over 19,874 GJs	Total
Rate Schedule 3/23	7,243	50	7,293
Rate Schedule 5/25	585	182	767
Total	7,828	232	8,080

Demand	Under 19,874 GJs	Over 19,874 GJs	Total
Rate Schedule 3/23 ¹	27,071,860	1,763,043	28,834,903
Rate Schedule 5/25	5,864,854	9,709,772	15,574,626
Total	32,936,714	11,472,815	44,409,529

¹2016 Weather Normalized Actuals

- 1 2
- 2

4 5

6

- - Please state the annual throughput required to make RS 5/RS 25 economic at a load factor of (i) 35 percent and (ii) 30 percent based on: (a) 2016 COSA rates; and (b) FEI's proposed rates.
- 7 8

9 Response:

- 10 The following table provides the annual throughput required to make RS 5/RS 25 economic at a
- 11 load factor of 35 percent and 30 percent based on the corrected (as discussed below) 2016
- 12 COSA rates and FEI's proposed rates:

Load	Economic Crossover					
Factor	2016 COSA Rates	Proposed Rates				
35%	20,020 GJ	(716,705) GJ				
30%	(83,029) GJ	(13,491) GJ				

13

As shown in the table above, at the 2016 COSA Rates there is no economic crossover at a 30 percent load factor as the mathematical solution results in a negative volume. At the FEI proposed rates there is no economic crossover at either 35 percent or 30 percent as the mathematical solution results in negative volumes.

In preparing this response FEI noticed that some of the rates used in the tables were incorrect.The corrected rate changes are the following:

- Table 9-7 RS 23 Delivery Charge was \$3.161, should have been \$3.188,
- Table 9-12 RS 23 Delivery Charge was \$3.175, should have been \$3.190,
- Table 9-12 RS 23 Monthly Charge was \$223.78, should have been \$184.78,



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
C	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 153

- Table 9-13 RS 23 Delivery Charge was \$3.175, should have been \$3.190; •
- Table 9-13 RS 23 Monthly Charges was \$223.78, should have been \$184.78; and
- 3

2

- Table 9-13 RS 25 Monthly Charges was \$665.00, should have been \$626.00.
- 4

5 Corrected tables are provided below. Table 9-7 and Table 9-13 have two additional rows added 6 to show the results at a load factor of 35 percent and 30 percent at the 2016 COSA rates 7 (corrected Table 9-7) and at FEI's proposed rates (corrected Table 9-13). Corrected Table 9-14 8 summarizes the corrected economic crossover volumes. Corrected Table 9-12 shows what the 9 load factors and economic crossover volumes would be using RS 23 Proposed Rates and RS 10 25 COSA charges using the 1.1 multiplier for the Peak Winter Month and associated Daily Demand.

11

12 13

Corrected Table 9-7: Large Commercial / General Firm Economic Crossover at Varying Load Factors at 2016 Approved Rates + Known and Measurable Changes

	RS 23			RS 25		
Monthly Charges (Basic + Adi	\$210.52	2		\$665.00		
Demand Charge		N / A			\$21.596	
Delivery Charge		\$3.188	3		\$0.887	
		Economic Cross-over (GJ/Year)	Dai Dema	ly and	Peak Winter Month With 1.25 multiplier	
	50%	6,191 GJ	34 (GJ	814 GJ	
	45%	7,541 GJ	46	GJ	1,102 GJ	
	40%	10,369 GJ	71 (GJ	1,704 GJ	
Load Factor	39%	11,351 GJ	80	GJ	1,914 GJ	
	38%	12,608 GJ	91 (GJ	2,182 GJ	
	37%	14,274 GJ 106 GJ		GJ	2,537 GJ	
	36%	16,589 GJ	126	GJ	3,030 GJ	
	35%	20,020 GJ	157	GJ	3,761 GJ	
	30%	(83,029) GJ	(758)	GJ	(18,198) GJ	



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 154

Corrected Table 9-13: Large Commercial / General Firm Economic Crossover at Varying Load Factors at Proposed Rates

		RS 23		RS 25			
Monthly Cha Admin. Fee)	arges (Basic + \$/Month	\$184.78	3 \$626.00				
Demand Ch	arge \$/GJ/Month	N / A		\$24.596		From Table 9-7 at 2016	
Delivery Cha	arge \$/GJ	\$3.190 \$0.887		\$0.887	COSA RATES		
		Economic Cross-over (GJ/Year)	Dail Dema	Peak Winter ly Month With and 1.1 multiplier		Daily Demand	Peak Winter Month With 1.25 multiplier
	50%	7,721 GJ	42 (GJ	1,154 GJ	34 GJ	814 GJ
	45%	10,463 GJ	64 (GJ	1,737 GJ	46 GJ	1,102 GJ
	40%	18,815 GJ	129 (GJ	3,515 GJ	71 GJ	1,704 GJ
	39%	23,063 GJ	162 (GJ	4,419 GJ	80 GJ	1,914 GJ
Load Factor	38%	30,253 GJ	218 (GJ	5,949 GJ	91 GJ	2,182 GJ
1 40101	37%	45,061 GJ	334 (GJ	9,100 GJ	106 GJ	2,537 GJ
	36%	93,232 GJ	710 (GJ	19,351 GJ	126 GJ	3,030 GJ
	35%	(716,705) GJ	(5,610)) GJ	(153,006) GJ	157 GJ	3,761 GJ
	30%	(13,491) GJ	(123)	GJ	(3,360) GJ	(758) GJ	(18,198) GJ

3

Corrected Table 9-14: Economic Crossover Volume at Proposed Rates (Corrected Table 9-13) Compared to at 2016 COSA Rates (Corrected Table 9-7)

Load Factor	Economic Crossover at Proposed Rates	Economic Crossover at 2016 COSA Rates
50%	7,721 GJ	6,191 GJ
45%	10,463 GJ	7,541 GJ
40%	18,815 GJ	10,369 GJ
39%	23,063 GJ	11,351 GJ
38%	30,253 GJ	12,608 GJ
37%	45,061 GJ	14,274 GJ
36%	93,232 GJ	16,589 GJ
35%	(716,705) GJ	20,020 GJ
30%	(13,491) GJ	(83,029) GJ



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 155

Corrected & Updated Table 9-12: Large Commercial / General Firm Economic Crossover at Varying Load Factors at Proposed Rates for RS 3/RS 23 but RS 5/RS 25 at 2016 COSA Rates with Proposed Multiplier

		RS 23			RS 25	
Monthly Charges (Basic + Admin.	\$184.78		\$665.00			
Demand Charge		N / A		\$21.596		
Delivery Charge		\$3.190)		\$0.887	
		Economic Cross-over (GJ/Year)	Daily Demand		Peak Winter Month With 1.1 multiplier	
	58.2%	5,800 GJ	27	GJ	745 GJ	
	52.0%	7,075 GJ	37	GJ	1,016 GJ	
	45.9%	9,785 GJ	58	GJ	1,593 GJ	
Load Factor	44.7%	10,755 GJ	66	GJ	1,799 GJ	
	43.4%	12,000 GJ	76 GJ		2,064 GJ	
	42.2%	13,680 GJ	89	GJ	2,422 GJ	
	41.0%	16,060 GJ	107	GJ	2,928 GJ	

31.3 Please explain if FEI considers that the customers with high annual volumes and with load factors below 40 percent should be classified as RS 3/RS 23.

Response:

No, FEI does not believe the relatively few customers with load factors below 40 percent but with sufficiently large enough loads to economically be better served under RS 5/25 should be forced to be served under RS 3/23. FEI believes what is important is to have an appropriate calculation of the Daily Demand and Demand Charge so that customers who have a sufficient annual demand and, generally, a load factor characteristic of 40 percent or higher would be best served under General Firm Service RS 5/25. Large volume customers that would be served in the General Firm rate schedules are sophisticated enough to determine which rate schedule would be the most appropriate for their needs and what it would cost them. The average Demand Charge of customers whose load factor is less than 40 percent will be higher than for a customer whose load factor is equal to or greater than 40 percent.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the C Information Request (IR) No. 1	ommission) Page 156

Please explain if the RS 5/RS 25 customers with high annual volumes

and with load factors below 40 percent would pay a higher overall bill if

- 1
- 2
- 3
- 4
- 5 **Response:**

31.3.1

Based on the 2015 Billed Data, using the proposed charges for RS 3/23 and RS 5/25, and using
the Current Method with the Updated Multiplier of 1.1 (Method 2) (Exhibit B-1, Table 9-8, Page
9-17), of the 26 customers whose load factors are less than 40 percent, 22 customers would
have a lower annual bill on RS 3/23 and 4 customers would have a higher annual bill on RS
3/23.

they were reclassified as RS 3/RS 23 customers.

After the Commission's Decision on this Application, FEI proposes to review the account history of all RS 3/23 and 5/25 customers to see if there are customers who should consider migrating from General Firm Service to Large Commercial Service or if there are Large Commercial Service customers who may be better off being served under General Firm Service. The discussions with customers will need to consider the customers' expected future consumption as well as their historical demand profile.

- 17
- 18
- 19
- 31.4 Please provide in table form, for each of RS 3 and RS 23, the percentage of
 costs recovered through all fixed charges using (i) existing charges; and (ii) FEI's
 proposed charges.
- 23

24 **Response:**

25 The requested information is provided in the table below. The percentage of revenue recovery

26 from the Basic Charge is higher for RS 3 than it is for RS 23 because of the lower average

27 annual use per customer for RS 3 sales customers. The result for RS 3 is a little over 13 percent

28 whereas for RS 23 customers it averages approximately 9.5 percent.

29 The existing rates are those approved by the Commission effective January 1, 2017.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 157

	Rate Schedule 3			Rate Sche	edule 23
	Existing	Proposed		Existing	Proposed
	Rates	Rates		Rates	Rates
Average Use per Customer (2015,					
Exhibit B-1, Figures 8-4 & 8-5, Page 8-6)	3,587	3,587		5,174	5,174
Basic Charge \$ / Day	\$ 4.3538	\$ 4.7895			
Basic Charge \$ / Month		-		\$ 132.52	\$ 145.78
Delivery Charge \$ / GJ	\$ 2.939	\$ 3.190		\$ 2.939	\$ 3.190
		-			
Basic Charge Revenue	\$ 1,590	\$ 1,749		\$ 1,590	\$ 1,749
Delivery Charge Revenue	10,542	11,443		15,206	16,505
Total Delivery Margin Revenue	\$ 12,132	\$13,192		\$ 16,797	\$ 18,254
% of Basic Charge Revenue to Total					
Delivery Margin Revenue	13.1%	13.3%		9.5%	9.6%



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
BC™	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 158

1	32.0	Reference:	RATE DESIGN FOR INDUSTRIAL CUSTOMERS
2 3			Exhibit B-1, Section 9.6.1, p. 9-25; Section 9.6.3.2, p. 26; Section 9.6.4, p. 9-21
4			RS 7 and RS 27 interruptible service charges
5		On page 9-2	5 of Exhibit B-1, FEI states:
6 7		Fl Spec	El offers the service at a discount from the General Firm Service rate. ifically, the existing delivery charges for RS 7/RS 27 are based on the
8 9		Gene plus f	the RS 5/RS 25 Delivery Charge The existing method has resulted in a
10 11		consi firm ı	stent discount of approximately 18% from the firm rate, where the effective rate is based on an 80% load factorFEI currently has a total of 113
12		custo	mers served under General Interruptible Service (sales and transport) that
13 14		incluc	les a wide range of industries such as asphalt plants, greenhouses,
15		avera	ige of 59,200 GJ per year. Figure 9-5 below shows that the annual demand
16		from	these customers ranges from about 5,000 GJ to 150,000 GJ.
17 18 19		32.1 Using disco using	FEI's proposed rates, please use calculations to show the percentage unt from the firm sales service would be for an interruptible sales customer an average of 59 200 GJ per year based on a RS 5/RS 25 Demand

- 19 20
- discount from the firm sales service would be for an interruptible sales customer using an average of 59,200 GJ per year based on a RS 5/RS 25 Demand Charge using a load factor of (i) 100 percent; (ii) 80 percent; (iii) 60 percent; and (iv) 40 percent.
- 21 22

23 Response:

The requested calculations and results are found in the table below. In responding to this IR for calculating the discount as a percent of Total Firm, it is important to note that for the customer Load Factor cases of 80 percent, 60 percent or 40 percent FEI made a simplifying assumption that the Peak Day is equal to the highest average day of any month. It is quite possible that a customer's actual peak day could be higher than the highest monthly average.



TC		FortisBC E 2016 Rate D	Submission Date: June 9, 2017				
15	BC.	Response to British Columbi Infor	sion) Page 159				
Line				Scen	arios		
No.		Particulars	i)	ii)	iii)	iv)	
1	Customer	s Operating Load Factor	100%	80%	60%	40%	
2	Proposed	Multiplier	1.1	1.1	1.1	1.1	
3	Effective	Load Factor	90.9%	72.7%	54.5%	36.4%	Line 1 / Line 2
4	Proposed	Demand Charge	\$ 24.596	\$ 24.596	\$ 24.596	\$ 24.596	
5	Months ir	n Year	12	12	12	12	
6	Days in Ye	ear	365	365	365	365	
7	Demand (Charge Effective Rate \$/GJ	\$ 0.889	\$ 1.112	\$ 1.482	\$ 2.224	Line 4 x Line 5 / Line 6 / Line 3
8	Delivery (Charge \$/GJ	0.887	0.887	0.887	0.887	
9	Total Effe	ctive Rate	\$ 1.776	\$ 1.999	\$ 2.369	\$ 3.111	Line 7 + Line 8
10	Proposed	Interruptible Rate \$ / GJ	\$ 1.443	\$ 1.443	\$ 1.443	\$ 1.443	
11	Different	ial \$ / GJ	\$ 0.333	\$ 0.556	\$ 0.926	\$ 1.668	Line 9 - Line 10
12	Discount	as a % of Total Firm	18.8%	27.8%	39.1%	53.6%	Line 11 / Line 9

2

1

3 As shown in the table above, a customer that takes natural gas service at a 100 percent Load 4 Factor, an 80 percent Load Factor, a 60 percent Load Factor or a 40 percent Load Factor will 5 have the following respective discounts as an Interruptible customer: 18.8 percent, 27.8 percent, 6 39.1 percent and 53.6 percent. A customer's annual volume has no effect on the determination 7 of the relative percentage of the discounts; it is the customer's demand profile (i.e., load factor) 8 and the proposed rates for firm and interruptible service that determine the percentage discount.

9 As such, FEI did not utilize the customer usage of 59,200 GJ stated in the question.

10 It is important to note that the multiplier of 1.1 will limit the effective load factor of a customer to 11 a maximum of 90.9 percent. The current methodology, using a multiplier of 1.25, will limit the 12 effective load factor of a customer to a maximum of 80 percent. As an example, if a customer 13 uses 100 GJ every day, i.e., operating at a 100 percent load factor, for billing under General 14 Firm Service, that customer will have a Daily Demand of 110 GJ (i.e., 100 GJ x 1.1). The 15 effective load factor then becomes 90.9 percent (average daily use of 100 GJ / Daily Demand of 16 110 GJ).

17 In conclusion, although the relative percentage of the discount significantly increases as the 18 load factor decreases, the RS 7/27 customers provide significant benefits that exceed the total 19 value of the discount. This is discussed in the response to BCUC-FEI IR 1.32.6.

20

21

FORTIS BC

TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
L	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 160

132.2Using FEI's proposed rates, please use calculations to show what the discount2from the firm transportation service would be for an interruptible transportation3customer using an average of 59,200 GJ per year based on a RS 5/RS 254Demand Charge using a load factor of (i) 100 percent; (ii) 80 percent; (iii) 605percent; and (iv) 40 percent.

7 **Response:**

6

8 The requested calculations and results are provided in the table below.

Line						
No.	Particulars	100%	80%	60%	40%	
1	Annual Volume (GJ)	59,200	59,200	59,200	59,200	
2	Days in Year	365	365	365	365	
3	Average Daily Volume	162.2	162.2	162.2	162.2	Line 1 / Line 2
4	Customers Operation Load Factor	100%	80%	60%	40%	
5	Customer's Operating Peak Day	162.2	202.7	270.3	405.5	Line 3 / Line 4
6	Proposed Multiplier	1.1	1.1	1.1	1.1	
7	Daily Demand	178.4	223.0	297.4	446.0	Line 5 x Line 6
8	Effective Load Factor	90.9%	72.7%	54.5%	36.4%	Line 3 / Line 7
9	Rate Schedules 5 & 25					
10	Proposed Demand Charge	\$ 24.596	\$ 24.596	\$ 24.596	\$ 24.596	
11	Proposed Delivery Charge	\$ 0.887	\$ 0.887	\$ 0.887	\$ 0.887	
12	Demand Charge Revenue	\$ 52,658	\$ 65,823	\$ 87,764	\$131,646	Line 7 x 12 months x Line 10
13	Delivery Charge Revenue	52,510	52,510	52,510	52,510	Line 1 x Line 11
	Total RS 5 / 25 Demand & Delivery					
14	Revenue	\$105 169	\$118 333	\$140 274	\$184 156	line 12 + line 13
	lievenide	<i>ų</i> 103,103	φ110,000	φ110, 2 /1	<i>ϕ</i> 10 1/100	
15	Pronosed Rate Schedules 7 & 27					
16	Proposed Delivery Charge S / GI	\$ 1//3	\$ 1//3	\$ 1 1/13	\$ 1//3	
10	riopsed benvery enarge \$7 G	у <u>1</u> .+-3	Ş 1.45	у <u>1</u> .+-Ј	у 1. у	
17	Delivery Charge Revenue	\$ 85 126	¢ 85 126	¢ 85 176	\$ 85 176	ling 1 x ling 16
1/	Derivery Charge Revenue	<u>3 83,420</u>	<u>3 83,420</u>	<u>3 03,420</u>	<u>3 83,420</u>	Line 1X Line 10
			+		+	
18	Dollar Value of Discount from Firm	Ş 19,743	\$ 32,908	\$ 54,849	\$ 98,731	Line 14 - Line 17
19	Discount % of Firm	18.8%	27.8%	39.1%	53.6%	

9 10

As shown in the table, for a customer using 59,200 GJ per year at load factors of 100 percent, percent, 60 percent or 40 percent the respective values of the discount would be \$19,743; \$32,908; \$54,849; or \$98,731. Line 19 shows the percentage of the discount which is the same as shown in the response to BCUC-FEI IR 1.32.1. In this case the size of the annual volume affects the dollar value of the discount, while not impacting the percentage of the discount. Although the value of the discount significantly increases as the load factor decreases, the RS



9

	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
-	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 161

7/27 customers provide significant benefits that exceed the total value of the discount. This is
 discussed in the response to BCUC-FEI IR 1.32.6.

5
6 32.3 Please state the number of customers that have switched from interruptible service (RS 7/RS 27) to firm service (RS 5/RS 25) for each year of the last five years of actual data.

10 Response:

For the period of January 1, 2012 to December 31, 2016 (last five years of actual data) there was 1 customer that switched from interruptible service (RS 7/27) to firm service (RS 5/25),

- 13 which was in 2014.
- 14
- 15
- 16

20

17 32.4 Please state the number of customers that have switched from firm service (RS
 18 5/RS 25) to interruptible service (RS 7/RS 27) for each year of the last five years
 19 of actual data.

21 **Response:**

As shown in the table below, for the period from January 1, 2012 to December 31, 2016 (the

23 last five years of actual data) there were 6 customers that switched from firm service (RS 5/25)

to interruptible service (RS 7/27).

Year	# Customers
2012	3
2013	1
2014	0
2015	1
2016	1
Total	6

25 26

20

27 28

29

32.5 Does FEI consider that, in general, the discount provided for interruptible service should be based on marginal rather than sunk costs? For example, being based



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 162

on the cost savings to the utility in providing interruptible instead of firm service

and/or the discount required to shift consumption away from peak periods and so

- 1
- 2
- 3
- 4

5 Response:

6 FEI does not consider that the marginal cost should be used as the basis for setting the Delivery 7 Charge discount for Interruptible service. The reason for this is the marginal costs for 8 interruptible service on the Transmission system and Distribution mains-related system are very 9 small or zero. For system planning on design day conditions, interruptible customers are a zero 10 load and no capacity and associated costs are incurred for serving interruptible customers. 11 Setting the pricing or discounts based on marginal costs would allow interruptible customers to 12 be a 'free rider' on FEI's Transmission system and Distribution Mains system. In order to have 13 interruptible customers contribute to the recovery of Transmission costs and Distribution Mains-14 related costs, the pricing for interruptible service should be based on a discount-based 15 approach from firm service rates, as has been the case since the 1996 Rate Design.

achieve cost savings for customers overall. Please explain.

- 16
- 17
- 18
- 19 32.6 Does FEI consider that it has optimized its use of interruptible rates to cost 20 effectively defer the need for new infrastructure investment? Please explain.
- 21

22 Response:

23 Yes, FEI considers it has optimized its use of interruptible rates to cost effectively defer the 24 need for new infrastructure investment. FEI has consistently shown in its 1993 Phase B, 1996, 25 2001 and 2016 Rate Design Applications that there is a net benefit in avoided capital costs and 26 the associated avoided cost of service from interruptible customers not receiving firm service.

27 In Exhibit B-1, Appendix 9-3, pages 2 and 3 show that the avoided capital cost of all interruptible 28 customers not taking firm service (Rate Schedules 7, 27 and 22) and RS 22 alone (page 3, Year 29 2016) is approximately \$134.2 million and \$40.15 million, respectively. From Table 9-19 on 30 page 9-30, FEI calculates the net benefit for all customers from RS 7/27 not taking firm service 31 to be \$5.0 million per year. The net benefit is derived by subtracting the \$2.3 million value of the 32 discount from the avoided cost of service of \$7.3 million.

- 33
- 34
- 35



אז	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 163

- On page 9-26 of Exhibit B-1, FEI states: "During the 1996 Rate Design, FEI established a discount for interruptible service from General Firm Service (RS 5/RS 25) based upon an 80% load factor."
- 4 5

2

3

- 32.7 Please state if FEI had proposed the use of an 80 percent factor in the 1996 Rate Design.
- 6

7 <u>Response:</u>

8 FEI confirms that FEI proposed the use of an 80 percent load factor in the 1996 Rate Design
9 Application – Amendments (Exhibit 2A) filed on September 30, 1996, Tab 3, pages 10 and 11.
10 The use of the 80 percent load factor was agreed to in the negotiated settlement that was
11 approved by the Commission.

- 12
- 13
- 14
 15 32.7.1 If FEI had originally proposed the use of an 80 percent load factor,
 16 please explain the reasons why and please discuss the applicability of
 17 these reasons to the proposals in the 2016 Rate Design Application.
- 18
- 19 **Response:**
- 20 FEI provides the rationale for both the 1996 and 2016 Rate Design Applications below.

21 Rationale in the 1996 Rate Design Application

In its 1996 Rate Design Application, FEI originally proposed a General Interruptible Rate at a discount from the average Firm Rate to reflect the customer costs of one to two days of curtailment per year or the risk of production losses. FEI proposed the following two adjustments to the Firm Rate:

- An alternative fuel adjustment, which deducted from the Firm Rate the incremental fuel cost of the customer for alternative fuels for two days of expected interruption over 365 days.
- An incentive adjustment, which deducted from the Firm Rate the estimated capital costs
 of the customer for equipment to burn the alternative fuel.

During the proceeding, FEI amended its proposal.²² Instead of making the two adjustments described above, FEI proposed to reach the same result by calculating the delivery charge for the General Interruptible rate as equivalent to the firm demand charge at an 80 percent load factor. As the General Firm customers' average load factor was 55 percent, using an 80 percent load factor for interruptible customers yielded a discount from the General Firm Rate.

²² 1996 Rate Design Application – Amended (Exhibit 2a).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 164

1 The proposed effective average firm rate for General Firm Service was \$0.78 per GJ, and the

2 proposed General Interruptible Service was \$0.64 per GJ. This resulted in a \$0.14 per GJ or 18

3 percent discount.²³ The load factor approach was agreed to in the negotiated settlement

4 approved by the Commission.

5 In summary, the General Interruptible rate was set at an 18 percent discount from the General

6 Firm rate to approximate the customer costs of one or two days of curtailment per year, and was

7 derived by using the equivalent of the firm demand charge at an 80 percent load factor for

8 interruptible customers.

9 Rationale in the 2016 Rate Design Application

10 FEI is proposing to maintain the discount that was approved in 1996 and again in 2001, by 11 updating the methodology to reflect the change in the Daily Demand formula (reducing the 12 multiplier from 1.25 to 1.10). As described on page 9-31 of the Application, after the change in 13 the Daily Demand formula, an 80 percent load factor firm customer would be a 90.9 percent 14 load factor customer. The Delivery Charge for the General Interruptible rate is therefore 15 proposed to be updated to be the equivalent of the firm demand charge at a 90.9 percent load 16 factor. As shown in Table 9-20 of the Application, this results in an 18.8 percent discount from 17 the firm rate.

18 19 20 21 22 32.7.2 If FEI had not originally proposed the use of an 80 percent load factor, 23 please explain the reasons that led to the use of the 80 percent load 24 factor. 25 26 **Response:** 27 Please refer to the response to BCUC-FEI IR 1.32.7.1. 28 29 30 31 32 On page 9-29 of Exhibit B-1, FEI states: 33 Over the past twenty years, interruptible customers have experienced a total of 34 approximately 19.5 days of capacity curtailment. On average, the annual curtailment is about one day per year. ... Based upon cold weather days where 35

²³ Exhibit 2A, 1996 Rate Design Application – Amended, Tab 5, Page 18).



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 165

- 1 all interruptible customers are curtailed, but not including capacity constrained 2 regions of the FEI system where partial curtailment happens every year, or for 3 FEI system maintenance related curtailment.
- 4
- 32.8 Please state the actual number of (i) partial curtailments in 2016; and (ii) FEI system maintenance related curtailments in 2016.
- 5 6

7 Response:

8 FEI does not track the statistics related to partial curtailments or to system maintenance related 9 curtailments as they often do not result in an actual curtailment for the customer, as described 10 further below.

11 When FEI referred to partial curtailments in the guoted passage, it was referring to limiting the 12 amount of interruptible capacity that may be available to those large industrial customers that 13 would be over their contracted firm capacity. These limitations may be above a customer's 14 expected consumption and therefore may not be a restriction at all.

15 FEI's system maintenance-related curtailments are usually in the summer and usually only 16 place limits on the amount of interruptible capacity that may be available during the 17 maintenance procedure. These limitations in many cases are above a customer's expected 18 consumption and therefore not a restriction at all. FEI also makes efforts to coordinate 19 maintenance procedures where possible with the large industrial customer's planned 20 maintenance outages.

- 21
- 22
- 23 24 32.9 Please state the average duration of (i) all partial curtailments in 2016; and (ii) all 25 FEI system maintenance related curtailments in 2016.
- 26
- 27 **Response:**
- 28 Please refer to the response to BCUC-FEI IR 1.32.8
- 29
- 30
- 31
- 32 32.10 Please state the longest duration of (i) all partial curtailments in 2016; and (ii) all 33 FEI system maintenance related curtailments in 2016.
- 34
- 35 Response:
- 36 Please refer to the response to BCUC-FEI IR 1.32.8



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 166

- 1
- 2
- 3 4

6

32.11 Please explain if some interruptible customers experience more curtailments than other interruptible customers due to their location on the FEI system.

7 Response:

8 Some interruptible customers may experience more curtailments, partial curtailments and/or 9 system maintenance-related curtailments than other interruptible customers due to their location 10 on the FEI system. The reasons for the curtailments could be numerous. Maintenance-related 11 curtailment work, for instance, may be more common for customers that are served directly off 12 of the transmission line, as system pressures in the transmission lateral must be reduced during 13 various aspects of the work for safety reasons, thereby reducing capacity on a temporary basis. 14 Another possible reason that some customers may experience more curtailments or partial 15 curtailments is that certain areas of the system are becoming or have become capacity 16 constrained.



BC [™]	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 167

1	33.0	Reference:	RATE DESIGN FOR INDUSTRIAL CUSTOMERS
2			Exhibit B-1, Section 9.8.1, Table 9-22, p. 9-37
3			Large volume transportation customers
4 5		Table 9-22 of and the 2016	on page 9-37 of Exhibit B-1 shows large volume transportation customers 3 annual demand forecast by rate schedule.
6 7 8 9 10		33.1 Pleas Colur Interr	e provide an updated version of Table 9-22 by splitting the Annual Demand nn into three showing the (i) Annual Firm Demand (TJ); (ii) Annual uptible Demand (TJ); and (iii) Annual Total Demand (TJ).

11 Response:

12 The requested information is provided in the following table.

				Firm	Interruptible	Total Annual	
		Rate Schedule	Customers	Demand	Demand	Demand	
		RS 22	26	732	12,457	13,189	
		RS 22A	9	10,878	0	9,030	
		RS 22B	5	4,215	1,061	5,277	
		Subtotal	40	15,825	13,518	27,496	
		Joint Venture	1	4,758	0	4,758	
		BC Hydro IG	1	16,425	0	16,425	
13		Total	42	37,008	13,518	48,679	
14			Units in th	e Demand	Columns are	all in TJ	
15							
16							
17							
18	33.2	Please state th	ne percer	tage of	the total F	El 2016 fo	recast throughput
19		represented by la	arge volum	ne transpo	rtation custo	mers, includii	ng RS 22, RS 22A,
20		RS 22B. VIGJV a	and BC Hv	dro IG.		-	0
21		,	,				

22 Response:

The forecast of volume for RS 22, 22A, 22B, VIGJV and BC Hydro IG as a percentage of the total 2016 forecast throughput is 22.7 percent.

Particulars	2016 Forecast (TJ)	% of Total Throughput
Total Sales Throughput (TJ)	121,772.2	58.6%
	10.404.0	0.00/
RS 22 Throughput (TJ)	13,164.9	6.3%
RS 22A Throughput (TJ)	9,048.5	4.4%



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 168

Particulars	2016 Forecast (TJ)	% of Total Throughput
RS 22B Throughput (TJ)	5,281.9	2.5%
VIGJV	4,758.0	2.3%
BC Hydro IG	<u>14,945.0</u>	<u>7.2%</u>
Subtotal – Large Industrial	47,198.3	22.7%
All Other T-Service Throughput (TJ)	<u>38,804.2</u>	<u>18.7%</u>
Total T-Service Throughput (TJ)	86,002.5	41.4%
Total Throughput (TJ)	<u>207,774.7</u>	<u>100.0%</u>



BC [∞]	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 169

1	34.0	Refere	nce: RATE DESIGN FOR INDUSTRIAL CUSTOMERS
2 3			Exhibit B-1, Section 9.8.1.4, p. 9-39; Section 9.8.3, pp. 9-42 to 9-43; Section 9.8.5.3, Table 9-27, p. 9-48; Section 9.8.5.4, p. 9-48
4			Large volume transportation – large industrial contract customers
5		On pag	ge 9-39 of Exhibit B-1, FEI states:
6 7 8 9 10			The VIGJV provides for the natural gas needs of five pulp mills and has a service contract for firm contract demand of 13,000 GJ per day which expires on December 31, 2017. FEI anticipates as an interim measure to extend the existing VIGJV contract until the Commission approved Rate Design becomes effective for RS 22.
11		On pag	ges 9-42 to 9-43 of Exhibit B-1, FEI states:
12 13 14 15 16 17 18 19 20 21 22			FEI reviewed the rate design for RS 22, the VIGJV and BC Hydro IG considering the rate design principles discussed above in Section 6.1, government policy and in light of the amalgamation of utilities. Based upon this review, FEI concluded that it should consider the potential for new cost-based firm and interruptible rates under RS 22 that would be applicable to all large industrial customers. Similar rates and rate structures for RS 22 and each of the VIGJV and BC Hydro IG may be more aligned with the fair apportionment of costs (Principle 2) and avoidance of undue discrimination among similar type customers (Principle 8). Large Industrial customers receiving similar service and having similar rates and rate structures would also be likely to improve customer understanding and acceptance (Principle 4).
23		On pag	ge 9-48 of Exhibit B-1, FEI states:
24 25 26 27 28 29			FEI will create a firm rate for RS 22, VIGJV and BC Hydro IG based on a cost allocation from the COSA model. Under this option, Tariff Supplement G-21 for Creative Energy would be terminated and the VIGJV could choose to become a RS 22 customer after its contract expires. The contract for BC Hydro IG would be included as a Tariff Supplement and, after the contract expires, BC Hydro could choose to become a RS 22 customer a RS 22 customer.
30 31		34.1	Please explain what FEI would do with the expiring VIGJC contract if the Commission does not approve FEI's proposal for RS 22 and VIGJV as described

34 **Response:**

in the preamble.

32

33

35 If the Commission does not approve FEI's proposal for RS 22 and the VIGJV, then FEI would 36 have to negotiate an extension or new agreement with the VIGJV that would need to be 37 submitted to the Commission for review and approval.



TN .	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 170

- 3 4 34.2 If the Commission does not approve FEI's proposal for large volume 5 transportation customers, could FEI establish new contracts based on costbased rates with VIGJV and BC Hydro IG after the expiration of the current 6 7 contracts? Please explain your response. 8 9 **Response:** 10 If the Commission does not approve FEI's proposal for large volume transportation customers, 11 then FEI could negotiate and try to establish new contracts based on cost-based rates with the 12 VIGJV and BC Hydro IG after the expiration of the current contracts. FEI's proposal for RS 22 13 firm service establishes cost-based firm rates for the current RS 22, VIGJV and BC Hydro IG 14 combined, based on cost allocation from the COSA model for this group of customers. 15 16 17 18 34.3 Please produce a table to discuss the similarities and differences between (a) the 19 average RS 22 customer, (b) Creative Energy, (c) VIGJV and (d) BC Hydro IG. 20 Please use figures where necessary and include a discussion for each on the: 21 annual throughput and expected changes in throughput over time; i. 22 ii. existing R:C ratios and M:C ratios before rate design proposals and 23 rebalancing; 24 iii. nature of the service (firm/interruptible) and the ability of the customer(s) 25 to manage interruptions in FEI's service; 26 iv. customer attributes (including load factor); 27 v. location on FEI's system and any special circumstances unique to that 28 customer or group of customers; and 29 vi. the incremental cost to FEI in providing service. 30 31 Response:
- 32 The following table provides the requested information:



TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
1	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 171

	Particulars	RS 22¹	Creative Energy	VIGJV	BCH IG
i)	# of Customers	26	1	1	1
''	Forecast Annual Throughput (T.I)	20			
	2016	11.441 TJ	1.748 TJ:	4.758 TJ	14.945 TJ
	2017	11.323 TJ	1.748 TJ	4.380 TJ	14.600 TJ
	2018	11.359 TJ	1.748 TJ	4.380 TJ	14.600 TJ
	2019	11.385 TJ	1.748 TJ	4.380 TJ	14.600 TJ
	2020	11,381 TJ	1,748 TJ	4,392 TJ	14,600 TJ
	The forecasted demand for all custor It should be noted that the VIGJV a contract demand. The VIGJV forecontract demand. The VIGJV forecontract TJ/day. BC Hydro IG is based on the Hydro IG is a dispatchable facility a RS 22 and Creative Energy include	omers in the ground BC Hydro IG ast is based upon the current firm conditioned the south on the facility on south interruptike	up is currently ex forecasted throu on their current fi ontract demand ly runs on certai ole and firm proje	opected to be sta oghput is currentl rm contract dem of 45 TJ/day; ho n days. The fore ected consumption	ble over time. y tied to their and of 13 wever, BC ecast for the on.
ii)	Before Rate Design Proposals				
,	R:C Ratio	142	5.5%	N/A	N/A
	M:C Ratio	1864	4.4%	N/A	N/A
	The R:C & M:C ratio for the VIGJV and BC Hydro IG is not applicable, but what is important is to VIGJV and BC Hydro IG are paying FEI for capacity on a take-or-pay basis. The interruptible F 22 customers are not allocated transmission and distribution costs on a peak day as they are deemed to be interrupted; therefore the M:C and R:C ratios are irrelevant.				portant is that erruptible RS they are
iii)	2016 Forecast Throughput (TJ)				
	Firm	Nil	732 TJ	4,758 TJ	14,945 TJ
	Interruptible	11,441 TJ	1,016 TJ	Nil	Nil
	Firm DTQ	Nil	2 TJ	13 TJ	45 TJ
	All these customers have an interruptible component to their agreement and need to be able to handle interruption of some capacity.				
iv)	Customers' Attributes				
	CP Load Factor ²⁾	N / A	100%	97.1%	3.8%
	NCP Load Factor (2016 Billed	66.4%	35.8%	48.8%	2.4%
	Actual) Other Attributes				
V)	Location on FEI's System & Special Circumstances	Lower Mainland	Lower Mainland	Vancouver	Vancouver
		Transmission	Transmission	Transmission	Transmission
		& Distribution	& Distribution	System	System
		System	System		
	Although the RS 22 customers are	all served off the	Lower Mainland	Distribution sys	tem, some of
	them are very close to the Transmission system and would generally all be served off larger distribution pipe. The VIGJV and BC Hydro are served from the Island Transmission system which is off of the FEI Lower Mainland transmission system.				



אז	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 172

	Particulars	RS 22 ¹	Creative Energy	VIGJV	BCH IG
vi)	Incremental Cost to Serve	Customer Stations, Measurement & Billing, Customer Relations, WINS & Gas Supply	Customer Station, Measurement & Billing, Customer Relations, WINS & Gas Supply	Customer Stations, Measurement & Billing, Customer Relations, WINS & Gas Supply	Customer Station, Measurement & Billing, Customer Relations, WINS & Gas Supply
	As all these customers are already customers is the ongoing O&M, tax	on the system, tl es and depreciat	ne only incremer ion.	ntal costs related	to serve these

¹ Includes only the RS 22 Non-Bypass customers, but also excludes Creative Energy which is a RS 22 Non-Bypass customer, as it is shown separately.

1 2

3

4

5

6

- ² CP Load Factor is calculated based on Firm Load consumption, i.e., it excludes interruptible volume. The NCP Load Factor includes all volumes, i.e., both firm and interruptible volume. The reason for excluding the interruptible volume from the CP Load Factor is that the Company's obligation for delivery is the firm DTQ less any peak shaving arrangement FEI has with the customer.
- 7 8
- 9
- 10
- 1134.4Considering FEI's response to the previous question, please explain how FEI's12proposal would be more aligned with the fair apportionment of costs and13avoidance of undue discrimination among similar type customers, as described in14the preamble.
- 15

16 **Response:**

As discussed in response to BCUC-FEI IR 1.34.8, moving towards postage stamps rates for large industrial customers would reduce the number of large industrial rate structures across the province and extend the principles of common rates to large industrial customers. FEI's cost allocation approach is transparent and consistent with the rate design principles of customer understanding and acceptance, fair apportionment of costs and avoidance of undue discrimination among similar types of customers.

23 Today, RS 22 serves a broad group of industrial customers (including Creative Energy). FEI's 24 proposal seeks to potentially add the VIGJV and BC Hydro IG, which already have similar 25 delivery rates, in order to streamline the number of industrial rates. The individual customer 26 sites within the VIGJV have consumption that is close to other existing RS 22 customers and. 27 prior to the termination of the agreement, BC Hydro's Burrard Thermal site was an RS 22 28 Bypass customer. FEI recognizes that today RS 22 is mostly an interruptible service, while the 29 VIGJV and BC Hydro have a combination firm and interruptible service. FEI's proposal will 30 allow industrials across the province more choice in contracting their gas supply requirements 31 with different combinations of firm and interruptible capacity.

FORTIS BC

1

6

7

	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
' -	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 173

2
3
3
4 34.5 Please discuss if any of the current RS 22 customers or contract customers
5 (Creative Energy, VIGJV, BC Hydro IG) had difficulty understanding and/or

accepting the existing structure.

8 **Response:**

9 None of the existing RS 22 customers or contract customers have expressed any difficulty in 10 understanding and/or accepting the existing rate structure. However, FEI is proposing to have 11 postage stamp cost-based rates and similar rate structures for all large volume transportation 12 industrial customers instead of relying on negotiated rates for contract customers. FEI believes 13 that moving to common rates and rate structures for these customers as proposed will be more 14 transparent and consistent with the rate design principle of customer understanding and 15 acceptance.

- 16
- 17
- 18 19

20

21

34.6 Please discuss if any of the affected contract customers (Creative Energy, VIGJV, BC Hydro IG) are supportive or opposed to FEI's proposal.

22 Response:

23 FEI has met with both the VIGJV and BC Hydro regarding its rate design proposals. Neither 24 party indicated either support or opposition to FEI's proposal. The proposed rates under RS 22 25 are very similar to the current approved rates of both the VIGJV and BC Hydro. The proposed 26 changes to the Transportation model are also linked to the proposed RS 22; therefore, the 27 VIGJV and BC Hydro have to look at these proposed changes in one package when comparing 28 them to their current contracts. FEI has not met directly with Creative Energy about the 29 proposal for RS 22; however, the proposed firm rates are lower than what their firm rates would 30 be based upon their current tariff supplement as described as Option 1 in FEI's RS 22 proposal. 31 Creative Energy's Shipper Agent is registered and participating in the proceeding on behalf of 32 Creative Energy and its other customers.

- 34
- 35
- 3634.7Please explain why Tariff Supplement G-21 for Creative Energy would be
terminated and Creative Energy would take firm service under the new charges



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 174

3

from RS 22 firm service while VIGJV and BC Hydro would be allowed to choose to become a RS 22 customer after their contracts expire.

4 <u>Response:</u>

5 Creative Energy is currently a RS 22 customer with a Commission-approved RS 22 Tariff 6 Supplement No. G-21 for firm transportation. Tariff Supplement G-21 was approved by Order 7 G-128-05, dated December 1, 2005, subject to the review of firm rates for RS 22 in the next FEI 8 rate design proceeding. If the Commission approves FEI's RS 22 firm rate proposal, there is no 9 longer any need for the tariff supplement as there would then be an approved firm rate 10 applicable to all RS 22 customers. Currently, the VIGJV and BC Hydro have separate contracts 11 and are not RS 22 customers, and therefore after their contracts expire they would have to elect 12 to sign up for service under RS 22 if FEI's rate design proposal is approved. If BC Hydro and 13 the VIGJV elect to become RS 22 customers, FEI may still need to negotiate an RS 22 tariff 14 supplement that would require Commission approval pertaining to any special terms that may 15 be required for these customers.

- 16
- 17
- 18
- 19 20
- 34.7.1 Please explain options available to VIGJV and BC Hydro IG if they choose not to become a RS 22 customer.
- 21

22 <u>Response:</u>

If the VIGJV chooses not to become a RS 22 customer, then FEI would have to negotiate a rate under a new agreement with the VIGJV that would be subject to Commission approval. If BC Hydro elects not to become a RS 22 customer, BC Hydro could elect to become an RS 50 customer, if they meet the requirements of that rate schedule. BC Hydro could also elect to extend their current agreement, which would require negotiation of a rate that would need to be approved by the Commission.

- 29
- 30

- 32 34.8 Does FEI consider that a decision to consolidate one or more customer classes 33 should consider whether they have similar revenue to cost ratios, customer 34 attributes (for example, load factor), customer service type (for example, firmness 35 of supply), and whether affected customers are supportive or opposed to the 36 merger? Please explain.
- 37



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 175

1 Response:

FEI believes that similar types of customers (i.e., customers with similar customer load and service characteristics [load factors, volume, types of end use]), should be grouped together in the COSA model for cost allocation purposes. FEI consolidated RS22, VIGJV and BC Hydro IG to derive firm rates based on cost of service allocation results. FEI believes that this cost allocation approach is transparent, and consistent with the rate design principles of customer understanding and acceptance, fair apportionment of costs and avoidance of undue discrimination among similar types of customers.

9 In addition to the reasons outlined above, the proposal to merge RS 22, VIGJV and BC Hydro 10 IG into a single rate schedule is related to amalgamation of FEI's Mainland, Vancouver Island / 11 Sunshine Coast and Whistler utilities into one utility and the establishment of common or 12 postage stamp rates, where possible, across FEI's service territory (except for Fort Nelson). 13 FEI's Mainland pre-amalgamation rate schedules for residential, commercial and firm or 14 interruptible general service were approved as the rate schedules for these types of services going forward for the amalgamated utility and as the basis for achieving the benefits of 15 16 amalgamation and common rates. The proposal with respect to RS 22, VIGJV and BC Hydro IG 17 seeks to extend these principles of amalgamation and common rates to large industrial 18 customers, and to achieve the related benefits among customers in that category.

- 19
- 20
- 21
- 22
- 23 24

34.9 Does FEI consider that there are broader BC benefits arising from the proposal regarding large industrial contract customers? For example: economic development; environmental benefits? If yes, please describe.

2526 Response:

27 FEI views its proposal with respect to large industrial transportation customers as an expansion 28 of the postage stamp rate methodology that resulted from the Reconsideration Decision on 29 FEI's Common Rates, Amalgamation and Rate Design Application (Order G-21-14, Decision). 30 Section 3.1 of the Amalgamation Reconsideration Decision (pages 12 to 16) cited various 31 benefits of amalgamation and postage stamp rates, including accepting or acknowledging the 32 submissions in the proceeding by the Ministry of Energy and Mines that amalgamation and 33 postage stamp rates would support the Province's Natural Gas Strategy, economic 34 development and job creation, regulatory efficiency and rate stability. With respect to large 35 industrial customers the proposal is to develop an industrial rate (RS 22) that is applicable to 36 Vancouver Island, the Sunshine Coast and Whistler as well as to the Mainland, thereby 37 expanding the postage stamp rate construct and creating the potential for these benefits to 38 develop in the industrial sector. However, for large industrial contract customers the rate design 39 proposal recognizes that FEI has existing contractual commitments that must be 40 accommodated before these customers can move to the proposed RS 22.



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
SC	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 176

34.9.1 Does FEI consider that there are efficiency benefits arising from the proposal, including the mitigation of uneconomic bypass risk? If yes, please describe.

Response:

FEI believes that efficiency benefits will generally accrue as its service offerings and the terms and conditions of service for industrial customers become standardized across the formerly separate Vancouver Island, Whistler and Mainland service territories. Traditionally, bypass risk has been more focused in the Interior of the province where some industrial customers are located in close proximity to the upstream pipelines. In the Lower Mainland and Vancouver Island areas bypass is generally not feasible, except possibly for the largest of customers. FEI established RS 50 to accommodate very large industrial loads, some of which may have bypass potential. Beyond the small group of Interior industrial customers already on bypass rates, FEI does not consider the bypass risk to be significant for the customers expected to take service under the proposed RS 22.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 177

1 35.0 Reference: RATE DESIGN FOR INDUSTRIAL CUSTOMERS

2 3

4

5

6

Exhibit B-1, Section 6.5.2, p. 6-36;

RS 22 R:C and M:C ratios

Table 6-19 on page 6-36 of Exhibit B-1 shows the R:C and M:C ratio results for rate schedules not set using COSA allocations. This includes an R:C ratio and M:C ratio of 1425.5% and 1864.4% respectively for Rate Schedule 22.

- 735.1Please provide a table showing the (i) Actual 2016 Revenues, (ii) Costs
determined through the COSA, (iii) the corresponding R:C Ratio and (iv) the
corresponding M:C Ratio for each of (a) Creative Energy, (b) VIGJV and (c) BC9Hydro IG:
- 11
- i. Under the existing rates and rate structure; and

ii. Using FEI's proposed rates and rate structures.

- 12
- 13

14 **Response:**

15 The COSA model does not have a separate cost allocation for Creative Energy; Creative 16 Energy is an RS 22 customer and is included with all other RS 22 customers when allocating 17 costs. Therefore, to derive an allocated cost for Creative Energy, FEI used 100 percent of the 18 demand related costs for RS 22 since Creative Energy is the only RS 22 non-bypass customer 19 with firm demand. FEI used 3.8 percent of the customer related costs and 13.3 percent of the 20 energy related costs as these percentages represent Creative Energy's contribution to the 21 customer and energy allocators. FEI has also included Creative Energy's allocation of UAF as 22 its cost of gas. FEI has used the cost allocations from the COSA for BC Hydro IG and VIGJV.

23 The following table uses 2016 Actual Revenues under the existing rates and the cost allocation 24 from the COSA. It is important to note that the costs from the COSA include known and 25 measurable changes of 7.4 percent, which creates a deficiency from the 2016 Actual revenues 26 in the table below. Also, on November 1, 2016, BC Hydro IG's rate and firm demand increased 27 by \$0.10 per GJ and 5 TJ/day firm demand (to 45 TJ/day total), respectively. The revenue in 28 the following table for BC Hydro IG reflects this; however, the cost allocations and revenues in 29 the COSA have been changed to show BC Hydro IG at \$0.958 per GJ and 45 TJ/day firm 30 demand for an entire year (not just November and December 2016), as this rate and level of 31 firm demand is representative of what they will experience in 2018 (when approved rate design 32 proposals are implemented).

To be comparable to the RS 22 R:C and M:C ratios referred to in the preamble of 1425.5 percent and 1864.4 percent respectively, FEI has presented the R:C and M:C ratios in Table 1 below, including IT revenue.

The R:C ratio is calculated by dividing Column 1 by (Column 2 plus Column 3), and M:C ratios are calculated by dividing (Column 1 less Column 2) by Column 3.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 178

\$000	Column 1	Column 2	Column 3	Column 4	Column 5
Customer	Total Revenue	Cost of Gas	Allocated Costs	R:C ratio	M:C Ratio
BC Hydro IG	13,097	0	14,530	90.1%	90.1%
Joint Venture	7,106	0	5,837	121.7%	121.7%
Creative Energy	1,648	15	654	246.3%	249.7%

Table 1

2

3 The following table uses the same revenue from the above table summed for BC Hydro IG,

4 VIGJV and Creative Energy compared to the cost allocated to RS 22 as proposed.

5

\$000	Column 1	Column 2	Column 3	Column 4	Column 5
Customer	Total Revenue	Cost of Gas	Allocated Costs	R:C ratio	M:C Ratio
RS 22 Firm	21,851	15	21,021	103.9%	103.9%

Table 2

6

7 It is important to note that while RS 22 Interruptible has high R:C and M:C ratios as described in 8 the preamble, these customers, because they are interruptible, cause very few demand-related 9 costs as their daily demand needs are not considered when planning and building the system. 10 However, as a group of customers, including BC Hydro IG and VIGJV, the firm daily demand of 11 this group now attracts demand related costs, thereby decreasing the R:C and M:C ratios. By 12 grouping these like customers together, a firm service can be offered and cost-based firm 13 charges can be derived.

- 14
- 15
- 16
- 1735.2Please compare and provide a discussion for any difference in R:C and M:C18ratios for RS 22 as presented in Table 6-19 and RS 22 using FEI's proposed19rates and rate structures. Please explain any shifts in revenues/costs that had an20impact on the R:C ratio.
- 21
- 22 Response:

RS 22 as presented in Table 6-19 is not the same rate schedule as RS 22 as proposed and found in Table 12-2. The two main differences between these two views of RS 22 is the treatment of interruptible revenue and the amount of cost allocation.

RS 22 in Table 6-19 is the RS 22 that is in place today. The revenue included in the R:C and M:C ratios includes both firm and interruptible revenues. There is a small firm component of the



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 179

revenues from Creative Energy based on 2 TJ firm demand per day, but the majority of the revenues are interruptible. The demand cost allocation for the existing RS 22 is small based on the fact that FEI has not built capacity to serve these customers on the peak day; consequently, RS 22 contributes only 2 TJ/Day or approximately 0.2 percent to peak day demand. As discussed in Section 6.5.2, the rates for FEI's existing RS 22 are not set based on COSA results.

7 RS 22 rates as proposed, for which the R:C ratio can be found in Table 12-2, are set based on 8 COSA results. To set the rates and assess the R:C ratio appropriately FEI had to treat the 9 interruptible revenues in a different manner. Because the interruptible revenues do not cause 10 demand costs, if they are not removed from the revenues for the R:C ratio calculation, the 11 revenues would not be comparable to the costs and the resulting R:C ratio would not be 12 informative (i.e., would not inform the reader whether the firm revenues collect the allocated 13 costs). Consequently, FEI treated the interruptible revenues (under the proposed RS 22) as 14 credits to the cost of service and allocated them to all non-bypass customers. The revenues that 15 are left are the firm revenues to be compared to the allocated costs. The proposed RS 22 has 16 higher allocated demand costs based on the higher firm demand of 60 TJ/Day which is 17 approximately 5 percent of peak day demand.

The different treatment of revenue and the different allocation of costs as described above result
 in the difference in the R:C (and M:C) ratios when comparing Table 6-19 and Table 12-2.

Under the proposed RS 22 there is a net decrease in revenue as compared to the existing RS
22 of \$754 thousand as described in Section 12.1.3 of the Application. However, approximately
5 percent more of the revenue under the proposed RS 22 is firm when compared to the existing
RS 22.


	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
DO⊮	2016 Rate Design Application (the Application)	June 9, 2017
BC.	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 180

1	36.0	Reference:	FEI COST OF SERVICE STUDY
2 3 4 5 6			FEI Certificate of Public Convenience and Necessity for Approval of Contracts and Rate for Public Utility Service to Provide Thermal Energy Service to Delta School District Number 37, Order G-31-12 and Decision dated March 9, 2012 (Delta School District 37 CPCN Decision), pp. 50 to 51
7			Low load factor "super-peaking" customers
8 9		In the Delta S 51:	School District 37 CPCN Decision, the Commission stated on pages 50 and
10		Energ	y Consumption and Customer Forecast
11 12 13 14 15 16 17 18 19 20 21 22		The S histori confin that m year a Once natura FEI sh GJ/da from 2 are of geoth	SD is the sole thermal energy customer and each of the 19 sites has cal and ongoing energy consumption patterns. Because the Project is ed to the 19 sites it does not include energy requirements of other sites hay be added later The current annual natural gas load is 58,607 GJ per and the current annual electricity load is equivalent to 4,684 GJ per year the proposed thermal energy systems are installed, the projected annual al gas load is projected to decrease by 77 percent to 13,641 GJ per year hows the peak day gas demand is projected to drop from 553 GJ/day to 301 y The impact on the load factor (utilization rate) for natural gas is a drop 29.0 percent to 12.1 percent because natural gas boilers at the GSHP sites hy required to provide supplemental energy on peak days. For sites with ermal installations, the natural gas load factors average 4.4 percent.
23		Comr	nission Determination
24 25 26 27		The F addre introd addre	Panel agrees that the Delta SD proceeding is not the appropriate forum for ssing poor load factor customer use and related issues such as the uction of a super-peaking rate. However, the Panel encourages FEI to ss these issues in a more suitable forum in the near future

- 28 36.1 Please explain if FEI has undertaken any research into the existence and impact 29 of super-peaking customers since the Delta School District 37 CPCN Decision 30 was issued in March 2012.
- 31

32 Response:

33 This response also addresses BCUC-FEI IRs 1.36.1.1 through 1.36.3.1.

34 FEI has conducted a high level review of customers that have access to an alternative energy 35 source or service that results in a high peak day requirement relative to their average day and 36 believes that it is still premature to analyze the impact of these customers on FEI's system. It is 37 not possible at this time to propose rate design alternatives for these types of customers as FEI



TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 181

- has only a small number of customers with limited history of these alternative energy systems to
 date. Also, many of these types of customers, particularly in district energy systems, have not
 converted from natural gas to their planned alternative energy source. Therefore, FEI will
 continue to use a tariff supplement approach until such time that FEI has suitable data available
- 5 that might be analyzed to support a separate rate schedule or rate design alternative.

6 FEI does have customers, other than Delta School District, which have back-up gas space 7 heating appliances that are used to provide supplemental energy on peak days. However, it is 8 not appropriate, in many instances, to classify these customers only as super-peaking because 9 they may have other gas loads, such as cooktops, barbeques, clothes drying, fireplaces, or 10 water heaters that are used year round and are not weather dependent.

11 12 13 14 15 36.1.1 If FEI has conducted research, please provide the results from the 16 reports/analysis completed. 17 18 **Response:** 19 Please refer to the response to BCUC-FEI IR 1.36.1. 20 21 22 23 36.2 Please explain the theoretical impact that super-peaking customers would have 24 on FEI's system. Please include a look at system capacities, utilization and 25 efficiency. 26 27 **Response:** 28 Please refer to the response to BCUC-FEI IR 1.36.1. 29 30 31 32 36.2.1 Please discuss how FEI could address any impact to its system caused 33 by super-peaking customers. 34



Response to British Columbia Utilities Commission (BCUC or the Commission) Page 182 Information Request (IR) No. 1 Page 182	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 182

1 Response:

- 2 Please refer to the response to BCUC-FEI IR 1.36.1.
- 3 4 5 6 36.3 Please state if FEI has customers, other than the Delta School District Number 7 37, that use natural gas only to provide supplemental energy on peak days 8 (super-peaking customers). 9 10 Response: 11 Please refer to the response to BCUC-FEI IR 1.36.1. 12 13 14 15 36.3.1 If FEI has other super-peaking customers, please describe the different 16 end-uses for these customers. 17 18 Response: 19 Please refer to the response to BCUC-FEI IR 1.36.1. 20 21 22
- 2336.3.2If FEI has super-peaking customers, please use the following template24to provide information, to the best of your ability, for super-peaking25customers for each year from 2011 to 2015.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 183

	-		-					-	
		Super-peaking customers							
	2011						2015		
	Number	Demand	Load Factor	Number	Demand	Load Factor	Number	Demand	Load Factor
Residential									
Commercial									
Industrial									
Contract									
Total									
FEI Total System Throughput									
Super-peaking Demand as a									
% of Total Throughput									
Notes									
(1) Number: The number of super	r-peaking cus	stomers.							
(2) Demand: The total annual demand for super-peaking customers.									

(3) Load Factor: The load factor for the group of super-peaking customers based on the average annual demand and the average peak demand.

2 Response:

FEI is unable to provide the requested information. The majority of FEI's customers, including nearly all residential and commercial accounts, do not have meters at their premises that would allow FEI to determine how much gas is being consumed in peak conditions as compared with non-peak conditions. FEI's billing information is based on monthly physical meter readings. This would provide an estimate of average daily consumption in the month but not the consumption increases or decreases due to weather variations within the month.

9 Another reason that FEI may not be able to identify customers that use gas only in peak 10 conditions is that customers can reconfigure their heating systems without having to let FEI 11 know that their gas use will be changing. For instance, a residential customer could install an 12 air-source heat pump as the main thermal energy source, but still retain their gas furnace or 13 possibly a heating fireplace for some of their thermal energy requirements. If they have a gas 14 water heater or other convenience appliances, there may be continued gas consumption 15 through the non-heating months of the year.

16

17 18 19

20

21

- 36.3.2.1 If FEI is unable to provide the information requested in the table above, please estimate the amount of effort in time and person-hours that would be required to complete the table.
- 23 **Response:**
- 24 Please refer to the response to BCUC-FEI IR 1.36.3.2.
- 25



CHAPTER 11 – FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES I. 1 2 FOR SERVICE 3 37.0 **Reference:** RATE DESIGN FOR RESIDENTIAL CUSTOMERS 4 Exhibit B-1, Section 11.3.2, p. 11-28 5 Overhead and Marketing (OH&M) charge updated calculation 6 On page 11-28 of Exhibit B-1, FEI states: 7 Using the 2016 and 2017 forecast volumes from the FEI Annual Review for 2017 8 Rates, Evidentiary Update filed October 5, 2016, the OH&M charge calculation in 9 Table 11-6 results in \$0.57/GJ. 10 ...Based on FEI's review and the updated calculation, FEI recommends the 11 OH&M charge for CNG and LNG fueling station customers remain unchanged at 12 \$0.52/GJ. 13 Please update Table 11-6 in Exhibit B-1, page 11-28 to include the actual OH&M 37.1 14 costs and volumes for 2012 to2016 and the average OH&M charge for 2012 to 15 2017. 16 17 **Response:**

18 The following provides the response for BCUC FEI IRs 1.37.1 and 1.37.2.

19 Table 11-6 in Exhibit B-1 on page 11-28 has been updated to include the actual OH&M costs

20 and volumes from 2012 to 2016, as well as the forecast costs and volumes for 2017. The

21 staffing resource costs are based on the previously determined resources and allocation of

those resources that were established by the BCUC in Order G-78-13.

	Actual 2012	Actual 2013	Actual 2014	Actual 2015	Actual 2016	Forecast 2017	Forecast Total
Staff Resources (\$000's)	560	611	715	713	760	782	4,140
Customer Education (\$000's)	0	54	93	58	27	100	331
Total Overhead (\$000's)	560	665	808	770	786	882	4,472
Projected Volumes (000's GJs)	187	295	736	957	1,098	1,354	4,627
Average Annual OH&M Cost (\$/GJ)	\$3.00	\$2.25	\$1.10	\$0.81	\$0.72	\$0.65	\$0.97
Annual OH&M Charge (\$/GJ)	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52

23

The table above indicates that the Average Annual OH&M Cost is \$0.97 per GJ for the period from 2012-2017.

26 B.C. Reg. 214/2016, which was deposited on August 21, 2016, extended the term of the 27 Greenhouse Gas Reduction Regulation to March 31, 2022.



FortisBC Energy Inc. (FEL or the Company)	Submission Date
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission)	Page 185

- 1 Table 11-6 has therefore been updated to provide the OH&M costs and volumes for the period
- 2 from 2012-2022.

	Actual 2012	Actual 2013	Actual 2014	Actual 2015	Actual 2016	Forecast 2017	Forecast 2018	Forecast 2019	Forecast 2020	Forecast 2021	Forecast 2022	Forecast Total 2012-2022
Staff Resources (\$000's)	560	611	715	713	760	782	806	830	855	881	907	8,419
Customer Education (\$000's)	0	54	93	58	27	100	130	150	150	100	50	911
Total Overhead (\$000's)	560	665	808	770	786	882	936	980	1005	981	957	9,330
Projected Volumes (000's GJs)	187	295	736	957	1,098	1,354	1,839	2,936	5,254	6,479	9,522	30,657
Average Annual OH&M Cost (\$/GJ)	\$3.00	\$2.25	\$1.10	\$0.81	\$0.72	\$0.65	\$0.51	\$0.33	\$0.19	\$0.15	\$0.10	\$0.30
Annual OH&M Charge (\$/GJ)	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52	\$0.52

4 As indicated in the table above, the Average Annual OH&M Cost over this period is forecast to 5 be \$0.30 per GJ. This table also shows that this cost has decreased from \$3.00/GJ in 2012 to 6 \$0.72/GJ in 2016. This trend will continue as NGT volumes are forecast to increase, further 7 decreasing the average OH&M cost. It can further be seen that the average OH&M cost is 8 forecast to be \$0.30/GJ over the GGRR period of 2012 to 2022 based on current demand 9 forecasts, which is lower than the current OH&M charge of \$0.52 per GJ. FEI therefore 10 recommends that the OH&M charge remain unchanged at \$0.52 per GJ at this time. FEI has 11 commenced a consultation process with NGT stakeholders to gather information and 12 considerations for the rate structures and rate offerings for NGT. FEI will also review the 13 appropriate level for the OH&M charge as part of that analysis, and report its findings as part of 14 an application to be filed in 2018.

15

- 16
- 17
- 18
- 37.2 In the same format as Table 11-6 in Exhibit B-1, page 11-28, please provide a forecast of the OH&M costs and volumes for 2018 to 2022. 19
- 20
- 21 **Response:**
- 22 Please refer to the response to BCUC-FEI IR 1.37.1.
- 23



1 2	38.0	Reference:	FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES FOR SERVICE
3 4			Exhibit B-1, Section 11.1.2.1, p. 11-5, Appendix 11-1, p. Original Page 14-1
5 6			General Terms and Conditions (GT&C), Access to Premises and Equipment
7 8		On original past section to the	age 14-1 of Exhibit B-1, Appendix 11-1, FEI proposes to add the following FEI General Terms and Conditions:
9		14.3 li	nstallation of Remote Meter
10 11			If a Customer fails to provide FortisBC Energy with access to the Customer's Premises as set out in Section 14.1 (Access to Premises) or
12			to FortisBC Energy's equipment as set out in Section 14.2 (Access to
13			Equipment), FortisBC Energy will be authorized to install a remote meter.
14			The Customer will be responsible for FortisBC Energy's full costs
15			(including overheads) associated with installing and maintaining the
16			remote meter.

- 17 38.1 Please provide the number of customers impacted by the proposed addition of
 18 section 14.3 to FEI's GT&C.
- 19

20 Response:

There are no customers expected to be impacted at this time by the proposed addition of Section 14.3.

This is because remote meters that are installed today are driven by Company requirements such as safety and efficiency. Where there are access issues to the premise that are driven by the customer, the Company works with the contractor and the customer to find options that allow for a successful meter read; however, this process can be challenging and may result in several estimated reads, several visits to the premise and, if a resolution cannot be achieved, ultimately may result in disconnection of service.

The inclusion of this provision would provide the Company and the customer with a final option before having to consider disconnection of service and thus it is expected that the need to implement Section 14.3 would be rare. FEI cannot estimate the number of customers that Section 14.3 would apply to, given the unique customer-specific circumstances where this would be required. The addition of Section 14.3 would provide FEI with the ability to recover the costs of installing such a meter, when required, from the individual customer on their bill.

35

36



		FortisBC Energy Inc. (FEI or the Company) Submiss							
	TN	2016 Rate Design Application (the Application)	June 9, 2017						
NI 13 BU	Respo	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1							
	38.1.1	Please discuss the communication and consultation customers.	with the affected						
Respons	se:								
Please re by propos	efer to the res	sponse to BCUC-FEI IR 1.38.1. There are no customers 14.3, thus communication or consultation was not necess	currently affected ary.						
38	3.2 Please	provide the estimated full cost of installing a remote	e meter including						
<u>Respons</u>	se:	aus and annual cost of maintaining the meter.							

The estimated installation cost is approximately \$180, including overhead and removal of the existing meter. There are no expected incremental annual operating costs associated with maintaining the remote meter. Further, there will be no meter reading cost reductions as the site will continue to be visited, as it is today, for meter reading purposes in order to receive the meter reading information from the remote meter signal.

38.3 Please provide the accounting treatment for the removal the inaccessible meter and the customer payments for "installing and maintaining the remote meter."

Response:

If the proposed amendment to the FEI GT&Cs to include Section 14.3 (Installation of Remote Meter) is approved by the Commission, the accounting treatment for the removal of an inaccessible meter from a customer's premise will be no different than the current accounting treatment for the removal of a meter from a customer's premise for other various reasons. When meters are removed from a customer premise, depending on the condition of the meter, they are either returned to inventory and reused at another premise or scrapped. In the case where meters are scrapped, they are retired from the Distribution Meters (478.10) asset class.

With respect to customer payments for installing and maintaining the remote meter, please refer to the response to BCUC-FEI IR 1.38.2.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 188

2 3

4

38.3.1 Please provide the depreciated cost of the inaccessible meters that FEI expects to remove.

5 **Response:**

6 As stated in the response to BCUC-FEI 1.38.1, there are no customers expected to be impacted 7 at this time by the proposed addition of Section 14.3; therefore, there are currently no

8 inaccessible meters that FEI expects to remove.



1 **Reference:** FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES 39.0 2 FOR SERVICE 3 Exhibit B-1, Appendix 11-1, p. S-1; Appendix 11-2, p. 1 4 GT&C, Standard Charges Schedule – Application Charge 5 On page S-1 of the General Terms and Conditions in Appendix 11-1, FEI proposes a reduction in all Application Charges from \$25 to \$15. In Appendix 11-2, FEI provides the 6 7 calculation of the proposed Application Charge. 8 39.1 Please provide justifications and calculations for the existing \$25 Application 9 Charge. 10 11 Response: 12 The current Application Fee of \$25 for existing installations was approved by the Commission 13 as part of the FEI Phase B RDA in the Decision and Order G-101-93 dated October 25, 1993, 14 and became effective January 1, 1994. Within that application, FEI proposed the consolidation 15 of three separate Application Fees for existing installations from the former Lower Mainland,

Inland and Columbia divisions. Please refer to Attachment 39.1 which includes Tab 12, pages 7 to 12 of the Phase B RDA and provides the justifications and calculations for the current Application Fee of \$25 for existing installations. Attachment 39.1 also contains page 49 of the Phase B RDA Decision which outlines the Commission approval of the FEI consolidated Control Terms and Conditions and aposition by the \$25 Application Fee for existing installations.

- General Terms and Conditions and specifically the \$25 Application Fee for existing installations,
 effective January 1, 1994.
- 22
- 23
- 24
- 2539.2Please explain, with calculations, why the Application Charge for a customer with
an Existing Installation is equal to the Application Charge for a customer required
a New Installation.
- 28

29 Response:

A single Application Charge applicable to both existing customers and new installations should
 remain for three reasons: 1) ease of understandability for customers; 2) ease of administration
 for FEI; and 3) the costs are comparable.

As shown in the table below, although a new installation may have a higher labour cost per transaction due to the longer call duration that is typically experienced, this is offset by meter reading costs that are not applicable to a new installation²⁴.

²⁴ Please refer to the response to BCUC-FEI IR 1.39.3 where FEI has updated the approximate customer service labour cost for moves and new service calls.

FORTIS BC^{**}

FortisBC Energy Inc. (FEI or the Company) Submission Date: 2016 Rate Design Application (the Application) June 9, 2017 Response to British Columbia Utilities Commission (BCUC or the Commission) Page 190 Information Request (IR) No. 1

Line Particulars for 2015 Moves New Install Notes Number of times Application Fee was charged 91,474 1 16.898 2 Approximate customer service labour cost for moves and new service calls 817,086 \$ 158,808 3 Approximate labour cost per transaction \$ 8.93 \$ Line 2 / Line 1 9.40 4 \$ 1.65 \$ 5 Credit check and ID validation per transaction 1.65 6 Off-Cycle move-in/move-out per transaction 4.5<u>0</u>\$ \$ -7 Line 3 + Line 5 + Line 6 8 Approximate Incremental Application Cost 15.08 \$ 11.05

2

1

3 FEI's specific costs related to new installations are captured in the Service Line Cost 4 Allowance.25

- 5
- 6

- 7
- 8 39.3
- 9 10

Please provide the calculations for the proposed application charge, in the same manner as on page 1 of Appendix 11-2, using figures from (i) 2014; (ii) 2013; (iii) 2012; and (iv) 2011.

11

12 Response:

13 As shown in the table below, analysis from 2012 through 2015 supports the proposed reduction 14 of the Application Fee from \$25 to \$15.

15 Information available for the stabilization period of 2012 and 2013 is limited and as such, an 16 estimate of call volumes and handle time has been used. Enhancements were made to the 17 queue management system in 2014 that allows FortisBC to be able to track calls by more 18 specific areas of customer call types. Prior to 2014, agents entered codes to indicate the reason for the call and these codes are what have been used to provide the estimates below. 19 20 While the results are reasonable, they may be lower than if detailed queue information was available²⁶ and, as such, may not be directly comparable to 2014 and 2015 data. 21

22 Finally, please note that comparable data for 2011 is not available as the Customer Service 23 function was outsourced at that time.

²⁵ FEI GT&Cs, Standard Charges Schedule, Original Page S-1.

²⁶ As a result of human error including, but not limited to, success rate of entering codes and code most applicable to the call.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 191

Basis for Calculation of Standard Charges Schedule

FEI Proposed Application Charge

Line	Particulars		2015 ¹		2014		2013		2012	Notes
1 2 3	Total number of applications charged for new service and changes to existing accounts (moves)	1	.08,372	1	02,277		94,368		98,032	
4	Customer service labour costs related to processing applications	\$9	75,894	\$9	86,909	\$7	88,671	² \$ 9	84,732	2
5										
6	Approximate avg customer service labour cost	\$	9.01	\$	9.65	\$	8.36	\$	10.05	Line 4 / Line 2
7										
8	TransUnion credit check and ID validation cost per transaction	\$	1.65	\$	1.51	\$	1.51	\$	1.40	
9										
10	Off-cycle move-in/move-out meter cost per transaction	\$	4.50	\$	4.50	\$	4.50	\$	2.00	3
11										
12	Approximate Incremental Application Cost	\$	15.16	\$	15.66	\$	14.37	\$	13.45	Line 6 + Line 8 + Line 10
13										
14										
15	FEI proposed Application Charge for new and existing customers	\$	15.00							
16										

¹⁷ ¹ Restated to reflect update in average handle time and total calls for Construction Services in 2015. Update results in decrease of labour cost of approximately \$5 thousand and corresponding decrease of \$0.04 to approximate cost per Application.

¹⁸² Total service costs have been estimated for 2012 and 2013 based on call wrap codes entered by agents, and as such estimates are less precise then 2014 and 2015. Prior to queue management enhancements made in 2014, calls by moves/construction queue are not available.

¹⁹ ³ Approximate average cost per read for 2012 as Services Contract with ABSU did not isolate a special read cost. Thus, likely understated and not directly comparable to off cycle read per read costs for 2013-2015.

2



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
BC™	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 192

1											
2	3	9.3.1	Please	explain	any	significant	changes	over t	the ye	ars in	the
3			approxir	nate av	rage	customer	service	labour	cost	related	to
4			process	ing appli	cations	s for new se	ervice and	change	s to aco	counts (l	Line
5			6).								
6											
7	Response:										

8 Please refer to the response to BCUC-FEI IR 1.39.3.



40.0 **Reference:** FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES 1 2 FOR SERVICE 3 Exhibit B-1, Appendix 11-1, p. S-1; Appendix 11-2, p. 2; FortisBC Inc. Electric Tariff BCUC No. 2. Sheet 39²⁷ 4 5 GT&C, Standard Charges Schedule – Returned Payment Charge 6 On page S-1 of the General Terms and Conditions in Appendix 11-1, FEI proposes a 7 reduction in the Returned Payment Charge (currently called "Dishonoured Cheque 8 Charge") from \$20 to \$8. On page 2 of Appendix 11-2, FEI provides the calculation of 9 the proposed Returned Payment Charge. 10 FBC's Returned Cheque Service Charge is \$19 as seen in the FBC Electric Tariff. 11 Please explain, with calculations, how the cost of return payments of \$3.91 40.1 12 (Appendix 11-2, p. 2, line 15) was calculated. 13 14 Response: 15 The cost per return payment (Appendix 11-2, page 2, line 15) is calculated by dividing the total annual billing department labour costs for processing returned payments by the total number of 16 17 returned payment items²⁸. 18 The following formula provides the calculation: $Labour \ Costs \ for \ Processing \ Returned \ Payments = Cost \ per \ Returned \ Payment$ 19 Total Number of Returned Payments 20 The following data provides the calculation supporting \$3.91 as the average cost of return payments for 2015. 21 $\frac{\$15,102}{3.862} = \3.91 22 23 As noted in Appendix 11-2, FEI added its bank processing charges and finance department 24 processing costs to the cost per returned payment item to inform its proposal to amend the 25 Returned Payment Charge to \$8.00. 26 27 28 29 40.2 If a customer living in an area served by both FEI and FBC, for example Trail, 30 BC, provides dishonoured cheques to both entities, please explain why FEI ²⁷ https://www.fortisbc.com/About/RegulatoryAffairs/ElecUtility/Documents/FortisBCElectricTariff.pdf. ²⁸ Includes returned payment items for both electronic fund transfer returns and returned cheques.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 194

3

would charge \$8 and FBC would charge \$19. Is the difference in costs stemming from the Finance Department or the Customer Service Billing Department?

4 Response:

5 FEI and FBC have their own cost structures upon which their fees and charges are based. 6 From time to time, the companies review these fees and charges. When necessary, the 7 companies apply to the Commission to amend these fees and charges when they no longer 8 reflect the costs which they are intended to offset. The Commission reviews applications for 9 changes to these fees and charges, and the currently approved fees and charges represent 10 what the utilities are each allowed to collect. The timing of the review of these fees and charges 11 differs for each utility.

12 If a customer's payment is returned (whether as a result of a dishonoured cheque or other 13 reason) to both utilities, then the customer would be charged the appropriate Commission 14 approved fee that was in place for each entity at the time.

15 FEI understands that FBC intends to review its Returned Cheque Service Charge as part of a

16 broader review of the FBC Electric Tariff in its next Rate Design Application, expected to be filed

17 late in 2017 or early 2018.



1 2	41.0 R	eference:	FEI GENERAL TERMS AND CONDITIONS AND RATE SCHEDULES FOR SERVICE
3			Exhibit B-1, Appendix 11-1, p. D-2
4			GT&C, Definitions – Carbon Offsets
5 6	O ha	n page D-2 as replaced t	of the General Terms and Conditions, when defining Carbon Offsets, FEI :he word <i>'will'</i> with <i>'may'</i> :
7 8 9 10 11 12		Carbo equiva purcha event reducti [emph	n Offsets Means the number of metric tons of carbon dioxide or its lent volume in other greenhouse gas(es) that FortisBC Energy may use as a mechanism to balance demand-supply for Biomethane in the of an undersupply of Biomethane in order to retain the greenhouse gas ions that Customers would have received from Biomethane supply. asis added]
13 14 15	4	1.1 Please supply	e explain if FEI has ever purchased carbon offsets to balance demand- for Biomethane in the event of an undersupply of Biomethane.
16	<u>Respons</u>	;e:	
17	No, FEI h	as never pu	rchased carbon offsets to balance demand and supply for Biomethane.
18 19	In the ev would no	ent that an a tify the Com	annual undersupply of biomethane occurs and FEI purchases offsets, FEI mission when filing the BVA Annual Report.
20 21			
22 23 24 25 26 27	Respons	41.1.1	If FEI has purchased carbon offsets in the event of a Biomethane undersupply, please state the number of occurrences and, if feasible, discuss the nature of each occurrence.
28	Please re	efer to the re	sponse to BCUC-FEI IR 1.41.1.
29 30			
31 32 33 34	4	1.2 Please	provide the rationale for this change.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 196

1 Response:

2 Due to evolving carbon markets, FEI is proposing a minor amendment to the definition of 3 Carbon Offsets to allow for more flexibility in the future, if alternative replacement options for 4 Biomethane become available. At this time FEI is not investigating any specific alternative to 5 the purchase of Carbon Offsets. However, should alternatives to the purchase of Carbon 6 Offsets become available in the future, prior to using such an alternative, FEI would ensure that 7 the costs and benefits for its customers were equal to or better than the purchase of Carbon 8 Offsets.

9 For clarity, the change to this section does not give FEI the authority to actually utilize options 10 other than carbon offsets. Pursuant to the Commission-approved Biomethane Program, FEI 11 only has approval to replace biomethane with carbon offsets. If FEI finds that there is another 12 reasonable option other than the purchase of carbon offsets, FEI will seek approval to use that

13 option as part of its Biomethane Program.



1 J. CHAPTER 12 – FEI FINAL COST OF SERVICE RESULTS AND REBALANCING

2 42.0 **Reference:** FEI FINAL COST OF SERVICE RESULTS AND REBALANCING 3 Exhibit B-1, Section 12.1.3, pp. 12-3 to 12-4; Section 6.5.1, p. 6-32; 4 Section 12.2.1, Table 12-2, p. 12-5; Section 9.8.5.3, Table 9-27, p. 9-48 5 RS 22 R:C ratio and rebalancing based on rate design proposal 6 On page 12-3 of Exhibit B-1, FEI states: 7 ... As a group, the R:C ratio for RS 22 customers is 103.5% before any 8 adjustments. As the RS 22 firm offering is a new service offering, FEI is 9 proposing to set the new offering at a 100% R:C ratio, in the middle of the 90% to 110% range of reasonableness. 10 11 On page 6-32 of Exhibit B-1, FEI states: 12 R:C ratios are assessed based on whether or not they fall within an established 13 "range of reasonableness". FEI believes that the appropriate range of 14 reasonableness for evaluating its R:C ratios is 90 per cent to 110 per cent. In 15 theory, the R:C ratio should equal 100% for each rate schedule, indicating that 16 the revenues recovered from each rate schedule would equal the indicated cost 17 to serve them. However, achieving unity implies a level of precision that does not 18 exist with any COSA. As a COSA study necessarily involves assumptions, 19 simplifications, judgments and generalizations, a range of estimates. 20 reasonableness is warranted and accepted when evaluating the appropriateness 21 of the R:C ratios.

42.1 Please explain why FEI is proposing to adjust RS 22 revenues to achieve an R:C
ratio of 100% when the R:C ratio is 103.5%, which is within FEI's R:C ratio range
of reasonableness of 90% to 110%.

26 **Response:**

27 Describing FEI as moving RS 22 to a 100 percent R:C is not accurate. FEI is creating a new 28 rate and rate structure for large volume industrial transportation customers, and has calculated 29 the new rate to collect the allocated costs. When creating a new rate schedule and rate 30 structure, there is no pre-existing R:C ratio and, therefore, no basis to set the R:C ratio at 31 anything other than 100 percent. If an R:C ratio other than 100 percent were to be adopted for 32 a new rate schedule or service, it would be equally as reasonable to propose 90 percent at the 33 lower end of the range of reasonableness as 110 percent at the upper end. These 34 considerations demonstrate that selecting 100 percent for the R:C ratio of a new service is the 35 sensible approach.

36

25



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 198

3

4

42.2 Please explain if FEI's level of precision in the COSA study supports the use of a target R:C ratio of unity for its proposed RS 22.

5 **Response:**

6 The level of precision in the COSA does not support a target of unity for any <u>existing</u> rate class 7 which is why FEI uses a range of reasonableness when considering rate rebalancing.

8 The proposal to set RS 22 rates on the basis of allocated costs is not related to a level of 9 precision for this class. Instead, it is based on the fact that developing the proposed RS 22 firm 10 rate as a <u>new</u> service offering is a different circumstance than looking at the R:C ratios for 11 existing rate classes for rebalancing purposes.

- 12
- . .
- 13
- 14
- 15

42.3 Please explain the impact to the COSA R:C ratios and M:C ratios if FEI were to leave the R:C ratio for the proposed RS 22 unchanged at 103.5%. Please include updated version of Tables 12-2 and 12-3 showing the R:C and M:C ratio results after the COSA, after the rate design proposals and after rebalancing.

20

21 Response:

Eliminating the revenue shift to RS 1 from setting the proposed RS 22 R:C ratio to 1.0 will reduce the RS 1 revenue shift by \$473 thousand from \$786 thousand to \$313 thousand. FEI has not eliminated all of the revenue shift from RS 22 because BC Hydro IG is under contract until 2022 and is paying a rate that is slightly below the proposed RS 22 rate: consequently there is a residual \$281 thousand²⁹ revenue shortfall that is from RS 22 that is still being picked up by RS 1 (as proposed). After this adjustment, the final R:C for the proposed RS 22 equals 102.2 percent and RS 1 equals 96.4 percent.

29 FEI has included Tables 12-2 and 12-3, revised as requested, below.

²⁹ Section 12.1.3, pages 12-3 and 12-4



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
,	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 199

Table 12-2 (Revised): COSA R:C and M:C Results after Rate Design Proposals

Rate Schedule	Initial	COSA	Revenue Shift (\$000)	Approximate Annual Bill Change	COSA after Prop	Rate Design losals
	R:C	R:C M:C			R:C	M:C
Rate Schedule 1	95.6%	03.1%	312.0	0.0%	96.4%	94 3%
Residential Service	33.070	55.170	512.5	0.070	30.470	34.370
Rate Schedule 2	101 20/	102 50/	(1 174 1)	0.5%	102.20/	10/ 10/
Small Commercial Service	101.5%	102.5%	(1,1/4.1)	-0.5%	102.2%	104.1%
Rate Schedule 3/23						
Large Commercial Sales and	101.6%	103.3%	1,174.1	0.6%	103.6%	107.6%
Transportation Service						
Rate Schedule 5/25						
General Firm Sales and	104.9%	112.2%	45.2	0.0%	106.3%	116.0%
Transportation Service						
Rate Schedule 6/6P	131.2%	150 1%			131 7%	160.4%
Natural Gas Vehicle Service		139.170			131.7 /0	100.478
Rate Schedule 22A						
Transportation Service (Closed)	109.5%	109.8%			113.0%	113.4%
Inland Service Area						
Rate Schedule 22B						
Transportation Service (Closed)	99.7%	99.7%			103.1%	103.1%
Columbia Service Area						
Rate Schedule 22						
Large Volume Transportation	1425.5%	1864.4%	(280.7)	-1.3%	102.2%	102.2%
Service						

Rate Schedule (rates not set using allocated costs)	Initial COSA		Revenue Shift (\$000)	Approximate Annual Bill Change	COSA after Prop	Rate Design oosals
	R:C	M:C		Ŭ	R:C	M:C
Rate Schedule 4	117 1%	550.0%	13.3	1.0%	150.2%	578 3%
Seasonal Firm Gas Service	147.470	550.97	13.5	1.970	150.2 /0	576.576
Rate Schedule 7/27						
General Interruptible Sales and	139.6%	712.3%	(90.7)	-0.3%	139.3%	713.6%
Transportation Service						



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 200

1 Table 12-3 (Revised): COSA R:C and M:C Results after Rate Design Proposals and Rebalancing

Rate Schedule	COSA after Rate Design Proposals		Rebalance Amount (\$000)	Approximate Annual Bill Change	COSA after Rate Design Proposals and Rebalancing	
	R:C	M:C			R:C	M:C
Rate Schedule 1	96.4%	01 3%	617	0.0%	96.4%	04 3%
Residential Service	90.4 /0	94.570	01.7	0.078	90.470	94.570
Rate Schedule 2	102.2%	10/ 10/			102 204	104 1%
Small Commercial Service	102.270	104.170			102.270	104.170
Rate Schedule 3/23						
Large Commercial Sales and	103.6%	107.6%			103.6%	107.6%
Transportation Service						
Rate Schedule 5/25						
General Firm Sales and	106.3%	116.0%			106.3%	116.0%
Transportation Service						
Rate Schedule 6/6P	121 70/	160 / %	(61.7)	16 5%	110.0%	110.0%
Natural Gas Vehicle Service	131.7 /0	100.4 /0	(01.7)	-10.570	110.0 %	119.0%
Rate Schedule 22A						
Transportation Service (Closed)	113.0%	113.4%			113.0%	113.4%
Inland Service Area						
Rate Schedule 22B						
Transportation Service (Closed)	103.1%	103.1%			103.1%	103.1%
Columbia Service Area						
Rate Schedule 22						
Large Volume Transportation	102.2%	102.2%			102.2%	102.2%
Service						

Rate Schedule (rates not set using allocated costs)	COSA after Rate Design Proposals		COSA after Rate Rebalance Approxima Design Proposals Amount Annual B (\$000) Change		e COSA after Rate Design Proposals and Rebalancing	
	R:C	M:C		Ū	R:C	M:C
Rate Schedule 4	150.20/	EZO 20/			150.00/	EZO 20/
Seasonal Firm Gas Service	150.2%	576.3%			130.2%	576.5%
Rate Schedule 7/27						
General Interruptible Sales and Transportation Service	139.3%	713.6%			139.3%	713.6%

On pages 12-3 and 12-4 of Exhibit B-1, FEI states:

8 When comparing the firm revenues for the current RS 22 customers and VIGJV
9 using the rates derived in Section 9.8 to the revenues embedded in the test year,
10 FEI will collect \$473 thousand less revenue. In addition, BC Hydro IG has



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 201

contract rates in place until 2022 that are marginally lower than they would pay
 under the new RS 22 service. This results in an additional \$281 thousand
 reduction in revenue. In total, after setting rates for this new service offering at
 allocated costs, FEI will collect \$754 thousand less revenue from these
 customers.

6 Table 9-27 on page 9-48 of Exhibit B-1 shows the "Summary of Change in Revenue and 7 Change in Rates for RS 22 and VIGJV." The final row of the table shows that under 8 current rates the total RS 22 and VIGJV revenue is approximately \$18,823 thousand 9 while under FEI's proposed rates and rate structure the total revenue is \$18,529 10 thousand, a difference of \$294 thousand.

- 11
- 12
- 13
- 42.4 Please provide calculations to show how the difference of \$473 thousand referenced in the preamble was calculated.
- 14

15 **Response:**

16 The \$473 thousand in the preamble is the required revenue shift to bring the RS 22 proposed to

17 a 100 percent R:C ratio.

Line	Particulars	Amount	Reference
1	BC Hydro IG rate (\$/GJ)	0.958	Section 9, Table 9-26
2	Proposed RS 22 rate (\$/GJ)	0.972	Section 9, Table 9-26
3	BC Hydro Firm Annual Volume (TJ)	16,425	
4	BC Hydro IG Revenue shortfall from retaining \$.958 until 2022 (\$000)	(\$236)	(Line 1 – Line 2) x Line 3 (small differences from rounding)
5	No Basic or Admin Charge from BC Hydro IG until 2022 (\$000)	(\$45)	
6	Total Revenue Shortfall from BC Hydro IG retaining existing rate until 2022 (\$000)	(\$281)	Line 4 + Line 5
7			
8	Revenue at 2016 Existing rates excluding known & measurable changes	22,183	Appendix 12-2, Schedule 1, Line 3
9	RS 22 Allocated Costs (\$000)	21,429	Appendix 12-2, Schedule 1, Line 9
10	R:C adjusted for rate design proposals	102.2%	(Line 6 + Line 8) / Line 9
11	Revenue Rebalancing Adjustment	(\$473)	(-Line 10 + 1) x Line 9 (small differences from rounding)

18

19



C™	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 202

- 1 2
- 42.5 Please explain why Table 9-27 shows a difference of \$294 thousand and on page 12-3 of Exhibit B-1, FEI states that there is \$473 thousand less revenue. Please include calculations in your response.
- 3 4

5 **Response:**

6 The difference calculated in Table 9-27 is not directly comparable to the rebalancing amount of 7 \$473 thousand on page 12-3. Table 9-27 is a calculation to show the difference between RS 22 8 existing customers plus VIGJV at existing rates compared to RS 22 existing customers plus 9 VIGJV at proposed rates. To do this, IT revenue and system gas are included and BC Hydro IG 10 revenues are excluded from the table. For the rebalancing amount referenced on page 12-3, BC 11 Hydro IG revenues are included and IT revenue is treated as a credit to the cost of service and 12 does not form part of the rebalancing calculation.

- 13
- 14

15

- 42.6 Please confirm, or otherwise explain, that regardless of the outcome of this
 Application, there will be no changes to BC Hydro IG rates or rate structure until
 its contract expires in 2022.
- 19
- 20 **Response:**
- 21 Confirmed.
- 22
- 23
- 24
- 25 26

- 42.6.1 If confirmed, please explain if the \$281 thousand is a theoretical reduction in revenue since BC Hydro IG and FEI have an agreement in place to pay a particular rate using a particular rate structure until 2022.
- 28
- 29 **Response:**
- 30 BC Hydro IG will continue to pay their contract rate until 2022. The \$281 thousand is a 31 theoretical reduction in revenue.
- 32
 33
 34
 35
 42.6.2 Please explain the impact to the COSA R:C ratios and M:C ratios if FEI were to omit the \$281 thousand related to BC Hydro IG. Please include updated version of Tables 12-2 and 12-3 showing the R:C and M:C ratio



TN .	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 203

results after the COSA, after the rate design proposals and after rebalancing.

- 34 <u>Response:</u>
- 5 If FEI omitted the \$281 thousand related to BC Hydro IG, the R:C and M:C ratios for RS 22 Firm
- 6 would increase to 101.3 percent. RS 22 charges would need to be amended upwards to
- 7 address the reduced rebalancing.



אז	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 204

1			FORT NELSON SERVICE AREA
2	К.	CHAPTER 1	3 – STAKEHOLDER ENGAGEMENT
3	43.0	Reference:	STAKEHOLDER ENGAGEMENT
4 5			Exhibit B-1-1, Section 13.3.2, pp. 13-10 to 13-12; Exhibit B-1, Appendix 4-5, p. 7
6			Survey methodology and scope
7		On page 7 of	Exhibit B-1, Appendix 4-5, FEI states:
8		The margin o	f error associated with each sample size is summarized below:

Region	Sample Size	Margins of Error (95% confidence level)
FEI	753	+/- 3.6%
Fort Nelson	65	+/- 12.2%

1043.1Please explain why Sentis did not increase the sample size of the Fort Nelson11residential customers to reduce the margin of error to a level comparable to the12FEI margin of error.

13

9

14 Response:

According to the 2016 Census, Fort Nelson has a population of 3,366 and has 1,682 private dwellings. To achieve the same margin of error for the Fort Nelson survey as comparable to the FEI margin of error would have required 500 complete surveys. Given the overall population size, attaining 500 complete surveys was not feasible.

- 19
- 20
- 21

28

- 22 On page 7 of Exhibit B-1, Appendix 4-5, FEI states:
- For Fort Nelson customers specifically, a telephone recruitment-to-online survey
 methodology (using a purchased list of Fort Nelson residential phone listings) was
 employed to obtain an oversample of Fort Nelson customers.
- 43.2 Please explain why Sentis used only residential phone listings to obtain theirsample of Fort Nelson customers.

29 Response:

- 30 Sentis only used residential phone listings to obtain their sample of Fort Nelson customers for
- 31 the following reasons:



3

8 9

TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 205

- Only residential customers were eligible for the survey.
 - Given the small overall population size, sufficient sample data was unavailable from online research panel providers.
- The only other options would be to mail letters out to Fort Nelson households asking for participation in the study or to hire interviewers to survey residents onsite in Fort Nelson.
 Both of these methodologies would have taken more time, adding several weeks to the study timeline, and would have been costlier.
- 101143.2.112Conducted in English and could only be completed on-line (i.e. the
customer did not have the option to complete a survey over the phone
or a hard copy survey).
- 16 **Response:**
- 17 Confirmed. The Fort Nelson survey was only available in English and could only be completed18 online.
- 19

15

- 20
- 20
- 21
- 22 23
- 43.2.2 Please confirm, or otherwise explain, that the sample was a random sample.
- 2425 Response:
- 26 Please refer to the response to BCUC IR 1.2.2.
- 27
- 28
- 29
- 3043.3Please explain if FEI considers the Fort Nelson customer survey to be31representative of the customers in Fort Nelson.
- 32
- 33 Response:

All survey methodologies exclude some element of the overall population. An online method scludes those without access to the internet, whereas a phone methodology excludes those



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 206

without a phone. In this instance those residents without a landline would not have been part ofthe overall sample pool.

3 The employment of a landline phone recruitment methodology likely contributed to an 4 underrepresentation of residents between the ages of 18 and 34. While these age groups 5 represent 36 percent of the Fort Nelson population, they only represent three percent of the 6 survey participants.

However, a comparison of results between FEI customers and Fort Nelson customers does not
reveal significant differences, which suggests that this under-representation had no material
impact on the results and that the survey was representative of the views of Fort Nelson
customers.

- 11
- 12

13

43.4 With regard to specific proposals in the Application, when does FEI take into
 consideration the Fort Nelson customer research survey results.

16

17 **Response:**

As described in Section 4.6 of the Application, the stakeholder engagement process, which includes the results of the residential customer research survey, is used as an input, among other rate design considerations, to conduct a full review of FEI's rate design for the residential rate class for all of FEI's service areas, including a survey specific to Fort Nelson.

22 The survey results for Fort Nelson are relatively supportive of FEI's proposal to transition from 23 the current bundled declining block rate structure to an unbundled flat rate structure similar to 24 the one that exists in the rest of the province. The survey results indicate that only 21 percent of 25 Fort Nelson customers prefer the current bundled rates, and 42 percent prefer a structure that 26 matches the rest of the province. In addition, as explained in Section 13.5.3 of the 27 Supplementary Filing (Exhibit B-1-1), the customer research survey results indicate that the flat 28 rate structure is preferred by the majority of Fort Nelson's residential customers as it received 29 the highest marks on all rate design considerations compared to other rate structure options.

- 30 31
- -
- 32
- 33 34

- 43.4.1 Please explain how much weight is given to the customer research survey results when determining rate design proposals.
- 3536 <u>Response:</u>
- 37 Please refer to the response to BCUC-FEI IR 1.2.6.1.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 207

1 L. CHAPTER 13 – FORT NELSON COST OF SERVICE ALLOCATION METHODOLOGY

2	44.0	Refer	ence: Fort Nelson Cost of Service Allocation Methodology
3 4			Exhibit B-1-1-1, Section 13.4.1.5.3, p. 13-17, Section 13.7.2, p. 13-55, Appendix 13-1, p. 2
5			Minimum System Study
6 7 8 9	<u>Respo</u>	44.1 onse:	Please provide a breakdown of mains installed in 2015 and 2016 in the same format as Table 1 in Exhibit B-1-1, Appendix 13-1, p. 2.

- 10 The table below shows the combined steel and plastic mains installed in Fort Nelson in 2015
- 11 and 2016. The Unit Cost, Weighted Cost and Minimum Size Cost are the replacement cost from
- 12 the Combined Steel and Plastic mains table in Appendix 13-1.

Diameter		Length	Unit Cost /	W	eighted	Mi (Al	nimum Size Cost I pipe valued
Inches	mm	in Meters	Length (\$/m)	Co	st	at	60 mm PE)
1.0	26	9	73.45	\$	661	\$	414
1.7	42	2	52.62	\$	118	\$	103
2.4	60	297	93.04	\$	27,636	\$	13,663
4.5	114	298	175.4	\$	52,213	\$	13,693

13

14 It is important to note that the table above includes only mains installed in 2015 and 2016, 15 whereas the table included in Appendix 13-1 includes all the mains in place in Fort Nelson up to 16 the time that the Minimum System Study was completed.

17
18
19
20 44.1.1 Please provide the percentage of mains that are equal to or less than 60 mm in Table 1 in Exhibit B-1-1, Appendix 13-1, page 2.
22

23 Response:

The percentage of all mains less than 60 mm from the referenced table equals 19.6 percent and less than or equal to 60 mm equals 83.9 percent. Further to the response to BCUC-FEI IR 1.44.1, only 1.8 percent of the mains installed in 2015 and 2016 were less than 60 mm.

27

28

FortisBC Energy Inc. (FEI or the Company) Submission Date: 2016 Rate Design Application (the Application) June 9, 2017 FORTIS BC^{**} Response to British Columbia Utilities Commission (BCUC or the Commission) Page 208 Information Request (IR) No. 1 44.1.2 Please revise the following tables to reflect a minimum main size of 42 1 2 mm: 3 i. Table 1 in Exhibit B-1-1, Appendix 13-1, p. 2 4 ii. Table 13-10 in Exhibit B-1-1-1, page 13-17 5 iii. Table 13-29 in Exhibit B1-1-1, page 13-55

7 **Response:**

8 The Minimum System Study using 60 mm pipe is the correct approach as it is FEI's minimum 9 standard in most cases when installing mains for the distribution system. As identified in the

10 response to BCUC-FEI IR 1.44.1, the number of 42 mm mains installed in 2015 and 2016 was

11 small compared to FEI's standard of 60 mm.

FEI has updated the Minimum System Study to use 42mm as the minimum as requested. FEI calculated the average replacement cost based on the 42 mm minimum system at \$40.80 per meter using geographic pricing. When using the 42 mm as the minimum, distribution mains are split 41 percent Customer and 59 percent Demand, as can be found in an adjusted Table 1 below.

10

6

17

Table 1: (Adjusted) Minimum System Results for All Mains

COMBINE	COMBINED STEEL & PLASTIC MAINS								
	Diame	eter		Unit Cost / Length				Mi	nimum Size Cost
Line No.	Inches	mm	Length in Meters		(\$/m)	V	Veighted Cost	(Al	l Pipe Valued at
	(1)	(2)	(3)		(4)		(5)		(6)
1	1.0	26	1,144	\$	65.15	\$	74,515.78	\$	46,666
2	1.7	42	19,282	\$	46.67	\$	899,964.07	\$	786,688
3	1.9	48	2,407	\$	124.20	\$	298,981.69	\$	98,216
4	2.4	60	74,794	\$	93.04	\$	6,959,133.02	\$	3,051,594
5	3.5	88	6,196	\$	165.10	\$	1,022,876.71	\$	252,780
6	4.5	114	12,148	\$	175.40	\$	2,130,834.89	\$	495,647
7	6.6	168	246	\$	354.13	\$	86,979.87	\$	10,021
8	L	Jnknown	80			\$	8,789.67	\$	3,255
9	Т	OTAL	116,296			\$	11,482,075.69	\$	4,744,867
10									
11	11 Customer Related Component			Lir	ne 9, Column (6)	/ Liı	ne 9, Column (5)		<u>41%</u>
12	12 Demand Related Component				1-	- Lin	e 11, Column (6)		<u>59%</u>

18

The PLCC is the peak load carrying capacity of the minimum system. When using a 42 mm minimum system, a new PLCC must be calculated. When the minimum system is reduced from 60 mm to 42 mm, the PLCC adjustment is reduced from 1.178 to 0.577.

22 The combination of the decreased customer component of distribution mains and the reduction

of the PLCC resulted in an increase in the allocation of costs to Rate 1 and a decrease in the cost allocated to Rate 2.1 and Rate 2.2.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 209

Table 13-10: (Adjusted) Delivery Cost of Service Allocation to Rates Summary

Rate	(\$000s)	% of total
1	\$1,354	54.4%
2.1	\$843	33.9%
2.2	\$158	6.3%
RS 25	\$134	5.4%
Total	\$2,489	100.0%

2

1

3 Three of FEI's rate classes for Fort Nelson would be affected by the reallocation of costs as 4 described above: Rate 1 and Rates 2.1 and 2.2. With more costs allocated to Rate 1 the R:C 5 ratio falls to 85.1 percent.

6 In rebalancing the above result, FEI has made the following assumptions which mirror the 7 approach taken in the Application.

8 Rate 1 revenue responsibility is increased by \$83 thousand, which brings Rate 1 to a 90 percent 9 R:C ratio. The added revenue responsibility is recovered through an increased Basic Charge. 10 As a group, Rate 1 customers will experience about a 7 percent annual bill increase from this 11 rebalancing and individual customers will experience annual bill changes between -30 percent 12 and +26 percent. The revenue shift from Rate 1 reduces the revenue responsibilities of Rate 2.1 13 and Rate 2.2. The \$83 thousand is split between Rates 2.1 and 2.2 so that their final R:C ratios 14 are similar at approximately 112 percent. Splitting the rebalancing dollar amount unevenly 15 between Rate 2.1 and Rate 2.2 necessitates a recalculation of the proposed basic and 16 volumetric charges for these two rates. FEI used the same parameters as described in Exhibit 17 B-1-1-1 Sections 13.5.5.3.3 and 13.7.1.4. Rate 2.1 customers will experience about a 1 percent 18 annual bill decrease from this rebalancing and individual customers will experience annual bill 19 changes between -16 percent and +5 percent. Rate 2.2 customers will experience about a 16 20 percent annual bill decrease from this rebalancing and individual customers will experience 21 annual bill changes between -22 percent and -3 percent. The table below presents the adjusted 22 rates.



TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 210

Table 13-29: (Adjusted) Fort Nelson Rate Proposal Summary

Rate Component	Rate 1	Rate 2.1	Rate 2.2	Rate 3.1	RS 25
Existing COSA Rates					
Minimum daily Charge incl. 1 st 2 GJ/month	\$0.5483	\$1.4337	\$1.4337		
Administration Charge (/month)					\$202
Next 28 GJ/month	\$4.885				
Excess over 30 GJ/month	\$4.782				
Next 298 GJ/ month		\$5.336	\$5.336		
Excess over 300 GJ/month		\$5.210	\$5.210		
Delivery Charge First 20 GJ/month				\$4.522	\$4.522
Delivery Charge Next 260 GJ/month				\$4.201	\$4.201
Excess over 280 GJ/month				\$3.450	\$3.450
Minimum Delivery Charge/month				\$1,826	\$1,826
Total Annual Bill: ³⁰	\$742	\$2,433	\$28,546	n/a ³¹	\$148,664
Proposed Rates					
Basic Charge/Day	\$0.3929	\$1.2078	\$7.7630		
Basic Charge (/Month)				\$600.00	\$600.00
Administration Charge (/Month)					\$39.00
Demand Charge (/GJ/Month)				\$28.727	\$28.727
Delivery Charge (\$/GJ)	\$3.512	\$3.888	\$2.691	\$1.000	\$1.000
Commodity Cost Recovery Charge (\$/GJ)	\$1.275	\$1.275	\$1.275	\$1.275	
Storage and Transport Charge (\$/GJ)	\$0.019	\$0.020	\$0.017	\$0.019	
Total Annual Bill:	\$792	\$2,421	\$24,070	n/a ³²	\$148,243

 ³⁰ Based on an average annual demand per customer of 135 GJ for Rate 1, 382 GJ for Rate 2.1 and 5,332 GJ for Rate 2.2 and 39,500 GJ for RS 25.
 ³¹ There are no customers taking service under Rate 3.1, therefore Total Annual Bill shows as n/a.



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 211

1	45.0	Refere	ence: FORT NELSON COST OF SERVICE ALLOCATION METHODOLOGY						
2			Exhibit B-1, Section 8.3.5, p. 8-14;						
3		Exhibit B-1-1, Section 13.4.1.4, p. 13-15							
4			Cost Allocation based on RS 25 Load Factor						
5		On pa	ge 13-15 of Exhibit B-1-1, FEI states:						
6 7 9 10 11 12 13 14			Currently, there is one customer that is taking service in Fort Nelson under RS 25 and that customer has a load factor of 27%. This low load factor is a result of the customer scaling back on its operations and only using gas for space heating purposes. As in FEI, Fort Nelson's Rate Schedule 25 is intended to serve process load customers As described in Section 9.5.1, customers with load factors less than 40% are more heat sensitive than a typical process load and should be taking service under the large commercial rate. To allocate costs in accordance with the intended use of Rate Schedule 25, FEI has used a load factor of 40% for this rate schedule.						
15		On pa	ge 8-14 of Exhibit B-1, FEI states:						
16 17 18			When reviewing existing rate design and setting rates, and according to the fair apportionment of cost principle [FEI Rate Design Principle 2], FEI seeks to align cost recovery with cost causality.						
19 20 21 22 23	Resp	45.1	Please explain why FEI is allocating costs in accordance with the intended use of RS 25, as opposed to the actual use of RS 25. Please include in your response a discussion regarding the fair apportionment of cost principle and cost causality.						

The single remaining RS 25 customer announced that it has permanently closed plant operations and has informed FEI that it will only be using gas for space heating for a few years to preserve its assets but will eventually no longer require gas. The customer's other site in Fort Nelson, which was formerly served under RS 25, also closed permanently in 2008 and has already gone to zero gas consumption as of December 2015, and has subsequently switched to Fort Nelson Rate 2.1.

FEI is allocating costs in accordance with the intended use of RS 25 as opposed to the actual use of RS 25 as the remaining customer could choose to switch to Fort Nelson Rate 2.1 or 2.2 at any time and is not representative of a RS 25 customer. Utilizing a customer with a heat sensitive load profile to design a rate intended for a process load would result in a rate and rate structure that would not be appropriate for any future customers.

FEI wants to maintain the RS 25 option for future customers based upon its intended use to maintain a rate structure for Fort Nelson that would support local economic development for a process load customer setting up business in the Fort Nelson community.



174	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 212

- 1
- 2
- 3 4

6

45.2 Please state the number of years that this RS 25 customer has only been using gas for space heating purposes.

7 <u>Response:</u>

8 The customer has only been using gas for space heating since 2008, so for approximately nine 9 years to the end of 2016. To the best of FEI's knowledge, when this customer ran its production 10 process, a biomass system provided the primary energy for the production process and gas was 11 used to supplement the biomass system, or to provide space heating. This explains why annual 12 gas consumption since the plant closed in 2008 has increased as is shown in the response to 13 BCUC-FEI IR 1.45.2.1.

- 14
- 15
- ...
- 16
- 17 18
- 45.2.1 For each of the years included in the response to the previous question, please provide this RS 25 customer's load factor.
- 19
 20 <u>Response:</u>
- 21 Please find in the table below the annual throughput and load factor for the three years prior to
- the customer ceasing production in 2008:

Year	Annual Throughput (GJ)	Load Factor
2005	39,647	21%
2006	28,050	15%
2007	25,997	14%

- 24 FEI also provides the response to BCUC-FEI IR 1.51.1 here, so that load factors can be easily
- 25 compared across years. Please find in the table below the annual throughput and load factor
- 26 for the customer since they permanently shut down production in 2008.

Year	Annual Throughput (GJ)	Load Factor
2008	38,418	27%
2009	46,111	25%
2010	37,616	20%
2011	37,460	20%



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 213

Year	Annual Throughput (GJ)	Load Factor
2012	37,610	20%
2013	40,642	22%
2014	47,039	25%
2015	39,684	28%
2016	41,110	26%

- 45.3 Please explain and quantify the impact to the COSA results and the R:C and M:C ratios of using the RS 25 customer's load factor of 27%, instead of 40%. Please include updated versions of Tables 13-7, 13-10 and 13-12 with your response.
- **Response:**

9 By using the lower load factor of 27% for RS 25, a larger peak day demand is calculated and
10 subsequently more costs are allocated to RS 25. With higher allocated costs both the R:C and
11 MiC ratios dealing. The undated tables are included below.

11 M:C ratios decline. The updated tables are included below.

12 Table 13-7 (revised): Customers, Annual Volume, Load Factor and Peak Day by Rate

Rate	Customers	Annual Volume (TJ)	Load Factor	Peak Day Demand (TJ)
1	1,961	259.9	35.7%	2.0
2.1	480	203.7	33.4%	1.7
2.2	7	56.7	40.5%	0.4
RS 25	1	39.5	26.5%	0.4
Total	2,449	559.8		4.6

Table 13-10 (revised): Delivery Cost of Service Allocation to Rates Summary

Rate	(\$000s)	% of total	
1	\$1,233	49.5%	
2.1	\$902	36.3%	
2.2	\$191	7.7%	
RS 25	\$163	6.5%	
Total	\$2,489	100.0%	



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 214

Table 13-12 (revised): Revenue to Cost and Margin to Cost Ratios

Rate	R:C	M:C
Rate 1	Q1 4%	89.0%
Domestic (Residential) Service	51.470	00.070
Rate 2.1	100 /%	112 2%
General (Small Commercial) Service	109.470	112.270
Rate 2.2	111 10/	110 9%
General (Large Commercial) Service	114.4 /0	119.070
Rate Schedule 25	02 40/	02 40/
General Firm Transportation Service	92.470	92.4%



1 M. CHAPTER 13 – RESIDENTIAL RATE DESIGN FOR FORT NELSON

2	46.0	Refere	ence: RESIDENTIAL RATE DESIGN FOR FORT NELSON
3			Exhibit B-1, Section 7.2.4 p. 7-8
4			Exhibit B-1-1, Section 13.5.4.2.3, p. 13-28
5			Residential customer characteristics
6		On pag	ge 13-28 of Exhibit B-1-1, FEI states:
7 8 9 10			As can be seen from the figure below, the 100–110 GJ annual consumption range has the highest density of customers (compared to 70–80 GJ for other FEI customers), followed closely by the 110–120 GJ and 120–130 GJ consumption ranges.
11 12 13		46.1	Please provide the standard deviation for the 2016 residential annual consumption.
14	<u>Respo</u>	nse:	
15 16	The sta provide	andard d in Fig	deviation for Fort Nelson's 2016 residential annual consumption per customer as gure 13-7 is approximately 70 GJ.
17 18			
19 20 21 22		46.2	Please provide the average consumption range for the top and bottom 10 percent of Fort Nelson's residential customers.
23	Respo	nse:	
24 25	The average annual consumption per customer for the bottom and top 10 percent of Fort Nelson's residential customers are approximately 44 GJ and 279 GJ respectively.		
26 27			
28 29 30 31		46.3	Please provide a residential customer scatter plot in the same format as Figure 7-8 in Exhibit B 1, p. 7-8.
32	<u>Respo</u>	nse:	
~~	T I2		d a setter what for East Malance as side with a set one on is were ideally below.

33 The requested scatter plot for Fort Nelson residential customers is provided below.


	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 216





FortisBC Energy Inc. (FEI or the Company)
2016 Rate Design Application (the Application)Submission Date:
June 9, 2017Response to British Columbia Utilities Commission (BCUC or the Commission)
Information Request (IR) No. 1Page 217





17 The figure below is the same format as Figure 7-11 in Exhibit B-1, but for Fort Nelson Rate 1.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 218

1 The table below is in the same format as Table 7-5 in Exhibit B-1, but for Fort Nelson Rate 1.

Type of Cost	Unit Cost Based on COSA Results	Current Average Monthly Basic Charge	Difference
Customer-related cost	\$29.18 per month		
Demand-related cost	\$17.08 per month		
Total fixed costs	\$46.26 per month	\$14.10 per month	\$32.16 per month



1 N. CHAPTER 13 – COMMERCIAL RATE DESIGN FOR FORT NELSON

2	48.0 Refe	erence:	COMMERCIAL RATE DESIGN FOR FORT NELSON
3			Exhibit B-1-1, Section 13.3.2, p. 13-23
4			Consultation
5	FEI	states on	page 13-10 of Exhibit B-1 that:
6 7 8 9		FEI re reside compo rate de	etained the services of Sentis to conduct an online survey to measure ntial customers' knowledge of Fort Nelson's existing rate structure and bill onents and to better understand customers' preference regarding various esign considerations.
10 11 12 13	48.1	Please custor results	e explain whether FEI has done any consultation with commercial ners on their preference on the rate structure. If yes, please provide the s. If not, why not?
14	Response:		
15 16 17 18 19 20	FEI has c sessions, s customers. in all the s customers o other rate d	onducted stakehold The Com stakehold was revie esign opt	a robust stakeholder engagement process consisting of information er workshops, and a customer research online survey for residential imercial Energy Association of British Columbia (CEC) participated actively er sessions, including the workshops where rate design for commercial ewed, including evaluations of the existing commercial rate structures and ions.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 220

1 49.0 Reference: COMMERCIAL RATE DESIGN FOR FORT NELSON

2

3

Exhibit B-1, Table 8-4, p. 8-14; Exhibit B-1-1, p. 13-41

Customer data

Table 13-20 shows the Comparison between Small & Large Commercial using 6000 GJ
Threshold, including the Average Customer-related Cost and Average Demand-Related
& Energy-related Cost.

- Table 8-4 on page 8-14 of Exhibit B-1 shows the comparison of fixed costs and fixed
 charge recoveries for RS 2 and RS 3/RS 23 customers.
- 9 49.1 Please replicate table 13-20 using the proposed 2000 GJ threshold.
- 10

11 Response:

- 12 The following table provides the same information as in Table 13-20, except it is adjusted for the
- 13 changes applicable to Commercial customers with a 2,000 GJ threshold.
- 14

Comparison between Small & Large Commercial using 2000 GJ Threshold

	Rate 2.1	Rate 2.2
Customer Weighting Factor	1.6	5.7
Use per Customer	383 GJ	5,332 GJ
Load Factor	32.8%	38.8%
Average Customer-related Cost / Customer / Day	\$1.394	\$3.657
Average Demand-Related & Energy-related Cost / GJ	\$3.306	\$3.203

- 15
- 16
- 17

1849.2Please show, similar to in Table 8-4 in Exhibit B-1, what percentage of the19functional unbundled costs allocated to commercial customers are recovered20from the corresponding proposed rates (basic charge, delivery charge) for RS 2.121and RS 2.2.

2223 Response:

24 The following table compares the proposed rates, based on the results from the update to the

25 PLCC factor for Fort Nelson (per Exhibit B-1-1-1), to the unit allocated costs in the final COSA

which has the threshold at 2,000 GJ between Rate 2.1 and Rate 2.2.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 221

	Proposed Rates	Unit Allocated Cost	% of Related Costs
Rate 2.1 Small Commercial			
Basic Charge / Customer \$ / Day	\$1.2008	\$1.394	86%
Delivery Charge / Demand & Energy \$ / GJ	\$3.989	\$3.306	121%
Rate 2.2 Large Commercial			
Basic Charge / Customer \$ / Day	\$3.1581	\$3.657	86%
Delivery Charge / Demand & Energy \$ / GJ	\$3.631	\$3.203	113%



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 222

1	50.0	Refere	ence: COMMERCIAL RATE DESIGN FOR FORT NELSON
2			Exhibit B-1, p. 8-12; Exhibit B-1-1, p. 13-42
3			Proposed rate change
4		FEI st	ates on page 13-42 of Exhibit B-1-1 that:
5 6 7			Moving the threshold from 6,000 GJ/year to 2,000 GJ/year and setting the rates to result in an economic crossover at 2,000 GJ results in the following range of bill impacts when compared to existing bills.
8 9 10		Table point I crosso	8-3 on page 8-12 of Exhibit B-1 shows the calculation of the economic crossover petween RS 2 and RS 3. Figure 8-12 on page 8-20 shows the shift in economic over point after the rate proposal.
11 12 13		50.1	Please replicate Table 8-3 and figure 8-20 in Exhibit B-1 for Fort Nelson's RS 2.1 and 2.2.
14	Resp	onse:	
15 16	An ec was ir	onomic Table (crossover for Rate 2.1 and Rate 2.2 at existing rates cannot be calculated as it 8-3 because both Rate 2.1 and 2.2 have the same monthly Basic Charge.

17 The table provided below at proposed rates demonstrates that the economic crossover is 18 approximately 2,000 GJ.

19

Economic Crossover Volume for Rate 2.1 and Rate 2.2

Rate Components	Rate 2.1	Rate 2.2	Difference
1. Basic Charge (per day)	\$1.2008	\$3.1581	
2. Times number of days	365.25	365.25	
3. = Basic Charge Revenue	\$438.59	\$1,153.50	\$714.91
4. Delivery Charge (\$/GJ)	\$3.989	\$3.631	
5. Plus Cost of Gas (\$/GJ) ³³	\$1.294	\$1.294	
6. = Total Variable Cost (\$/GJ)	\$5.283	\$4.925	\$0.358
7. Economic Crossover Point (Line 3/Line 6)			1,997 GJ

The following graph, similar to Figure 8-12 on Page 8-20 of Exhibit B-1, shows the average effective cost using the proposed rates for Rate 2.1 and Rate 2.2. Although visually it is difficult

to see exactly where the crossover is, at 2,000 GJ the average effective rate for both Rate 2.1

24 and Rate 2.2 is \$5.502 per GJ.

²⁰

³³ For the purpose of this calculation, FEI uses the gas costs from the compliance filing for the Annual Review for 2016 Rates (Order G-193-15).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 223





1 O. CHAPTER 13 – INDUSTRIAL RATE DESIGN FOR FORT NELSON

2 51.0 **Reference:** INDUSTRIAL RATE DESIGN FOR FORT NELSON 3 Exhibit B-1-1, Section 13.5.6.2 and 13.5.6.3, p. 13-45 4 Fort Nelson industrial customer characteristics and bill impact 5 On page 13-45 of Exhibit B-1-1, FEI states: 6 Fort Nelson has only one industrial customer taking service under RS 25 ... The 7 customer is no longer operating its production facility, but is still using natural gas 8 for space heating to protect facilities and equipment from extreme cold weather 9 damage. The customer's 2018 forecast demand is 40 TJ and its three year 10 average load factor is 27%. ... FEI is proposing to adopt the same rate structure 11 for Fort Nelson as exists in FEI's other service areas. The charges included for 12 the two industrial rate schedules would be: a Basic Charge, Demand Charge, 13 and a Delivery Charge. ... The proposed 2018 rates will be designed to collect 14 the same revenue as was forecast in Fort Nelsons 2017-2018 Revenue 15 Requirement so that no other Rate Schedules are affected by this change. 16 51.1 For the sole RS 25 industrial customer, please state the annual throughput and 17 load factor for each of the three years prior to the customer ceasing operations at 18 its production facility. 19 20 Response: 21 Please refer to the response to BCUC-FEI IR 1.45.2.1. 22 23 24 25 Please explain if, and when, FEI is expecting this RS 25 customer to resume 51.2 26 regular operations in the near future. 27 28 Response: 29 FEI was informed by this RS 25 customer that the facility is closed permanently. FEI is hopeful 30 an industrial customer will take over this site or that of the other former RS 25 customer site 31 next door. FEI would like to maintain an Industrial Rate in Fort Nelson for economic

- 32 development reasons to help attract a new customer and employer to this community.
- 33
- 34
- 35



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
BC	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 225

51.3	Please produce a table to show:
	 the calculation of the total annual bill of the sole industrial's customer using the existing rate structure and rates, if the customer is operating its production facility;
	 the calculation of the total annual bill of the sole industrial's customer using FEI's proposed rate structure and rates (as seen in Table 13-24), if the customer is operating its production facility; and
	iii. the difference between the total bills calculated in response to (i) and (ii) above.
	For your response, please use the average annual throughput of the three years prior to the customer ceasing operations at its production facility.
	51.3

14 **Response:**

15 Please find below a comparison of the total annual bill under FEI's proposed rate structure and 16 rates against the current rate structure and rates. Please refer to years 2005-2007 for the 3 17 years the facility was operating prior to it ceasing operations in 2008. Years 2009-2016 provide 18 a comparison of the rates since they have ceased operations. It can be seen in the table that 19 under FEI's proposed rates, the customer would have paid higher rates per GJ in the years 20 when they were operating, as this particular facility had a biomass system that was the main 21 source of heat for the process, and natural gas provided a backup and space heating role in 22 those years. Since ceasing operations, the customer used gas only for space heating which 23 has a slightly better load factor than when gas was also used as backup to the biomass system. 24 If a new RS 25 customer came into Fort Nelson with a relatively flat load profile, their effective 25 rate per GJ could be even lower, which is why FEI's proposed rate supports economic development to help attract potential new industry to this community. 26

	FN R25	5 Proposed			FN	FN R25 Current		ırrent		Difference	Annual Volume
Year	Year (\$)		(\$	5/GJ)	(\$)		(\$/GJ)		(\$)		(GJ)
2005	\$	168,940	\$	4.26	\$	136,076	\$	3.43	\$	32,865	39647
2006	\$	120,194	\$	4.29	\$	100,231	\$	3.57	\$	19,964	28050
2007	\$	112,403	\$	4.32	\$	92,252	\$	3.55	\$	20,151	25997
2008	\$	168,532	\$	4.39	\$	133,269	\$	3.47	\$	35,263	38418
2009	\$	184,241	\$	4.00	\$	158,481	\$	3.44	\$	25,760	46111
2010	\$	158,862	\$	4.22	\$	131,618	\$	3.50	\$	27,244	37616
2011	\$	149,481	\$	3.99	\$	131,735	\$	3.52	\$	17,745	37460
2012	\$	148,131	\$	3.94	\$	132,130	\$	3.51	\$	16,001	37610
2013	\$	156,000	\$	3.84	\$	139,471	\$	3.43	\$	16,530	40642
2014	\$	167,846	\$	3.57	\$	161,842	\$	3.44	\$	6,004	47039
2015	\$	148,239	\$	3.74	\$	138,666	\$	3.49	\$	9,574	39684
2016	\$	155,379	\$	3.78	\$	143,196	\$	3.48	\$	12,183	41110

FORTIS BC

FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 226

- 1
- 2

3 4

5

6

- 51.4 Please produce a table to show the percentage of the total RS 25 revenues collected through (i) fixed charges; and (ii) variable charges using (a) the existing rate structure and rates; and (b) FEI's proposed rate structure and rates, based on the sole RS 25 customer's 2018 forecast demand of 40 TJ.
- 7 8

9 Response:

- 10 The following table provides the fixed charge revenue, variable revenue, total revenue and the
- 11 percentage of fixed charge revenues to total revenues at existing and proposed rates. The
- 12 second table shows the existing rates and proposed rates.

	Existing		Proposed
Particulars		Rates	Rates
Fixed Charges Revenue	\$	27,936	\$ 108,559
Variable Charge Revenue		114,333	39,684
Total Revenue	\$	142,269	\$ 148,243
% of Fixed Charges			
Revenue to Total Revenue		20%	73%

Existing Rates		
Minimum Delivery Charge \$ / Mo.	\$1	L,826.00
1st 20 GJ \$/ GJ	\$	4.186
Next 260 GJ \$/ GJ	\$	3.884
Excess over 280 GJ \$/GJ	\$	3.179
Administration Charge \$ / Mo	\$	502.00
Proposed Rates		
Basic Charge \$ / Mo.	\$	600.00
Demand Charge \$/GJ of Daily Demand	\$	28.727
Delivery Charge \$/GJ	\$	1.000
Administration Charge S / Mo.	Ś	39.00

13

14

Under the existing rates, the fixed charge revenue includes the minimum Delivery Charge of \$1,826.00 plus the administration charge of \$502.00. The minimum Delivery Charge would include monthly volumes up to 510.4 GJ – effectively a take-or-pay. The variable revenue is the volume in each month that exceeds 510.4 GJ times the trailing block rate of \$3.179. Every month the customer would have to consume 510.4 GJ for the Delivery Charges to equal the minimum Delivery Charge of \$1,826.00. The first two blocks (i.e., 20 GJ plus 260 GJ) plus 230.4 GJ of the trailing block are embedded in the Minimum Delivery Charge.



TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 227

Under proposed rates, the fixed charges include the monthly Basic Charge plus the Administration Charge plus the Demand Charge Revenue. The Demand Charge revenue is the Daily Demand volume times the Demand Charge of \$28.727. The Daily Demand is the highest month average day in the preceding year times 1.1. For forecast 2017, the highest average day use is 266.1 in December which results in a Daily Demand of 292.7 GJ. The variable revenue is the 2018 forecast sales volumes times the Delivery Charge of \$1.000 per GJ.

- 8
- 9

15

- 1051.5Please produce a table to show the percentage of the total RS 25 revenues11collected through (i) fixed charges; and (ii) variable charges using (a) the existing12rate structure and rates; and (b) FEI's proposed rate structure and rates, based13on the average annual throughput of the three years prior to the customer14ceasing operations at its production facility.
- 16 **Response:**

17 The following table provides the fixed charge revenue, variable revenue, total revenue and the

- 18 percentage of fixed charges revenues to total revenues at existing and proposed rates. The
- 19 second table shows the existing rates and proposed rates.

		Existing		Proposed	
Particulars			Rates		Rates
Fixed Charges Revenue		\$	27,936	\$	93,418
Variable Charge Revenue			84,590		31,231
Total Revenue		\$	112,526	\$:	124,650
% of Fixed Charges			25%		75%

Existing Rates		
Minimum Delivery Charge \$ / Mo.	\$1	1,826.00
1st 20 GJ \$/ GJ	\$	4.186
Next 260 GJ \$/ GJ	\$	3.884
Excess over 280 GJ \$/GJ	\$	3.179
Administration Charge \$ / Mo	\$	502.00
Proposed Rates		
Basic Charge \$ / Mo.	\$	600.00
Demand Charge \$/GJ of Daily Demand	\$	28.727
Delivery Charge \$/GJ	\$	1.000
Administration Charge \$ / Mo.	\$	39.00



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 228

1 The fixed charge revenue includes the minimum Delivery Charge of \$1,826.00 plus the 2 administration charge of \$502.00. The minimum Delivery Charge would include monthly 3 volumes up to 510.4 GJ – effectively, a take-or-pay. The variable revenue is the volume in each 4 month that exceeds 510.4 GJ times the trailing block rate of \$3.179.

5 Under proposed rates, the fixed charges include the monthly Basic Charge plus the 6 Administration Charge plus the Demand Charge Revenue. The Demand Charge revenue is the 7 Daily Demand volume times the Demand Charge of \$28.727. The Daily Demand is the highest 8 month average day (for April through October the average day is multiplied by 0.5) in the 9 preceding year times 1.1. For the average of 2005 through 2007 the highest average day use is 10 226.1 in January which results in a Daily Demand of 248.8 GJ. The variable revenue is the 11 three-year average (2005 to 2007) sales volumes times the Delivery Charge of \$1.000 per GJ.

- 12
- 13
- 14

51.6 Please explain whether FEI has done any consultation with the RS 25 customer
on their preference on the rate structure. If yes, please provide the results. If not,
why not?

18

19 Response:

FEI has not done any specific consultation with the RS 25 customer on their preference on rate structure as the customer notified FEI that they are closed permanently. The customer will be

22 using gas only for space heating for a period of time to preserve their assets and will eventually

23 no longer require gas service at all.

FORTIS BC[®]

Ρ.

REBALANCING

1

2

BC^{**} Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1 Page 229 CHAPTER 13 – FORT NELSON FINAL COST OF SERVICE RESULTS AND

3 52.0 **Reference:** FORT NELSON FINAL COST OF SERVICE RESULTS AND 4 REBALANCING 5 Exhibit B-1-1-1, Section 13.7.1.4, p. 13-51 6 Impact to Rate 1 from rebalancing Rate 2.2 7 On page 13-51 of Exhibit B-1-1-1, FEI provides Table 13-27 and describes as follows: 8 Fort Nelson rates must be adjusted to account for the shift in revenue 9 responsibility. For Rate 1, FEI will increase the Basic Charge to \$0.3003 per day 10 so that the \$16 thousand in revenue shift is recovered from all residential 11 customers equally. FEI chose to collect all of the revenue shift through the Rate 1 12 Basic Charge because the lowest consuming customers receive the greatest rate 13 reductions to their annual bills through the unbundling of Fort Nelson residential 14 rates. Before rebalancing, a customer with annual consumption of 34 GJ (one 15 guarter of the average) will experience a 7% decrease to their annual bill. By applying the adjustment only to the Basic Charge, FEI moderates the decrease 16 17 to lower consuming customers making the adjustments more equitable between

- low and high consumers in Rate 1. This also results in Fort Nelson collecting
 more of its customer-related charges through the Basic Charge. Fort Nelson will
 collect approximately 19% of its revenue from Rate 1 through the Basic Charge;
 the customer-related costs in the COSA equal 62%
- 2252.1Please produce a table to show the percentage of the total Rate 1 revenues23collected through (i) fixed charges; and (ii) variable charges using (a) the existing24rate structure and rates; and (b) FEI's proposed rate structure and rates.

26 Response:

Please refer to the table below for the requested information. The numbers in the table excludethe cost of gas.

	Existing Rate	Proposed Rate
\$000	Structure	Structure
Fixed Revenue	30%	19%
Variable Revenue	70%	81%
Total	100%	100%

29

25

30

31



TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 230

52.2 Please calculate the Rate 1 Basic Charge if half of the \$16 thousand in revenue shift was recovered through the Rate 1 Basic Charge and the other half was recovered through the variable charge.

5 Response:

6 The Fort Nelson Rate 1 Basic Charge would equal \$0.2893 per day if only half of the \$167 thousand revenue shift was recovered through the Basic Charge.

8

1

2

3

4

9

10

- 1152.3Please state the annual percentage bill impact that a Rate 1 customer with12annual consumption of 34 GJ will experience after rebalancing, as proposed by13FEI.
- 14

15 **Response:**

As discussed in Section 13.5.4.4 of the Supplemental Filing, due to the 2 GJ monthly threshold for the minimum daily charge calculations and the declining block rate structure of Fort Nelson's existing residential rates, the bill impact on individual customers will depend on their monthly consumption pattern. As such, customers with the same annual consumption level may experience different bill impacts. FEI reviewed nine customers consuming between 32 and 36 GJ per year and the annual bill impacts ranged from an increase of 0.5 percent to a decrease of 20.8 percent, and the group average is a decrease of 6.8 percent.

- 23
- 24
- 25
- 2652.4Please state the annual percentage bill impact that a Rate 1 customer with
annual consumption of 34 GJ will experience after rebalancing, if half of the \$1628thousand in revenue shift was recovered through the Rate 1 Basic Charge and
the other half was recovered through the variable charge.
- 30
- 31 **Response:**

As discussed in the response to BCUC-FEI IR 1.52.3, due to the 2 GJ monthly threshold for the minimum daily charge calculations and the declining block rate structure of Fort Nelson's existing residential rates, the bill impact on individual customers will depend on their monthly consumption pattern. As such, customers with the same annual consumption level may experience different bill impacts.

FEI reviewed the same nine customers as in BCUC-FEI IR 1.52.3, consuming between 32 and 36 GJ per year. After rebalancing and recovering one half of the \$16 thousand revenue shift



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 231

- 1 through the Basic Charge and one half through the Delivery Charge, these customers will
- 2 experience between a 0.6 percent and 21.7 percent decrease to their annual bills.



4

1 53.0 Reference: FORT NELSON FINAL COST OF SERVICE RESULTS AND 2 REBALANCING

Exhibit B-1-1, Section 13.7.1.4, p. 13-51

Rebalancing RS 25

5 On page 13-51 of Exhibit B-1-1, FEI provides Table 13-27 as follows:

Table 13-27: Revenue to Cost and Margin to Cost Ratios after rebalancing

Rate Schedule	COSA after Rate Design Proposals		Rebalance Amount	Approximate Annual Bill	COSA after Rate Design Proposals and Rebalancing	
	R:C	M:C	(\$000)	Change	R:C	M:C
Rate 1	90.9%	88.4%	16.0	1.9%	91.9%	89.7%
Domestic (Residential) Service	50.570	00.470	10.0	1.570	01.070	00.170
Rate 2.1	107.2%	109.4%			107 2%	100 / %
General (Small Commercial) Service					107.270	103.470
Rate 2.2	114 50/	110 10/	9.40/ (16.0)	2.00/	100.0%	112.6%
General (Large Commercial) Service	114.0%	110.4 %	(10.0)	-3.270	109.9%	
Rate Schedule 25	444.00/ 44	111 0%			111 004	111.0%
General Firm Transportation Service	111.0% 111.0%			2000000	111.0%	111.0%

6

The table shows that after the rate design proposals the R:C ratios for Rate 2.2 (114.5%)
and RS 25 (111%) are outside of FEI's R:C ratio range of reasonableness of 90% to
110%.

53.1 Please explain why FEI does not propose to rebalance RS 25 to within the R:C ratio range of reasonableness.

12

13 Response:

FEI considers the range of reasonableness of 90 percent to 110 percent to be a guideline, and not a rule that needs to be adhered to in all situations.

In this case, the sole RS 25 customer has ceased operations permanently and is using gas only for space heating purposes (and temporarily only), the proposed RS 25 rates are being designed for future RS 25 customers. In view of those circumstances, FEI does not consider it necessary that RS 25 has its rates lowered to achieve a 110 percent R:C ratio, particularly since any revenue shift would be allocated to the residential rate schedule.

- 21
- 22
- 23
 24 53.2 Please provide an updated version of Table 13-27 which includes the
 25 rebalancing of RS 25 to within the R:C ratio range of reasonableness.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 233

- Please explain if Rate 1, 2.1 or 2.2 would experience rate shock when combining
 the rate design proposals within the Application with the rebalancing of RS 25 to
 within the R:C ratio range of reasonableness.
- 5 **Response:**

6 The requested update to Table 13-27 is found below.

7 Rebalancing RS 25 to within the range of reasonableness (90 percent - 110 percent) does not

8 cause rate shock for any other rates. However, with the combined rate impact from the RRA for

9 2018 plus the Rate Design Application, the rate impact for residential customers is closer to 10

10 percent with this additional rebalancing. FEI has included an updated Table 13-27 below.

Rate Schedule	COSA a Design P R:C	fter Rate roposals M:C	Rebalance Amount (\$000)	Approximate Annual Bill Change	COSA after Rate Design Proposals and Rebalancing R:C M:C		
Rate 1	90.9%	88.4%	17.4	2.0%	92.0%	89.8%	
Domestic (Residential) Service				,			
Rate 2.1	107.2%	100.4%			107.2%	100.4%	
General (Small Commercial) Service	107.270 109.470			107.270	100.470		
Rate 2.2	111 50/	440.40/	(16.0)	2.00/	100.0%	110.00/	
General (Large Commercial) Service	114.3%	110.4%	(10.0)	-3.2%	109.9%	112.0%	
Rate Schedule 25	111.00/	111.00/	(1.4)	1.00/	110.00/	110.00/	
General Firm Transportation Service	111.0%	111.0%	(1.4)	-1.0%	110.0%	110.0%	

12



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 234

1			TRANSPORTATION SERVICE REVIEW
2	Q.	CHAP	TER 10 – TRANSPORTATION SERVICE REVIEW
3	54.0	Refer	ence: TRANSPORTATION SERVICE REVIEW
4			Exhibit B-1, Section 10.3.6, pp. 10-14, 10-15
5			Impact on core customers
6 7 8 9		In Fig aggree custor daily v	ure 10-5 on page 10-15 of Exhibit B-1, FEI provides a graph that shows the gate actual daily supply and aggregate actual daily demand for transportation mers. On page 10-15, FEI also discusses how a number of factors contribute to the variance between supply and demand. On page 10-14 of Exhibit B-1 FEI states:
10 11 12 13			As seen in Figure 1-15 below, gas supply frequently deviates from demand by as much as 50,000 GJ/day once the day comes to a close These imbalances require FEI to use midstream resources to withdraw or inject quantities of gas, often on an intraday basis to balance the entire System.
14 15 16	Respo	54.1 onse:	Please provide the data points for Figure 10-5 in a working Excel spreadsheet.
17 18 19	Please and d year 2	e refer emand 2015.	to Attachment 54.1 for the requested data points. The data points include supply from both daily and monthly balanced transportation customers combined, in the
20 21			
22 23 24 25 26 27		54.2	Please provide a graph showing the difference between the aggregate actual daily supply and aggregate actual daily demand for transportation customers on a daily basis for 2015. On the same graph please add a line showing the Sumas daily index price.
28	<u>Resp</u>	onse:	
29 30	The g aggre	graph t gate ac	below shows the difference between the aggregate actual daily supply and tual daily demand for transportation customers as well as the Sumas daily index

31 for 2015.

	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
I OKI IS BC	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 235



FEI is unable to draw correlations from the Sumas daily price to the daily variances. Like FEI,
Shipper Agents forecast gas requirements for their transportation customers 24 hours in
advance. Factors like weather and changes in consumption for process load customers may
impact the supply/demand imbalance on a daily basis.

- -

54.2.1 Please discuss the extent to which the Sumas daily price is correlated to the daily variance between the transportation customers' supply and demand.

- **Response:**
- 14 Please refer to the response to BCUC-FEI IR 1.54.2.

- 171854.2.2Please populate the following table for each of the ten days in 2015 with19the greatest absolute difference between the aggregate actual daily20supply and aggregate actual daily demand for transportation service21customers.2224



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission)	Page 236

Date	Aggregate actual daily supply for transportation customers (GJ)	Aggregate actual daily demand for transportation customers (GJ)	Transportation customer daily imbalance (GJ)	Total daily system demand (GJ)	Daily transportation customer imbalance expressed as percentage of total system demand for day(%)	Description of midstream resources used to balance the FEI system on the day

2 Response:

3 The table below provides the ten days in 2015 with the greatest absolute difference between the 4 aggregate actual daily supply and aggregate actual daily demand for both daily and monthly 5 balanced transportation service customers combined. Positive imbalances reflect oversupply 6 while negative imbalances reflect undersupply by Shipper Agents. The daily imbalance for 7 transportation service customers varies from negative 10 percent to positive 14 percent of total 8 system demand. The midstream resources used to balance the FEI system include Aitken 9 Creek Storage, Mist Storage, Jackson Prairie Storage, and Westcoast (WEI) OBA. These 10 storage assets help to balance the system as a whole because there are limited intra-day markets to buy and sell gas on the Westcoast system. Other ways to manage imbalances 11 12 included adding to or subtracting from FEI line pack on a given day. More than likely, on any 13 particular day, more than one of these resources/activities was used to manage the imbalance 14 for the system as a whole.

Date	Aggregate actual daily supply for transportation customers (GJ)	Aggregate actual daily demand for transportation customers (GJ)	Transportation customer daily imbalance (GJ)	Total daily system demand (GJ)	Daily transportation customer imbalance expressed as percentage of total system demand for day(%)	Description of midstream resources used to balance the FEI system on the day
12/29/2015	362,507	263,835	98,672	960,833	10%	Aitken Creek Storage, Mist Storage, WEI OBA
1/1/2015	302,556	214,027	88,529	921,239	10%	Aitken Creek Storage, Mist Storage, WEI OBA
12/17/2015	221,049	304,730	(83,681)	996,778	-8%	Mist Storage, Jackson Prairie Storage, WEI OBA
10/31/2015	271,097	198,674	72,423	501,832	14%	WEI OBA
12/24/2015	293,822	223,560	70,262	871,932	8%	Aitken Creek Storage, Mist Storage, WEI OBA
12/16/2015	214,426	279,365	(64,939)	950,567	-7%	Mist Storage, Jackson Prairie Storage, WEI OBA
3/25/2015	196,617	260,433	(63,816)	666,286	-10%	WEI OBA
11/8/2015	253,862	191,644	62,218	588,859	11%	Aitken Creek Storage, WEI OBA
12/25/2015	276,326	215,628	60,698	886,939	7%	Aitken Creek Storage, WEI OBA
10/30/2015	273,926	213,271	60,655	492,381	12%	Aitken Creek Storage, WEI OBA



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9. 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 237

1	55.0	Refere	ence: TRANSPORTATION SERVICE REVIEW
2			Exhibit B-1, Section 10.3.6, pp. 10-15 to 10-16
3			Shipper Agent inventory levels
4 5		In Figu of day	ure 10-6 on page 10-16 of Exhibit B-1, FEI provides a graph showing the number s of supply held on behalf of all Shipper Agents on FEI's system.
6		On pa	ges 10-15 to 10-16 of Exhibit B-1, FEI states:
7 8 9 10 11			Under normal circumstances, FEI requests that shipper agents holding both daily and monthly balanced groups keep to a 2 to 3 day pack/draft balancing inventory level, which FEI has deemed to be reasonable to manage the System as a whole. The 2 to 3 days of inventory is based on the average consumption of the daily and monthly balanced customer groups divided by the total inventory held.
12 13 14 15 16		55.1	Please clarify whether FEI requests that Shipper Agents stay within a tolerance band of 2 to 3 days of pack inventory level to 2 to 3 days of draft inventory level or whether FEI requests that Shipper Agents hold a pack inventory of 2 to 3 days inventory level.
17	<u>Respo</u>	onse:	

18 FEI requests that Shipper Agents' overall inventory levels are maintained within a tolerance 19 band of 2 to 3 days of pack, to 2 to 3 days of draft. Shipper Agents holding monthly balanced 20 groups have the ability to both pack and/or draft the system, so the 2 to 3 days of tolerance 21 applies to both pack or draft circumstances. The rules in the tariff generally incent Shipper 22 Agents holding daily balanced groups to pack. For daily balanced customers, drafting is not 23 permitted, and in cases where under-deliveries occur on the day, FEI balances the Shipper 24 Agent's group by selling day gas. Given this, the 2 to 3 day pack only applies to Shipper Agents 25 holding daily balanced groups.

As demonstrated by this response, there are different bandwidths or tolerances for monthly and daily balanced customers. Some Shipper Agents hold and pool monthly balanced customers exclusively, some hold both daily and monthly balanced groups, and some marketers pool monthly balanced customers within a daily balanced group. In the interests of consistency, ease of administration and leveling the playing field, FEI would like to move to one set of rules to apply to all transportation customers.

32 33		
34		
35	55.1.1	If FEI requests the former please explain why, as shown in Figure 10-6,
36		the inventory levels are typically at least one day of pack inventory in
37		aggregate for Shipper Agents (i.e. never in an aggregate draft position).



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 238

16

20

21

2 Response:

3 Figure 10-6 shows the days of inventory which are calculated by taking the aggregate of all 4 Shipper Agents' inventory and dividing it by daily demand to determine how many days of 5 supply the marketers have banked on the system at a given point in time. It does not go 6 negative (below zero), as overall Shipper Agents almost always have positive inventory 7 balances due to the net effect of packed or oversupply of daily customers and drafted or under-8 supplied monthly customers. For the days in which Shipper Agents run a net negative 9 imbalance, this is represented by a negative slope or decline, rather than a subzero value.

10 The following charts show the days of inventory for daily and monthly balanced groups 11 separately in 2014 and 2015. In general, daily balanced groups tend to pack while monthly 12 balanced groups tend to draft the system. Although the aggregated inventory levels are typically 13 positive, as indicated in the below figures, the two balancing practices clearly incent different 14 behavior. FEI would like to remove monthly balancing provisions to incent consistent balancing

15 behaviors across all Shipper Agents.



55.2 Does FEI request that Shipper Agents holding only a daily balanced group(s) keep to a 2 to 3 day pack/draft balancing inventory?



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 239

1 2	<u>Response:</u>	
3	Please refer t	o the response to BCUC-FEI IR 1.55.1.
4 5		
6 7 8 9 10	55.3 <u>Response:</u>	Does FEI request that Shipper Agents holding only a monthly balanced group(s) keep to a 2 to 3 day pack/draft balancing inventory?
11	Please refer t	o the response to BCUC-FEI IR 1.55.1.
12 13		
14 15 16	The ir somet	iventory levels in Figure 10-6 show that FEI typically holds at least one day and imes as much as four days of pack inventory for Shipper Agents on FEI's system.
17 18 19 20 21	55.4 <u>Response:</u>	Please add a line to each of the graphs in Figure 10-6 showing the Sumas daily index price.
22 23	The following price for 2014	graphs provide the aggregate inventory of Shipper Agents and Sumas daily index 4 and 2015. FEI is not able to identify a correlation between the inventory and the

24 Sumas price.



- M	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 240





FORTIS BC^{*}

FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 241

- 1
- 2
- 3

4

5

6

7

55.5 Please explain whether FEI physically holds this pack inventory on the FEI system or, alternatively, whether the core sales customer inventory is necessarily in a corresponding draft inventory position in order to balance the FEI system.

8 <u>Response:</u>

9 When a Shipper Agent leaves excess gas on the system, FEI holds this packed or banked 10 supply as inventory on their behalf. FEI returns this supply on a later date by way of the 11 imbalance return mechanism. This is a paper transaction. Physically, FEI manages the system 12 as a whole; when system imbalances occur from either sales or transportation customers. FEI 13 responds in a timely manner to balance the system. Imbalances are managed using FEI's 14 midstream resources including upstream and downstream storage, Westcoast OBA, or the 15 buying and/or selling of gas on the day. In general FEI trends its overall OBAs³⁴ with upstream 16 pipelines to zero on a daily basis.

If the core sales customer inventory is put in a corresponding draft

inventory position to balance the transportation customers' pack

inventory, please describe the midstream resources used and the

nature of the associated costs borne by the core customers as a result.

- 17
- 18
- 10
- 19
- 20
- 21
- 22 23
- 24

25 **Response:**

55.5.1

As indicated in the response to BCUC-FEI IR 1.55.5, irrespective of the entity responsible for the imbalance (whether core customers or transportation customers), fixed resources acquired under the ACP on behalf of core customers are used to balance the system as a whole. Core customers pay for these resources through the Midstream charge. Thus, the imbalance for the whole system is managed by FEI line pack, OBAs, FEI storage accounts, or through sales/purchases of gas with counterparties. All of these activities and their associated costs flow through the Midstream account.

33 On any particular day, the Shipper Agents may pack the system as a whole, which may help to 34 meet the core customers' load on that day. There is no savings to the core customer because

³⁴ FEI's has an Operational Balancing Agreement, or "OBA" with Spectra which requires daily balancing. Imbalances that exceed the threshold at each interconnect point require the utilization of resources on FEI's System, typically by injecting excess gas into storage or withdrawing gas from storage in order to meet imbalance swings.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 242

- 1 the underlying fixed costs of the ACP resources are there regardless. Further, the gas of the
- 2 Shipper Agents has to be returned to them at some point in the future.

In the Application to Amend the Monthly Balancing Charges for Rate Schedules 23, 25, 26 and
27, the Commission directed FEI to evaluate the extent to which FEI uses core gas cost
resources to balance the overall transportation service imbalances for each day and the cost to
the sales customers. The research and analysis to derive an associated cost borne by core

7 customers is included in Section 10.7.4 of the Application.



4

FortisBC Energy Inc. (FEI or the Company)	Submission Date
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 243

156.0Reference:TRANSPORTATION SERVICE REVIEW2Exhibit B-1, Section 10.6.1, p. 10-24

Evolution of technology and systems

On page 10-24 of Exhibit B-1 FEI states:

- 5 The combination of improved technology and increased nomination cycles has 6 resulted in greater ability for market participants to match supply and demand 7 more closely on a daily basis. Transportation customers have access to tools 8 to amend gas requirements on the day to reflect changes in load. For example, 9 over the past several years, there have been technology improvements such as 10 wireless metering, 178 which allow shipper agents to access and track supply and 11 daily consumption by customer more closely. Through FEI's Web Information 12 and Nomination System (WINS), shipper agents have access to historical daily 13 consumption which helps to forecast customer load under varied weather 14 conditions.
- 15 56.1 Please describe, for a typical Shipper Agent with a group of monthly balanced 16 transportation customers, the evolution of the metering accuracy and the method 17 and frequency of communicating the customers' daily consumption data to 18 Shipper Agents since the time of inception of the transportation model in 1993 to 19 today, including the timing and nature of significant improvements over this time. 20 In particular, describe the degree of improvement in accuracy and reliability and 21 in the amount of elapsed time from the time a Gas Day ends to the time the 22 Shipper Agent receives its customers' consumption data for that Gas Day.

2324 **Response:**

To clarify, Shipper Agents representing monthly balanced customers have the same access to customer consumption data as Shipper Agents with daily balanced customers. All Shipper Agents today have access to WINS, which is a self-serve web based application to view individual customer and group demand by day, historical customer consumption, authorized supply from the interconnects, system inventory and imbalances. All Shipper Agents also have the ability to make intraday nomination changes to reflect changes in demand caused by weather or customer behaviour.

32 Since the inception of the transportation services model in 1993, automatic meter readers or 33 AMRs have routinely been installed at the transportation customers' sites (both daily and 34 monthly balanced) in order for FEI to record the daily consumption of each individual customer. 35 Historically the AMR required a dedicated fax line and power line to facilitate the daily upload of 36 data to FEI. While the majority of the sites reported metered data on a daily basis, some 37 locations were not successful, due to AMR failure, disconnected fax or power line. These 38 trouble sites required a manual read at the end of the month, so, based on this, FEI would 39 provide both metered and estimated daily consumption data to Shipper Agents.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 244

1 With respect to improvements made to metering and reporting over time, through the 1990s,

2 FEI provided daily consumption data for the month to date to Shipper Agents twice a week by

fax or email. Shipper Agents would communicate nomination or supply requirements via fax and
 FEI would manually enter supply requests into a database. Imbalance positions would be

5 relayed back to the Shipper Agent with requests to adjust supply accordingly via fax or email.

6 In 2001, Gas Connect was the first application made available to Shipper Agents which allowed 7 users to manage nominations and scheduling. Subsequent phases of this application added 8 functionality to perform balancing for customer groups, a front-end screen for Shipper Agents to 9 manage their business and insert gas requests/nominations within the gas day cycles, intra-day 10 nominations were introduced, electronic data exchange (EDI) for delivery of gas requests and 11 receipt of scheduled results, all in accordance with the NAESB standards. Gas Connect was 12 supported by external spreadsheets and a MS Access database application for manual 13 workarounds.

14 In 2005/06, Gas Connect was replaced by the current WINS system, which provided Shipper Agents with a central portal to view metered consumption by customer and in group aggregate, 15 16 historical data, and demand/supply imbalances. Authorized supply from the interconnecting 17 pipelines was updated in WINS via EDI. The system supported nomination flexibility over four 18 gas cycles, the system would email cut reports by cycle to Shipper Agents, and users could 19 retrieve consumption and inventory reports as required. Updated metered consumption is 20 updated on a daily basis. Shipper Agents can access their customer consumption in WINS 24 hours a day, 7 days a week. 21

22 In 2009 FEI's measurement department started a project to install wireless (cellular) AMR 23 devices at sites where the customer-provided phone line (land line) was not functioning or 24 access to a phone line was never provided by the customer. The adoption of wireless (cellular) 25 AMR technologies has eliminated the reliance on a customer-supplied telephone connection for 26 the transmission of the AMR device data to FEI's data collection servers. As a result of this 27 project, there have been significant improvements in the sites that report actual metered data 28 within 24 hours of the end of the gas day. Since 2009, the percentage of sites reporting actual 29 metered data, as opposed to estimates, has increased from 75 percent to greater than 95 30 percent. FEI does not anticipate any further significant technology enhancements to its 31 measurement gathering system at this time.

Regarding lapsed time in WINS, Shipper Agents have access to metered data two days previous to the current (48-hour time lag). The day immediately prior to the current day is always an estimated quantity, as the time over which the data is collected for the 24-hour gas day is not complete. For Shipper Agents that use the previous day's consumption for specific customers to forecast the next day's load, FEI recommends contacting the customers directly.

With respect to how FEI forecasts gas demand, Gas Control forecasts load based on the weather forecast, degree day calculation (also based on the forecast) and historical loads or trends. For the large industrial RS 22 customers on FEI's system, Gas Control also uses the SCADA system which provides consumption for the previous day, previous hourly flow and



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 245

1 cumulative hourly flow of the current day. SCADA is a telemetered tool, which tracks the 2 movement of gas across FEI's transmission system. FEI's large volume customers are 3 monitored to this degree due to their location on FEI's system, their large volume swings which 4 can impact pressure, and for the purpose of maintaining overall system integrity.

5 To assist in managing large volume customers on FEI's system, Shipper Agents have been 6 provided with access to SCADA. Currently there are seven Shipper Agents accessing real time 7 hourly flows for thirty-nine large volume customers. For Shipper Agents managing these 8 customers, the elapsed reporting time in WINS is made up in the real time flow data in SCADA, 9 which includes the previous day, and current day information.

With the available access to historical data from WINS, and from SCADA for large volume customers, combined with taking into account weather forecast and direct communication with customers, Shipper Agents have the tools to forecast customer demand within a tighter tolerance. As shown in Table 10-8 of the Application, nearly half of Shipper Agents today manage within a 10 percent tolerance, the majority of which have exclusively daily balanced groups. Based on this, the proposed changes to the transportation services model are reasonable and achievable.

- 17
- ...
- 18
- 19
- 20
- 21
- 22
- 23

24 Response:

The table below shows the nomination deadline in Pacific Standard Time for each cycle in a given gas day.

Task	TIMELY	EVENING	ID1	ID2	ID3
Nomination Deadline	11:00	16:00	08:00	12:30	17:00

Please provide a timeline showing, for a particular Gas Day, the

nomination timelines and the timing for when supply and consumption

data, respectively, are available to the Shipper Agent through WINS.

Timely and Evening cycles are done prior to the gas day which starts at 7 am the next day. ID1,
ID2 and ID3 (i.e., intra-day 1, 2 and 3) cycles occur in the current gas day.

30 Shipper Agents can make changes to their nominated or requested supply up to five times for 31 each gas day to best match their forecasted demand. Over time, more nominations cycles have 32 been added to help the industry manage its business as the pipeline systems in general have 33 moved to daily balanced systems. For example, the ID3 cycle was added just recently on April 34 0010 to the WEL number of the time.

34 1, 2016 to the WEI nomination system.

56.1.1

²⁷



	FortisBC Energy Inc. (FEI or the Company)	Submission Date:
TN	2016 Rate Design Application (the Application)	June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 246

- 1 At Station #2 (WEI) and Huntington trading hubs, the majority of the gas is traded between 5:00-
- 2 7:00 am for the Timely cycle, which is 24 to 26 hours prior to the beginning of the particular Gas
- 3 Dav.
- 4 Shipper Agents can view their authorized supply each cycle in WINS approximately 3-4 hours 5 after the cycle deadline indicated above. Supply information would also be available directly 6 from WEI nomination system as the cycles become authorized.
- 7 Consumption data for each transportation customer is uploaded daily within WINS at 8 approximately 4 am each day for each day previous to the current. The data from the previous 9 day relative to the current is always an estimate, as at the time data was collected, the 24-hour 10 gas day is not complete. All other days contain metered daily consumption.
- 11 Please refer to BCUC-FEI IR 1.56.1 which details how FEI forecasts gas demand. Consumption 12 data is one of a few factors that Shipper Agents should be monitoring to help them better 13 forecast customer consumption. For customers with heat sensitive loads, other factors such as 14 weather forecasts, degree day data and historical data or trends should also be considered.
- 15
- 16
- 17
- 18 56.1.2 For a Shipper Agent with a monthly balanced group, describe the 19 current degree of accuracy of the consumption data provided to the 20 Shipper Agent through WINS. In particular, to what degree are 21 imbalances in a Shipper Agent's monthly balanced group due to 22 inaccuracies in the consumption data initially communicate via WINS?
- 23

24 Response:

25 With respect to the accuracy of consumption data in WINS, please refer to the response to 26 BCUC-FEI IR 1.56.1.

27 Shipper Agents holding monthly balanced groups have the same access to customer data as 28 those managing daily balanced groups. Shipper Agents holding daily and/or monthly balanced 29 groups are required to make best efforts to nominate appropriately based on their forecast 30 customer demand. As indicated in response to BCUC-FEI IR 1.56.1, of the approximate 2,400 31 transportation customer sites, 99 percent of all customer sites have Automatic Meter Reading 32 (AMR) which reports metered consumption daily.

33 In its response to BCUC-FEI IR 1.55.1.1, FEI provides graphs showing the packing and drafting 34 patterns for Shipper Agents holding daily balanced groups and monthly balanced groups 35 separately. The graphs show a drafting (i.e., negative imbalance) pattern for Shipper Agents holding monthly balanced groups, and a packing (i.e., positive imbalance) pattern for Shipper 36 37 Agents holding daily balanced groups. FEI does not believe the data available to Shipper



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 247

1 Agents in WINS is responsible for the drafting pattern of Shipper Agents holding monthly 2 balanced groups.

3 Given that many monthly balanced transportation customers have heat sensitive loads, 4 historical consumption data combined with a weather forecast should enable Shipper Agents to 5 derive a load forecast. Shipper Agents managing customers with more process driven loads 6 should be in contact with those customers on a daily basis to forecast their gas requirements. 7 SCADA consumption data by itself will not produce a reasonable forecast for customers 8 because some of the customers within a Shipper Agent group are heat sensitive. FEI believes 9 that Shipper Agents should be using weather forecasts, such as from Environment Canada for 10 example, to produce better demand forecasts for their customers or groups.

- 11
- 12
- 13 14 To the extent there is a lag time in communicating customer consumption data to 56.2 15 Shipper Agents after the end of the Gas Day, please describe any technological 16 changes FEI anticipates will be available to implement in the next few years to 17 further improve the accuracy and access to daily consumption data.
- 18

19 Response:

20 As the Gas Day goes from 7 am to 7am, daily consumption data is available to the Shipper 21 Agents on less than a 24-hour time lag. The gas trading window is done a day out, usually 22 between 5:30 am to 7:30 am, so Shipper Agents are usually trading for tomorrow's gas day prior 23 to yesterday's gas day being completed. FEI does not anticipate any technological changes in 24 the next few years to further improve the accuracy and access to daily consumption data.

- 25
- 26

32

- 27 28 56.2.1 Please elaborate on the extent to which FEI currently has SCADA data 29 for monthly balanced transportation customers and the possibility and 30 associated costs of providing such real-time SCADA data to Shipper 31 Agents.
- 33 **Response:**

34 Generally, Shipper Agents managing monthly balanced customers would not have access to 35 SCADA data for any of their monthly balanced customers. The SCADA system is primarily 36 used by the Gas Control department to manage FEI's natural gas delivery system. Real-time 37 SCADA data is usually only available for large gas users, namely RS 22 customers that are 38 primarily served off the transmission pressure system. Gas Control has no need to have



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 248

SCADA data at all of the sites of monthly balanced customers, as they are usually too small individually and are mostly served off of the distribution system. As such, FEI has no plans to move towards that kind of SCADA solution. Monthly balanced customers are typically more heat sensitive, so the combination of historical data and trends and weather forecast information should be sufficient for Shipper Agents to determine the daily load requirements for these

6 customers.



BC™	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 249

1	57.0	Refere	ence:	TRANSPORTATION SERVICE REVIEW
2				Exhibit B-1, Section 10.6.1, pp. 10-22; Exhibit B-1, Section 10.9, p.
3				10-41
4				Implementation of proposed changes
5		On pa	ge 10-2	2 of Exhibit B-1, FEI states:
6			FEI ob	oserves that shipper agents with a daily and monthly balanced group at the
7			same	location (i.e. the Lower Mainland) typically over-supply their daily group,
8			and g	row a positive inventory through the month to avoid daily balancing
9			charge	es. These same shipper agents also typically under-supply their monthly
10			group	as there are no balancing tolerances on the day for the monthly balanced
11 12			cusion	ners, and in doing so grow a negative inventory through the month. The
13			month	ly to their daily group to avoid imbalance charges at month end.
14		On pa	age 10	-41 of Exhibit B-1 FEI summarizes the changes it proposes to the
15		transp	ortation	service balancing provisions as follows:
16		•	Elimin	ate the existing monthly balanced provisions entirely for the transportation
17			model	and require all transportation customers in all service areas to balance
18			daily.	
19		•	Amen	d the balancing tolerance from 20% to 10%, and implement a tiered charge
20			approa	ach whereby charges increase as tolerance ranges are exceeded.
21		57.1	Please	e discuss the pros and cons of a phased approach whereby monthly
22			baland	ing is eliminated in the first phase and the tolerances are adjusted in a
23			secon	d phase with the opportunity to first assess the extent to which elimination
24			of mo	onthly balancing has reduced the magnitude of the aggregate daily
25			transp	ortation service imbalances on the FEI system.
26	_			
27	Kesp	onse:		

28 FEI believes that Shipper Agents can and should be able to manage to the change of moving to 29 exclusive daily balancing, as research conducted by Black & Veatch indicates this is a common 30 industry standard. Implementing the proposed changes together should be a relatively seamless 31 change, as many marketers today already hold exclusive daily groups and balance under a 10 32 percent tolerance. From FEI's perspective, there are no major pros or cons to a phased 33 approach, as FEI is able to make these changes to the WINS system at any time. Implementing 34 the changes in a phased approach would be acceptable to FEI, but given that some existing 35 shippers are already managing within the 10 percent tolerance, FEI's preference would be to implement both changes at the same time. 36



3 4

5

6

7

8

9

10

TN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 250

57.1.1 Please discuss the extent to which FEI anticipates that elimination of the opportunity for Shipper Agents to transfer balances from daily balanced groups to monthly balanced groups to avoid balancing charges will incent Shipper Agents to balance more closely each day thereby reducing the magnitude of the aggregate daily transportation service imbalances on the FEI system.

11 Response:

12 FEI believes that imposing daily balancing provisions across all service areas will incent Shipper

13 Agents to match supply and demand more closely on a daily basis. As shown in Table 10-8 in

14 the Application, some marketers today hold exclusive daily balanced groups and maintain

15 system imbalances within the 2 to 3 pack bandwidth.



TM	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 251

1 58.0 Reference: TRANSPORTATION SERVICE REVIEW

- 2 Exhibit B-1, Section 10.9, p. 10-41;
- Exhibit A2-3, FEI Application to Amend the Balancing Charges for
 RS 23, RS 25, RS 26 and RS 27 (Monthly Balancing Charge
 Application), Exhibit A-4, BCUC IR No. 2, Attachment ;
- 6 FEI Monthly Balancing Charge Application, Exhibit B-5, BCUC 2.7.1
 7 and 2.7.3;
- 8 FEI Monthly Balancing Charge Application, Order G-187-14 and
 9 Decision dated December 1, 2014, p. 14
- 10

Implementation of proposed changes

- 11 In BCUC IR 2.7 in the FEI Monthly Balancing Charge Application, Commission staff 12 constructed a table showing details of an example month of hypothetical data for a 13 hypothetical Shipper Agent with a monthly balanced group with four scenarios of 14 behavior using the same hypothetical load and Sumas Daily Index prices for each 15 scenario. The details were set out in Attachment 1 in a PDF version of the data and live 16 spreadsheet. A copy of the PDF version of Attachment 1 has been entered by 17 Commission staff as evidence in this proceeding as Exhibit A2-3.
- 18 The four scenarios are:
- 19Scenario 1 The Shipper Agent makes an effort to balance to the group's load20requirements on a daily basis and incurs a Balancing Gas quantity of 2,281 GJ.
- 21Scenario 2 The Shipper Agent generally makes an effort to balance to the22group's load requirements on a daily basis with the exception of four days where23the Shipper Agent does not increase its supply during a high price period, makes24up part of the shortfall later in the month and incurs a Balancing Gas quantity of252,281 GJ.
- 26Scenario 3 The Shipper Agent generally makes an effort to balance to the27group's load requirements on a daily basis with the exception of four days where28the Shipper Agent does not increase its supply during a high price period, makes29up the entire monthly shortfall later in the month and incurs a Balancing Gas30quantity of zero GJ.
- 31Scenario 4 The Shipper Agent generally makes an effort to balance to the32group's load requirements on a daily basis with the exception of four days where33the Shipper Agent significantly reduces its supply during a high price period,34makes up part of this shortfall later in the month and incurs a Balancing Gas35quantity of 2,281 GJ.
- 36The example in Attachment 1 calculates Balancing Gas Charges and shows the Shipper37Agent's cost of gas assuming the Shipper Agent pays the Daily Sumas Index Price for


FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 252

the purposes of showing the relative magnitude of the Balancing Gas Charges relative to
 the cost of gas. The example also shows the amount of Balancing Gas Charges for a
 number of Balancing Gas Charge alternatives.

In response to BCUC IR 2.7.1 in the Monthly Balancing Charge Application, FEI
confirmed that the calculations and assumptions in regard to Balancing Gas volumes
and Balancing Gas costs in Attachment 1 were accurate. In response to BCUC IR 2.7.3
FEI confirmed that, at that time, FEI would not have taken any action with the Shipper
Agent in Scenario 4 in regard to in response to the decrease in the marketer nomination
on the 8th day of the month.

- 10 In the FEI Monthly Balancing Charge Decision accompanying Order G-187-14, the 11 Panel in that proceeding was of the view that FEI presently has the tools to ensure 12 compliance with the tariff under sections 7.3 and 7.6 of the monthly balanced 13 transportation service tariffs and FEI should endeavour to better utilize these tools and 14 amend business practices to ensure compliance with the intent of the tariff.
- 58.1 Please describe any changes FEI has implemented to its business practices in regard to increased use of the tools already available to FEI under the monthly balanced transportation service tariffs since the Monthly Balancing Charge Decision was issued and the effectiveness of such changes.

20 Response:

Since the Monthly Balancing Gas Charge Decision was issued, FEI has exercised the tools within the tariff to ensure balancing compliance for both over-deliveries and under-deliveries. FEI has acted on its ability to change nominated quantities and issued notices specifically advising that Shipper Agents limit supply to no greater than 20 percent above forecasted demand.

Prior to the Decision, in monitoring imbalances on the system, FEI took the approach of leaving the nominating practices in the hands of the Shipper Agent in adhering to balancing requests from FEI. Following the Decision, FEI has on a few occasions taken a more "hands-on" approach, and intervened when FEI believed it was necessary.

30

19

31

32

38

3358.2Does FEI agree that Scenario 4 in Exhibit A2-1 is evidence that, under the
current transportation service tariffs for monthly balanced transportation service,
a Shipper Agent with a monthly balanced group behaving like the Shipper Agent
in Scenario 4 could "game" the FEI system even without a daily balanced group
to transfer balances from at the end of the month? If not, please explain.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 253

1 Response:

Given that Shipper Agents have the ability to draft under the current monthly balancing provisions without incurring any charges, FEI agrees that under Scenario 4 a Shipper Agent could game the system. Some Shipper Agents today hold exclusive monthly balanced groups at a single location and despite requests made by FEI, these shipper agents bring varied supply on throughout the month. The Shipper Agents operating this way typically have varied balances at month end and often incur monthly balancing gas to balance the group.

8

- 9
- 10

15

- 1158.3Does FEI anticipate that FEI's proposal, on page 10-41 in Exhibit B-1, to12eliminate monthly balancing and require daily balancing for all transportation13service rate schedules would prevent the "gaming" exhibited by the Shipper14Agent in Scenario 4? If not, please elaborate.
- 16 **Response:**

FEI cannot say with certainty that the elimination of monthly balancing would eliminate the potential for Shipper Agents to game the system. The shift to exclusive daily balancing will reduce "gaming", but it is difficult to say to what extent, as FEI is unaware of the commercial arrangements made by the Shipper Agents for themselves or on behalf of their customers. The proposals in the Application would effectively level the playing field across all Shipper Agents and their customers, and provide everyone with one set of rules to manage.



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 254

1 59.0 Reference: TRANSPORTATION SERVICE REVIEW

2

3

Exhibit B-1, Section 10.7.2, pp. 10-30 to 10-31;

Review of other jurisdictions

In Section 10.7.2 of Exhibit B-1, FEI describes industry balancing practices for local distribution companies across North America as per the research conducted for FEI by
Black & Veatch. Figure 10-11 on page 10-31 shows a map of North America listing the utilities sampled by Black & Veatch and the Monthly deadband and Daily deadband
tolerances expressed as percentages. On page 10-30 FEI states:

- 9 Industry-wide, balancing provision can vary substantially between local 10 distribution companies (LDCs) based on regional infrastructure differences. For 11 example, balancing provisions can be relatively stringent for LDCs (such as FEI) 12 with service territories adjacent to major natural gas market hubs in order to 13 reduce the opportunity for shipper agents to profit from price swings by running 14 imbalances to transport gas in excess of their contracted transportation quantity.
- 59.1 Please identify those LDCs sampled by Black & Veatch that FEI believes are
 most similar to FEI in terms of regional infrastructure and market hub
 characteristics.
- 18

19 Response:

20 Black & Veatch provides the following response.

21 FEI's regional infrastructure and market hub characteristics are as follows: diverse geographic 22 service territory; served by one pipeline, limited connectivity to upstream pipelines and 23 moderately liquid supply hubs with limited intraday gas trading activity. While one could argue 24 that all LDCs are unique in one respect or another, the LDCs that are most similar are those 25 within the Pacific Northwest, specifically Northwest Natural, Avista, Cascade and Puget Sound 26 Energy. For example, Puget Sound Energy and Northwest Natural are served primarily by one 27 pipeline, Northwest Pipeline, and indirectly by another, Gas Transmission Northwest. However, 28 these LDCs are somewhat dissimilar as the primary upstream pipeline, Northwest Pipeline, 29 does not require daily balancing, as well as their proximity and rights to underground storage. 30 Balancing tolerances for these Pacific Northwest LDCs range between 3 percent and 5 percent 31 depending on the time of year.

- 32 33
- 34
- 59.1.1 Please describe the balancing tolerances and applicable penalties of the transmission pipelines that FEI interconnects with.
- 36 37



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 255

1 Response:

2 An Operational Balancing Agreement (OBA) exists between natural gas pipelines and 3 receipt/delivery parties at interconnects to describe the balancing provisions between nominated 4 and scheduled quantities. OBAs are in place to help manage and maintain efficient pipeline 5 operations on a daily basis for both upstream and downstream parties. OBAs are not a firm 6 physical resource. The use of OBAs as a source of supply for either party is subject to daily 7 operating conditions and availability by either Gas Control group to accommodate the request. 8 FEI has no history of incurring penalties when OBA agreement balancing tolerances are 9 exceeded.

10 FEI interconnects with the Westcoast Energy Inc. (WEI) system at four locations as listed below.

11 Described with each OBA Location are the Daily Tolerance (DT) and Cumulative Tolerance 12 (CT) in GJ:

131. BC Gas Interior Division (aggregated location including BC Gas Interior Off-line)

- 14 DT 15,000 GJ
- 15 CT 15,000 GJ
- 16 2. BC Gas Lower Mainland
- 17 DT 25,000 GJ
- 18 CT 50,000 GJ
- 19 3. Fort Nelson
- 20 DT 3,000 GJ
- 21 CT 6,000 GJ
- 22 4. Kingsvale
- 23 DT 10,000 GJ
- 24 CT 15,000 GJ
- 25

On a relative basis, the OBA tolerances between FEI and WEI are small, given the amount of gas that flows through those interconnect points on a daily basis. Therefore, while FEI does employ the use of OBAs to assist in balancing the system on a daily basis; the amount of supply available can be marginal and not always a reliable source. Overall, FEI works with upstream pipelines on a daily basis to trend these OBAs to zero.

31 FEI interconnects with the TransCanada Foothills BC system at seven locations as listed below.

FEI does not have OBA tolerances at these interconnect locations, requiring FEI to continuallytrend to a zero imbalance.

34 1. East Kootenay Exchange



C.⊾	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 256

- 1 2. Cranbrook Sales Tap
- 2 3. Elko Sales Tap
- 3 4. Fernie Sales Tap
- 4 5. Galloway Sales Tap
- 5 6. Sparwood Sales Tap
- 6 7. Yahk Sales Tap

8 The FEI system interconnects with the Williams Northwest Pipeline system at Huntingdon. FEI 9 does not have OBA tolerances at this interconnect location, requiring FEI to continually trend to 10 a zero imbalance.

11

7

- 12
- 13
 14 59.2 Do any of the LDCs sampled by Black & Veatch allow transfers between daily
 15 and monthly balanced groups? If so, which ones?
- 16

17 Response:

18 The following response was provided by Black & Veatch.

19 Based on the research conducted on over 20 LDCs by Black & Veatch, FEI's current policy 20 allowing Shipper Agents to transfer imbalances between daily and monthly accounts is unique. 21 In fact, most LDCs that have balancing provisions with daily and monthly timeframes require all 22 Shipper Agents to balance both on a daily and monthly basis rather than one or the other. Black 23 & Veatch notes that Pacific Gas & Electric allows customers to choose between monthly 24 balancing and a daily balancing service (which it calls the "Self-Balancing" service), but it does 25 not appear that Shipper Agents are permitted to participate in both schemes simultaneously or 26 transfer imbalances between monthly and daily accounts.

Given FEI's proposal to eliminate monthly balancing and move exclusively to daily balancing,
the transfers that occur today between daily and monthly groups will effectively go away if
monthly balanced groups are eliminated.

- 30
- 31
- 32
- Are the balancing tolerances of other LDCs typically applicable to both packing
 and drafting the LDC? Please identify any LDCs that have tolerances only
 applicable for drafting.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 257

1 <u>Response:</u>

- 2 Black & Veatch provides the following response.
- 3 Black & Veatch has observed that LDCs' balancing tolerances are typically applicable for both
- 4 positive imbalances (packing) and negative imbalances (drafting). In its review of the balancing
- 5 provisions of more than 20 LDCs, Black & Veatch did not note any that had a tolerance that
- 6 applied only to negative imbalance quantities.



BC [∞]	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 258

1	60.0	Reference:	TRANSPORTATION SERVICE REVIEW
2			Exhibit B-1, Section 3.4, p. 3-18; Section 10, p. 10-35;
3 4			FEI Monthly Balancing Charge Application, Order G-187-14 and Decision dated December 1, 2014, p. 25;
5 6			Exhibit A2-4, FEI 2015–2016 Rate Schedule 14A Gas Purchases & Sales Summary – BCUC Order G-152-12 Compliance Filing
7			FEI as Shipper Agent/Rate Schedule 14A
8 9		On page 25 of follows:	of the FEI Monthly Balancing Charge Decision, the Commission directed as
10 11		The I Servio	Panel directs that when FEI makes its Monthly Balanced Transportation ce rate design application that it is to include a review of the impact of FEI
12 12		acting	as a Shipper Agent supplying gas under Rate Schedule 14A to Monthly
14		descr	ibe, in the context of Monthly Balanced Transportation Service, how FEI as
15		a Shi	pper Agent procures gas under Rate Schedule 14A, how its practices are
16		simila	r and dissimilar to other Shippers/Shipper Agents, how it impacts the costs
17		to the	e core, and to provide information on FEI's use of Balancing Gas in a
18		mann	er similar to all other Shippers/Shipper Agents.
19		In Table 3-4	on page 3-18 of Exhibit B-1, FEI provides a summary of the directives from
20		past Commis	ssion decisions relevant to the Application. This table does not include the
21		directives sho	own above from the FEI Monthly Balancing Charge Decision.

60.1 Please provide a reference to where in the FEI 2016 Rate Design Application FEI
 addresses the above directives. If FEI has not addressed these directives in the
 Application, please provide the review as directed in the Monthly Balancing
 Charge Decision.

27 Response:

26

FEI's proposal to eliminate monthly balancing altogether and move to daily balancing for all Shipper groups would include RS 14A. As a result of this daily balancing proposal, FEI inadvertently did not address the above directive and includes a review below as directed in the Monthly Balancing Charge Decision.

32 FEI as a Shipper Agent procures gas under RS 14A through FEI's gas supply department.

Rate Schedule 14A provides a positive benefit to the costs of the core. A market factor premium of the greater of \$0.06 CDN per GJ or cost is added to all gas purchases. The core market receives any proceeds from the spread between market factor premium and actual costs, which are reported in the annual RS 14A Purchase and Sales Summary to the Commission. In Order G-64-04 regarding Rate Schedules 7, 10, 14 and 14A for the 2004/05



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 259

Gas Year, the Commission determined that the market factor premium should be increased from the greater of \$0.05/GJ or cost, to the current level of the greater of \$0.06/GJ or cost. The increase in the market factor premium of \$0.01/GJ was to cover any potential core administration cost related to RS 14A gas supply purchases. The market factor premium is in addition to the RS 14A management fees of \$0.05-\$0.09 per GJ for administering RS 14A.

6 FEI's contracting practices in its role as a Shipper Agent supplying RS 14A customers are 7 generally reflective of industry practice that would apply to other Shippers and/or Shipper 8 Agents acquiring gas supply for delivery to the FEI system at Huntingdon. The main difference 9 is that FEI is the sole source for purchases (including any incurred Balancing Gas) on behalf of 10 RS 14A customers, whereas Shipper and/or Shipper Agents can purchase from the 11 marketplace or potentially under-deliver on their supply requirements and incur Balancing Gas 12 supply from FEI. Additionally, when compared to FEI, depending on the credit worthiness of 13 the Shipper and or/ Shipper Agent, the Shipper/Shipper Agent may get charged higher 14 premiums or prices in the marketplace because of this risk. Further, the number of 15 counterparties available to supply gas to the Shipper and/ or Shipper Agents may be more 16 limited.

Below is an update to the table provided in response to BCUC IR 2.1.3³⁵ from the FEI Monthly Balancing Charge Application that shows FEI's use of balancing gas under RS 14A. As discussed further in response to BCUC-FEI 1.60.9.2, FEI has improved its nomination processes in recent years. As a result, as shown in the table below, FEI has reduced balancing gas amounts, while also maintaining minimal month end inventory levels.

³⁵ FEI has identified that the original response to BCUC IR 2.1.3 for the FEI Monthly Balancing Charge Application contained a few errors. The confidential response to BCUC IR 2.1.4 had the correct numbers for FEI Rate Schedule 14A. This table now corrects and updates the information originally filed in BCUC IR 2.1.3 for the FEI Monthly Balancing Charge Application.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 260

RS 14A Historical Balancing Gas and Month End Balance

	Monthly	Balancing Gas	Backstonning	Month Fnd	Balancing Gas as
Date	Load (GJ)	(GJ)	Gas (GJ)	Balance	% of Total Load
2012	987,041	72,191	-		7.31%
January	116,738	16,280		-	13.95%
February	102,310	16,781		-	16.40%
March	105,811	12,780		-	12.08%
April	75,669	1,709		-	2.26%
May	63,056	1,931		-	3.06%
June	50,350	598		202	1.19%
July	34,086	864		88	2.53%
August	31,662	2,621		-	8.28%
September	39,701	6,561		-	16.53%
October	81,412	2,097		-	2.58%
November	131,328	7,198		-	5.48%
December	154,918	2,771		-	1.79%
2013	1,475,482	76,608	-		5.19%
January	175,009	3,653		-	2.09%
February	158,606	9,830		-	6.20%
March	149,672	854		-	0.57%
April	130,440	4,969		3,882	3.81%
May	87,332	6,908		-	7.91%
June	62,741	2,832		-	4.51%
July	50,282	4,632		10	9.21%
August	47,639	5,346		-	11.22%
September	66,493	16,988		-	25.55%
October	135,683	11,233		-	8.28%
November	177,194	3,139		-	1.77%
December	234,391	6,224		-	2.66%
2014	1,534,479	40,722	-		2.65%
January	219,452	12,154		-	5.54%
February	225,678	16,629		-	7.37%
March	185,718	2,544		3,052	1.37%
April	139,100	210		566	0.15%
May	97,455	379		566	0.39%
June	77,988	1,107		140	1.42%
July	59,137	351		87	0.59%
August	56,458	31		38	0.05%
September	65,681	1,165		9	1.77%
October	102,526	-		284	0.00%
November	147,428	3,553		397	2.41%
December	157,859	2,598		843	1.65%



FortisBC Energy Inc. (FEI or the Company)	Submission Date
2016 Pate Design Application (the Application)	
	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 261

	84	Palanting C	Destates		
	Wonthly	Balancing Gas	Backstopping	Nonth End	Balancing Gas as
Date	LOad (GJ)	(GJ)	Gas (GJ)	Balance	% of Iotal Load
2015	1,263,635	6,278	-		0.50%
January	154,504	1,292		451	0.84%
February	124,387	272		1,609	0.22%
March	124,187	-		2,268	0.00%
April	110,218	233		1,603	0.21%
May	79,113	183		12,257	0.23%
June	63,547	24		1,366	0.04%
July	56,603	-		205	0.00%
August	57,328	130		892	0.23%
September	76,631	858		296	1.12%
October	101,905	187		272	0.18%
November	153,432	1,509		17	0.98%
December	161,781	1,591		17	0.98%
2016	1,333,094	16,209	-		1.22%
January	160,655	1,104		17	0.69%
February	136,231	-		3,196	0.00%
March	129,759	-		3,444	0.00%
April	92,569	1,273		146	1.38%
May	84,286	427		26	0.51%
June	71,191	-		625	0.00%
July	65,414	6,064		399	9.27%
August	68,992	5,440		488	7.88%
September	83,080	737		-	0.89%
October	121,388	999		132	0.82%
November	127,008	167		201	0.13%
December	192,522	-		4,827	0.00%
2017					
January	187,936	16,135		6,467	8.59%
February	161,141	3,216		358	2.00%
March	161,719	-		2,700	0.00%
April	129,835	215		471	0.17%

60.2 Does FEI intend for the changes in the balancing rules proposed in the Application to apply to FEI in its role as Shipper Agent? If not, why not.



W	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 262

1 Response:

- Confirmed. The proposed changes in the Application will apply to all Shipper Agents, including
 FEI acting as a Shipper Agent under RS 14A.
- 4 5 6 7 8 Exhibit A2-4 is FEI's 2015–2016 Rate Schedule 14A Gas Purchases & Sales Summary 9 filed in compliance with Commission Order G-152-12. In the table included in the report, 10 FEI reports purchases and sales to the transportation service customers that FEI 11 procures supply for in three categories: Index Rate Option; Term Fixed price Option and 12 RNG Option. 13 Please confirm, or otherwise explain, that the purchases listed in the categories 60.3 14 of "Index Rate Option" and "Term Fixed Rate Option" are supplied to FEI's 15 transportation service customers under Rate Schedule 14A. 16 17 **Response:** 18 Confirmed. 19 20 21 22 Please confirm, or otherwise explain, that the purchases listed in the categories 60.4 of "RNG Option" are biomethane (also referred to as renewable natural gas or 23 24 RNG) purchases supplied to FEI's transportation service customers under Rate 25 Schedule 11B. 26 27 Response: 28 Confirmed. 29 30 31
- 32 60.5 Is the supply and demand for FEI's transportation service customers tracked in
 33 WINS? If not, why not?
- 34



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 263

1 **Response:**

- 2 Confirmed. Supply and demand data for all customers managed by Shipper Agents, including 3 those under RS 14A, is tracked in WINS.
- 4
- 5

6 7

8

- 60.6 Do Shipper Agents other than FEI supply transportation service customers with RNG purchased under Rate Schedule 11B?
- 9 10 Response:
- 11 All individual transportation customers and/or their Shipper Agents have the opportunity to 12 purchase RNG under RS 11B.
- 13 In addition to FEI, at present there are three other Shipper Agents representing six 14 transportation customers that are actively purchasing RNG volumes from FEI under RS 11B.
- 15
- 16

- 17

21

- 18 Is RNG supply captured in WINS on a daily basis and is it factored into the daily 60.7 19 balance of supply and demand for a Shipper Agent's Group for the purpose of 20 determining balancing charges? If not, why not?
- 22 Response:

23 The RNG supply is captured in WINS and can be viewed in the Shipper Agent's Inventory 24 Report.

25 In order to facilitate an RNG sale to a transportation customer, FEI transfers the purchased 26 quantity of RNG into the Shipper Agent's group in which the transportation customer resides. 27 RNG sale quantities are typically transferred into the group once a month in a lump sum. The 28 purchased RNG supply is factored into the overall inventory of the group. It is not factored into 29 or added to the direct physical supply on the day, but, given that the supply inflates the Shipper 30 Agent's inventory, it does impact the determination of balancing charges.

31

32

33



	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
sc	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 264

- In Table 10-8 on page 10-35 of Exhibit B-1, FEI provides data on the balancing
 performance of "shipper agents operating under the transportation service model today"
 listing individual Shipper Agents and identifying each with a unique letter.
- 4 60.8 Does Table 10-8 include all Shipper Agents active on the FEI system during this
 5 timeframe? If not, why not?

7 <u>Response:</u>

6

8 Table 10-8 includes all Shipper Agents with the exception of FEI acting as a Shipper Agent
9 supplying gas under RS 14A. Please refer to the updated table in response to BCUC-FEI IR
10 1.60.9.1, which includes FEI.

- 11 12 13 14 60.9 Please identify which Shipper Agent in Table 10-8 is FEI. 15 16 Response: 17 Please refer to the response to BCUC-FEI IR 1.60.8. 18 19 20 21 60.9.1 If FEI is not included in Table 10-8, please explain why not and please 22 provide a revised version of Table 10-8 including lines for FEI with FEI 23 identified as "Shipper Agent FEI." 24 25 **Response:**
- FEI was not included. FEI as a Shipper Agent is now included in the revised Table 10-8 below and is identified as Shipper Agent FEI.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 265

1

Table 10-8 (Revised): Imbalance Date under a 10% Tolerance

Shipper Agent	Daily / Monthly	Service Area	# Imb Days / Year	Annual Volume in Excess	Volume in Excess / Day	Demand / Day	Volume in Excess / Demand
Shipper Agent N	М	INL	287	-2,010	-6	8	-67%
Shipper Agent FEI	М	LML	262	-526,205	-1,442	3,291	-44%
Shipper Agent N	М	LML	219	-30,843	-85	230	-37%
Shipper Agent M	М	LML	216	-74,312	-204	467	-44%
Shipper Agent I	М	INL	210	-28,100	-77	414	-19%
Shipper Agent E	D & M	INL	203	-209,596	-574	2,128	-27%
Shipper Agent C	D & M	LML	185	-848,871	-2,326	13,829	-17%
Shipper Agent O	М	LML	170	-4,442	-12	124	-10%
Shipper Agent D	М	INL	169	-210,408	-576	3,401	-17%
Shipper Agent D	М	LML	161	-652,440	-1,788	14,446	-12%
Shipper Agent E	D & M	LML	149	-691,630	-1,895	13,008	-15%
Shipper Agent A	D & M	LML	137	-256,193	-702	19,970	-4%
Shipper Agent C	D & M	INL	115	-143,545	-393	8,173	-5%
Shipper Agent I	D & M	LML	109	-56,657	-155	2,591	-6%
Shipper Agent FEI	М	INL	85	-24,633	-67	502	-13%
Shipper Agent H	D	INL	17	-21,248	-58	5,293	-1%
Shipper Agent B	D & M	INL	12	-13,784	-38	15,191	0%
Shipper Agent A	D & M	INL	11	-59,806	-164	10,978	-1%
Shipper Agent F	D	INL	7	-22,161	-61	14,602	0%
Shipper Agent B	D	LML	5	-7,141	-20	15,641	0%
Shipper Agent K	D	INL	4	-2,767	-8	1,199	-1%
Shipper Agent L	D	LML	3	-2,049	-6	1,155	0%
Shipper Agent H	D	LML	1	-405	-1	3,027	0%
Shipper Agent G	D	INL	1	-921	-3	9,830	0%
Shipper Agent I	D	IMI	1	-69	0	1.435	0%

- 2
- 3
- 4
- 5 6
- 7
- 8
- 9

10

Response:

11 Over the two-year period from 2014 and 2015, Shipper Agent FEI is above the red line. As the 12 table shows, however, the volume in excess on the day above the 10% threshold is relatively 13 small for both groups.

and demand on a daily basis.

If FEI is one of the Shipper Agents above the red line on Table 10-8,

please explain why FEI was not able to more closely balance supply

60.9.2

Since the FEI monthly balancing gas charge proceeding in the second half of 2014, FEI has adjusted its nomination processes for RS 14A and is now more closely managing supply and demand. FEI has reduced balancing gas amounts, while also maintaining minimal month-end inventory levels. This trend can be seen in the table provided in response to BCUC-FEI IR 1.60.1, which shows a number of years of data for FEI, including 2016 and part of 2017.



FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 266

1 61.0 Reference: TRANSPORTATION SERVICE REVIEW

2

3

4

5

6

7

8

13

Exhibit B-1, Section 10.8, p. 10-39;

T-South allocation to transportation service customers

In Section 10.8 of Exhibit B-1, FEI discusses the allocation of 40 TJ/day of firm transportation service from Spectra Energy south to the Huntingdon Delivery area (T-South Long-Haul) that FEI secured in late 2015. This capacity was contracted by FEI to provide additional T-South Long-Haul capacity for transportation service customers potentially seeking to return to bundled sales service.

9 61.1 Please confirm, or otherwise explain, that FEI contracted for the 40 TJ/day of 10 excess T-South Long-Haul capacity primarily on the basis that should 11 transportation service customers revert to bundled sales service FEI would have 12 an obligation to serve these customers as bundled sales customers.

14 <u>Response:</u>

15 Confirmed. It is also important to note that the 40 TJ/day of T-South Long-Haul capacity³⁶ 16 makes up only a portion of the transportation service customers' total load requirements. If all 17 transportation service customers return back to the bundled service, FEI would have to contract 18 additional resources.

19

20

21

26

61.2 Have any transportation service customers reverted to bundled sales service
since FEI contracted the 40 TJ/day of excess T-South Long-Haul capacity? If so,
what amount of the 40 TJ/day of excess capacity is required to serve these
customers.

27 <u>Response:</u>

Since FEI contracted the 40 TJ/day of excess T-South Long-Haul capacity, the customer movement between the bundled and unbundled sales service has been minimal, which is consistent with past years. However, as the chart below shows, the potential for customers to return to bundled service may increase compared to the past, as FEI's rate has been lower than the Sumas daily price over the past year. In recent weeks, FEI has been fielding inquiries from customers asking about returning to bundled service for November 1, 2017.

³⁶ T-South Long-Haul is also known as T-South Huntingdon Delivery Area, which is defined on page 1.11 of Westcoast Energy's General Terms and Conditions as the area comprised of the Export Delivery Area and the Lower Mainland Delivery Area.



FortisBC Energy Inc. (FEI or the Company)	Submission Date:
2016 Rate Design Application (the Application)	June 9, 2017
Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 267

1

FEI's RS 5&7 vs Sumas Daily Price



2

The risk that transportation service customers will come back to the bundled service remains, as customers relying on the Huntingdon market continue to face an uncertain future. The potential new incremental load from industrial projects within the Lower Mainland and the Pacific Northwest, and a fully contracted T-South pipeline, risks leaving transportation service customers without adequate gas supply or they will need to pay significantly higher commodity prices at Huntingdon. This risk will likely persist if this incremental new demand arrives before any additional infrastructure is completed.

- 10
- 11
- 10
- 12
- 61.3 Please confirm, or otherwise explain, that T-South Long-Haul capacity is
 currently fully contracted and not available on a firm basis regardless of Spectra
 Energy's credit requirements for contracting firm service.
- 16

17 <u>Response:</u>

18 Confirmed. Historically, Spectra offered up to 1,700 MMcf/d of contractible T-South Long-Haul 19 firm 365-day transportation service based on the winter design capacity of its system. In 20 October 2014, Spectra reduced contractable firm service to 1,450 MMcf/d based on its expected 21 system capacity in the summer months. The announcement resulted in shippers, including FEI, contracting for the last remaining contractable 365-day Westcoast T-South Long-Haul capacity. 22 23 In November 2016, upon receiving NEB approval, Westcoast conducted an open season to 24 contract for 160 MMcf/d of T-South Long-Haul capacity for each November to March winter 25 period commencing November 1, 2017 (Winter Firm Service). The Winter Firm Service capacity



NTN	FortisBC Energy Inc. (FEI or the Company) 2016 Rate Design Application (the Application)	Submission Date: June 9, 2017
	Response to British Columbia Utilities Commission (BCUC or the Commission) Information Request (IR) No. 1	Page 268

was 160 MMcf/d of the remaining 250 MMcf/d of T-South Long-Haul capacity. This capacity
was purchased by Northwest Innovations Works, a Company that is proposing to construct
methanol production plants in Washington and Oregon State, and Painted Pony, a major
producer in Northeast BC.

5 Westcoast is currently conducting an Open Season for the remaining 90 MMcf/d of 365-day 6 contractable portion of T-South Long-Haul capacity, and a small 100 MMcf/d expansion. This 7 Open Season will be the last firm contractable pipeline capacity available to secure until a larger 8 scale pipeline expansion is built, which would likely occur no earlier than 2023. However, that 9 timeline could be pushed out even further, as there are still a number of uncertainties with a 10 large scale pipeline expansion, such as new customers willing to underwrite the expansion and 11 environmental/regulatory challenges.

- 12
- 13
- 14 15

61.4 Has FEI allocated any of the excess T-South Long-Haul capacity to FEI as Shipper Agent? If so, please elaborate.

16 17

18 **Response:**

19 FEI has not allocated any of the excess T-South Long-Haul capacity to FEI as a Shipper Agent.

Attachment 7.1

Attachment 7.1

INFORMATION BULLETIN

Bulletin # (See Standard GEN 01-13)	Effective Date (Yr/Mth/Day)	Cross-references, if any (e.g., Operating Procedure SAF 30-7, dated 26 Jan 96)		
2008-43	08/11/03	See standards listed below		
Audience (e.g., engineers, install managers, cons	truction personel)	Replaces (e.g., Information Bulletin CS/1995/12, dated 10 Sept 96)		
See list below		course contents to appropriately reflect this diat		
Originating Group or Department				
Engineering		is bandware and they appeared WH Construction with the environment of		
Subject Matter Expert Name		Subject Matter Expert Signature		
Frank Mostad		Smath and		
Owner's Title	Owner's Name	Owner's Signature		
Dist Assets & Improvements Mana	ger Gary Johnson	Att		
Note: The origina	l of every Information Bull	etin <u>must</u> always be sent to Engineering Governance		

This bulletin is being issued to convey information of an urgent nature, of a general nature or information that clarifies, but does not alter the intent of, an approved policy or procedure.

Elimination of 88 mm PE Pipe and Restricted Usage of 42 mm PE Pipe

This bulletin is mandatory reading for the following job functions: Engineering, Capacity Planning, Procurement, Inventory Analysts, Training, Planning & Design, Install Managers, Install Crews, Contract Crews, and Project Management.

Effective immediately, 88 mm PE pipe is to be discontinued for all upgrades and new installations. 42 mm PE pipe is to be restricted to single services without branches only, and will no longer be permitted for use as mains or service headers.

88 mm PE pipe was deemed to be an impractical size that will be replaced by 114 mm PE pipe. Short 2 metre lengths of 88 mm PE pipe will continue to be stocked for emergency purposes only. These short lengths will be protected from UV radiation and contamination. See part number 250-2219.

42 mm PE pipe has had a number of fusion melt-through issues and therefore represents a greater risk in our system. This size will be replaced by 60 mm PE pipe.

For existing planned jobs where 42 mm PE pipe and 88 mm PE pipe has been nominated, these jobs do not need to be redesigned and may proceed as planned providing that there is sufficient material in inventory.

Parts affected within the Approved Product Catalogue include:

- 42 mm PE Coil Pipe, Part Number 250-2208 use to depletion then delete part
- 42 mm PE Straight Pipe, Part Number 250-2230 small quantities will be stocked for repairs and short replacements only, as well as single non-branched services
- 88 mm PE Straight Pipe, Part Number 250-2216 use to depletion then delete part
- 88 mm PE Reeled Pipe, Part Number 250-2217/250-2218 parts to be deleted
- 88 mm PE Straight Pipe, 2 m length, Part Number 250-2219 new part

Designers, Planners, Install Managers, Install Crews, and Contractors are advised to start utilizing 60 mm PE pipe and 114 mm PE pipe instead of 42 mm PE pipe and 88 mm PE pipe in their planning, construction and maintenance jobs. The Training Department should revise their course contents to appropriately reflect this change.

Most of the related PE fittings will be stocked in quantities appropriated to these changes and these quantities will be adjusted as we gain experience through usage.

Impact on Standards

The following standards must be reviewed for impact and updated as required, prior or by June 1st, 2009, to reflect the changes.

Standard No.	Standard Name	Owner	SME
MAT 05-01	Polyethylene Pipe and Tubing	F Pataki	F Mostad
MAT 06-30	Polyethylene Ball and Plug Valves	F Pataki	F Mostad
DES 04-01-01	Distribution System Piping Design	F Pataki	F Mostad
DES 04-01-04	PE Pipe	F Pataki	F Mostad
DES 04-01-17	Use and Installation of PE Pipe	F Pataki	F Mostad
DES 04-02-06	PE Service Material	F Pataki	F Mostad
DES 04-01-08	Sizing of Distribution Pipe – Mains and Services	F Pataki	T Penner
CON 04-02	Socket Fusion	S Barbour	M Walls
CON 04-07	Fusion and Hot Tapping of Side Saddles	S Barbour	R Arnott
CON 04-09	Fusion Tests, Certificates, and Inspections	S Barbour	S Barbour
CON 06-02	Installing Stub Services – PE Main	D Johnston	R Arnott
CON 06-03	Installing PE Services	D Johnston	P Kropp
CON 06-06	Completing Stubbed Services	D Johnston	R Arnott
CON 11-04	Handling PE Pipe and Fitting	R Samels	V Triggiano
CON 11-08	Inserting PE into Ducts	S Barbour	P Tassie

If you have any questions with regards to these changes, please contact Gary Johnson in Asset Management at 604-592-7739, or Frank Mostad in Engineering Services at 604-592-7899.

2476 05/06

Fairs of Sectod within the Approved Preduce Catalogue matrice.

41 mm Pd. Coll Phys. Fur. Number, 270–2103 – and in depiction theor little's pirt
42 mm Pd. Straight Phys. Fur. Number, 259 -22230 – and in depiction theor little's pirt
43 mm Pd. Straight Phys. Fur. Number, 259 -22330 – and in depiction theor little's pirt
83 mm Pd. Straight Phys. Rev. D 1000 (2000) 2107 – 210 -22300 – and pressoration in all be atordered (pirture state) and pressoration in all be atordered (pirture state) (pirt) (pirt)

Attachment 25.10



TARIFF SUPPLEMENT NO. E-2

FURTHER AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT RATE SCHEDULE 25

BETWEEN

DUNKLEY LUMBER LIMITED

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc.)

Effective November 1, 2004

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT ("Agreement") made as of the 11th day of March, 2004, with effect as and from November 1, 2004 is a further amendment and restatement of an agreement made as of June 1st, 2002 with effect as and from November 1st, 2002, itself being an amendment and restatement of an agreement made as of December, 1989 between the parties noted below.

BETWEEN:

TERASEN GAS INC. (formerly known as BC Gas Utility Ltd. and BC Gas Inc.), a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "Terasen Gas")

OF THE FIRST PART

AND:

DUNKLEY LUMBER LIMITED, a company incorporated under the laws of British Columbia having its registered office at P. O. Box 173, Prince George, British Columbia

(hereinafter called "Dunkley")

OF THE SECOND PART

RECITALS

WHEREAS:

- A. Dunkley operates sawmill facilities near Prince George, British Columbia, and requires Gas for its sawmill operations located at Dunkley Road adjacent to Highway 97 North.
- B. Terasen Gas owns and operates a Gas transmission pipeline which is connected to the sawmill operations of Dunkley near Prince George, British Columbia.
- C. Dunkley entered into an agreement with BC Gas Inc. (now known as Terasen Gas Inc.) dated December, 1989 effective January 1, 1990 (the "1990 Bypass Agreement"), which allows Dunkley to receive transportation service from Terasen Gas at rates and on terms and conditions that are based on the principles of the British Columbia Utilities Commission decision of October 2, 1987 respecting a bypass pipeline application by Northwood Pulp and Timber Limited.
- D. On April 15, 1993, BC Gas Inc. filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Diane Roy, Director Regulatory Services

- E. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service.
- F. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties entered into an amended and restated Agreement to clarify their respective rights and obligations effective November 1, 1993.
- G. The forecast annual volume in the 1990 Bypass Agreement (Schedule 1) and in the 1995, effective November 1, 1993, Restated and Amended Agreement was 157,050 gigajoules (GJ). The Daily Transportation Quantity (DTQ) in that same Restated and Amended agreement was 617 gigajoules (GJ).
- H. For the Further Amended and Restated Bypass Transportation Agreement dated June 1, 2002 and effective November 1, 2002 (the "2002 Agreement") the annual forecast volume was increased to 250,000 GJ and the DTQ was increased to 1,540 GJ. The daily capacity of the Bypass Pipeline was 1,855 GJ under the 2002 Agreement. The increased annual volumes and DTQ used for the 2002 Agreement exceeded 120 percent of the forecast volumes and DTQ in the Restated and Amended Agreement effective November 1, 1993. To deliver the increased annual volumes and the DTQ, additional line pressure equipment would have been incurred for the Bypass Pipeline. No incremental operating and maintenance expense and property tax was incurred. The resulting incremental annual cost to Dunkley under the 2002 Agreement was \$36,327.
- For this Agreement, the annual forecast volumes are 460,000 GJ and the DTQ is 4,153 GJ. The daily capacity of the Bypass Pipeline is 4,984 GJ. The increased annual volumes and DTQ used for this Agreement exceed 120 percent of the forecast volumes and DTQ in the 2002 Agreement. To deliver the increased annual volumes and the DTQ required under this Agreement, a 4.2 km loop and upgraded metering facilities would have been incurred for the Bypass Pipeline. Incremental operating and maintenance expense and property tax will be incurred under this Agreement.
- J. Resulting incremental annual cost to Dunkley is \$12,527 under this Agreement. Part of the annual cost will be recovered by the Delivery Charge that is related to the annual cost in Schedule 1 of the 1990 Bypass Agreement times the incremental volumes of 210,000 GJ (460,000 GJ 250,000 GJ). \$0.046 of the \$0.07 Delivery Charge pertains to the annual cost of the 1990 Bypass Pipeline Agreement: \$9,660 (210,000 GJ x \$0.046). The incremental annual fixed cost under this Agreement is \$2,867 (\$12,527 \$9,660) or \$239 per Month.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Diane Roy, Director Regulatory Services

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

> "Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by Dunkley in order to allow Dunkley to bypass the Terasen Gas transmission and distribution Gas system.

"DTQ" means Daily Transportation Quantity as defined in the Rate Schedule 25 Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which Dunkley may elect to receive service.

"Terms and Conditions of the Transportation Rate Schedule" means the terms of the applicable transportation Rate Schedule and Transportation Agreement thereunder, and the General Terms and Conditions of Terasen Gas, as approved by the British Columbia Utilities Commission by its Decision of October 25, 1993 or any subsequent transportation Rate Schedule accepted for filing by the British Columbia Utilities Commission.

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Rate Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 Terasen Gas will provide firm transportation service to Dunkley for its sawmill operations near Prince George, British Columbia and Dunkley will accept such transportation service in accordance with the then prevailing provisions of Terasen Gas Rate Schedule 25 and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Rate Schedule conflicts or is inconsistent with the terms and conditions set out in this Agreement, this Agreement governs.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Diane Roy, Director Regulatory Services

2.03 Dunkley shall be entitled to take service under a transportation Rate Schedule other than Rate Schedule 25 subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.

ARTICLE 3

TERM OF AGREEMENT

- 3.01 Subject to Article 9, the initial term of this Agreement will be for a period of 10 Years, effective the 1st Day of November 2004 up to the 1st Day of November 2014.
- 3.02 The term of this Agreement shall be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon Dunkley providing written notice to Terasen Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 Months prior to the then current termination date and Terasen Gas agreeing to such extension of the term of this Agreement. Terasen Gas shall not unreasonably withhold agreement to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension under this Agreement will be for a period of not less than 1 contract Year.

ARTICLE 4

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for, Dunkley will each Month pay to Terasen Gas for services provided hereunder from November 1, 2004 to the expiry of this Agreement the following rates:
 - (i) A monthly charge of \$11,571.60;
 - (ii) A Delivery Charge of \$0.07 / gigajoule; and.
 - (iii) Any charges pursuant to Articles 4.02, 4.03 and 4.04.
- 4.02 Upon changes in its DTQ pursuant to Article 5 or in the annual Gas quantity consumed under this Agreement in excess of 120 percent of those forecast in Recital I, the monthly charge will be adjusted by Terasen Gas to reflect any changes in costs which would have been incurred by Dunkley as a result of the change in the DTQ or annual Gas quantity had Dunkley constructed and operated the Bypass Pipeline.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Diane Roy, Director Regulatory Services

- 4.03 The monthly charge and the Delivery Charge as specified under Article 4.01 will be adjusted effective November 1 of each contract Year on the following basis to compensate Terasen Gas for the following changes in its costs:
 - (i) the operating, maintenance and property tax expenses included in the monthly charge as at November 1, 2004 are \$2,447.54 per Month and will be subject to the annual percentage change in the Consumer Price Index for the City of Vancouver for October, up to the contract Year expiring November 1, 2005. Thereafter, the Consumer Price Index for the Month of August will be used. The monthly charge will be adjusted by the calculated change in costs. The adjusted operating, maintenance and property tax expenses will form the basis for the following contract Year adjustment; and
 - (ii) any new or increased tax related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Dunkley will be calculated on a monthly basis (the annual cost divided by 12) and added to the monthly charge specified in Section 4.01 (i).
- 4.04 Unauthorized Overrun Gas will be charged for and paid by Dunkley in accordance with the rate set out in Rate Schedule 25.

ARTICLE 5

ADJUSTMENTS TO DTQ

5.01 The DTQ for the contract Year commencing November 1, 2004 is agreed to be 4,153 gigajoules. Dunkley will be responsible to ensure that its DTQ is adjusted on November 1 of each contract Year during the term of this Agreement by requesting an amended DTQ upon not less than 2 Months prior to the end of the contract Year then in effect.

ARTICLE 6

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any transportation Rate Schedule under which Dunkley takes service, Dunkley will not be entitled to any monthly charge credits from Terasen Gas as a result of Force Majeure as defined in the Rate Schedule 25 General Firm Transportation Service Agreement.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Diane Roy, Director Regulatory Services

FORTISBC ENERGY INC. RATE SCHEDULE 25 SUPPLEMENT

ARTICLE 7

ARBITRATION

- 7.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the Commercial Arbitration Act of British Columbia or successor legislation, save as expressly provided herein.
- <u>7.02</u> Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 7.03 The parties will have 10 Days from receipt of the demand referred to in section 7.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors, permitted assigns or affiliates, any customer or supplier of Dunkley or Terasen Gas.
- 7.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed by the arbitrator, within 45 Days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 Days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 7.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

ARTICLE 8

FRANCHISE FEES

8.01 Dunkley acknowledges that Terasen Gas is obligated to pay Franchise Fees to certain municipalities it serves. In the event that Terasen Gas is required by statute, agreement, regulation, an arbitrator, a court of competent jurisdiction, or the British Columbia Utilities Commission to pay Franchise Fees or like fees on revenues received from Dunkley, or in relation to services provided, in addition to those calculated and paid by Terasen Gas at the commencement of this Agreement, Dunkley will pay an amount equal to such additional fees to Terasen Gas and the rates and charges payable by Dunkley as set out in Article 4 will be adjusted to reflect this change.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Diane Roy, Director Regulatory Services

ARTICLE 9

TERMINATION

9.01 In addition to any other rights either party may have to terminate this Agreement:

- (i) Notwithstanding Article 3.01, Terasen Gas may terminate this Agreement if at any time the British Columbia Utilities Commission disallows Terasen Gas the recovery from the other customers of Terasen Gas of any revenue shortfalls, resulting from the negotiated rates under this Agreement;
- (ii) Notwithstanding Article 3.01, Dunkley may terminate this Agreement by giving Terasen Gas at least 1 Year written notice of its intention to terminate and upon payment in full by Dunkley of the following amounts:

If termination is after November 1, 2003 and on or	
before November 1, 2004	\$102,960
If termination is after November 1, 2004 and on or	
before November 1, 2005	\$90.090
If termination is after November 1, 2005 and on or	
before November 1, 2006	\$77,220
If termination is after November 1, 2006 and on or	
before November 1, 2007	\$64,350
If termination is after November 1, 2007 and on or	
before November 1, 2008	\$51,480
If termination is after November 1, 2008 and on or	
before November 1, 2009	\$38,610
If termination is after November 1, 2009 and on or	
before November 1, 2010	\$25,740
If termination is after November 1, 2010 and on	
or before November 1, 2011	\$12,870

- (iii) Notwithstanding Article 9.01(ii), Dunkley may terminate this Agreement at any time after November 1, 2011, without any payment under this Article.
- 9.02 Upon not less than 12 Months prior notice, any termination pursuant to Article 9.01 will take effect on the next November 1 following the giving of notice in writing by the party terminating this Agreement.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Diane Roy, Director Regulatory Services

FORTISBC ENERGY INC. RATE SCHEDULE 25 SUPPLEMENT

ARTICLE 10

NOTICES

10.01 If in any Year an executed Transportation Agreement is not in place, then the notice provisions of the latest executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 11

MISCELLANEOUS

- 11.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 11.02 Notwithstanding Article 11.01, Terasen Gas may assign, without the consent of Dunkley, Terasen Gas' rights and obligations under this Agreement to a party which acquires all or substantially all of Terasen Gas' Gas utility operations.
- 11.03 Notwithstanding Article 11.01, Dunkley may assign, without the consent of Terasen Gas, Dunkley's rights and obligations under this Agreement to any party which acquires all or substantially all of Dunkley's sawmill operations served by Terasen Gas pursuant to this Agreement.
- 11.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns including, without limitation successors by merger, amalgamation or consolidation.
- 11.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 11.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of this Agreement by the British Columbia Utilities Commission, the 2002 Agreement will be cancelled and deemed to have been replaced and superseded in its entirety by this Agreement.
- 11.07 In this Agreement the words, phrases or expressions which are not defined herein or in the Terasen Gas General Terms and Conditions or in the Rate Schedule 25 General Firm Transportation Service Agreement and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Diane Roy, Director Regulatory Services

FORTISBC ENERGY INC. RATE SCHEDULE 25 SUPPLEMENT

IN WITNESS WHEREOF the parties have executed this Agreement.

AGREED TO AND ACCEPTED:

This 18 day of March , 2004. This_

This la day of Maech 2004.

TERASEN GAS INC.

By Signature)

DUNKLEY LUMBER LIMITED

AGREED TO AND ACCEPTED:

By: (Signature)

Rick Parnell (Name – Please print) Director, Large Commercial & Industrial Markets Blair Mayes (Name – Please print) General Manager

Order No.: G-35-04

Effective Date: November 1, 2004

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Diane Roy, Director Regulatory Services

Attachment 25.10

FORTISBC ENERGY INC. RATE SCHEDULE 25 SUPPLEMENT



PHONE (250) 998-4421 FAX (250) 998-4513

P.O. BOX 173 PRINCE GEORGE, B.C. CANADA V2L 4S1

October 7, 2013

Rajoo Jagtap FortisBC 16705, Fraser Highway, Surrey, BC V4N 0E8

Re: Bypass Agreement extension request - E2 Dunkley Lumber Ltd. (Prince George)

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further period of 2 years for expiry November 1st, 2016. The current agreement expires November 1st, 2014.

Thank you.

Yours truly, DUNKLEY LUMBER LTD.

*

Jony Magues

Tony Mogus General Manager

Letter Dated: October 7, 2013

Effective Date: November 1, 2014

BCUC Secretary: Original signed by Erica Hamilton

Issued By: Diane Roy, Director Regulatory Services

Attachment 25.10

FORTISBC ENERGY INC. RATE SCHEDULE 25 SUPPLEMENT



PHONE (250) 998-4421 FAX (250) 998-4513

P.O. BOX 173 PRINCE GEORGE, B.C. CANADA V2L 4S1

Original Page 11

September 4, 2015

Rajoo Jagtap FortisBC 16705, Fraser Highway, Surrey, BC V4N 0E8

Sent via Email: rajoo.jagtap@fortisbc.com

Re: Bypass Agreement extension request - E2 Dunkley Lumber Ltd. (Prince George)

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further period of 2 years for expiry November 1st, 2018. The current agreement expires November 1st, 2016.

Thank you.

Yours truly, DUNKLEY LUMBER LTD.

Tory Mogues

Tony Mogus General Manager

Order No.:	G-35-04	Issued By: Diane Roy, Vice President, Regulatory Affairs
Effective Date:	November 1, 2016	Accepted for Filing: <u>May 4, 2017</u> Tariff Supplement E-2

BCUC Secretary: Original signed by Patrick Wruck



TARIFF SUPPLEMENT NO. E-5

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT RATE SCHEDULE 25

BETWEEN

TOLKO INDUSTRIES LTD. SODA CREEK DIVISION (Formerly Riverside Forest Products Limited and Timberwest Forest Limited)

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: <u>Original signed by E.M. Hamilton</u>

THIS AGREEMENT made as of the 21st day of December 1994 with effect as and from November 1, 1993 is an amendment and restatement of an agreement made as of December, 1989 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

TIMBERWEST FOREST LIMITED, a company incorporated under the laws of British Columbia having its registered office at 9th Floor, 700 West Georgia Street, Vancouver, British Columbia

(hereinafter called "Timberwest")

OF THE SECOND PART

WHEREAS:

- A. Timberwest operates sawmill facilities in Williams Lake, British Columbia, and requires Gas for its Pinette & Thieran, Williams Lake Division sawmill operations located at North McKenzie Avenue;
- B. BC Gas owns and operates a Gas transmission pipeline which is connected to the sawmill operations of Timberwest in Williams Lake, British Columbia;
- C. Timberwest entered into an agreement with BC Gas' predecessor company, BC Gas Inc., dated December, 1989 effective January 1, 1990, which allows Timberwest to receive transportation service from BC Gas at rates and on terms and conditions that are based on the principles of the British Columbia Utilities Commission decision of October 22, 1987 respecting a bypass pipeline application by Northwood Pulp and Timber Limited.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- D. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service.
- E. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service.
- F. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amended and restated Agreement to clarify their respective rights and obligations.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by Timberwest in order to allow Timberwest to bypass the BC Gas transmission and distribution Gas System.

"DTQ" means Daily Transportation Quantity as defined in the Rate Schedule 25 Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which Timberwest may elect to receive service.

"Terms and Conditions of the Transportation Rate Schedule" means the terms of the applicable transportation Rate Schedule and Transportation Agreements thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation Rate Schedule accepted for filing by the British Columbia Utilities

Letter Dated: July 31, 1995

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith
Commission.

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Rate Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm transportation service to Timberwest for its sawmill operations in Williams Lake, British Columbia and Timberwest will accept such transportation service in accordance with the then prevailing provisions of BC Gas Rate Schedule 25 and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Rate Schedule conflicts or is inconsistent with the terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 Timberwest will be entitled to take service under a transportation or Sales Rate Schedule other than Rate Schedule 25 subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.

ARTICLE 3

TERM OF AGREEMENT

- 3.01 Subject to Article 9, the initial term of this Agreement shall be for a period of 6 years, effective the 1st Day of November 1993 up to the 1st Day of November 1999.
- 3.02 The term of this Agreement, shall be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon Timberwest providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date and BC Gas agreeing to such extension of the term of this Agreement. BC Gas shall not unreasonably withhold agreement to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension will be for a period of not less than 1 Contract Year.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for, Timberwest will each month pay to BC Gas for services provided hereunder from November 1, 1993 to the expiry of this Agreement the following rates:
 - A monthly charge of \$8,709.51.
 - (ii) A Delivery Charge of \$0.059/gigajoule.
 - (iii) Any charges pursuant to Articles 4.02, 4.03 and 4.04.
- 4.02 Upon changes in its DTQ pursuant to Article 5 or in the annual Gas quantity consumed under this Agreement in excess of 120 percent of those forecast in Schedule 1, the monthly charge will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by Timberwest as a result of the change in the DTQ or annual Gas quantity had Timberwest constructed and operated the Bypass Pipeline.
- 4.03 The monthly charge and the Delivery Charge as specified under Article 4.01 will be adjusted effective November 1 of each Contract Year on the following basis to compensate BC Gas for changes in its costs:
 - (i) the operating, maintenance and property tax expenses included in the monthly charge as at November 1, 1993 are \$1,427.51 per month and will be subject to the annual percentage change in the Consumer Price Index for the City of Vancouver for October, up to the Contract Year expiring November 1, 1994. Thereafter, the Consumer Price Index for the month of August will be used. The monthly charge will be adjusted by the calculated change in costs. The adjusted operating, maintenance and property tax expenses will form the basis for the following Contract Year adjustment.
 - (ii) any new or increased tax related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Timberwest will be calculated on a monthly basis (the annual cost divided by 12) and added to the monthly charge.

Letter Dated: July 31, 1995

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

Effective Date: November 1, 1993

4.04 Unauthorized Overrun Gas will be charged for and paid by Timberwest in accordance with the rate set out in Rate Schedule 25.

ARTICLE 5

ADJUSTMENTS TO DTO

5.01 The DTQ for the Contract Year commencing November 1, 1993 is agreed to be 667 gigajoules. Timberwest will be responsible to ensure that its DTQ is adjusted on November 1 of each Contract Year during the term of this Agreement by requesting an amended DTQ upon not less than 2 months prior to the end of the Contract Year then in effect.

ARTICLE 6

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any transportation Rate Schedule under which Timberwest elects to receive service, Timberwest will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure.

ARTICLE 7

ARBITRATION

- 7.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 7.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 7.03 The parties will have 10 days from receipt of the demand referred to in section 7.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of Timberwest or BC Gas.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

- 7.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 7.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

FRANCHISE FEES

8.01 Timberwest acknowledges that BC Gas is obligated to pay Franchise Fees to certain municipalities it serves. In the event that BC Gas is required by statute, agreement, regulation, an arbitrator, a court of competent jurisdiction, or the British Columbia Utilities Commission to pay Franchise Fees or like fees on revenues received from Timberwest, or in relation to services provided, in addition to those calculated and paid by BC Gas at the commencement of this Agreement, Timberwest will pay an amount equal to such additional fees to BC Gas and the rates and charges payable by Timberwest as set out in Article 4 will be adjusted to reflect this change.

ARTICLE 9

TERMINATION

- 9.01 In addition to any other rights either party may have to terminate this Agreement:
 - Notwithstanding Article 3.01, BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission disallows BC Gas the recovery from the other customers of BC Gas of any revenue shortfalls, resulting from the negotiated rates;
 - (ii) Notwithstanding Article 3.01, Timberwest may terminate this Agreement by giving BC Gas at least 1 year written notice of its intention to terminate and upon payment in full by Timberwest of the following amounts:

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

- (a) if termination is effective on November 1, 1995, the payment will be \$122,751;
- (b) if termination is effective on November 1, 1996, and each November 1 following, the payment specified in Article 9.01 (ii)(a) will be reduced by \$13,550 per year. No payment will be required under this Article for termination effective on November 1, 1999 or on any subsequent extension of this Agreement.
- (iii) Notwithstanding Article 9.01(ii), Timberwest may terminate this Agreement at any time after January 1, 1995, without further payment under this Article, if Timberwest, on a permanent basis, closes its operations at or near the site of its current sawmill facilities.
- 9.02 Any termination pursuant to Article 9.01 will take effect on the next November 1 following the giving of notice in writing by the party terminating this Agreement.

NOTICES

10.01 If in any year an executed Transportation Agreement is not in place, then the notice provisions of the latest executed Transportation Agreement and Rate Schedule will apply to this Agreement.

Effective Date: November 1, 1993

MISCELLANEOUS

- 11.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 11.02 Notwithstanding Article 11.01, BC Gas may assign, without the consent of Timberwest, BC Gas' rights and obligations under this Agreement to a party which acquires all or substantially all of BC Gas' Gas utility operations.
- 11.03 Notwithstanding Article 11.01, Timberwest may assign, without the consent of BC Gas, Timberwest's rights and obligations under this Agreement to any party which acquires all or substantially all of Timberwest's sawmill operations served by BC Gas pursuant to this Agreement.
- 11.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns including, without limitation successors by merger, amalgamation or consolidation.
- 11.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 11.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of this Agreement by the British Columbia Utilities Commission, the agreement made between the parties as of December, 1989 will be cancelled.
- 11.07 In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

Letter Dated: July 31, 1995

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

Effective Date: November 1, 1993

Į

IN	WITNESS	WHEREOF	the	parties	have	executed	this	Agreement.
----	---------	---------	-----	---------	------	----------	------	------------

BC GAS UTILITY LTD TIMBERWEST FOREST LIMITED By: ignatu

6, DINTER

(Name - please print)

Signath 165

(Name please print

95 Date: Date:

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

Sep-16-98 09:34A CPNI

604 682 6447

ACSIMILE MESS

GAGAN

Re: BC Gas Bypass Transportation Agreement Effective January 1, 1990 to November 1, 1999 Riverside Forest Products Ltd. Williams Lake Division & Sode Creek #1 Division

Please accept this facsimile message as notification to extend the terms and rates of the above noted Transportation Agreements as amended from time to time, in accordance with clause 3.02. We would request that the Transportation Agreements be extended for _____ years to November 1, _____P

We understand that you will be preparing the necessary documentation to affect this extension.

We thank you for your past services provided and look forward to a mutually satisfactory arrangement in future.

Peter Fac

Riverside Forest Products Ltd.

Coast Pacific Management Inc.

Letter Dated: July 31, 1995

Effective Date: November 1, 1999

Issued By: D.M. Masuhara, Vice President Legal, Regulatory & Logistics

BCUC Secretary: Original signed by R.J. Pellatt

FACSIMILE MESSAGE

Riverside - Sola Creek P

To: Ken Fuhr BC Gas Utility Ltd.

Date: October 20,1999

Fax No: 443-6770

- 1

Pages (including this one) 1

Re: BC Gas Bypass Transportation Agreement Effective November 1, 1998 to November 1, 2000

Please accept this facsimile message as notification to extend the terms and rates of the above noted Transportation Agreement as amended from time to time, in accordance with your Bypass Agreement. We would request that the Transportation Agreement be extended for $\underline{\partial}_{-}$ years to November 1, $\underline{\partial}_{-}$ 3.

We understand that you will be preparing the necessary documentation to affect this extension.

We thank you for your past services provided and look forward to a mutually satisfactory arrangement in future.

ca: Harry Chivers Avista Energy Canada, Ltd.

Letter Dated: July 15, 1995

Effective Date: November 1, 2000

Issued By: D.M. Masuhara, Vice President Legal, Regulatory & Logistics

BCUC Secretary: Original signed by R.J. Pellatt



Soda Creek Division RR#3, 5000 Soda Creek Road Williams Lake, B.C. Canada V23 1M3 Telephone (250) 398-3600 Fax (250) 398-3649

October 23, 2001

Ms. Melissa Philion BC Gas Utility Ltd. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No: (604) 592-7894

Dear Melissa:

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and BC Gas Utility Ltd. for a further two years for expiry November 1, 2004. The current agreement expires November 1, 2002.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours truly,

lan Fillinger Manager

cc; Spence Brigden Peter Fagan Dennis Szalkai Dan Johnson Richard Crowell Mary McCordic



Letter Dated: July 31, 1995

Effective Date: November 1, 2002

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Dietz Kellmann, Director Finance Development Services



Soda Creek Division (vitist, 6000 Soda Creek Road Williams Lave, B.C. Canada V20 1M3 Telephone (250) 398-3600 Fax (250) 398-3600

October 14, 2003

Ms. Melissa Phillon BC Gas Utility Ltd. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No: 604-592-7894

Dear Melissa:

RE: Bypass Agreement Extension Request

Ptease accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2005. The current agreement expires November 1, 2004.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours truly,

Richard Crowell Cariboo Region Operations Manager Riverside Forest Products Ltd.

cc: Mary McCordic, Avista Energy Canada, Ltd. (fax no: 604-682-6447)



Letter Dated: July 31, 1995

Effective Date: November 1, 2004

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

> Tariff Supplement E-5 Original Page 13

BCUC Secretary: Original signed by R.J. Pellatt

Terasen Gas Rate Schedule 25 Supplement

Cariboo Manufacturing 180 Hodgson Road Williams Lake, BC Canada V2G 3P6 T 250.392.3371 Fax 250,398,3909 www.riverside.bc.ca



October 27, 2004

Mr. Gordon Doyle Terasen Gas 16705 Fraser Highway Surrey, BC V3S 2X7 Fax No.: 604.592.7894

Dear Gordon:

Re: Bypass Agreement Extension Request Reference No. Tarif Supplement E3 – Williams Lake Reference No. Tarif Supplement E9 – Soda Creek

Please accept this letter as notice of our desire to extend the existing Bypass Agreements as noted above between our firm and Terasen Gas for a further year for expiry November 1, 2006. The current agreement expires November 1, 2005.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours truly,

RIVERSIDE FOREST PRODUCTS LIMITED

Rob Therrien Regional Administration Manager

RT/wa

Mary McCordic, Avista Energy Canada, Ltd. (fax no: 604.682.6447)

Letter Dated: July 31, 1995

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 2005

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>



180 Hodgson Road Williams Lake, BC Canada V2G 3P6 Maln: (250) 392-3371 Fax: (250) 398-3909

October 6, 2005

Mr. Gordon Doyle BC Gas Utility Ltd. 16705 Fraser Highway Surrey BC V3S 2X7 Fax: 604-592-7894

Dear Gord,

Re: Bypass Agreement Extension Request

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2007. The current agreement expires November 1, 2006.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada Ltd.

Thank you.

Yours truly,

RIVERSIDE FOREST PRODUCTS LTD.

Rob Therrien Regional Controller

cc: Mary McCordic, Avista Energy Canada Ltd. (Fax: 604-682-6447)

Letter Dated: July 31, 1995

Effective Date: November 1, 2006

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and Chief Financial Officer Tariff Supplement E-5 Original Page 15

BCUC Secretary: Original signed by R.J. Pellatt



VIA FAX (604.592.7894)

October 18, 2006

Kevin Hodgins BC Ges Utility Ltd. 16705 Fraser Highway Surrey BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2008. The current agreement expires November 1, 2007.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada Ltd.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Rob Therrien Regional Controller

RT/wd

cc: Mary McCordic, Aviste Energy Canada Ltd. (fax 604.682.6447)

150 Hodgion, Bend Williem Lake, BC Capital V2G 3P6

www.tolles.com

Marketing and manufacturing specially forest products

Letter Dated: July 31, 1995

Effective Date: November 1, 2007

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and Chief Financial Officer Tariff Supplement E-5 Original Page 16



VIA FAX (604.592.7894)

August 10, 2007

Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request – E5 Williams Lake Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2009. The current agreement expires November 1, 2008.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Coral Energy Canada Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey

Regional Controller

JH/wd

cc: Mary McCordic, Coral Energy Canada Inc. (fax 604.682.6447)

180 Hodgian Rosd Williams Lake, BC Canada: V2G 3P6

www.jalka.com

Letter Dated: July 31, 1995

Effective Date: November 1, 2008

BCUC Secretary: Original signed by E. M. Hamilton

Issued By: Scott Thomson, Vice President Regulatory Affairs and Chief Financial Officer Tariff Supplement E-5 Original Page 17



VIA FAX: 604-592-7894

August 26, 2008

Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request – E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2010. The current agreement expires November 1, 2009.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Shell Energy North America (Canada) Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

JH/jc

cc: Mary McCordic, Shell Energy North America (Canada) Inc. (fax: 604-682-6447)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6

www.tolko.com

Marketing and manufacturing specialcy forest products

Letter Dated: July 31, 1995

Issued By: Tom Loski, Chief Regulatory Officer

Effective Date: November 1, 2009

BCUC Secretary: Original signed by E.M. Hamilton

VIA FAX: 604-592-7894

October 13, 2009

Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request -- E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2011. The current agreement expires November 1, 2010.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Shell Energy North America (Canada) Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Mary McCordic, Shell Energy North America (Canada) Inc. (fax: 604-682-6447)

180 Hodgson Roac Williams Lake, BC Canada V2G 3Pt www.tolko.con

Marketing and manufacturing specialty forest products

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2010

BCUC Secretary: Original signed by E.M. Hamilton



VIA email to Kevin.Hodgins@terasengas.com

October 12, 2010

Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2012. The current agreement expires November 1, 2011.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Shell Energy North America (Canada) Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Mary McCordic, Shell Energy North America (Canada) Inc. (fax: 604-682-6447)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6

www.tolko.com

Marketing and manufacturing specialty forest products

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2011

BCUC Secretary: Original signed by Alanna Gillis



VIA email to Kevin.Hodgins@fortisbc.com

October 31, 2011

Kevin Hodgins FortisBC 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2013. The current agreement expires November 1, 2012.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absolute-energy.ca)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6 www.tolko.com

Marketing and manufacturing specialty forest products

Letter Dated:

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamiltion

October 31, 2011



VIA email to Kevin.Hodgins@fortisbc.com

October 31, 2012

Kevin Hodgins FortisBC 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2014. The current agreement expires November 1, 2013.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absolute-energy.ca)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6

www.tolko.com

Aarketing and manufacturing specialty forest products Letter Dated: October 31, 2011

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2013

BCUC Secretary: Original signed by E.M. Hamilton



Main: 250 392 3371 Fax: 250 398 3909 180 Hodgson Road Williams Lake, BC V2G 3P6

VIA email to Rajoo.Jagtap@fortisbc.com

October 31, 2013

Rajoo Jagtap FortisBC 16705 Fraser Highway Surrey, BC V4N 0E8

Dear Rajoo,

Re: Bypass Agreement Extension Request - E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2015. The current agreement expires November 1, 2014.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absoluteenergy.ca)

www.tolko.com



Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2014

BCUC Secretary: Original signed by Erica Hamilton



Main: 250 392 3371 Fax: 250 398 3909 180 Hodgson Road Williams Lake, BC V2G 3P6

VIA email to Rajoo.Jagtap@fortisbc.com

October 31, 2014

Rajoo Jagtap FortisBC 16705 Fraser Highway Surrey, BC V4N 0E8

Dear Rajoo,

Re: Bypass Agreement Extension Request - E5 Soda Creek Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2016. The current agreement expires November 1, 2015.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey

Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absoluteenergy.ca)

www.tolko.com



Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2015

BCUC Secretary: Original signed by Erica Hamilton



Main: (250) 545-4411 Fax: (250) 549-5353

September 8, 2015

Rajoo Jagtap Key Account Manager FortisBC 16705 Fraser Highway Surrey, BC

RE: Extension of Bypass Transportation Agreements

Rajoo,

This letter will serve as notice that Tolko Industries Ltd. wishes to extend the two Amended and Restated Bypass Transportation Agreements below for one year. This will extend the expiration date of these agreements to November 1, 2017.

E-5 Soda Creek Division E-6 Quesnel Division

If you have any questions regarding this notice, please do not hesitate to contact me.

Sincerely,

make on

Michael Towers Manager, Energy Supply and Systems Tolko Industries Ltd.

cc: Fred Dupas, Tolko

PO Box 39 3000 - 28th Street Vernon, BC Canada V1T 6M1

www.tolko.com

Marketing and manufacturing specialty forest products

Order No.: G-101-93

Issued By: Diane Roy, Vice President, Regulatory Affairs

Effective Date: November 1, 2016

Accepted for Filing: May 4, 2017

BCUC Secretary: Original signed by Patrick Wruck



TARIFF SUPPLEMENT NO. E-6

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT RATE SCHEDULE 25

BETWEEN

TOLKO INDUSTRIES LIMITED

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the 21st day of December 1994 with effect as and from November 1, 1993 is an amendment and restatement of an agreement made as of December, 1989 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

TOLKO INDUSTRIES LIMITED, a company incorporated under the laws of British Columbia having its registered office at 17th Floor, 1075 West Georgia Street, Vancouver, British Columbia

(hereinafter called "Tolko")

OF THE SECOND PART

WHEREAS:

- A. Tolko operates dry kiln facilities in Quesnel, British Columbia, and requires Gas for its Quest Wood Products Division dry kiln operations located at 2 Mile Flat;
- B. BC Gas owns and operates a Gas transmission pipeline which is connected to the dry kiln operations of Tolko in Quesnel, British Columbia;
- C. Tolko entered into an agreement with BC Gas' predecessor company, BC Gas Inc., dated December, 1989 effective January 1, 1990, which allows Tolko to receive transportation service from BC Gas at rates and on terms and conditions that are based on the principles of the British Columbia Utilities Commission decision of October 22, 1987 respecting a bypass pipeline application by Northwood Pulp and Timber Limited.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- D. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service.
- E. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service.
- F. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amended and restated Agreement to clarify their respective rights and obligations.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by Tolko in order to allow Tolko to bypass the BC Gas transmission and distribution Gas System.

"DTQ" means Daily Transportation Quantity as defined in the Rate Schedule 25 Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which Tolko may elect to receive service.

"Terms and Conditions of the Transportation Rate Schedule" means the terms of the applicable transportation Rate Schedule and Transportation Agreement thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation Rate Schedule accepted for filing by the British Columbia Utilities Commission.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Rate Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm transportation service to Tolko for its dry kiln operations in Quesnel, British Columbia and Tolko will accept such transportation service in accordance with the then prevailing provisions of BC Gas Rate Schedule 25 and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Rate Schedule conflicts or is inconsistent with the terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 Tolko will be entitled to take service under a transportation or Sales Rate schedule other than Rate Schedule 25 subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.

ARTICLE 3

TERM OF AGREEMENT

- 3.01 Subject to Article 9 the initial term of this Agreement will be for a period of 6 years, effective the 1st Day of November 1993 up to the 1st Day of November 1999.
- 3.02 The term of this Agreement shall be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon Tolko providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date and BC Gas agreeing to such extension of the term of this Agreement. BC Gas shall not unreasonably withhold agreement to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension under this Agreement will be for a period of not less than 1 Contract Year.

ARTICLE 4

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for, Tolko will each month pay to BC Gas for services provided hereunder from November 1, 1993 to the expiry of this Agreement the following rates:
 - (i) A monthly charge of \$3,998.04.
 - (ii) A Delivery Charge of \$0.064/gigajoule.
 - (iii) Any charges pursuant to Articles 4.02, 4.03 and 4.04.
- 4.02 Upon changes in its DTQ pursuant to Article 5 or in the annual Gas quantity consumed under this Agreement in excess of 120 percent of those forecast in Schedule 1, the monthly charge will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by Tolko as a result of the change in the DTQ or annual Gas quantity had Tolko constructed and operated the Bypass Pipeline.
- 4.03 The monthly charge and the Delivery Charge as specified under Article 4.01 will be adjusted effective November 1 of each Contract Year on the following basis to compensate BC Gas for changes in its costs:
 - (i) the operating, maintenance and property tax expenses included in the monthly charge as at November 1, 1993 are \$668.04 per month and will be subject to the annual percentage change in the Consumer Price Index for the City of Vancouver for October, up to the Contract Year expiring November 1, 1994. Thereafter, the Consumer Price Index for the month of August will be used. The monthly charge will be adjusted by the calculated change in costs. The adjusted operating, maintenance and property tax expenses will form the basis for the following Contract Year adjustment.
 - (ii) any new or increased tax related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Tolko will be calculated on a monthly basis (the annual cost divided by 12) and added to the monthly charge.
- 4.04 Unauthorized Overrun Gas will be charged for and paid by Tolko in accordance with the rate set out in Rate Schedule 25.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

ADJUSTMENTS TO DTO

5.01 The DTQ for the Contract Year commencing November 1, 1993 is agreed to be 630 gigajoules. Tolko will be responsible to ensure that its DTQ is adjusted on November 1 of each Contract Year during the term of this Agreement by requesting an amended DTQ upon not less than 2 months prior to the end of the Contract Year then in effect.

ARTICLE 6

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any transportation Rate Schedule under which Tolko takes service, Tolko will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure.

ARTICLE 7

ARBITRATION

- 7.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 7.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 7.03 The parties will have 10 days from receipt of the demand referred to in section 7.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of Tolko or BC Gas.
- 7.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed the arbitrator, within 45 days after the

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.

7.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

ARTICLE 8

FRANCHISE FEES

8.01 Tolko acknowledges that BC Gas is obligated to pay Franchise Fees to certain municipalities it serves. In the event that BC Gas is required by statute, agreement regulation, an arbitrator, a court of competent jurisdiction, or the British Columbia Utilities Commission to pay Franchise Fees or like fees on revenues received from Tolko, or in relation to services provided, in addition to those calculated and paid by BC Gas at the commencement of this Agreement, Tolko will pay an amount equal to such additional fees to BC Gas and the rates and charges payable by Tolko as set out in Article 4 will be adjusted to reflect this change.

ARTICLE 9

TERMINATION

- 9.01 In addition to any other rights either party may have to terminate this Agreement:
 - (i) Notwithstanding Article 3.01, BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission disallows BC Gas the recovery from the other customers of BC Gas of any revenue shortfalls, resulting from the negotiated rates;
 - (ii) Notwithstanding Article 3.01, Tolko may terminate this Agreement by giving BC Gas at least 1 year written notice of its intention to terminate and upon payment in full by Tolko of the following amounts:
 - (a) if termination is effective on November 1, 1995, the payment will be \$55,816;

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- (b) if termination is effective on November 1, 1996, and each November 1 following, the payment specified in 9.01 (ii)(a) will be reduced by \$6,202 per year. No payment shall be required under this Article for termination effective on November 1, 1999 or on any subsequent extension of this Agreement;
- (iii) Notwithstanding Article 9.01(ii), Tolko may terminate this Agreement at any time after January 1, 1995, without any payment under this Article, if Tolko, on a permanent basis, closes its operation at or near the site of its current dry kiln facilities.
- 9.02 Any termination pursuant to Article 19.01 will take effect on the next November 1 following the giving of notice in writing by the party terminating this Agreement.

NOTICES

10.01 If in any year an executed Transportation Agreement is not in place, then the notice provisions of the latest executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 11

MISCELLANEOUS

- 11.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 11.02 Notwithstanding Article 11.01, BC Gas may assign, without the consent of Tolko, BC Gas' rights and obligations under this Agreement to a party which acquires all or substantially all of BC Gas' Gas utility operations.
- 11.03 Notwithstanding Article 11.01, Tolko may assign, without the consent of BC Gas, Tolko's rights and obligations under this Agreement to any party which acquires all or substantially all of Tolko's dry kiln operations served by BC Gas pursuant to this Agreement.
- 11.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns including, without limitation successors by

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

merger, amalgamation or consolidation.

- 11.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 11.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of this Agreement by the British Columbia Utilities Commission, the agreement made between the parties as of December, 1989 will be cancelled.
- 11.07 In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

IN WITNESS WHEREOF the parties have executed this Agreement.

BC GAS UTILITY LTD.	TOLKO
Br Jule	By:
(signature)	
(Name - please print)	identitation temp
Date: JAN 30, 1995	Date:

FOLKO INDUSTRIES LIMITED

ame - please prin

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs



Phone: (250) 545-4411 Confidential Fax: (250) 545-4783

FACSIMILE MESSAGE

DATE: September 16, 1998

TO: Ken Fuhr BC Gas Utility Ltd.

FAX NO.: 443-6770

PAGES (including this one) 1

Re: BC Gas Bypass Transportation Agreemant Effective January 1, 1990 to November 1, 1999 Tolko Industries Ltd. -- Quest Wood Division

Please accept this fansimile message as notification to extend the terms and rates of the above noted Transportation Agreement as amended from time to time, in accordance with clause 3.02. We would request that the Transportation Agreement to be extended for one (1) year to November 1, 2000.

We understand that you will be preparing the necessary documentation to affect this extension.

We thank you for your past services provided and look forward to a mutually satisfactory errangement in future.

and the

Fred Stahle Tolko Industries Ltd.

cc: Hany Chivers – Coast Pacific Management Inc. Ame Hanson – Tolko Industries Ltd. - Quest Wood Division

> P.O. Box 37 3803 - 30th Avenue Vennon, B.C. Cavado VIT 6543

Mailoting and sumafacturing talli wood products

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

		Main O	Elan (250) 545-4		
		Confidencial Four (250) \$4			
DATE:	October 29, 1999				
TO:	Ken Fuhr - BC Gas Utility Ltd.				
FAX NO.	443-6770				
PAGES:	(including this one) 1				
		she take	307		
RE: BC	Gas Bypass Transportation Agreement Chie November 1, 1996 to November	nt 1,2000	039240		
Please ac rates of th time, in ar Transport	cept this facsimile message as notifical above noted Transportation Agreement cordance with your Bypass Agreement stion Agreement be extended for <u>2</u> y	Son to extend the torms and ont as amended from time to L. We would request that the rears to November 1, <u>2007</u>			
We under this exten	stand that you will be preparing the nec- tion.	sessary documentation to affect			
We thank satisfactor	you for your past services provided and y amongement in future.	d look forward to a mutually			
Yours truit	beits of existences in some of				
TOLKO IN	DUSTRIES LTD.				
Free	stahl.	Second or foregoing state			
Mr. Fred H Secretary	. Stohie Treasurer				
"HS/eh					
FHS/eh c.c.: Har Avis	y Chivera ta Energy Canada, Ltd.				
FHS/eh c.c.: Har Avis	y Chivers ta Energy Canada, Ltd.				
FHS/eth c.c.: Har Avie	y Chivers ta Energy Canada, Ltd.				

1203 - 30th Assess Vernon, B.C. Canada VIT 654:

Merketing and surrayfurnaring solid wood products

Letter Dated:

Effective Date:

July 31, 1995

BCUC Secretary: Original signed by C. Smith

November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs



Phanes (250) 545-5411 Candidential Face (250) 545-6783

August 15, 2000

B.C. Gas Utility Ltd. Industrial Sales Representative 9th Floor – 1111 West Georgia Vancouver, B.C. V6E 4M3

Attention: Mr. Greg Simmons

Re: Extension of Bypass Agreement for Tolko – Quest Division

Dear Mr. Simmons,

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and B.C. Gas Utility Ltd. for a further 2 (two) years for expiry November 1, 2003. The current agreement expires November 1, 2001.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, as well as by a copy sent to our agent, Avista Energy Canada, Ltd. We look forward to your early response.

Yours very truly,

TOLKO INDUSTRIES LTD. Per:

(M Stal Tree

Fred H. Stehle Secretary / Treasurer FHS/tv

cc: Avists Energy Canada Ltd. 1006 – 1168 Alberni Street Vancouver, B.C. V6E 3Z3

> P.O. Rox 39 3203 - 30th Avenue Vernor, R.C. Casada VIT 6443

Markesing and manufacturing specialty forest (voluci)

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs



Phone: (250) 545-4411 Confidential Fax: (250) 545-4783

September 26, 2002

Via Courier

B.C. Gas Utility Ltd. Industrial Sales Representative 9th Floor - 1111 West Georgia Vancouver, B.C. V6E 4M3

Attention: Mr. Greg Simmons

Dear Mr. Simmons:

Re: Extension of Bypass Agreement for Tolko - Quest Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and B.C. Gas Utility Ltd. for a further 2 (two) years for expiry November 1, 2005. The current agreement expires November 1, 2003.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, as well as by a copy sent to our agent, Avista Energy Canada Ltd. We look forward to your response in due course.

Yours very truly,

TOLKO INDUSTRIES LTD. Per:

14 Stahl

Fred H. Stehle Secretary / Treasurer FHS/bm

cc: Avista Energy Canada Ltd. 1006 - 1166 Alberni Street Vancouver, B.C. V6E 3Z3

> PO Box 39 3203 - 30th Avenue

Vemon, BC Canada VIT 6M1

Marketing and manufacturing specialty forest products

Letter Dated: July 31, 1995

Effective Date: November 1, 2003

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs


Telephone: (250) 992-1700 Fax: (250) 992-1701

October 5, 2004

Mr. Gordon Doyle BC Gas Utility Ltd. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No: 604-592-7894

Dear Gordon:

RE: Bypass Agreement Extension Request

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2006. The current agreement expires November 1, 2005.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours truly,

WA.

Marina Browne, C.G.A. Divisional Controller

cc: Mary McCordic, Avista Energy Canada, Ltd. (fax no: 604-682-6447)

1879 Brownmiller Road Quesnel, BC Canada V2] 6R9

Letter Dated: July 31, 1995

Effective Date: November 1, 2005

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs



QUEST WOOD DIVISION

Main: (250) 992-1700 Fax: (250) 992-1701

September 16, 2005

Mr. Gordon Doyle Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No. 604-592-7894

Dear Gord:

Re: Bypass Agreement Extension Request

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2007. The current agreement expires November 1, 2006.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you,

Yours truly,

Marina Browne, C.G.A. **Divisional** Controller

Cc Mary McCordic, Avista Energy Canada, Ltd. (Fax # 604-682-6447)

> 1879 Brownmiller Road Quesnel, BC Canada V2J 6R9

> > olko.com

Montestine und al dammer & QQQ 'ON much

ARASE MOON ARASHAL

0.1101

Letter Dated: July 31, 1995

Effective Date: November 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and **Chief Financial Officer** Tariff Supplement E-6 **Original Page 14**

0007



Main: (250) 992-1700 Fax: (250) 992-1701

October 18, 2006

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No: 604-592-7894

Dear Kevin:

RE: Bypass Agreement Extension Request

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terssen Gas for a further year for expiry November 1, 2008. The current agreement expires November 1, 2007.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours truly,

cc: Mary McCordic, Avista Energy Canada, Ltd. (fax no: 604-682-6447)

1819 Brownneller Road Quernel. BC Canada V2J 689 www.tolko.com

Marketing and manufacturing speaking forest products

Letter Dated: July 31, 1995

Effective Date: November 1, 2008

BCUC Secretary: Original signed by E. M. Hamilton

Issued By: Scott Thomson, Vice President Regulatory Affairs and Chief Financial Officer Tariff Supplement E-6 Original Page 15



QUEST WOOD DIVISION

October 9, 2007

Terasen Gas 16705 Fraser Highway Surrey, BC

Attention Kevin Hodgins

RE: Bypass Transportation Agreement Renewal

Further to our discussion this morning, please accept this confirmation of Tolko Industries Ltd., Quest Wood Division's extension of our Bypass Agreement with Terasen Gas.

Sincerply ER NY

Rob Beard, COA Divisional Controller Quest Wood Division Tolko Industries Ltd.

1879 Brownmiller Road Quesnel, BC Canada V2J 689

www.tolko.com

Marketing and manufacturing spacealty forest products

Letter Dated: July 31, 1995

Effective Date: November 1, 2008

BCUC Secretary: Original signed by E. M. Hamilton

Issued By: Scott Thomson, Vice President Regulatory Affairs and Chief Financial Officer Tariff Supplement E-6 Original Page 16



Phone: (250) 992-1700 Fax: (250) 992-1701

October 15, 2008

Terasen Gas 16705 Fraser Highway Surrey, BC V3S 2X7

Attention Kevin Hodgins

RE: Bypass Transportation Agreement Renewal

Further to our discussion this morning, please accept this confirmation of Tolko Industries Ltd., Quest Wood Division's extension of our Bypass Agreement with Terasen Gas.

Sincerely,

Hank Randrup

Hank Randrup Plant Manager Quest Wood Division Tolko Industries Ltd.

cc: File. Terasen Gas

1879 Brownmiller Road Quesnel, BC Canada V2J 6R9 www.tolko.com

Marketing and manufacturing specialty forest products

Letter Dated: July 31, 1995

Effective Date: November 1, 2009

BCUC Secretary: Original signed by E.M. Hamilton

Issued By: Tom Loski, Chief Regulatory Officer



Telephone: (250) 992-1700 Fax: (250) 992-1701

October 31, 2009

Terasen Gas Inc. 16705 Fraser Highway Surrey, BC V3S 2X7

Attn: Kevin Hodgins

Dear Sirs:

Re: <u>Amended and Restated Bypass Transportation Agreement between Tolko</u> Industries Limited and Terasen Gas Inc for Rate Schedule 25 (E-6 Quesnel)

Please accept this letter as notice of our desire to extend the bypass agreement between our firm and Terasen Gas for a further year for expiry November 1, 2011. The current agreement expires November 1, 2010.

Yours Truly,

and fandings

Hank Randrup Plant Manager, Tolko Industries Ltd. Quest Wood Division

1879 Brownmiller Road Quesnel, BC Canada V2] 6R9

Marketing and manufacturing specialty forest products

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2010

BCUC Secretary: Original signed by E.M. Hamilton



Phone: (250) 392-3371 Fax: (250) 398-3909

VIA email to Kevin.Hodgins@terasengas.com

October 12, 2010

Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E6 Quesnel Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2012. The current agreement expires November 1, 2011.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Shell Energy North America (Canada) Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey

Regional Controller

cc: Mary McCordic, Shell Energy North America (Canada) Inc. (fax: 604-682-6447)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6

www.tolko.com

Marketing and manufacturing specialty forest products

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2011

BCUC Secretary: Original signed by Alanna Gillis

TOLKO

Phone: (250) 392-3371 Fax: (250) 398-3909

VIA email to Kevin.Hodgins@fortisbc.com

October 31, 2011

Kevin Hodgins FortisBC 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E6 Quesnel Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2013. The current agreement expires November 1, 2012.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absolute-energy.ca)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6

www.tolko.com

Marketing and manufacturing specialty forest products

Letter Dated:

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamiltion

July 31, 1995



Phone: (250) 392-3371 Fax: (250) 398-3909

VIA email to Kevin.Hodgins@fortisbc.com

October 31, 2012

Kevin Hodgins FortisBC 16705 Fraser Highway Surrey, BC V3S 2X7

Dear Kevin,

Re: Bypass Agreement Extension Request - E6 Quesnel Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2014. The current agreement expires November 1, 2013.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absolute-energy.ca)

180 Hodgson Road Williams Lake, BC Canada V2G 3P6

www.tolko.com

Marketing and manufacturing specialty forest products Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2013

BCUC Secretary: Original signed by E.M. Hamilton



Main: 250 392 3371 Fax: 250 398 3909 180 Hodgson Road Williams Lake, BC V2G 3P6

VIA email to Rajoo.Jagtap@fortisbc.com

October 31, 2013

Rajoo Jagtap FortisBC 16705 Fraser Highway Surrey, BC V4N 0E8

Dear Rajoo,

Re: Bypass Agreement Extension Request - E6 Quesnel Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2015. The current agreement expires November 1, 2014.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absoluteenergy.ca)

www.tolko.com



Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2014

BCUC Secretary: Original signed by Erica Hamilton



Main: 250 392 3371 Fax: 250 398 3909 180 Hodgson Road Williams Lake, BC V2G 3P6

VIA email to Rajoo.Jagtap@fortisbc.com

October 31, 2014

Rajoo Jagtap FortisBC 16705 Fraser Highway Surrey, BC V4N 0E8

Dear Rajoo,

Re: Bypass Agreement Extension Request - E6 Quesnel Division

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and FortisBC for a further year for expiry November 1, 2016. The current agreement expires November 1, 2015.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Absolute Energy Inc.

Thank you.

Yours truly,

TOLKO INDUSTRIES LTD.

Jay Harvey

Regional Controller

cc: Peter Kresnyak, Absolute Energy Inc. (Fax: 604-982-0467, e-mail Peter@absoluteenergy.ca)

www.tolko.com



Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2015

BCUC Secretary: Original signed by Erica Hamilton



Main: (250) 545-4411 Fax: (250) 549-5353

September 8, 2015

Rajoo Jagtap Key Account Manager FortisBC 16705 Fraser Highway Surrey, BC

RE: Extension of Bypass Transportation Agreements

Rajoo,

This letter will serve as notice that Tolko Industries Ltd. wishes to extend the two Amended and Restated Bypass Transportation Agreements below for one year. This will extend the expiration date of these agreements to November 1, 2017.

E-5 Soda Creek Division E-6 Quesnel Division

If you have any questions regarding this notice, please do not hesitate to contact me.

Sincerely,

na 0

Michael Towers Manager, Energy Supply and Systems Tolko Industries Ltd.

cc: Fred Dupas, Tolko

PO Box 39 3000 - 28th Street Vernon, BC Canada V1T 6M1

www.tolko.com

Marketing and manufacturing specialty forest products

Order No.: G-101-93

Nevershard 0040

Issued By: Diane Roy, Vice President, Regulatory Affairs

Effective Date: November 1, 2016

Accepted for Filing: May 4, 2017

BCUC Secretary: Original signed by Patrick Wruck



TARIFF SUPPLEMENT NO. E-8

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT RATE SCHEDULE 25

BETWEEN

WEST FRASER MILLS LIMITED

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the 21st day of December 1994 with effect as and from November 1, 1993 is an amendment and restatement of an agreement made as of December, 1989 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

WEST FRASER MILLS LIMITED, a company incorporated under the laws of British Columbia having its registered office at P.O. Box 4360, Williams Lake, British Columbia

(hereinafter called "West Fraser")

OF THE SECOND PART

WHEREAS:

- A. West Fraser operates sawmill facilities in Williams Lake, British Columbia, and requires Gas for its sawmill operations located at McKenzie Avenue and Rottacker Road;
- B. BC Gas owns and operates a Gas transmission pipeline which is connected to the sawmill operations of West Fraser in Williams Lake, British Columbia;
- C. West Fraser entered into an agreement with BC Gas' predecessor company, BC Gas Inc., dated December, 1989 effective January 1, 1990, which allows West Fraser to receive transportation service from BC Gas at rates and on terms and conditions that are based on the principles of the British Columbia Utilities Commission decision of October 22, 1987 respecting a bypass pipeline application by Northwood Pulp and Timber Limited.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- D. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service.
- E. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service.
- F. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amended and restated Agreement to clarify their respective rights and obligations.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by West Fraser in order to allow West Fraser to bypass the BC Gas transmission and distribution Gas System.

"DTQ" means Daily Transportation Quantity as defined in the Rate Schedule 25 Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which West Fraser may elect to receive service.

"Terms and Conditions of the Transportation Rate Schedule" means the terms of the applicable transportation Rate Schedule and Transportation Agreement thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation Rate Schedule accepted for filing by the British Columbia Utilities Commission.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Rate Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm transportation service to West Fraser for its sawmill operations in Williams Lake, British Columbia and West Fraser will accept such transportation service in accordance with the then prevailing provisions of BC Gas Rate Schedule 25 and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Rate Schedule conflicts or is inconsistent with the terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 West Fraser will be entitled to take service under a Transportation or Sales Rate Schedule other than Rate Schedule 25 subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.

ARTICLE 3

TERM OF AGREEMENT

- 3.01 Subject to Article 9 the initial term of this Agreement will be for a period of 6 years effective the 1st Day of November 1993 up to the 1st Day of November 1999.
- 3.02 The term of this Agreement, shall be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions upon West Fraser providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date. BC Gas shall not unreasonably withhold agreement to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension will be for a period of not less than 1 Contract Year.

Letter Dated: July 31, 1995

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

ARTICLE 4

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for, West Fraser will each month pay to BC Gas for services provided hereunder from November 1, 1993 to the expiry of this Agreement the following rates:
 - (i) A monthly charge of \$5,843.85.
 - (ii) A Delivery Charge of \$0.069/gigajoule.
 - (iii) Any charges pursuant to Articles 4.02, 4.03 and 4.04.
- 4.02 Upon changes in its DTQ pursuant to Article 5 or in the annual Gas quantity consumed under this Agreement in excess of 120 percent of those forecast in Schedule 1, the monthly charge will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by West Fraser as a result of the change in the DTQ or annual Gas quantity had West Fraser constructed and operated the Bypass Pipeline.
- 4.03 The monthly charge and the Delivery Charge as specified under Article 4.01 will be adjusted effective November 1 of each Contract Year on the following basis to compensate BC Gas for changes in its costs:
 - (i) the operating, maintenance and property tax expenses included in the monthly charge as at November 1, 1993 are \$843.85 per month and will be subject to the annual percentage change in the Consumer Price Index for the City of Vancouver for October, up to the Contract Year expiring November 1, 1994. Thereafter, the Consumer Price Index for the month of August will be used. The monthly charge will be adjusted by the calculated change in costs. The adjusted base operating, maintenance and property tax expenses will form the basis for the following Contract Year adjustment.
 - (ii) any new or increased tax related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by West Fraser will be calculated on a monthly basis (the annual cost divided by 12) and added to the monthly charge.

Letter Dated: July 31, 1995

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

4.04 Unauthorized Overrun Gas will be charged for and paid by West Fraser in accordance with the rate set out in Rate Schedule 25.

ARTICLE 5

ADJUSTMENTS TO DTQ

5.01 The DTQ for the Contract Year commencing November 1, 1993 is agreed to be 481 gigajoules. West Fraser will be responsible to ensure that its DTQ is adjusted on November 1 of each Contract Year during the term of this Agreement by requesting an amended DTQ upon not less than 2 months prior to the end of the Contract Year then in effect.

ARTICLE 6

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any Ttransportation Rate Schedule under which West Fraser takes service, West Fraser will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure.

ARTICLE 7

ARBITRATION

- 7.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 7.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 7.03 The parties will have 10 days from receipt of the demand referred to in section 7.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of West Fraser or BC Gas.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- 7.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 7.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

ARTICLE 8

FRANCHISE FEES

8.01 West Fraser acknowledges that BC Gas is obligated to pay Franchise Fees to certain municipalities it serves. In the event that BC Gas is required by statute, agreement, regulation, an arbitrator, a court of competent jurisdiction, or the British Columbia Utilities Commission to pay Franchise Fees or like fees on revenues received from West Fraser, or in relation to services provided, in addition to those calculated and paid by BC Gas at the commencement of this Agreement, West Fraser will pay an amount equal to such additional fees to BC Gas and the rates and charges payable by West Fraser as set out in Article 4 will be adjusted to reflect this change.

ARTICLE 9

TERMINATION

- 9.01 In addition to any other rights either party may have to terminate this Agreement:
 - Notwithstanding Article 3.01, BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission disallows BC Gas the recovery from the other customers of BC Gas of any revenue shortfalls, resulting from the negotiated rates;
 - (ii) Notwithstanding Article 3.01, West Fraser may terminate this Agreement by giving BC Gas at least 1 year written notice of its intention to terminate and upon payment in full by West Fraser of the following amounts:

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- (a) if termination is effective on November 1, 1995, the payment will be \$83,716;
 - (b) if termination is effective on November 1, 1996, and each November 1 following, the payment specified in 9.01 (ii)(a) will be reduced by \$9,302 per year. No payment shall be required under this Article for termination effective on November 1, 1999 or on any subsequent extension of this Agreement;
- (iii) Notwithstanding Article 9.01(ii), West Fraser may terminate this Agreement at any time after January 1, 1995, without any payment under this Article, if West Fraser, on a permanent basis, closes its operations at or near the site of its current sawmill facilities.
- 9.02 Any termination pursuant to Article 9.01 will take effect on the next November 1 following the giving of notice in writing by the party terminating this Agreement.

ARTICLE 10

NOTICES

10.01 If in any year an executed Transportation Agreement is not in place, then the notice provisions of the latest executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 11

MISCELLANEOUS

- 11.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 11.02 Notwithstanding Article 11.01, BC Gas may assign, without the consent of West Fraser, BC Gas' rights and obligations under this Agreement to a party which acquires all or substantially all of BC Gas' Gas utility operations.
- 11.03 Notwithstanding Article 11.01, West Fraser may assign, without the consent of BC Gas, West Fraser's rights and obligations under this Agreement to any party which acquires all or substantially all of West Fraser's sawmill operations served by BC Gas pursuant to this Agreement.

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by C. Smith

- 11.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns including, without limitation successors by merger, amalgamation or consolidation.
- 11.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 11.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of this Agreement by the British Columbia Utilities Commission, the agreement made between the parties as of December, 1989 will be cancelled.
- 11.07 In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

IN WITNESS WHEREOF the parties have executed this Agreement.

BC GAS UTILITY LTD. WEST FRASER MILLS.LIMITED By: Signature Ignated (Name - please print please print Date: JANUARY 31 Date:

Letter Dated: July 31, 1995

Effective Date: November 1, 1993

BCUC Secretary: Original signed by C. Smith

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

West Fraser Mills Ltd

WILLIAMS LAKE DIVISION

P.O. Box 4360 Williams Lato, R.C. Canada V2G 2V4 Phone: (250) 392-7384 FAX: (250) 392-7016

October 7, 1998

Mr. Ken Fuhr BC Gas Utility Ltd. 1111 West Georgia Street Vancouver, BC V6E 4M4

Dear Mr. Fuhr:

RE: BC GAS BYPASS TRANSPORTATION AGREEMENT EFFECTIVE JANUARY 1, 1990 TO NOVEMBER 1, 1999 - REVISED

Please accept this facsimile message as notification to extend the terms and rates of the above noted Transportation Agreement as amended from time to time, in accordance with clause 3.02. we would request that the Transportation Agreement be extended for 2 years to November 1, 2001.

We understand that you will be preparing the necessary documentation to affect this extension.

We thank you for your past service provided and look forward to a mutually satisfactory arrangement in future.

Sincerely,

Brad Hunt West Fraser Mills Ltd. Williams Lake Division

cc: Harry Chivers Coast Pacific Management

Letter Dated: July 31, 1995

Effective Date: November 1, 1999

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

West fraser Mills Ltd

1250 Brownmiller Road Quesnel, BC Canada V2J 6P5 Telephone: (250) 992-9244 Fax: (250) 992-6233

NGC 1 (Createrial)

August 25, 2000

Mr. Greg Simmons BC Gas Utility Ltd. 1111 West Georgia St. Vancouver, BC V6E 4M4

Cc: Bill Legrow, West Fraser, Vancouver Brad Hunt, West Fraser, Williams Lake

Re: BC Gas By-Pass Transportation Agreement; Williams Lake Division

Dear Sir,

The attached By-Pass Agreement, effective January 1, 1990 to November 1, 1999, was revised and extended to November 1, 2001.

In accordance with clause 3.02 we request that this agreement be extended for a period of two years, to November 1, 2003.

Sincerely,

David Humber Manager, Technical Development West Fraser Mills Ltd.

Letter Dated: July 31, 1995

Effective Date: November 1, 2001

Issued By: D.M. Masuhara, Vice President Legal and Regulatory Affairs

> Tariff Supplement E-8 Original Page 10

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

1000-1100 MoMile Street Vancouver, B.C. Canada V6E 4A5 Talephone: (504) 895-2700 Fax: (504) 691-6061

West Fraser Timber Co. Ltd

September 25, 2002

BC Gas Utility Ltd. 16705 Fraser Highway Surrey, BC V3S 2X7

Attention: Melissa Philion Account Manager Fax: 604-592-7894

Dear Melissa:

Please extend the existing by pass agreement between BC Gas Utility Ltd. and West Fraser Mills, Williams Lake Division for a period of 2 years.

Yours truly,

David Humber Manager, Technical Development

Letter Dated: July 31, 1995

Effective Date: November 1, 2003

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs



West Fraser Mills Ltd

1250 Brownmiller Road Quesad, B.C. Canada V2J 625

Phone: (250) 992-9244 Faz: (250) 992-9233

August 13, 2004

Terasen Gas Corporate Office 16705 Fraser Highway Surrey, BC V3S 2X7

Attention: Gord Doyle Account Manager Fax: (604) 592-7894

Re: Terasen Gas By-Pass Transportation Agreement – Williams Lake Division Effective January 1, 1990 to November 1, 1999 - Revised

Dear Sir,

In accordance with clause 3.02, we request that the above noted agreement be extended for a period of two years to November 1, 2007.

Sincerely,

Kreshka Young Energy Manager West Fraser Mills Ltd.

Letter Dated: July 31, 1995

Effective Date: November 1, 2005

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt



October 30, 2006

Terasen Gas Corporate Office 16705 Fraser Highway, Surmy V3S 2X7

Attention: Kevin Hodgins Key Account Manager

RE: Bypass Transportation Agreement - Williams Lake division.

Dear Kevin:

As per clause 3.02 of the above stated agreement, West Fraser Mills Ltd would like to request that the Bypass agreement be extended for a further period of 2 years to November 1st 2009.

Sincerely,

//*

Andrew K. Kim

Project Engineer West Fraser Mills LTD.

1250 Brownmiller Road + Questel + B.C. + Canada + V2J 6P5 + (250) 992-0809 + (250) 988-0807 + Mp//www.wepthawnfimber.ca

Letter Dated: July 31, 1995

Effective Date: November 1, 2007

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and Chief Financial Officer Tariff Supplement E-8 Original Page 13



September 22, 2008

Terasen Gas Corporate Office 16705 Fraser Highway, Surrey V3S 2X7

Attention: Kevin Hodgins Key Account Manager

RE: Bypass Transportation Agreement - Williams Lake division.

Dear Kevin:

As per clause 3.02 of the above stated agreement, West Fraser Mills Ltd would like to request that the Bypass agreement be extended for a further period of **3 years** to November 1st 2012.

Sincerely,

Andrew K. Kim

Project Engineer West Fraser Mills LTD.

1250 Brownmiller Road • Quesnel • B.C. • Canada • V2J 6P5 • (250) 992-0609 • (250) 992-0807 • http://www.westfrasertimber.ca

Letter Dated: July 31, 1995

Issued By: Tom Loski, Chief Regulatory Officer

Effective Date: November 1, 2009

BCUC Secretary: Original signed by E.M. Hamilton



West Fraser Mills Ltd

October 26, 2011

Terasen Gas Corporate Office 16705 Fraser Highway, Surrey V3S 2X7

Attention: Kevin Hodgins Key Account Manager

RE: Bypass Transportation Agreement - Williams Lake division.

Dear Kevin:

As per clause 3.02 of the above stated agreement, West Fraser Mills Ltd would like to request that the Bypass agreement be extended for a further period of **2 years** to November 1st 2014.

Sincerely,

Verth the

Veikko Paivinen

Financial Manager, Energy & Carbon West Fraser Mills Ltd.

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamiltion



1250 Brownmiller Road Quesnel, B.C. Canada V2J 6P5

Phone: (250) 992-9244 Fax: (250) 992-9233

October 31, 2013

VIA Email,

Attention Rajoo Jagtap,

Key Account Manager, Industrial Energy Solutions, FortisBC

Dear Rajoo,

Re: E-8 West Fraser Mills Ltd. (Williams Lake)-Amended and Restated Bypass Transportation Agreement Rate Schedule 25 - dated November 1, 1993

Please be advised that west Fraser Mills Ltd. wishes to extend the Bypass agreement for another year, to November 1, 2015.

Yours Truly,

Vill R

Veikko Paivinen, CA Financial Manager, Energy & Carbon

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2014

BCUC Secretary: Original signed by Erica Hamilton



1250 Brownmiller Road Quesnel, B.C. Canada V2J 6P5

Phone: (250) 992-9244 Fax: (250) 992-9233

October 15, 2014

VIA Email,

Attention Rajoo Jagtap,

Key Account Manager, Industrial Energy Solutions, FortisBC

Dear Rajoo,

Re: E-8 West Fraser Mills Ltd. (Williams Lake)-Amended and Restated Bypass Transportation Agreement Rate Schedule 25 - dated November 1, 1993

Please be advised that west Fraser Mills Ltd. wishes to extend the Bypass agreement for another 5 years, to November 1, 2020.

Yours Truly,

Vull Pi

Veikko Paivinen, CA Financial Manager, Energy & Carbon

Letter Dated: July 31, 1995

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2015

BCUC Secretary: Original signed by Erica Hamilton



TARIFF SUPPLEMENT NO. G-5

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT FOR RATE SCHEDULE 22A

BETWEEN

CANADIAN FOREST PRODUCTS LTD. Prince George Pulp and Paper Limited

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the <u>28th</u> day of <u>February</u>, 2001, with effect as and from November 1, 1993, is an amendment and restatements of an agreement made as of October 20, 1988 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

CANADIAN FOREST PRODUCTS LTD. incorporated under the laws of British Columbia having its registered office at 1055 Dunsmuir Street, Vancouver, British Columbia

(hereinafter called "Canfor")

OF THE SECOND PART

WHEREAS:

- A. Canfor operates pulp and paper mills in the City of Prince George, British Columbia, previously operated by Prince George Pulp and Paper Limited, which company has amalgamated with Canfor, and requires Gas for its operations; for the purposes of this Agreement, references to "Canfor" include all predecessor corporations including Prince George Pulp & Paper Limited;
- B. BC Gas owns and operates a Gas transmission pipeline, which is connected to the pulp and paper mills' operations of Canfor in Prince George, British Columbia;
- C. Canfor made an application to the Province of British Columbia for an Energy Project Certificate pursuant to British Columbia Order-In-Council No. 552 dated March 19, 1987, which, if granted, would have allowed Canfor to construct a Bypass Pipeline (hereunder defined) from the Westcoast Energy Inc. pipeline system mainline to Canfor's facilities at Prince George;

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

- D. Canfor entered into an agreement with BC Gas' predecessor company, Inland Natural Gas Co. Ltd. ("Inland") dated October 30, 1987 which was superseded and replaced by an agreement between Canfor and Inland dated October 20, 1988 effective November 1, 1987 which allowed Canfor to receive transportation service from BC Gas at rates reasonably equivalent to the costs that would have been incurred by Canfor had it constructed the Bypass Pipeline hereinafter defined in 1987;
- E. The British Columbia Utilities Commission has endorsed the concept of negotiated rates that are competitive with the Bypass Pipeline alternative;
- F. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service;
- G. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service; and
- H. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amended and restated Agreement to clarify their respective rights and obligations.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by Canfor in order to provide Gas service to their pulp mill operations in Prince George, British Columbia.

Order No.: G-33-2003

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

"ITR" means the annual total revenue received by BC Gas from Canfor for Gas transportation service and sales service to Canfor's Prince George, British Columbia pulp and paper mills operations.

"DTQ" means the Firm DTQ and Interruptible DTQ as defined in the Rate Schedule 22A Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which Canfor may elect to receive service.

"Terms and Conditions of the Transportation Schedule" means the terms of the applicable Transportation Rate Schedule, and the Transportation Agreements thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation or Gas sales service Rate Schedules accepted for filing by the British Columbia Utilities Commission.

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm and interruptible transportation service to Canfor for its pulp and paper mills operations in Prince George, British Columbia and Canfor will accept such transportation services in accordance with the then prevailing provisions of BC Gas Rate Schedule 22A and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Schedule conflicts or is inconsistent with the rates, terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 Canfor will be entitled to elect to take transportation service under a Transportation Rate Schedule other than Schedule 22A subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.

Order No.: G-33-03

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

2.04 If Canfor in any Contract Year elects to take transportation service pursuant to a Rate Schedule under which monthly Gas balancing provisions result in benefits to Canfor which would not have been available if the Bypass Pipeline had been constructed, the rates specified in Article 4, will be increased for that Contract Year by an amount equal to Canfor's savings resulting from the provision of BC Gas' monthly Gas balancing on the tolls that would have been incurred on the Westcoast pipeline system had Canfor constructed the Bypass Pipeline. The said amount will be determined by BC Gas and paid to BC Gas at the end of the Contract Year.

ARTICLE 3

TERM OF AGREEMENT

- 3.01 Subject to Article 8, the initial term of this Agreement will be for a period of four (4) Contract Years, effective the 1st Day of November, 1993, up to the 1st Day of November 1997.
- 3.02 The term of this Agreement will be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon Canfor providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date and BC Gas agreeing to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension of the term of this Agreement will be for a period of not less than one year.

ARTICLE 4

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for and based on a DTQ of 220 10³m³, Canfor each month will pay to BC Gas for services provided hereunder, from November 1, 1993 to the expiry of the Agreement, the following rates:
 - A monthly charge of \$9,973.00
 - (ii) Any charges pursuant to Articles 4.02, 4.03 and 4.04

Order No.: G-33-03

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: Original signed by R.J. Pellatt

- 4.02 The rates will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by Canfor if it had constructed and operated the Bypass Pipeline as a result of an increase in the volume of Gas actually transported for Canfor to the extent that such increase would have necessitated an increase in the capacity of the Bypass Pipeline or a modification or addition to the Bypass Pipeline facilities.
- 4.03 In addition to the foregoing rates, Canfor will pay to BC Gas at the end of each Contract Year an annual surcharge for increases, if any, in BC Gas costs. The surcharge will be determined as the sum of the following:
 - (i) cost changes to the estimated 1987 costs set out in Schedule 1 for operation and maintenance expenses that would have been incurred had Canfor constructed and operated the Bypass Pipeline; without limiting the generality of the foregoing, such costs include odorant costs, heating fuel costs, cathodic protection costs and labour costs; and
 - (ii) costs changes in municipal, provincial or federal taxes and fees, including new taxes or fees, but excluding taxes on taxable income, related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Canfor from the 1987 tax costs set out in line 10 of Schedule 1. For greater clarity, the parties agree that changes in costs of either debt or equity capital and taxes on taxable income are not to be included in the surcharge.
- 4.04 Any disputes arising hereunder as to the amount of the annual surcharge or the appropriateness of including costs under this Article will be referred to arbitration in accordance with Article 6 of this Agreement.

ARTICLE 5

FORCE MAJEURE

5.01 Notwithstanding any of the provisions contained herein or in any Transportation Rate Schedule, Canfor will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure, as defined in BC Gas Rate Schedule 22, after November 1, 1993.

Order No.: G-33-03

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>
ARBITRATION

- 6.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 6.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 6.03 The parties will have 10 days from receipt of the demand referred to in section 6.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of Canfor or BC Gas.
- 6.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 6.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

ARTICLE 7

FRANCHISE FEES

7.01 Canfor acknowledges that BC Gas is obligated to pay Franchise Fees to the City of Prince George in an amount equal to 3.09% of the revenues received in each calendar year by BC Gas for Gas consumed within the boundary limits of the City of Prince George.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

7.02 The rates payable by Canfor pursuant to Article 4 hereof include a component for Franchise Fees which is calculated as follows:

 $F = ITR \times .0309$

Where F = the annual Franchise Fees payable by BC Gas in respect of the Gas consumed by Canfor.

- 7.03 BC Gas takes the position that the above calculation represents the correct amount for Franchise Fees payable to the City of Prince George in respect of Gas consumed by Canfor and will take all reasonable efforts to defend this position.
- 7.04 In the event that BC Gas is required by statute, regulation, a court of competent jurisdiction or the British Columbia Utilities Commission to pay the City of Prince George any additional or lesser amount for Franchise Fees in respect of the Gas consumed by Canfor, Canfor will pay BC Gas such additional amounts or BC Gas will refund Canfor such lesser amount and the rate payable by Canfor as set out in Article 4 respect of Franchise Fees will be adjusted to reflect this change.

ARTICLE 8

TERMINATION

- 8.01 In addition to any other rights to terminate this Agreement:
 - (i) BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission determines that a revenue shortfall exists on service under this Agreement and disallows BC Gas the recovery from other customers of BC Gas, of any revenue shortfall resulting from these negotiated rates;
 - (ii) Canfor may terminate this Agreement if at any time the British Columbia Utilities Commission sets rates for the Canfor pulp and paper mills at a level in excess of the negotiated rates set out or provided for herein.
- 8.02 Any termination pursuant to Article 8.01 will take effect on the next November 1 following the date of notice in writing by the party terminating this Agreement.

Order No.: G-33-03

Effective Date: November 1, 1993

NOTICES

9.01 If in any year an executed Transportation Agreement is not in place, then the notice provisions of the last executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 10

MISCELLANEOUS

- 10.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 10.02 Notwithstanding Article 10.01, BC Gas may assign, without the consent of Canfor, BC Gas' rights and obligations under this Agreement to a party, which acquires all or substantially all of BC Gas' utility operations.
- 10.03 Notwithstanding Article 10.01, Canfor, may assign, without the consent of BC Gas, Canfor's rights and obligations under this Agreement to a party which acquires all or substantially all of Canfor's Prince George pulp mill operations served by BC Gas pursuant to this Agreement.
- 10.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns, including without limitation, successors by merger, amalgamation or consolidation.
- 10.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 10.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of the Agreement by the British Columbia Utilities Commission, the Agreement made between the parties as of October 20, 1988 will be cancelled.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

10.07 In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement.

AGREED TO AND ACCEPTED:

AGREED TO AND ACCEPTED:

This 27 day of 7. any ____. 2002

This 8 day of May , 2002

BC GAS UTILITY LTD.

Jill Dennesser BY:

Marketing Services Thomas

BILL HENNEDSGY

CANADIAN FOREST PRODUCTS LTD. BY: (Signard)

David M Calabrigg

G-33-03 Order No.:

Effective Date: November 1, 1993

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

ň			SCHEDULE 1 PAGE 1 OF 4
ULE) DF 4 DF 4 DF 4	Secol A, PS1, Dow	All (11, 11, 11) (11, 11) (11, 11, 11) (1	
PAGE 1	144	90,01 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10,00 10	
	581 581	10,444 (5,464) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,414) (10,41	10°24
1	1991	10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10.01 10	20, 21
	1001	84.41 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614.42 614	11. 12.
	9 9 800 ⁻ 0118		π.a
	900/41/4 1482	66,00 86,00 87,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,00 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,000 81,0000 81,0000 81,0000000000	24,22
	1000 to 1000	100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,400,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40 100,40000000000	99 V
2	100 L 100	80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,42 80,420,42 80,420 80,420 80,420,420 80,420,420 80,420,420 80,420,420 80,420,420 80,420,420 80,420,420 80,420,420,420,420 80,420,420,420,420,420,420,420,420,420,42	20,465
2 C	NITE AND	10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10,45 10	840°06
NUM BO	1000 1000 1000 1000 1000 1000		11,42° 10,740
			10,200
	Aunti adure Aunti adure Aunti adure	Territ evolution perverting i mericionide reports trans reports trans protocol brin 144 correction for income brin for correction for correct	interest & 152. Aff al net cars from YHT
	111	A A A A A A A A A A A A A A A A A A A	and a Mar

Order No.: G-33-03

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

	11.00	here. An one data	-	3,407,944			(12,001)		នផ្តិន ខត្តិទ		1,251,271	111110				SCHÉDULE PAGE 2 OF	14
SCHEDUL		000		111,400	1997 A			LECTED IN	dir/mo	'n.	10. A	- A	31,466	000	0		
				007/011	000,000	612 012	0,800		dir. and	12,865	R R	10,000	37.4605	15,885 (15,885) 0	1,82,1		
		1		119,441	100/10	1941) 1941) 1941) 1941)	dialo D		100°C	522	Ē	6H44,815	27,000	200 B	3,742		
		200 200	9 9	116, 865	(not 'Nt)		(FL)		Gui, ag	•12 2 2	10, 001	a a	211.6141	12,200 11,200 11,200	27		
		100	007/0111	111,000	(001,62)		(21,10)		C R R	11,525	19,78	33,475	27, 6915	979 979	9 2 3		
		202	1111,000	110,460	(Here'year)	17.413 19.742	0.000		0007 (B)	9 K.9	20,404	STOR A	2047-24	12,441 42,557	98472		
		R	000/0114	119,860	021,023	0, P12) 70, 759	1997.00	Ê	100 CON	ч Н П	40,756	015,010	1007/22	6148 1118 1118	9,644		
	90	2002	1110,460	10,000	(mov/wp)	(BW) 0	en, año	0001100		2	E S	Surg Surg	27.4905	SIL.A	11,005		
×	16. 66 Peri-4 IV	and the second s	007/01120	111,000	000,400	0,9U5 97,9T	(12, 60)	10,110	875 8 5	an'a	e in	Constant Constant	17.400	111,412 111,412	10,450		
	NING BO	-	000/0100	111,000	(N, 100)	5, 12 17, 12	(M, 123)	(17,833)			10000	(107'11) (107'11)	17.485	942,517 (15,514) the,545	11,122		
	ALTER COLUMN	1941	000'0114 V	10,000	(24, m01	200	(10,403)	Carlos and		tation of the second	1000	(115,004) (115,004)	10.00	10,10 (11,10)	611,61	2 .	
		10110	100,117,111 1	1.000 000 0	Complete I	001/00		019,700	98. 28.	682 5 ⁴ 1	10.45	018,400				15-0015	
						-		-				-	•		io.com	1000	
	T state	Particular	availabed (s) weathed) south	perved with contra	Deter Americanis	Aroutty Table	Convects I m Per	copital, cost allowers	tendis Incom terrection File	income Tex incustment Tex Credit contract Contral Constra	feature from Augh Flags	George Property and	Met each flow Income text rais - duntent	bedr represent & finiterest bibt - Bidl.F. Represent over 23 years here - 6.0.F.	incorect &	are of put cash flow 1887 -	
-		261	- 54	6-100	40 K. I			no sit	1212	855	1717	an star	88.85	REARSS	88.95	29.2	
Order No.:	(G-33-0	3									ls	sued	By: Sco Fina	tt The	omson, Vice and Regula	e Presiden tory Affairs

Attachment 25.10 **Terasen Gas** Rate Schedule 22A Supplement

Effective Date: November 1, 1993

BCUC Secretary: Original signed by R.J. Pellatt

TO CAE No: G 3302 House the set of the set o					SCHEDULE 1 PAGE 3 OF 4
1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000 1000	3 OF 4	AN IN CASE	11,412,400	11, 721, 204 4	
Image: 1 <td< td=""><td>SCHE</td><td>170,071</td><td>1945 3,141,200 3,141,200</td><td>40,000 0,000 0,000 0,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,0000 1,0000 1,0000 1,0000 1,00000000</td><td></td></td<>	SCHE	170,071	1945 3,141,200 3,141,200	40,000 0,000 0,000 0,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,0000 1,0000 1,0000 1,0000 1,00000000	
Order No: 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03 0-33-03			100 1,141,200 1,141,200	1,2001 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,2000 1,20000 1,20000 1,20000 1,20000 1,2000 1,2000 1,20000 1,200000000	
Order No: 6-33-03 6-33-03 1000000000000000000000000000000000000			3, 101, 200 3, 101, 200 3, 111, 200	0,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,1 1,000,	
The second secon			9, MJ, 200	M. (00) L. 000 C. 000 C	
Order No: 6-33-03 EState Br. Scott Thomson Vice Pr			1,111,200	400,00 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0	
Order No: G-33-03 Ested By: Scott Thomson Vice Print			1 11, 111, 200	80.000 2.0000 2.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000	
Order No.: G-33-03 Issued By: Scott Thomson Vice Print		2	101, 200 6 7 101, 200	0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000	
Order No.: G-33-03 Issued By: Scott Thomson Vice Print		Bud 1 Pure	4445 0 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1,000,1 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,0000 1,0000 1,0000 1,0000 1,00000000	
Order No.: G-33-03 Issued By: Scott Thomson Vice Pre-		1 645 80. UP	1, 141, 100 3, 141, 100 3, 141, 200	10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,00000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,00000000	
Order No.: G-33-03		PLAN MALA	1,141,250	100200 000000 00000 00000 00000 00000 00000 0000	
Order No.: G-33-03			5		ч.
Order No.: G-33-03 Issued By: Scott Thomson Vice Pro		in the second se	haritadare sug vouetr - bi ues - tirlla edjetoert	ANT INCLUSE BOTTO 4/14 / 100	
	Order No	G-:	33-03		ott Thomson. Vice Pro

Effective Date: November 1, 1993

resident Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

SCHEDULE 1 PAGE 4 OF 4	INVERSE AND AN ANALYSIA ANALANA A	element - espectio deit M-400-401 cont	Mail Fed. 187 198 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 200 2	MA.COMI MA.COMI <t< th=""><th>51,721,04 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 <t< th=""><th>SCHEDULE 1 PAGE 4 OF 4</th></t<></th></t<>	51,721,04 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 1119,460 <t< th=""><th>SCHEDULE 1 PAGE 4 OF 4</th></t<>	SCHEDULE 1 PAGE 4 OF 4
		é nga é	Line Arritodars 1.000 Villes 1.000 Villes 1.0000 Villes 1.0000 Villes 1.0000 Villes 1.0000 Vil	contribution coeffit 1/122 - Unit coef issuescent, - Vot	Na etheron Amount contra Auriced contra Cont insurred Fourier Fourier	
Order No.:	:	G	-33-03		Issued By: S	cott Thomson, Vice President inance and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: Original signed by R.J. Pellatt



Canadian Forest Products Ltd.

and affiliated companies

October 31, 1996

File: 281.032

Mr. Henry Dinter B.C. Gas 1111 West Georgia Street Vancouver, B.C. V6E 4M4

Dear Sirs:

Re: B.C. Gas Bypass Arrangement

Canadian Forest Products Ltd. hereby provide notice, pursuant to Article 3.02 of the Agreement, that it wishes to extend the Prince George Pulp and Paper Mills Bypass Agreement for an additional 10 year period commencing November 1, 1997.

Yours truty,

CANADIAN FOREST PRODUCTS LTD.

an

A.K. MacMillan, P.Eng Vice-President, Environment and Energy

cc: Mr. Ken Fuhr

Order No.: G-33-03

Effective Date: November 1, 1997

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Canfor Pulp Holding Inc.



October 27, 2006

Mr. Kevin Hodgins Terasen Gas Inc. 18705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins:

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulpmill) - dated November 1, 1987.

This is to inform you that Canfor Pulp Limited Partnership ("CPLP"), pursuant to Clause 3.02 of the above agreement, wishes to extend the agreement for a period of one year commencing November 1, 2007 to October 31, 2008. It is CPLP's intent to re-examine the agreement in the coming twelve months to decide on further renewal.

Please acknowledge receipt of this letter.

Sincerely,

CANFOR PULP LIMITED PARTNERSHIP by its General Partner, Canfor Pulp Holding Inc.

Thomas Sitar, Chief Financial Officer and Secretary

Cc: Ms. Marie Moffat

1700 West (%⁴ Avenue, Vancouver, British Columbia, Canada V6P 602 Telephone 504-501-5259 Fax 504-601-5258 info@perforpulp.com www.camlorpulp.com

Order No.: G-33-03

Effective Date: November 1, 2007

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and Chief Financial Officer Tariff Supplement G-5 Original Page 15



a business of Canfor Pulp Limited Partnership

September 25, 2007

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulpmill) - dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2008 to November 1, 2009. It is CPLP's intent to re-examine the agreement in the coming twelve months to decide on further renewal.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Carlor Pulp Holding Inc.

Marie Moffat Buyer/Energy Co-ordinator

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia, V2N 2K3 Telephone 250-553-0161, Fax 250-551-3021, info@cenforpulp.com, www.canforpulp.com

Order No.: G-33-03

Effective Date: November 1, 2008

BCUC Secretary: Original signed by E. M. Hamilton

Issued By: Scott Thomson, Vice President Regulatory Affairs and Chief Financial Officer Tariff Supplement G-5 Original Page 16



a business of Canfor Pulp Limited Partnership

September 9, 2008

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulpmill) – dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2009 to November 1, 2010.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

CANFOR PULP LIMITED PARTNERSHIP, by its General Partner, CANFOR PULP HOLDING INC. By: <u>Pro</u>

Rick Remesch, CA, Corporate Controller

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Tom Loski, Chief Regulatory Officer

Effective Date: November 1, 2009

BCUC Secretary: Original signed by E.M. Hamilton



a business of Canfor Pulp Limited Partnership

September 29, 2009

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulpmill) – dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2010 to November 1, 2011.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

RM En

Rick Remesch, CA Corporate Controller

cc. Marie Moffat, Canfor Pulp

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2010

BCUC Secretary: Original signed by E.M. Hamilton



a business of Canfor Pulp Limited Partnership

October 29, 2010

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulpmill) – dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2011 to November 1, 2012.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Rick Remesch

cc. Marie Moffat, Canfor Pulp

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2011

BCUC Secretary: Original signed by Alanna Gillis



a business of Canfor Pulp Limited Partnership

October 19, 2011

Mr. Kevin Hodgins Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulpmill) – dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2012 to November 1, 2013.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Rick Remesch

cc. Marie Moffat, Canfor Pulp Clarke Anderson, Canfor Pulp

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamilton



a business of Canfor Pulp Limited Partnership

September 25, 2012

Mr. Kevin Hodgins Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulp mill) – dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2013 to November 1, 2014.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Rajoo Jagtap, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2013

BCUC Secretary: Original signed by E.M. Hamilton



a business of Canfor Pulp Limited Partnership

October 31, 2013

Mr. Kevin Hodgins Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulp mill) – dated November 1, 1987.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2014 to November 1, 2015.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Rm 7

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Rajoo Jagtap, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 Info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2014

BCUC Secretary: Original signed by Erica Hamilton



a business of Canfor Pulp Limited

October 2nd, 2014

Mr. Rajoo Jagtap Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulp mill) - dated November 1, 1987.

Canfor Pulp Limited, "CPL" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPL wishes to extend the agreement for a period of one year commencing November 1, 2015 to November 1, 2016.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited by its general partner Canfor Pulp Holding Inc.

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Kevin Hodgins, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2015

BCUC Secretary: Original signed by Erica Hamilton



a business of Canfor Pulp Limited

September 10, 2015

Mr. Doug Tufts Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Prince George Pulp mill) - dated November 1, 1987.

Canfor Pulp Limited, "CPL" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPL wishes to extend the agreement for a period of one year commencing November 1, 2016 to November 1, 2017.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Kevin Hodgins, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

G-33-03

Order No.:

Issued By: Diane Roy, Vice President, Regulatory Affairs

Effective Date: November 1, 2016 Accepted for Filing: May 4, 2017

BCUC Secretary: Original signed by Patrick Wruck



TARIFF SUPPLEMENT NO. G-6

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT FOR RATE SCHEDULE 22A

BETWEEN

WEST FRASER MILLS LTD.

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the <u>21st</u> day of <u>November</u>, 2002, with effect as and from November 1, 1993, is an amendment and restatements of an agreement made as of October 28, 1988 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 16705 Fraser Highway, Surrey, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

WEST FRASER MILLS LTD. Having an office at 2 Mile Flat, 1000 Fining Road, Quesnel, British Columbia

(hereinafter called "West Fraser")

OF THE SECOND PART

WHEREAS:

- A. West Fraser operates a pulp mill in the City of Quesnel, British Columbia, and requires Gas for its operations;
- B. BC Gas owns and operates a Gas transmission pipeline, which is connected to the pulp mill operations of West Fraser in Quesnel, British Columbia;
- C. West Fraser entered into an agreement with BC Gas' predecessor company, Inland Natural Gas Co. Ltd. ("Inland") dated October 30, 1987 which was superceded and replaced by an agreement between West Fraser and Inland dated October 26, 1988 effective November 1, 1987 which allowed West Fraser to receive transportation service from BC Gas at rates reasonably equivalent to the costs that would have been incurred by West Fraser had it constructed the Bypass Pipeline hereinafter defined in 1987;
- D. The British Columbia Utilities Commission has endorsed the concept of negotiated rates that are competitive with the Bypass Pipeline alternative;

Letter Dated: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

- E. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service;
- F. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service; and
- G. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amended and restated Agreement to clarify their respective rights and obligations.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by West Fraser in order to provide Gas service to their pulp mill operations in Quesnel, British Columbia.

"ITR" means the annual total revenue received by BC Gas from West Fraser for Gas transportation service and sales service to West Fraser's Quesnel, British Columbia pulp mill operations.

"DTQ" means the Firm DTQ and Interruptible DTQ as defined in the Rate Schedule 22A Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which West Fraser may elect to receive service.

Letter Dated: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

"Terms and Conditions of the Transportation Schedule" means the terms of the applicable Transportation Rate Schedule, and the Transportation Agreements thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation or Gas sales service Rate Schedules accepted for filing by the British Columbia Utilities Commission.

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm and interruptible transportation service to West Fraser for its pulp mill operations in Quesnel, British Columbia and West Fraser will accept such transportation services in accordance with the then prevailing provisions of BC Gas Rate Schedule 22A and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Schedule conflicts or is inconsistent with the rates, terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 West Fraser will be entitled to elect to take transportation service under a Transportation Rate Schedule other than Schedule 22A subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.
- 2.04 If West Fraser in any Contract Year elects to take transportation service pursuant to a Rate Schedule under which monthly Gas balancing provisions result in benefits to West Fraser which would not have been available if the Bypass Pipeline had been constructed, the rates specified in Article 4, will be increased for that Contract Year by an amount equal to West Fraser's savings resulting from the provision of BC Gas' monthly Gas balancing on the tolls that would have been incurred on the Westcoast pipeline system had West Fraser constructed the Bypass Pipeline. The said amount will be determined by BC Gas and paid to BC Gas at the end of the Contract Year.

Letter Dated: G-33-2003

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

TERM OF AGREEMENT

- 3.01 Subject to Article 8, the initial term of this Agreement will be for a period of four (4) Contract Years, effective the 1st Day of November, 1993, up to the 1st Day of November 1997.
- 3.02 The term of this Agreement will be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon West Fraser providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date and BC Gas agreeing to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension of the term of this Agreement will be for a period of not less than one year.

ARTICLE 4

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for and based on a DTQ of 3090 gigajoules, West Fraser each month will pay to BC Gas for services provided hereunder, from November 1, 1993 to the expiry of the Agreement, the following rates:
 - (i) A monthly charge of \$7,437.00
 - (ii) Any charges pursuant to Articles 4.02, 4.03 and 4.04
- 4.02 The rates will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by West Fraser as a result of: a material increase in the annual Gas volume actually transported to West Fraser; or an increase in the capacity of the Bypass Pipeline to meet West Fraser's DTQ requirements; or modification or addition to facilities which may reasonably be required for any other reason, had West Fraser constructed and operated the Bypass Pipeline.

Letter Dated: G-33-2003

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

- 4.03 In addition to the foregoing rates, West Fraser will pay to BC Gas at the end of each Contract Year an annual surcharge for increases, if any, in BC Gas costs. The surcharge will be determined as the sum of the following:
 - (i) cost changes to the estimated 1987 costs set out in Schedule 1 for operation and maintenance expenses that would have been incurred had West Fraser constructed and operated the Bypass Pipeline; without limiting the generality of the foregoing, such costs include odorant costs, heating fuel costs, cathodic protection costs and labour costs; and
 - (ii) costs changes in municipal, provincial or federal taxes and fees, including new taxes or fees, but excluding taxes on taxable income, related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by West Fraser from 1987 tax costs set out in line 10 of Schedule 1. For greater clarity, the parties agree that changes in costs of either debt or equity capital and taxes on taxable income are not to be included in the surcharge.
- 4.04 Any disputes arising hereunder as to the amount of the annual surcharge or the appropriateness of including costs under this Article will be referred to arbitration in accordance with Article 6 of this Agreement.

FORCE MAJEURE

5.01 Notwithstanding any of the provisions contained herein or in any Transportation Rate Schedule, West Fraser will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure, as defined in BC Gas Rate Schedule 22, after November 1, 1993.

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

ARBITRATION

- 6.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 6.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 6.03 The parties will have 10 days from receipt of the demand referred to in section 6.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of West Fraser or BC Gas.
- 6.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 6.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

ARTICLE 7

FRANCHISE FEES

7.01 West Fraser acknowledges that BC Gas is obligated to pay Franchise Fees to the City of Quesnel in an amount equal to 3.09% of the revenues received in each calendar year by BC Gas for Gas consumed within the boundary limits of the City of Quesnel.

Letter Dated: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

7.02 The rates payable by West Fraser pursuant to Article 4 hereof include a component for Franchise Fees which is calculated as follows:

 $F = ITR \ge .0309$

Where F = the annual Franchise Fees payable by BC Gas in respect of the Gas consumed by West Fraser.

- 7.03 BC Gas takes the position that the above calculation represents the correct amount for Franchise Fees payable to the City of Quesnel in respect of Gas consumed by West Fraser and will take all reasonable efforts to defend this position.
- 7.04 In the event that BC Gas is required by statute, regulation, a court of competent jurisdiction or the British Columbia Utilities Commission to pay the City of Quesnel any additional or lesser amount for Franchise Fees in respect of the Gas consumed by West Fraser, West Fraser will pay BC Gas such additional amounts or BC Gas will refund West Fraser such lesser amount and the rate payable by West Fraser as set out in Article 4 respect of Franchise Fees will be adjusted to reflect this change.

ARTICLE 8

TERMINATION

- 8.01 In addition to any other rights to terminate this Agreement:
 - (i) BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission determines that a revenue shortfall exists on service under this Agreement and disallows BC Gas the recovery from other customers of BC Gas, of any revenue shortfall resulting from these negotiated rates;
 - (ii) West Fraser may terminate this Agreement if at any time the British Columbia Utilities Commission sets rates for the West Fraser pulp mill at a level in excess of the negotiated rates set out or provided for herein.
- 8.02 Any termination pursuant to Article 8.01 will take effect on the next November 1 following the date of notice in writing by the party terminating this Agreement.

Letter Dated: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

NOTICES

9.01 If in any year an executed Transportation Agreement is not in place, then the notice provisions of the last executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 10

MISCELLANEOUS

- 10.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 10.02 Notwithstanding Article 10.01, BC Gas may assign, without the consent of West Fraser, BC Gas' rights and obligations under this Agreement to a party, which acquires all or substantially all of BC Gas' utility operations.
- 10.03 Notwithstanding Article 10.01, West Fraser, may assign, without the consent of BC Gas, West Fraser's rights and obligations under this Agreement to a party which acquires all or substantially all of West Fraser's Quesnel pulp mill operations served by BC Gas pursuant to this Agreement.
- 10.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns, including without limitation, successors by merger, amalgamation or consolidation.
- 10.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 10.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of the Agreement by the British Columbia Utilities Commission, the Agreement made between the parties as of October 20, 1988 will be cancelled.

Letter Dated: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

10.07 In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement.

AGREED TO AND ACCEPTED:

This 3 day of Felzowers 2003

AGREED TO AND ACCEPTED:

This I day of Alderaker 200. 2

BC GAS UTILITY LTD.

BY:

WEST FRASER MILLS LTD

BY:

General Strength

Bill Hennessey, PEng. Marketing Services Manager Transportation and Marketing Services BC Gas Utility Ltd. (Name - Please Print)

PEUGOMENT

LUPP.ENG

Order No.: G-33-2003

Effective Date: November 1, 1993

- 17 14 16.	arr u ôuder rend mear said 1	teat.	81,234,335	ci, 343, 6141	905'216	1101, 2011		1000-1000	(111) (111)			6.35	001/100	200,000					
SCHEDUL 943E 1 0	100	No.	1019,010	(00)/02)	40,461	418,0005	(1971) 1971	10,410	121210	(1915) (1915) (1915)	1995) 1997 1997	New Y	09,1001	2 2	0.000	100 (s) 100 (s)	11, 808	10,004	
-		544	1219,014	(00,000)	10. M)	1000 (01)	10.00	1244,441	(HERO	10 10 10 10 10 10 10 10 10 10 10 10 10 1	1	in the second se	(101)(40)	HP H	41,1500	197,491 197,491	100,738	10, 100	
			10,042	007102	112,00	0.01,000	21,02 21,02 21,02	100,000	0.000		100		04,180	10.42	41,1101	139,000 (5),110	110, 011	10,000	
		500 million	019,346	00700	10,451	C18, 890	50°0	121.015	(9,212) 10,212	18 58 58	200		ch, (191)	44	12, 1300		115,000	12,634	
		1	110,010	(000,000)	115,04	000102	Single Second	001,700	64.10 10	201 F20 F201 F20 F201 F20	 	19.2	1000/10	200	10,000	117,145 19,1457	th, 160	01,10	
		5	114,912	1004/9000	100,244	14,000		CHANNE	60°.00	ef Te			049.50	12	12.23%	92 92	10,100	11,454	
		1999	110,000	1002'0021	115,00	018,000		(18,572)	100,000	68 28			0.000	1997 L	41,1563	127 NO	3 2 -	111, 111	
	. 2.	1993	010,000	(00,303)	107-10	018,0000	1997) 1978) 1979	Une 'est	(11/10)	20 32	*** 5 5		29,1205		44,3345	18,481 18,481	111 VI	10,430	
	V 648 E5, VI Benefician MBN Constitution MBN	NW.	1019,916	(01/102)	12,21	00001945	0,053	Citt. AND	12710	1000	0-07 :	The second se	19,160		42.,5955	11.00 0.18	100,000	817,11	inter et al la constante et al constant
	A PARTY AND A PART	200	101,420	(00,300)	100,000	018,0000	1,000	1007-000					69-1100		NUMBER OF	101 (101 101 (101	12, 21	1177	6,5500
	- 4															100/100		100'01	1001
	11-041-05	Auto I col a to	mainten for another market	incurred voil, pasta	foto and table	Concertion 4, on Information	frequety turk	Internal and a	correction free lands in the second s	and a france	Eventional trail theol to Devention the	tagetter Gent Alleinande	Approximation inter-	set tark film	PARAME TAX, AND 1 - AMPTIMUT	bent reserved & Murrest Set - 3.5.7. Uppresid and 20 pairs	dent = 0.010	District & 100	any at yes apprilled 1948 -
		ξį.			an a	174 JU		ant a	22	528	2.5		1713	12:22	2.5	ARXE	82.2	(2.22	1912

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

SCHEDULE 1 PAGE 1 OF 4

	-
2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.0000 2014.000 2014.000 2014.000 2014.000 2014.000 2014.	
2011,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200 11,200	Ē
	ġ.
40, 100, 200, 200, 200, 200, 200, 200, 20	ž,
	497°C
2000, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010, 2010,	5
1001 (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (1001) (100	5,484
2000 (10) (10) (10) (10) (10) (10) (10) (4, 41
2010 10 10 10 10 10 10 10 10 10 10 10 10	8114
	1,70
10, 100 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	81'Y 8
10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	10,4804
	H 100
T control T control	100 million
res 2 and 12 million for a first and the analysis of the anal	district B
2 : 2 2 & 2 100 1000 0 0 1 0 2 : 2 2 & 2 2 100 1000 0 0 1 0 0 2 : 2 2 & 2 2 100 1000 0 0 0 0 0 0 0 0 0 0 0 0 0	

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

												SCH PAGE	EDULE 3 OF	1 4		
	3 OF 4	an u h.M. Indiater			1.100,000	0.0	7, 100, 271				10,63,03	(\$1,585,404)	507°294			
	SCHE	200,7122			1001	•	124/64	1017-100 0.05510	\$1.40515 (17.0071)	111	13/4/31	120, 201	116,941			
					1855	0	114.62	No. 1785	00,101 00,1015	20.11	115'4-51	(100,303)	10,91			
					200 UN 1984		10,01	10,1190 6,3071	142F-0	10, 114	210,544	1212, 2021	110,000			
					14		70, 00	0411.114	Intel.0	6.10	345,9161	(001,000)	11,43			
					100	1917/101	201 W.	200	14250	ц. 195	116,910	1201,202	40, M			
					185	0	40,00	9011/108 1000/10	1651	41,1155	14,410	(00,300)	169, N1			
					5	0 125 ¹ 910	104,407	807.08 109	1927.00	11.11	197-0128	(200,552)	N ² -01			
		å	3	Control Door Pri-	199	0 (10) 0 (10)	709,927	1017-04 1017-04	1957-0 1957-0	2411.04	029-044	(10,411)	140,041			
		11 YE 10	is Maket met	NO 111 NO	2	18,817 4	TW/107	1117-04	1464.0	2	111.013	007700	14,51			
		Dilute setters	TTAUE ANALTS	INTRA - ND	2011	744, 627 0, 05	104° 801		tion a	41.772	put tan	1202,002	144,139			
					P. Contanta	1	11.10	110 - 121			AN CALL					
				tange b	,	suits visualit ves - strike all access	100 Stirl 30	21900 0001			LENGER MAKE	daat indum	Testal.			
					$\frac{\xi}{2\pi}$			-*9:55	92923		2014/0187	2.8.8.2	1531513			
Order No.:	G-:	33-20	03							Issued	By: Sc	ott T	homso	n, Vice	Pres	siden
Effective Date:	No	vemb	er 1	, 19	93						LIL	ance		eguiai	JULY A	malls

BCUC Secretary: Original signed by R.J. Pellatt

				a 6 ini	PAGE 4 OF
** *** 5%	of the second se	100 U		10,00,10 6,635,65 81,675,65	
SCHEDUL Må£ 4 (80 100 000	201/01	6.511 6.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.516 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.517 0.51700000000000000000000000000000000000	101,012	
e.		70, 101		00.50 034,905	
		10,010 11,010	N11.40 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C125.1 C1	1 9 12 1 8 1	
		2000 0 10 - 10 - 10 - 10 - 10 - 10 - 10 -		100,000	
		710,/027 710,/027 710,/027		1001,002	
		Tay of a	6.561 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.	106,910 106,600	
			8,118 2,815 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118 1,118	101,015 101,015	
		2000 N9, 927 Tun, 937		40°,00	
	AUP Sound liters	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	81,110 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120 1,120	IN AND	
	, cu ci. 19 deni. col en el rec	100,607 10,607 310,402	811 10 10 10 10 10 10 10 10 10 10 10 10 1	100,0031	
	100,150 100,150 110,00,150	100 M. 10	6,118 1,118 1,118 1,118 1,118	10,000 10,000	
		1,491,194		00,00,00 00,00,00	
æ		5			
		n Marcadara 1. Noundi - Ga 1. Noundi - Ga 1. Noundi - Ga 1. Noundi - Ga	100 - 1001 8183 819 100 - 1001 8183 819	acts matteries (1915 anniant cotts card forwards foot	

BC Gas Rate Schedule 22A Supplement

Effective Date: November 1, 1993

Order No.:

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

G-33-2003

Telephone: (250) 992-8919 Facsimile: (250) 992-2612



OUESNEL RIVER PULP COMPANY

1000 FINNING ROAD, QUESNEL, BRITISH COLUMBIA, CANADA, V2J 6A1

August 13, 1997

Mr. Heary L. Dinter Industrial and Transportation Services Manager BC Gas 1111 West Georgia Street Vancouver, BC

Rc: QRP/BCG Bypass Agreement

As per clause 3.02 of the above stated agreement, this letter will confirm QRP's intention to renew the bypass agreement for a further period of ten years.

QRP recognizes that the agreement anticipates a renewal notice of one year. This late notice is a result of the unique nature of this agreement in requiring explicit renewal and the long period since this agreement was an issue between the two parties. In addition, QRP is of the understanding that the amortization period that derived the rates charged was twenty years and believes this would support QRP's election of a ten year extension.

Could you please forward an acknowledgement of our election for our files?

Ifany further information is required, please contact me directly.

Sincerely,

Steve Weatherall Manager of Maintenance and Engineering Quesnel River Pulp Company

SW/sdk File: P-05

Order No.: G-33-2003

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1997

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>



October 30, 2006

Terasen Gas Corporate Office 16705 Fraser Highway, Surrey V3S 2X7

Attention: Kevin Hodgins Key Account Manager

RE: Bypass Transportation Adreement - Quesnel River Pulp division,

Dear Kevin:

As per clause 3.02 of the above stated agreement. West Fraser Mills Ltd would like to request that the Bypass agreement be extended for a further period of 5 years to November 1st 2012.

Sincerely,

A

Andrew K. Kim

Project Engineer West Fraser Mills LTD.

1250 Browsmiller Road + Queensi + S.C. + Genade + V2J 6P5 + (350) 592-0008 + (550) 592-0807 + http://www.westhusertiviber.ce

Order No.: G-33-03

Effective Date: November 1, 2007

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and Chief Financial Officer Tariff Supplement G-6 Original Page 15



October 26, 2011

Terasen Gas Corporate Office 16705 Fraser Highway, Surrey V3S 2X7

Attention: Kevin Hodgins Key Account Manager

RE: Bypass Transportation Agreement - Quesnel River Pulp division.

Dear Kevin:

As per clause 3.02 of the above stated agreement, West Fraser Mills Ltd would like to request that the Bypass agreement be extended for a further period of **5 years** to November 1st 2017.

Sincerely,

Veite fain

Veikko Paivinen

Financial Manager, Energy & Carbon West Fraser Mills Ltd.

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamilton


TARIFF SUPPLEMENT NO. G-7

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT FOR RATE SCHEDULE 22A

BETWEEN

CANADIAN FOREST PRODUCTS LTD. Northwood Pulp and Timber Limited

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the <u>28th</u> day of <u>February</u>, 2001, with effect as and from November 1, 1993, is an amendment and restatement of an agreement made as of May 25, 1992 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

CANADIAN FOREST PRODUCTS LTD., incorporated under the laws of British Columbia having its registered office at 1055 Dunsmuir Street, Vancouver, British Columbia

(hereinafter called "Canfor")

OF THE SECOND PART

WHEREAS:

- A. Canfor operates a pulp mill in the City of Prince George, British Columbia, and requires Gas for its operations (for the purposes of this Agreement all references to Canfor will include all predecessor corporations including Northwood Pulp and Timber Ltd.);
- B. BC Gas owns and operates a Gas transmission pipeline, which is connected to the pulp mill operations of Canfor in Prince George, British Columbia;
- C. Canfor made an application (the "Bypass Application") on August 24, 1987 to the Province of British Columbia for an Energy Project Certificate pursuant to British Columbia Order-in-Council No. 552 dated March 19, 1987, which, if granted, would have allowed Canfor to construct a Bypass Pipeline (hereunder defined) from the Westcoast Energy Inc. pipeline system mainline to Canfor facilities at Prince George;

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

- D. The Bypass Application was heard by the British Columbia Utilities Commission at a hearing in Prince George commencing September 9, 1987.
- E. Following the completion of the hearing of the Bypass Application, the British Columbia Utilities Commission submitted its report and recommendations dated October 22, 1987, to the Lieutenant-Governor in Council. The report recommended that "an Energy Project Certificate not be granted to Canfor providing that BC Gas, (formerly Inland Natural Gas Co. Ltd.) promptly offers Canfor a transportation contract with the rates and conditions specified in Section 6 of the report."
- F. By Order-in-Council dated December 10, 1987, the Lieutenant-Governor in Council accepted the October 22, 1987 report of the British Columbia Utilities Commission.
- G. Canfor entered into an Agreement with BC Gas dated May 25, 1992 which allows Canfor to receive transportation service from BC Gas at rates and on terms and conditions that are in accordance with the British Columbia Utilities Commission decision of October 22, 1987.
- H. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for re-design of its Gas tariff Rate Schedules which, among other things, proposed the consolidation of BC Gas' operating divisions (excluding Fort Nelson), the implementation of "postagestamp" rates for residential, commercial and small industrial customers (excluding those in Fort Nelson) and the implementation of common general terms and conditions of service to combine and eliminate certain tariffs and rates
- I. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the consolidation and "postage stamp" rates in certain service areas of BC Gas and approved the implementation of new tariffs effective November 1, 1993 for industrial and general service and January 1, 1994 for residential and commercial tariffs.
- J. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amending and re-stating Agreement to clarify their respective rights and obligations.

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1997

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by Canfor in order to provide Gas service to their pulp mill operations in Prince George, British Columbia.

"TTR" means the annual total revenue received by BC Gas from Canfor for Gas transportation service and sales service to Canfor's Prince George, British Columbia pulp mill operations.

"DTQ" means the Firm DTQ and Interruptible DTQ as defined in the Rate Schedule 22A Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which Canfor may elect to receive service.

"Terms and Conditions of the Transportation Schedule" means the terms of the applicable Transportation Rate Schedule, and the Transportation Agreements thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation or Gas sales service Rate Schedules accepted for filing by the British Columbia Utilities Commission.

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Schedule.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm and interruptible transportation service to Canfor for its pulp mill operations in Prince George, British Columbia and Canfor will accept such transportation services in accordance with the then prevailing provisions of BC Gas Rate Schedule 22A and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of the Transportation Schedule conflicts or is inconsistent with the rates, terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 Canfor will be entitled to elect to take transportation service under a Transportation Rate Schedule other than Schedule 22A subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.
- 2.04 If Canfor in any Contract Year elects to take transportation service pursuant to a Rate Schedule under which monthly Gas balancing provisions result in benefits to Canfor which would not have been available if the Bypass Pipeline had been constructed, the rates specified in Article 4, will be increased for that Contract Year by an amount equal to Canfor's savings resulting from the provision of BC Gas' monthly Gas balancing on the tolls that would have been incurred on the Westcoast pipeline system had Canfor constructed the Bypass Pipeline. The said amount will be determined by BC Gas and paid to BC Gas at the end of the Contract Year.

ARTICLE 3

TERM OF AGREEMENT

3.01 Subject to Article 8, the initial term of this Agreement will be for a period of four (4) Contract Years, effective the 1st Day of November, 1993, up to the 1st Day of November 1997.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

3.02 The term of this Agreement may be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon Canfor providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date and BC Gas agreeing to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension of the term of this Agreement will be for a period of not less than one year.

ARTICLE 4

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for and based on a DTQ of 226.5 10³m³, Canfor each month will pay to BC Gas for services provided hereunder, from November 1, 1988 to the expiry of the Agreement, the following rates:
 - (i) A monthly charge of \$9,781.00
 - (ii) Any charges pursuant to Articles 4.02, 4.03 and 4.04
- 4.02 The rates will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by Canfor if it had constructed and operated the Bypass Pipeline, as a result of an increase in the volume of gas actually transported for Canfor to the extent that such increase would have necessitated an increase in the capacity of the Bypass Pipeline or a modification or addition to Bypass Pipeline facilities.
- 4.03 In addition to the foregoing rates, Canfor will pay to BC Gas at the end of each Contract Year an annual surcharge for increases, if any, in BC Gas costs. The surcharge will be determined as the sum of the following:
 - (i) cost changes to the estimated 1987 costs set out in Schedule 1 for operation and maintenance expenses that would have been incurred had Canfor constructed and operated the Bypass Pipeline; without limiting the generality of the foregoing, such costs include odorant costs, heating fuel costs, cathodic protection costs and labour costs; and

Order No.: G-33-03

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

Effective Date: November 1, 1993

- (ii) costs changes in municipal, provincial or federal taxes and fees, including new taxes or fees, but excluding taxes on taxable income, related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Canfor from the 1987 tax costs set out in line 10 of Schedule 1. For greater clarity, the parties agree that changes in costs of either debt or equity capital and taxes on taxable income are not to be included in the surcharge.
- 4.04 Any disputes arising hereunder as to the amount of the annual surcharge or the appropriateness of including costs under this Article will be referred to arbitration in accordance with Article 6 of this Agreement.

BYPASS FACILITIES

- 5.01 The rates herein are based upon the construction of a hypothetical 6.1 kilometers of 6 NPS steel Gas pipeline to service the Canfor, Northwood pulp mill located on the Fraser River in Prince George. The facilities include a measurement and pressure regulation (M & R) station at the mills site.
- 5.02 The firm capacity of the hypothetical pipeline is 40 10³M³ per Hour in order to provide 175 psig pressure to the M & R station. The design capacity of the M & R station is 24 10³M³ per Hour at 60 psig delivery pressure to the pulp mill.
- 5.03 The Firm DTQ permitted hereunder is the lesser of the firm pipeline capacity or M & R station design capacity expressed on a daily basis. In order to meet an hourly delivery obligation of 5% under Schedule 22A, the Firm DTQ is therefore 480 10³M³ per Day as at the date of this Agreement (24 Hours divided by the hourly percentage obligation of 0.05. The Firm DTQ shall be adjusted in accordance with changes in capacity as reflected in the rates, or changes in the Hourly delivery obligation under Schedule 22A.

Order No.: G-33-03

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

Effective Date: November 1, 1993

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any Transportation Rate Schedule, Canfor will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure, as defined in BC Gas Rate Schedule 22, after November 1, 1993.

ARTICLE 7

ARBITRATION

- 7.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 7.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.
- 7.03 The parties will have 10 days from receipt of the demand referred to in section 7.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of Canfor or BC Gas.
- 7.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforescen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 7.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

FRANCHISE FEES

- 8.01 Canfor acknowledges that BC Gas is obligated to pay Franchise Fees to the City of Prince George in an amount equal to 3.09% of the revenues received in each calendar year by BC Gas for Gas consumed within the boundary limits of the City of Prince George.
- 8.02 Canfor acknowledges that the City of Prince George has disputed the method by which BC Gas will determine Franchise Fees payable to the City in respect of the Gas consumed by Canfor in any calendar year.
- 8.03 The rates payable by Canfor pursuant to Article 4 hereof include a component for Franchise Fees which is calculated as follows:

F = ITR x .0309

Where F == the annual Franchise Fees payable by BC Gas in respect of the Gas consumed by Canfor.

- 8.04 BC Gas takes the position that the above calculation represents the correct amount for Franchise Fees payable to the City of Prince George in respect of Gas consumed by Canfor and will take all reasonable efforts to defend this position.
- 8.05 In the event that BC Gas is required by statute, regulation, a court of competent jurisdiction or the British Columbia Utilities Commission to pay the City of Prince George any additional or lesser amount for Franchise Fees in respect of the Gas consumed by Canfor, Canfor will pay BC Gas such additional amounts.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

TERMINATION

- 9.01 In addition to any other rights to terminate this Agreement:
 - (i) BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission determines that a revenue shortfall exists on service under this Agreement and disallows BC Gas the recovery from other customers of BC Gas, of any revenue shortfall resulting from these negotiated rates;
 - (ii) Canfor may terminate this Agreement if at any time the British Columbia Utilities Commission sets rates for the Canfor pulp mill at a level in excess of the negotiated rates set out or provided for herein.
- 9.02 Any termination pursuant to Article 9.01 will take effect on the next November 1 following the date of notice in writing by the party terminating this Agreement.

ARTICLE 10

NOTICES

10.01 If in any year an executed Transportation Agreement is not in place, then the notice provisions of the last executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 11

MISCELLANEOUS

- 11.01 This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 11.02 Notwithstanding Article 11.01, BC Gas may assign, without the consent of Canfor, BC Gas' rights and obligations under this Agreement to a party, which acquires all or substantially all of BC Gas' utility operations.

Order No.: G-33-03

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

BCUC Secretary: Original signed by R.J. Pellatt

Tariff Supplement G-7 Original Page 9

100

- 11.03 Notwithstanding Article 11.01, Canfor, may assign, without the consent of BC Gas, Canfor's rights and obligations under this Agreement to a party which acquires all or substantially all of Canfor's Prince George pulp mill operations served by BC Gas pursuant to this Agreement.
- 11.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns, including without limitation, successors by merger, amalgamation or consolidation.
- 11.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- 11.06 This Agreement is of no force or effect until accepted for filing by the British Columbia Utilities Commission. Upon acceptance for filing of the Agreement by the British Columbia Utilities Commission, the Agreement made between the parties as of May 25, 1992 will be cancelled.
- 11.07 In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement.

AGREED TO AND ACCEPTED: AGREED TO AND ACCEPTED: This 8 day of <u>May</u>, 2002-This 29 day of Prag., 2002 CANADIAN FOREST PRODUCTS LTD. BC GAS UTILITY LTD. BY:BY: Jeamore MARKETING SEAVICES MANAGER (Total)

Corprate Secretary David M Calabriga

BILL HENNEBSCY (Pane - Pieze Print)

> Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

> > Tariff Supplement G-7 Original Page 10

Effective Date: November 1, 1993

G-33-03

Order No.:

Ling you at a trooped to the trooped to the trooped to the transfer to the tra	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	100 0.452 4.619 1.466 1.461 0.411 0.4 100 0.452 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461 1.461
	Mail Mail <th< td=""><td>1,200 1,400 1,5 1,300 1,500 1,5 1,300 1,500 1,500</td></th<>	1,200 1,400 1,5 1,300 1,500 1,5 1,300 1,500 1,500
us artes, as to the Un. or address as to the Market assessment of the	MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL MOL	8, 2007 10 144, 2009 1, 40 144, 2009 1, 40
101 101 101 101 101 101 101 101 101 101	Arritold Photocols and a construction of a const	a of set and the 1964 - 1997 I

Effective Date: November 1, 1993

BCUC Secretary: Original signed by R.J. Pellatt

SCHEDULE 1 PAGE 1 OF 4

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

Tariff Supplement G-7

		SCREDULE 1 PAGE 2 OF 4
016 J		x
PAGE 2	And the second s	
	2010 000 000 000 000 000 000 000 000 000	
	111 111 111 111 111 111 111 111 111 11	1977 1977 1977
	and the second s	
	11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,111 11,1111 11,1111 11,1111 11,1111 11,1111 11,1111 11,11111 11,111111	1111 1111 1111 1111 1111 1111 1111 1111 1111
	101 (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (15,17) (1	101 101 101 101 101 101 101 101 101 101
	100 100 100 100 100 100 100 100 100 100	1911 B.
	1000 1000, 100 1000, 1000, 100 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1000, 1	11, 110 11, 110 11, 110 11, 110 11, 110 11, 110 11, 110 11, 110 11, 110
. 2 3m	11111111111111111111111111111111111111	100 100 100 100 100 100 100 100 100 100
LL OHE ID. L LL INVERSION D	881 10,101 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,0000 10,00000000	100, 101 101, 101 100
100 0001	100/10 100/10 100/10 100/10 100/10 100/10 100/10	10, 10 10, 10 11, 10 1, 10
		11,000
		- 10,000 family
	Antipology and and antipology and	reserved. A locatest and the fit period and the fit period and the fit period with the fit of the fit of the set and the fit of the - Jac-
100		N N

Effective Date:

November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

		SCHEDULE 1 PAGE 3 OF 4
OLE 1 3 OF 4 so in a.rsu transmins	14441, 159,159,000 16,139,000 16,139,000	10,000,000
PAGE 3 PAGE 3	100 1 1 1 1 1 1 1 1 1 1 1 1 1	607'040 607'040
	1,011,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,012,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,000 1,00000000	CER-042
	1,001,000 1,001,000 3,001,000 0,0000 0,0000 0,0000 0,0000	212/214 (212/11/1
	600 (11) 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,021,000 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1,020 1	10,000
	1,412,100 1,412,100 1,412,100 1,412,000 0,120,007 0,120,007 0,120,007	100,000 (200,000
	1,001,000 1,001,000 2,012,000 0,1010 0,1010 0,1010 0,1010 0,1010 0,000	607-100 1007-100
	3, 123, 000 3, 123, 000 1, 000, 900 1, 000, 900 1, 000, 900 1, 000, 900 1, 000, 900 1, 000, 900 1, 000, 900	871,036 (777,433)
<i>.</i>	3,001,F09 3,001,F09 2,113,F00 0,5550 0,5550 01,0550 01,0550	E20'5161
a, sus do, Lif	3, II3, WH 3, II3, WH 4, CH1, MH 1, CH1, MH 1, CH1, MH	8915,026 (TTT_A251) 8117,473
utha autra	1,011,000 1,012,000 6,1203 6,1203 (6,1203 (6,1203) (6,1203)	11,419,279 (197,435) 36(2,445
	Mariante Mariana Version 10 - 10 10 - 10 - 10 10 - 10 - 10 10 - 10 -	th Hundburdt south diend water is South red total
		5 1 5 9 1020020000000000000000000000000000000
	361	

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

			e	SCREDULE 1 AGE 4 DE 4
E 1 if 4 inversion Mail 2.1	1904 41, 895, 905	12,48,75		819,210,411 014,733,2150 81,560,756
SCHEOUL AGE 4 0	1,117,000	100 Lan	100 M	530/1151
a.	20.01.000	94 BOO'S	N 100 100 100 100 100 100 100 100 100 100	CENTRAL CENTRAL
	1005 1000,100 0	8	A 2014	100°440
	1,001,004	1,02,00	1000 H	019,070 019,770 019,770
	1007 100 1	1,421,700	6, 105 6, 203 6, 203 7,	CEP'342
	2004 100.4	1,101,903	ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING ALLING AL	6113, 164 CTTP, J423
	1,011, MO	1,011,000	1,000	100 (M) 100 (M)
	5,682,998	3,000,000	90,201 0,201 0,001 0,001 0,001 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000 0,000000	1016,000 1016,000
é	0. 1,411,000 1,411,000	A.012./701	1000 1000 1000 1000 1000	4911,0144 (1911,0144) 1011,014
A, 64 01 1	0000 041 M	244 22016	AND A TROUBLE	011,015 (M1,050
Africa suff	And in the second secon	1,001,000	2012 2012 2012 2012 2012 2012 2012 2012	CIPAIN NOTION
	141, 144,	36,529,000		00/00/0
	An E Particulari Martines Martines Martines Martines	1 PTTH WINDS - 41	0015 0001 000 100	cucta Ambudis GMTs verilond carts cost toourced form
	a : 45 561			-

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

> Tariff Supplement G-7 Original Page 14



PO Box 9000 Prince George, BC Canada V2L 4W2 Tel (604) 962-3611 * Pax (604) 962-3582

August 13, 1997

Henry Dinter Manager, Industrial Services BC Gas Utility 1111 West Georgia Vancouver, B.C. V6E 4M4

Re: Transportation Agreement dated November 1, 1987

Dear Henry:

This letter will confirm Northwood Pulp's intention to reacw the above referenced bypass agreement for a further period of ten years.

In reviewing this agreement clause 3.02 suggests that a twelve month notice period is to be provided. I hope that this later notice is not going to result in any difficulties to BC Gas. While there have been a number of personnel changes with both parties to the agreement I believe it is fair to say that there was a tacit understanding that Northwood's intention would be to renew the agreement.

If any further a formation is required please contact me directly.

Yours the C.MMana ñng

Order No.: G-33-03

Effective Date:

November 1, 1997

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs and

BCUC Secretary: Original signed by R.J. Pellatt

Cario D

Canfor Pulp Holding Inc.

October 27, 2006

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V38 2X7

Dear Mr. Hodgins:

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulpmill) – dated November 1, 1988.

This is to inform you that Canfor Pulp Limited Partnership ("CPLP"), pursuant to Clause 3.02 of the above agreement, wishes to extend the agreement for a period of one year commencing November 1, 2007 to October 31, 2008. It is CPLP's intent to re-examine the agreement in the coming twelve months to decide on further renewal.

Please acknowledge receipt of this letter.

Sincerely,

CANFOR PULP LIMITED PARTNERSHIP by its General Partner, Canfor Pulp Holding Inc.

moo dik

Thomas Sitar, Chief Financial Officer and Secretary

Cc: Ms. Marie Moffat

1700 West 75⁸ Avenue, Vancouver, British Columbia, Canada VBP 652 Talephone 504-651-5250 Fax 504-851-5228 Info@conferpuip.com www.canferpuip.com

Order No.: G-33-03

.. . . .

Effective Date: November 1, 2007

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and Chief Financial Officer Tariff Supplement G-7 Original Page 16



a business of Canfor Pulp Limited Partnership

September 25, 2007

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulpmill) - dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2008 to November 1, 2009. It is CPLP's intent to re-examine the agreement in the coming twelve months to decide on further renewal.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Marie Moffat Buyer/Energy Co-ordinator

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3621 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Effective Date: November 1, 2008

BCUC Secretary: Original signed by E. M. Hamilton

Issued By: Scott Thomson, Vice President Regulatory Affairs and Chief Financial Officer Tariff Supplement G-7 Original Page 17



a business of Canfor Pulp Limited Partnership

September 9, 2008

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulpmill) – dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2009 to November 1, 2010.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

CANFOR PULP LIMITED PARTNERSHIP, by its General Partner, CANFOR PULP HOLDING INC. By: Rom

Rick Remesch, CA, Corporate Controller

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Tom Loski, Chief Regulatory Officer

Effective Date: November 1, 2009

BCUC Secretary: Original signed by E.M. Hamilton



a business of Canfor Pulp Limited Partnership

September 29, 2009

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulpmill) – dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2010 to November 1, 2011.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

RM1-

Rick Remesch, CA Corporate Controller

cc. Marie Moffat, Canfor Pulp

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2010

BCUC Secretary: Original signed by E.M. Hamilton



a business of Canfor Pulp Limited Partnership

October 29, 2010

Mr. Kevin Hodgins Terasen Gas Inc. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulpmill) – dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Terasen that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2011 to November 1, 2012.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp, Holding Inc.

Rick Remesch

cc. Marie Moffat, Canfor Pulp

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2011

BCUC Secretary: Original signed by Alanna Gillis



a business of Canfor Pulp Limited Partnership

October 19, 2011

Mr. Kevin Hodgins Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulpmill) – dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2012 to November 1, 2013.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Rick Remesch

cc. Marie Moffat, Canfor Pulp Clarke Anderson, Canfor Pulp

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamilton



a business of Canfor Pulp Limited Partnership

September 25, 2012

Mr. Kevin Hodgins Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulp mill) – dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2013 to November 1, 2014.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holdingvinc.

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Rajoo Jagtap, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Telephone 250-505-0161 Pax 250-561-5921 molgicalitorpulp.com www.canic

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2013

BCUC Secretary: Original signed by E.M. Hamilton



a business of Canfor Pulp Limited Partnership

October 31st, 2013

Mr. Kevin Hodgins Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulp mill) – dated November 1, 1988.

Canfor Pulp Limited Partnership, "CPLP" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPLP wishes to extend the agreement for a period of one year commencing November 1, 2014 to November 1, 2015.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited Partnership by its general partner Canfor Pulp Holding Inc.

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Rajoo Jagtap, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Oluel INU.. G-33-03

ושטע שני שומוים הטא הווברוטו, הבעטומוטוא שבואוכלצ

Effective Date: November 1, 2014

BCUC Secretary: Original signed by Erica Hamilton



a business of Canfor Pulp Limited

October 2nd, 2014

Mr. Rajoo Jagtap Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulp mill) - dated November 1, 1988.

Canfor Pulp Limited, "CPL" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPL wishes to extend the agreement for a period of one year commencing November 1, 2015 to November 1, 2016.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited by its general partner Canfor Pulp Holding Inc.

Rick Remesch, Corporate Controller

cc. Clarke Anderson, Canfor Pulp Kevin Hodgins, Fortis BC

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Services

Effective Date: November 1, 2015

BCUC Secretary: Original signed by Erica Hamilton



a business of Canfor Pulp Limited

September 10, 2015

Mr. Doug Tufts Fortis BC 16705 Fraser Highway Surrey, B.C. V3S 2X7

Dear Mr. Hodgins

RE: Amended and Restated Bypass Agreement for Rate Schedule 22A (Northwood Pulp mill) – dated November 1, 1988.

Canfor Pulp Limited, "CPL" is writing to inform Fortis BC that pursuant to Clause 3.02 of the above agreement CPL wishes to extend the agreement for a period of one year commencing November 1, 2016 to November 1, 2017.

Please acknowledge receipt of this letter so that I may add to my file.

Yours truly,

Canfor Pulp Limited

Rick Remesch, Corporate Controller

CC.	Clarke Anderson, Canfor Pulp	2
	Kevin Hodgins, Fortis BC	

2533 Prince George Pulpmill Road, Post Office Box 6000, Prince George, British Columbia V2N 2K3 Telephone 250-563-0161 Fax 250-561-3921 info@canforpulp.com www.canforpulp.com

 Order No.:
 G-33-03
 Issued By: Diane Roy, Vice President, Regulatory Affairs

 Effective Date:
 November 1, 2016
 Accepted for Filing: <u>May 4, 2017</u>

 Tariff Supplement G-7

BCUC Secretary: Original signed by Patrick Wruck



TARIFF SUPPLEMENT NO. G-8

AMENDED AND RESTATED BYPASS TRANSPORTATION AGREEMENT FOR RATE SCHEDULE 22A

BETWEEN

CARIBOO PULP & PAPER CO.

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

Effective November 1, 1993

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the <u>23rd</u> day of <u>March</u>, 20<u>01</u>, with effect as and from November 1, 1993, is an amendment and restatement of an agreement made as of October 26, 1988 between the parties noted below.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

OF THE FIRST PART

AND:

CARIBOO PULP & PAPER CO., a joint venture of WELDWOOD OF CANADA LTD. and DAISHOWA-MARUBENI INTERNATIONAL LIMITED, having their respective offices at 1055 and 1066 West Hastings Street, Vancouver, British Columbia

(hereinafter called "Cariboo")

OF THE SECOND PART

WHEREAS:

- A. Cariboo operates a pulp mill in the City of Quesnel, British Columbia, and requires Gas for its operations;
- B. BC Gas owns and operates a Gas transmission pipeline, which is connected to the pulp mill operations of Cariboo in Quesnel, British Columbia;
- C. Cariboo entered into an agreement with BC Gas' predecessor company, Inland Natural Gas Co. Ltd. ("Inland") dated October 30, 1987 which was superceded and replaced by an agreement between Cariboo and Inland dated October 26, 1988 effective November 1, 1987 which allowed Cariboo to receive transportation service from BC Gas at rates reasonably equivalent to the costs that would have been incurred by Cariboo had it constructed the Bypass Pipeline hereinafter defined in 1987;
- D. The British Columbia Utilities Commission has endorsed the concept of negotiated rates that are competitive with the Bypass Pipeline alternative.

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

- E. On April 15, 1993, BC Gas filed an application (the "Phase B Rate Design Application") with the British Columbia Utilities Commission for redesign of its Gas tariff Rate Schedules which, among other things, proposed the implementation of common Rate Schedules and General Terms and Conditions of service.
- F. Following the completion of the hearing of the Phase B Rate Design Application, the British Columbia Utilities Commission issued its decision dated October 25, 1993, which, in part, approved the implementation of new Rate Schedules effective November 1, 1993 for industrial and general service.
- G. As a result of the British Columbia Utilities Commission decision dated October 25, 1993, the parties desire to enter into this amended and restated Agreement to clarify their respective rights and obligations.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

<u>1.01</u> Except where the context expressly states another meaning, the following words will have the following meaning:

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by Cariboo in order to provide Gas service to their pulp mill operations in Quesnel, British Columbia.

"ITR" means the annual total revenue received by BC Gas from Cariboo for Gas transportation service and sales service to Cariboo's Quesnel, British Columbia pulp mill operations.

"DTQ" means the Firm DTQ and Interruptible DTQ as defined in the Rate Schedule 22A Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which Cariboo may elect to receive service.

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

"Terms and Conditions of the Transportation Schedule" means the terms of the applicable Transportation Rate Schedule, and the Transportation Agreements thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission by its decision of October 25, 1993 or any subsequent transportation or Gas sales service Rate Schedules accepted for filing by the British Columbia Utilities Commission.

<u>1.02</u> Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 BC Gas will provide firm and interruptible transportation service to Cariboo for its pulp mill operations in Quesnel, British Columbia and Cariboo will accept such transportation services in accordance with the then prevailing provisions of BC Gas Rate Schedule 22A and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, where anything in the Terms and Conditions of Transportation Rate Schedule conflicts or is inconsistent with the rates, terms and conditions set out in this Agreement, this Agreement governs.
- 2.03 Cariboo will be entitled to elect to take transportation service under a Transportation Rate Schedule other than Schedule 22A subject to the terms and conditions of such Rate Schedule. In such case, Article 2.01 and 2.02 will apply in the same way to the elected Rate Schedule.
- 2.04 If Cariboo in any Contract Year elects to take transportation service pursuant to a Rate Schedule under which monthly Gas balancing provisions result in benefits to Cariboo which would not have been available if the Bypass Pipeline had been constructed, the rates specified in Article 4, will be increased for that Contract Year by an amount equal to Cariboo's savings resulting from the provision of BC Gas' monthly Gas balancing on the tolls that would have been incurred on the Westcoast pipeline system had Cariboo constructed the Bypass Pipeline. The said amount will be determined by BC Gas and paid to BC Gas at the end of the Contract Year.

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

TERM OF AGREEMENT

- 3.01 Subject to Article 8, the initial term of this Agreement will be for a period of four (4) Contract Years, effective the 1st Day of November, 1993, up to the 1st Day of November 1997.
- 3.02 The term of this Agreement will be extended beyond the initial termination date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the pricing provisions, upon Cariboo providing written notice to BC Gas of its desire to extend the term of the Agreement for a further specified period, at least 12 months prior to the then current termination date and BC Gas agreeing to such extension of the term of this Agreement. Notwithstanding the foregoing, any extension of the term of this Agreement will be for a period of not less than one year.

ARTICLE 4

RATES AND CHARGES

- 4.01 Subject to the adjustments hereinafter provided for and based on a DTQ of 3,552 gigajoules, Cariboo each month will pay to BC Gas for services provided hereunder, from November 1, 1993 to the expiry of the Agreement, the following rates:
 - (i) A Demand Charge of \$5,868.00
 - (ii) Any charges pursuant to Articles 4.02, 4.03 and 4.04
- 4.02 The rates will be adjusted by BC Gas to reflect any changes in costs which would have been incurred by Cariboo as a result of: a material increase in the annual Gas volume actually transported to Cariboo; or an increase in the capacity of the Bypass Pipeline to meet Cariboo's DTQ requirements; or modification or addition to facilities which may reasonably be required for any other reason, had Cariboo constructed and operated the Bypass Pipeline.
- 4.03 In addition to the foregoing rates, Cariboo will pay to BC Gas at the end of each Contract Year an annual surcharge for increases, if any, in BC Gas' costs. The surcharge will be determined as the sum of the following:

Order No.: G-33-2003

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

- (i) cost changes to the estimated 1987 costs set out in Schedule 1 for operation and maintenance expenses that would have been incurred had Cariboo constructed and operated the Bypass Pipeline; without limiting the generality of the foregoing, such costs include odorant costs, heating fuel costs, cathodic protection costs and labour costs; and
- (ii) costs changes in municipal, provincial or federal taxes and fees, including new taxes or fees, but excluding taxes on taxable income, related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Cariboo from the 1987 tax costs set out in line 10 of Schedule 1. For greater clarity, the parties agree that changes in costs of either debt or equity capital and taxes on taxable income are not to be included in the surcharge.
- 4.04 Any disputes arising hereunder as to the amount of the annual surcharge or the appropriateness of including costs under this Article will be referred to arbitration in accordance with Article 6 of this Agreement.

FORCE MAJEURE

5.01 Notwithstanding any of the provisions contained herein or in any Transportation Rate Schedule, Cariboo will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure, as defined in BC Gas Rate Schedule 22, after November 1, 1993.

ARTICLE 6

ARBITRATION

- 6.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- <u>6.02</u> Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

- 6.03 The parties will have 10 days from receipt of the demand referred to in section 6.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of Cariboo or BC Gas.
- 6.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed by the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment, then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- $\underline{6.05}$ The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

FRANCHISE FEES

- 7.01 Cariboo acknowledges that BC Gas is obligated to pay Franchise Fees to the City of Quesnel, in an amount equal to 3.09% of the revenues received in each calendar year by BC Gas for Gas consumed within the boundary limits of the City of Quesnel.
- <u>7.02</u> The rates payable by Cariboo pursuant to Article 4 hereof include a component for Franchise Fees which is calculated as follows:

F = ITR x .0309

Where F = the annual Franchise Fees payable by BC Gas in respect of the Gas consumed by Cariboo.

7.03 BC Gas takes the position that the above calculation represents the correct amount for Franchise Fees payable to the City of Quesnel in respect of Gas consumed by Cariboo and will take all reasonable efforts to defend this position.

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

7.04 In the event that BC Gas is required by statute, regulation, a court of competent jurisdiction or the British Columbia Utilities Commission to pay the City Quesnel any additional or lesser amount for Franchise Fees in respect of the Gas consumed by Cariboo, Cariboo will pay BC Gas such additional amounts or BC Gas will refund Cariboo such less amount and the rate payable by Cariboo as set out in Article 4 respect of Franchise Fees will be adjusted to reflect this change.

ARTICLE 8

TERMINATION

- <u>8.01</u> In addition to any other rights to terminate this Agreement:
 - BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission determines that a revenue shortfall exists on service under this Agreement and disallows BC Gas the recovery from other customers of BC Gas, of any revenue shortfall resulting from these negotiated rates;
 - (ii) Cariboo may terminate this Agreement if at any time the British Columbia Utilities Commission sets rates for the Cariboo pulp mill operations at a level in excess of the negotiated rates set out or provided for herein.
- 8.02 Any termination pursuant to Article 8.01 will take effect on the next November 1 following the date of notice in writing by the party terminating this Agreement.

ARTICLE 9

NOTICES

<u>9.01</u> If in any year an executed Transportation Agreement is not in place, then the notice provisions of the last executed Transportation Agreement and Rate Schedule will apply to this Agreement.

Effective Date:

MISCELLANEOUS

- <u>10.01</u> This Agreement will not be assigned without the written consent of the other party hereto, which consent will not be unreasonably withheld.
- 10.02 Notwithstanding Article 10.01, BC Gas may assign, without the consent of Cariboo, BC Gas' rights and obligations under this Agreement to a party, which acquires all or substantially all of BC Gas' utility operations.
- 10.03 Notwithstanding Article 10.01, Cariboo, may assign, without the consent of BC Gas, Cariboo's rights and obligations under this Agreement to a party which acquires all or substantially all of Cariboo's Quesnel pulp mill operations served by BC Gas pursuant to this Agreement.
- <u>10.04</u> This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns, including without limitation, successors by merger, amalgamation or consolidation.
- <u>10.05</u> This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia as to nature, validity and interpretation.
- <u>10.06</u> This Agreement is of no force or effect until accepted for filing by the British
 Columbia Utilities Commission. Upon acceptance for filing of the Agreement by the
 British Columbia Utilities Commission, the Agreement made between the parties as of
 October 26, 1988 will be cancelled.
- <u>10.07</u> In this Agreement the words, phrases or expressions which are not defined herein and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has an accepted meaning will have that meaning.

Order No.: G-33-2003

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: November 1, 1993

IN WITNESS WHEREOF, the parties hereto have executed this Agreement.

AGREED TO AND ACCEPTED:

This 21 day of May , 2002

AGREED TO AND ACCEPTED:

This 2 day of Amil_, 2001

BC GAS UTILITY LTD.

BY:

BILL HENNESSEY (Name - Please Print)

WELDWOOD OF CANADA LTD.

BY:

Tide)

PAUL RICHARDS (Name - Please Print)

DAISHOWA-MARUBENI INTERNATIONAL LTD.

BY:

÷.

PRESIDENT

(Title)

Tokiro Kawamura

(Name - Please Print)

Order No.: G-33-2003

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

> Tariff Supplement G-8 Original Page 9
SCHEDULE 1

Rate Schedule 22A Supplement

Attachment 25.10

PAGE 1 OF 4

CPP 11 0.706 INVESTMENT PADE 1 (160,000) (14,510) 984 ¢9884 Total 219,217 412,230 201,001 227, 509 (81,200) 191,000 ŝ (18,000) (1,451) (3,932) (8,120) 81[,]E3 42.5300 1,20 1,20 1,20 1,20 1,20 8,6 29,62 -----10.50 đ, 932 (18,000) (10,15) (17/16) (8,120) 69,61 8,6 10,922 24,40 21 12.3305 11, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31, 15 31 021.6 4°24 Ē -----5.35 z ¥. 607°EL# (18,000) 1997 (955,01) 8,1203 1000.53 8, e 50,936 11,12 10,556 ł (18,000) 10,40 11,451) 111,3661 (8,120) 12.130 ŝ 89, E 田記 11 60 (51 120) E, E 11,366 2. 2 5 Ę -192 607"0.1 (16,000) 12,150) (021, E) 42.330K 897°8 19710 1,994 121,800 (8,120) 12,180 00,938 Ę ŝ 12.247 1 1651 (18,000) 639°E4 (134,15) 3, 120) 42.3301 87, E (12,992) 12,003 27,345 129,928 (6,120) 21,800 12,992 5 8 (18,000) 69 'R 897°R 11,451 113,604 (6,120) 12.5305 (021,85) (021,8) ŝ 11,419 12.2 13, 604 13.410 27, 107 8,8 8 6**9**64 (11,000) (919,41) 12.3305 507°E 11.4511 120) (8, 180 (8, 120) 70°, 60 14,616 12,97 5,53 856'0 ř, 38,040 CARIBOO PULP & PAPER CONRECTION FEE NUTURAL CAS CO. LTD. 5961 (18,000) (15,426) (8,120) 570,409 \$ 2 1.431 26,654 12.3300 51,200 (51,120) 15,428 146,140 14.541 1961 (1,631) 2007 2007 (8,120) 43.670X 18.000 18.800 18.800 \$274,802 274, 602 (18,000) 163,603 155,252 (6,120) (6,120) 16,240 ŝ 54,280 10.00 10.00% 20 years Interest Particulars Awolded Intendants costs income tax rate - current Operating cash flow Operating 4 meintenand ection fee tal Cost Alloweron income before tax Cost Allowance Tax Credit mourned WCTL costs fotal available 11-0et-58 Range I repayment 4 1 muble troom eperment over bebt + £.0.T Property taxes Interest & 10% bebt repayment wet cash flow nteres ž 100 ebt oot ÷.

Order No.: G-33-2003

SCHEDULE

Effective Date: November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

ŝ

9.8505

MPV of het cash flow 1986 - 1997

BC Gas

BCUC Secretary: Original signed by R.J. Pellatt

BAL Aven Mar STOR, 100 SOL, 403 SOL, 403 S

Order No.: G-33-2003

Effective Date: November 1, 1993

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

> Tariff Supplement G-8 Original Page 11

- A-----

Rate Schedule 22A Supplement

Attachment 25.10 BC Gas Rate Schedule 22A Supplement

SCHEDULE 1

ŧ

PAGE 3 OF 4

PA	1913			1951 Mai 1959 1961 Mai 1961 Mai 1960	34'1 MI'295'1 MI'296	40,0515 140,0515 50, 0,0000 0,0000 0,0	a.cots 0.cots 0. a.cots 0.cots 0. a.cots 40.cots 40.	72 031,072 001,072 0 0 72 031,072 001,073
				0	1 11/2001	0.000	0.0315 0.0315 50.0315	617/613
				9 1,756,1	11,786,1	3,0000	8.000 8.000 8.000	6 6 6 6 7 7 7 7 6
-				1991 772,736,1	125,786,1	80,0015 0,0000	0.0515 0.0005 0.0515	8)'Es 8)'Es
				1,367,174	1,347,174	0,000	0.0015 0.0000 1170.04	6 6 6 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
		PAPER.	MARE COSTS	11,734,1	1,367,174	\$1,0515 0,000	8,0515 8,0515	87 ⁴ 87 6
	1, 045 00, 175	211100 PLCP 1	NCED BUE PUBL	1944	1,146,1	0,000	0.05 0.000 0.000 0.05	607'945 607'945
	UNUTAR RUTUR	TANKA SEATT	NEWENCE - INDO	1,347,174	1,207,174	0102-04	0.2018 0.0000 0.0000	504, M22
	_	-	-	e.01	а	184 MCT	ţ	E
			5 egre	Particut ALES VOLVES - CJ 434 - Stribs edjustment	- \$34000A \$3145 13	- ro/s staco cadiov	out increased	ISBUCED FURCHASE 000 Avoided conta Cast Incurred Focal

Order No.: G-33-2003

Effective Date:

November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

ULE 1 OF 4	CAN TI 0,704 INVESTMENT MARK 2.1		Totat	20,710,454	24,710,454					254,556,12		11,662,952	SCHEDULE 1 PAGE 4 OF 4
SCHED PAGE 4	CASE 2911,000		2002	1,367,174	1,747,174	\$120.02 0.0000	0.000	5150.64		607°045	•	10,10	
			2002	1,367,174	1,74,746,1	90.0515 0.0000	0.00	90.0515		607'015	•	23,43	
۰.			2002	11,746,1	1,367,174	\$0.0515 0.0000	0.0515	\$0.0515		\$19°603	a, ·	89,68 8	с. 19
			2004	1,367,174	1,92,11	0,000	0.0005	10.0515		\$7.E3	•	10,459	
			1002	1,742,174	1,96,174	\$1,000 0,0000	0.0015 0.0005	1978		5)°E	•	81 ⁶ 6	
			200	1,347,174	1,32,134	90000'e	9-0212 9-0212	\$150'94		697°23	•	607'048	
			2001	11,762,1	1,326,1	\$0,0515 0,0000	0.0515	10.01		10,409	•	81°83	
			2000	1,367,174	11,752,1	0,0000	0,0515	8.03 12		607 "028	•	40) ¹ 025	
	a.	CHART COSTS	6661	, 347, 74E, 1	11,764,1	300,0015 0,0000	8-0515 8-0005	M.015		607 ⁴ 81	•	\$97°EL	
	L 645 (0. L1	NA 210 000	1995	, 367, 174 0	11,96,1	80.0515 0.0000	0.0000	51-02-02 12		407"045	•	407'925	
	NUMBER OF STREET	ENTINE - MED	1461	1,347,574	1,786,1	\$0,0515 0.0000	0.0000	19.0		#10,409	•	\$10,400	Ψ. ·
		-	Bat. Pod.	0,47,770,6	ov, ra, t			•••		527 9245	•	507,6614	
	1			8.9								•	
311		tanga é	Perticutars	sales voumes - 64 Less - Strike adjustment	ket skiles volumes - GJ	Avolote costs A/64 - 140	cost INCLUSED . VCT		AENCED PROMAE COSTS	Avolded costs	Cost incurred	total	

Order No.:

Effective Date:

November 1, 1993

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

G-33-2003

Tariff Supplement G-8 Original Page 13

Attachment 25.10 BC Gas Rate Schedule 22A Supplement

P Cariboo

October 24, 1996

BC Gas Utility Ltd. 1111 West Georgia Street Vancouver, BC V6E 4M4

Attention: Mr. David M. Masuhara, Vice President Legal and Regulatory Affairs

Dear Sir:

Pursuant to Clause 3.02 of the Service Agreement (Agreement) for Schedule 22 between Inland Natural Gas Co. Ltd. and Cariboo Pulp & Paper Company dated November 1, 1987, Cariboo hereby serves written notice of its desire to extend the term of the Agreement by three (3) years. The first extension Agreement term shall end at 0800 hrs. on November 1, 2000. Cariboo reserves the right to make further extensions pursuant to the provisions of Clause 3.02 of the Agreement.

Yours truly,

CARIBOO PULP & PAPER COMPANY

Alma

Colin Almond, Controller

CKA/muk

VABCGAgenut.sam

Order No.: G-33-2003

Effective Date: November 1, 1997

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

FACSIMILE MESSAGE

To: Ken Fuhr BC Gas Utility Ltd.

Date: October 20,1999

Fax No: 443-6770

Pages (including this one) 1

Re: BC Gas Bypass Transportation Agreement Effective November 1, 1998 to November 1, 2000

Please accept this facsimile message as notification to extend the terms and rates of the above noted Transportation Agreement as amended from time to time, in accordance with your Bypass Agreement. We would request that the Transportation Agreement be extended for $\underline{2}$ years to November 1, $\underline{2.002}$.

We understand that you will be preparing the necessary documentation to affect this extension.

We thank you for your past services provided and look forward to a mutually satisfactory arrangement in future.

iboo Ruly & Capor

cc: Harry Chivers Avista Energy Canada, Ltd.

Order No.: G-33-2003

Effective Date: November 1, 2002

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt



October 24, 2001

Ms. Melissa Philion BC Gas Utility Ltd. 16705 Fraser Highway Surrey, B.C. V3S 2X7

Fax No: (604) 592-7894

Dear Ms. Philion:

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and BC Gas Utility Ltd. for a further two years for expiry November 1, 2004. The current agreement expires November 1, 2002.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you strill Robert C. Hesketh

Materials Manager Cariboo Pulp & Paper Company PO Box 7500 Quesnel B.C. V2J 3J6

Castoon Pulp & Paper Compony, P.O. Box 7500. Quesnet, S.C. V2/ 3/6 Telephone: (250) 992-0000 Telefox: (250) 992-2164

Order No.: G-33-2003

Effective Date:

November 1, 2002

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt



October 14, 2003

Ms. Melissa Philion BC Gas Utility Ltd. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No: 604-592-7894

Dear Melissa:

RE: Bypass Agreement Extension Request

Please accept this letter as notice of our desire to extend the existing Bypass-Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2005. The current agreement expires November 1, 2004.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours-truly, £ CAL ež.

Robert C. Hesketh Materials Manager

cc: Mary McCordic, Avista Energy Canada, Ltd. (fax no: 604-682-6447)

Costboo Pulp & Paper Company, RO, Box 7500, Queenel, B.C. 1/2/ 3/6 Tetephoner (250) 992-0200 Tetefox: (250) 992-2164

Order No.: G-33-2003

Effective Date: November 1, 2004

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt

Cariboo

October 5, 2004

Mr. Gordon Doyle BC Gas Utility Ltd. 16705 Fraser Highway Surrey, B.C. V3S 2X7 Fax No: 604-592-7894

Dear Gord:

RE: Bypass Agreement Extension Request

Please accept this letter as notice of our desire to extend the existing Bypass Agreement between our firm and Terasen Gas for a further year for expiry November 1, 2006. The current agreement expires November 1, 2005.

Please confirm receipt of this letter and the extension of the agreement in writing to the undersigned, along with a copy to our agent, Avista Energy Canada, Ltd.

Thank you.

Yours truly, Nobul C Meduth MATERIALS MANAGER

cc: Mary McCordic, Avista Energy Canada, Ltd. (fax no: 604-682-6447)

Carlboo Pulp & Paper Company, P.O. Box 7500, 600 North Star Road, Quesnel, B.C. V2J 336 Phone: (250) 992-0200 Fax: (250) 992-2164

Order No.: G-33-03

Effective Date: November 1, 2005

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: Original signed by R.J. Pellatt



Terasen Gas 16705 Fraser Highway Surrey, British Columbia V3S 2X7 Fax: (604) 592-7894

September 23, 2005

Attention: Kevin Hodgins

Re: Amended and Restated Bypass Transportation Agreement for Rate Schedule 22A Between Cariboo Pulp & Paper Co. and Terasen Gas Inc.

Dear Sir,

In accordance with Article 3.02, we request the above noted agreement be extended for a period of three (3) years, effective November 1, 2006.

Please do not hesitate to contact me if you have any questions or require further information.

Sincerely,

Bob Hesketh Materials Manager Cariboo Pulp & Paper Co. Phone: (250) 992-0291

Carlboo Pulp & Paper Company, P.O. Box 7500, 600 North Star Road, Quesnel, B.C. V2J 3J6 Phone: (250) 992-0200 Fax: (250) 992-2164

Order No.: G-33-03

Effective Date: November 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance & Regulatory Affairs and **Chief Financial Officer Tariff Supplement G-8 Original Page 19**



P.O. Box 7500 Quesnel, B.C. V2J 3J6

August 22, 2008

Terasen Gas 16705 Fraser Highway Surrey, British Columbia V3S 2X7

Attention : Mr. Kevin Hodgins:

Re: Amended and Restated Bypass Transportation Agreement for Rate Schedule 22A Between Cariboo Pulp & Paper Company and Terasen Gas Inc..

Dear Mr. Hodgins,

In accordance with Article 3.02, we request the above noted agreement be extended for a period of three (3) years, effective November 1, 2009.

Please do not hesitate to contact me if you have any questions or require further information.

Regards,

Mike Pagurut Controller Cariboo Pulp & Paper Company Phone : 250-992-0204 *CC. ANDREW KIM*

Order No.: G-33-03

Issued By: Tom Loski, Chief Regulatory Officer

Effective Date: November 1, 2009

BCUC Secretary: Original signed by E.M. Hamilton

Cariboo Pulp & Paper Company

P.O. Box 7500 50 North Star Road Quesnel, B.C. V2J 3J6

October 17, 2011

Terasen Gas Corporate Office 16705 Fraser Highway, Surrey V3S 2X7

Attention: Kevin Hodgins Key Account Manager

RE: Bypass Transportation Agreement - Cariboo Pulp & Paper Company

Dear Kevin:

As per clause 3.02 of the above stated agreement, Cariboo Pulp & Paper Company would like to request that the Bypass agreement be extended for a further period of **5** years to November 1st 2017.

Sincerely,

Mike Part

Mike Pagurut

Controller, Cariboo Pulp & Paper Company

Order No.: G-33-03

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 2012

BCUC Secretary: Original signed by Erica Hamilton



TARIFF SUPPLEMENT NO. G-10

BYPASS TRANSPORTATION AGREEMENT FOE RATE SCHEDULE 22

DATED MAY 1, 1998

BETWEEN

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc. and BC Gas Utility Ltd.)

AND

WEST FRASER MILLS LTD. (WESTPINE)

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT made as of the <u>1st</u> day of <u>May</u>, 19<u>98</u>, with effect as and from November 1, 1996.

BETWEEN:

BC GAS UTILITY LTD., a company incorporated under the laws of British Columbia having its registered office at 1111 West Georgia Street, Vancouver, British Columbia

(hereinafter called "BC Gas")

AND:

OF THE FIRST PART

WEST FRASER MILLS LTD., a company incorporated under the laws of British Columbia having its head office at Suite 1000, 1100 Melville Street, Vancouver, British Columbia

(hereinafter called "WestPine")

OF THE SECOND PART

WHEREAS:

- A. WestPine operates an MDF plant in Quesnel, British Columbia, (the "MDF Plant"), and requires Gas for its operations;
- B. BC Gas owns and operates a Gas transmission pipeline which is connected to the MDF Plant;
- C. WestPine desires to enter into an agreement with BC Gas which allows WestPine to receive transportation service from BC Gas at rates reasonably equivalent to the costs that would have been incurred by WestPine had it constructed the Bypass Pipeline (hereinafter defined) in 1996;
- D. The British Columbia Utilities Commission has endorsed the concept of negotiated rates that are competitive with a Bypass Pipeline alternative.

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"British Columbia Utilities Commission" means the British Columbia Utilities Commission established pursuant to the <u>Utilities Commission Act</u> or other regulatory tribunal having authority over the tariffs and rates of BC Gas.

"Bypass Pipeline" means those facilities which, but for this Agreement, would have been constructed by WestPine in order to provide Gas service to its MDF Plant.

"Contract Year" means a period of 12 consecutive months commencing on November 1 and ending on the next succeeding 1st day of November.

"DTQ" means the Firm DTQ as defined in the Rate Schedule 22 Transportation Agreement or other transportation agreement related to a Rate Schedule for transportation under which WestPine may receive service for the MDF Plant.

"Operating and Maintenance Expenses" means the expenses incurred by BC Gas to operate, maintain and administer its facilities.

"Property Taxes" means the payment required by municipalities or the provincial government to pay for services they deliver consistent with the legislative authority to do so.

"Terms and Conditions of the Transportation Schedule" means the terms of the applicable Transportation Rate Schedule as described in Article 2 and the Transportation Agreement thereunder, and the General Terms and Conditions of BC Gas, as approved by the British Columbia Utilities Commission effective November 1, 1996, or any subsequent Transportation Rate Schedules or Gas Sales Service Rate Schedules accepted for filing by the British Columbia Utilities Commission.

"Westcoast" means Westcoast Energy Inc.

1.02 Except for those terms defined under Article 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Schedule.

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

TRANSPORTATION SERVICE

ADD AND CHARLE

- 2.01 BC Gas will provide firm and interruptible transportation service to WestPine for its MDF Plant, and WestPine will accept such transportation service in accordance with the then prevailing provisions of BC Gas Rate Schedule 22 and the accompanying Transportation Agreement, for the term of this Agreement and such provisions, except where specifically stated otherwise herein, are incorporated herein by reference.
- 2.02 Notwithstanding the provisions of Article 2.01, if anything in the Terms and Conditions of the Schedule conflicts or is inconsistent with the rates, terms and conditions set out in this Agreement, this Agreement shall govern.
- 2.03 The following terms and conditions set out in Rate Schedule 22 do not apply, and are not incorporated by reference, to this Agreement

-section 5 (Table of Charges items a), b) and g) relating to Basic, Delivery and Administration Charges) -section 6 (Minimum Charge)

2.04 WestPine will be entitled to take transportation service under a Transportation Rate Schedule other than Rate Schedule 22 subject to the terms and conditions of such Rate Schedule. In such case, Articles 2.01, 2.02 and 2.03 will apply in the same way to the elected Rate Schedule.

ARTICLE 3

TERM OF AGREEMENT

- 3.01 Subject to Article 8, the initial term of this Agreement will be for a period of 20 Contract Years, effective the 1st Day of November, 1996, up to the 1st Day of November 1, 2016.
- 3.02 The term of this Agreement will continue from year to year after the expiry of the initial term unless cancelled by either BC Gas or WestPine upon not less than 6 months notice prior to the end of the Contract Year then in effect.

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

RATES AND CHARGES

ENVIRTE BOUTWERD BUILD

- 4.01 Subject to the adjustments hereinafter provided, WestPine will each month pay to BC Gas for services as defined in Rate Schedule 22 provided hereunder the following rates:
 - a monthly charge composed of a monthly facilities charge, operating, maintenance and property taxes as specified in Schedule 1 attached hereto ("the Monthly Charge");
 - (ii) Franchise Fees, as defined in article 7 below, as required to be paid by BC Gas, as specified in Schedule 1 attached hereto; and
 - (iii) any charges pursuant to Articles 4.02, 4.03.
- 4.02 The rates contained herein are based upon capacity and demand assumptions for the Bypass Pipeline, as specified in Schedule 2 attached hereto. Should WestPine consumption exceed the contracted demand of 5,000 Gigajoules per day on more than ten days in the Contract Year, BC Gas may at its sole discretion adjust the rates on a go forward basis to reflect the incremental costs of a correspondingly larger pipeline and facilities.
- 4.03 The rates under 4.01 (b)(i) shall be adjusted on November 1 of each year of the Term or any renewal thereof, on the following basis to compensate BC Gas for changes in its cost, including;
 - the Operating and Maintenance expenses shall be subject to the annual percentage change in the British Columbia Consumer Price Index. The Monthly Charge shall be adjusted by the calculated change in costs. The adjusted Operating and Maintenance expenses shall form the basis for the following Contract Year adjustment;
 - (ii) the Property Taxes shall be subject to adjustment based upon the most recent assessed values from the British Columbia Assessment Authority which would be applicable to the Bypass Pipeline multiplied by the most recent rates; and
 - (iii) any amounts to recover cost changes in Municipal, Provincial or Federal taxes and fees, including new taxes or fees, but excluding taxes

Order No.: G-68-98

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

on taxable income, related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by WestPine, shall be added to the Monthly Charge.

- 4.04 Any disputes arising hereunder as to any costs, rates, or charges under Article 4.02 and 4.03 will be referred to arbitration in accordance with Article 5 or this Agreement.
- 4.05 The rates under 4.01 were based on the assumptions as specified in Schedule 2 attached hereto.

ARTICLE 5

ARBITRATION

- 5.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 5.02 Either party may commence arbitration proceedings by sending to the other part a demand for arbitration setting out the nature of the dispute.
- 5.03 The parties will have 10 days from receipt of the demand referred to in section 7.2 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors or affiliates, any customer or supplier of WestPine or BC Gas.
- 5.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed by the arbitrator, within 45 days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 5.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof.

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any transportation or sales Rate Schedule, WestPine will not be entitled to any monthly charge credits from BC Gas as a result of Force Majeure, as defined in BC Gas Rate Schedule 22.

ARTICLE 7

FRANCHISE FEES

- 7.01 WestPine acknowledges that BC Gas is obligated to pay Franchise Fees to Quesnel in an amount equal to 3% of the amount received in each calendar year by BC Gas for Gas consumed within the boundary limits of such municipality. It is acknowledged that the WestPine operations are within Quesnel. BC Gas covenants that it is obligated to pay franchise fees to Quesnel.
- 7.02 The rates payable by WestPine pursuant to Article 4 hereof include a component for franchise fees which is calculated as follows:

F= 0.03 x AC / .097

Where F = the annual franchise fees payable by BC Gas in respect of the gas consumed by WestPine; and

Where AC = the annual total of the Monthly Charge, operating and maintenance expense and property, taxes as outlined in Article 4.

7.03 In the event that BC Gas is required by agreement, statute, regulation, an arbitrator, a court of competent jurisdiction or the British Columbia Utilities Commission to pay Quesnel any additional or lesser amount for franchise fees in respect of the gas consumed by WestPine, WestPine shall pay BC Gas such equal amount, and the rate payable by WestPine as set out in Article 4 shall be adjusted to reflect this additional or lesser charge.

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

CONDITIONS PRECEDENT

8.01 All obligations of the parties to this Agreement are subject to formal execution of this Agreement and acceptance for filing by the British Columbia Utilities Commission of the rates and terms and conditions set out herein.

ARTICLE 9

TERMINATION

- 9.01 In addition to any other rights to terminate this Agreement:
 - BC Gas may terminate this Agreement if at any time the British Columbia Utilities Commission determines that a revenue shortfall exists on service under this Agreement and disallows BC Gas the recovery from other customers of BC Gas, of any revenue shortfall resulting from these negotiated rates;
 - (ii) WestPine may terminate this Agreement if at any time the British Columbia Utilities Commission sets rates for the MDF Plant at a level in excess of the negotiated rates set out herein; and
 - (iii) WestPine may terminate this Agreement if the operations of its MDF Plant permanently ceases or if the MDF Plant permanently ceases to use gas.
- <u>9.02</u> Any termination pursuant to Article 9.01 will take effect on the November 1 following the date of notice in writing by the party terminating this Agreement.
- <u>9.03</u> Termination of this Agreement by WestPine, unless pursuant to Article 9.01 (ii) will require immediate payment by WestPine to BC Gas of an amount equal to the monthly charge at the time of termination multiplied by the number of months remaining in the term of this Agreement.

Notwithstanding any such payment, all facilities serving the MDF Plant will remain the property of BC Gas.

<u>9.04</u> If WestPine fails or neglects at any time to perform the obligations imposed upon WestPine under this Agreement, and does not, within 30 days after BC Gas has given notice of such failure or neglect, commence to remedy with due diligence and

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

thereafter continue to remedy the matter in which it is in default, then BC Gas may discontinue the transportation of Gas to WestPine under this Agreement; but such discontinuance will not relieve WestPine from its obligations under this Agreement. BC Gas may resume the transportation of Gas under this Agreement after such failure or neglect has been fully remedied; or may at its sole discretion, terminate this Agreement.

- <u>9.05</u> If WestPine becomes bankrupt or insolvent or commits or suffers an act of bankruptcy or of insolvency or if a receiver is appointed over the operations of the MDF Plant pursuant to a statute or under a debt instrument or if WestPine seeks protection from the demands of its creditors pursuant to any legislation enacted for that purpose, BC Gas will have the right, at its sole discretion, to terminate this Agreement by giving notice in writing to WestPine and thereupon BC Gas may cease further transportation of Gas to the MDF Plant under this Agreement and BC Gas may terminate this Agreement.
- <u>9.06</u> Termination of this Agreement by BC Gas pursuant to Article 9.04 or 9.05 will entitle BC Gas to require immediate payment by WestPine to BC Gas of an amount equal to the monthly charge at the time of termination multiplied by the number of months remaining in the term of this Agreement.

Notwithstanding any such payment, all facilities serving the MDF Plant will remain the property of BC Gas.

- 9.07 Except if termination of this Agreement is pursuant to Article 9.05, any termination payment to be paid by WestPine in full in accordance with Articles 9.03 or 9.06 may be subject to a discount to reflect the earlier receipt by BC Gas of payments from WestPine due under this Agreement. The reduced amount of the payment will be based upon discounting the future payments to the date payment is made at BC Gas' average cost of funds. BC Gas' average costs of funds will be its weighted average before tax cost of short and long term debt and after tax return on preferred and common equity, at the time of termination.
- <u>9.08</u> It is understood that payment made under Articles 9.03, 9.06 or 9.07 is assumed to include any de-commissioning costs that WestPine would have incurred if WestPine owned the facilities and no further payment for de-commissioning costs will be required.

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

GAS BALANCING

- 10.01 The MDF Plant will be subject to the Gas balancing provisions set out in Rate Schedule 22.
- 10.02 If WestPine in any Contract Year elects to take transportation service pursuant to a Rate Schedule under which the Gas balancing provisions result in benefits to WestPine which would not have been available if the Bypass Pipeline had been constructed, the rates specified in Article 4 will be increased for that Contract Year by an amount equal to WestPine's savings resulting from the provision of BC Gas' daily or monthly balancing on the tolls that would have been incurred on the Westcoast system. The said amount shall be determined and paid to BC Gas at the end of the Contract Year then in effect.
- 10.03 Notwithstanding Article 9.02 and provided that BC Gas waives its right to collect WestPine's benefits under Article 9.02, if at any time during the term of this Agreement the British Columbia Utilities Commission approves a specific rate for monthly Gas balancing, then WestPine agrees to pay such approved rate in addition to the rates specified in Article 4.

ARTICLE 11

NOMINATION

11.01 The MDF Plant will be subject to the nomination provisions set out in Rate Schedule 22.

ARTICLE 12

LIMITATION ON LIABILITY AND INDEMNITY

12.1 WestPine will indemnify, defend and hold harmless each of BC Gas, its officers, directors, employees, contractors and agents from all liability, adverse claims, losses, suits, actions, judgements, demands, debt, accounts, damages, costs, penalties and expenses (including all legal fees and disbursements) for injury to, or death, of any and all persons; and for damage, destruction or loss (except consequential damages or loss) to or of any and all property, real or personal (including but not limited to damage to the facilities servicing the MDF Plant and the cost of any Gas loss due to the damage), resulting directly or indirectly from WestPine's negligent, wrongful or

Order No.: G-68-98

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

unlawful acts or omissions. WestPine will provide all necessary assistance and assurances as may be required to remit and recover any claims for loss or damage to facilities owned or operated by BC Gas, whether or not covered by insurance.

- 12.2 BC Gas will indemnify, defend and hold harmless each of WestPine, its officers, directors, employees, contractors and agents from all liability, adverse claims, losses, suits, actions, judgements, demands, debt accounts, damages, costs, penalties and expenses (including all legal fees and disbursements) for injury to, or death, of any and all persons; and for damage, destruction or loss (except consequential damages or loss) to any and all property, real or personal (including but not limited to damage to the facilities serving the MDF Plant and the cost of any Gas loss due to the damage), resulting directly or indirectly from BC Gas' negligent, wrongful or unlawful acts or omissions as related to BC Gas' provision of transportation services herein. BC Gas will provide all necessary assistance and assurances as may be required to remit and recover any claims for loss or damage to facilities owned by WestPine.
- 12.3 The indemnities in Article 12 shall survive the termination of this Agreement.

Order No.: G-68-98

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

MISCELLANEOUS

13.01 Miscellaneous terms of the applicable transportation Rate Schedule will apply to this Agreement. If in any year an executed Transportation Agreement is not in place, then the terms of the latest executed Transportation Agreement and Rate Schedule including the notice provisions will apply to this Agreement.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement.

AGREED TO AND ACCEPTED:	AGREED TO AND ACCEPTED:
This 25 day of <u>may</u> 1998	This 25 day of <u>mAY</u> , 19 <u>98</u>
BC GAS UTILITY LTD.	WEST FRASER MILLS LTD. BY: Journ Humbu
(Signature)	(Signature)
Manager, Marketing Services (Tide)	(Tide)
Steve Connelly (Name - Please Print)	(Name - Please Print)
057.102	

Order No.: G-68-98

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

Attachment 25.10

SCHEDULE 1

BYPASS PIPELINE - MONTHLY CHARGES

TOTAL MONTHLY COST

Monthly Facilities Charge.	
for the year ending October 31, 1997	\$9,069
Operating and Maintenance	BILIN SABILITIES
for the year ending October 31, 1997	\$1,066
Property Taxes	
for the year ending October 31, 1997	\$1,475
Monthly Charge for the year ending October 31, 1997	\$11,611
Franchise Fees @ 3.09% on the Monthly Facilities Charge, and Operating, Maintenance and Property Taxes,	
for the year ending October 31, 1997	<u>\$ 359</u>
. Dradawah dasard	Second really
Total charge including Facilities Charge, Operating, Maintenance and Property Taxes and Franchise Fees	
for the year ending October 31, 1997	\$ <u>11,970</u>

The rates hereunder shall be subject to adjustment as provided under Article 4.

Order No.: G-68-98

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt

SCHEDULE 2

BYPASS ASSUMPTIONS

Bypass Pipeline Characteristics:

Length	4.62 km
Diameter	88 mm (3 inch)
Pipeline Pressure (PSIG)	Inlet 500
	Outlet 260
Regulating Station at WestPine Pressure (PSIG)	207
Pineline Maximum Canacity (MSCFH)	233
Daily Contract Demand (Gigaioules per Day)	5.000
Expected Annual Consumption (Gigaioules per Year)	385,000
Growth Rate of Consumption	Nil
Glowin faite of Colmaniption	
Financial Assumptions:	
<u>Emancial Assumptions</u>	
Capital Investment Required	\$684,276
Operating and Maintenance Expenses (Year 1)	\$12,797
Property Taxes (Year 1)	\$17,699
Franchise Fees Percentage Gross Up	3.09%
Income Tax Rate	38.60%
Large Corporations Tax Rate	0.225%
Corporate Capital Tax Rate	0.300%
Capital Cost Allowance Rate	4.0%
Depreciation Rate	5.0%
Inflation rate included in Model	Nil
Amortization Period (& Contract Life)	20 years
	-
Capital Structure Assumptions:	
Debt Component	50.0%

Equity Component Cost of Debt Return on Equity

Order No.: G-68-98

Issued By: Diane Roy, Director, Regulatory Affairs

50.0%

7.0%

13.0%

Effective Date: November 1, 1996

BCUC Secretary: Original signed by R.J. Pellatt



TARIFF SUPPLEMENT NO. G-20

BYPASS TRANSPORTATION AGREEMENT RATE SCHEDULE 22

BETWEEN

HUSKY ENERGY MARKETING INC.

AND

FORTISBC ENERGY INC. (Formerly Terasen Gas Inc.)

Effective February 1, 2006

Order No.: G-28-11

Issued By: Diane Roy, Director, Regulatory Affairs

Effective Date: March 1, 2011

BCUC Secretary: Original signed by E.M. Hamilton

THIS AGREEMENT ("**Agreement**") made as of the 1st day of June, 2005, with effect as of and from February 1, 2006.

BETWEEN:

TERASEN GAS INC. (Inc. No. 368861), a company incorporated under the laws of British Columbia having an office at 24th Floor, 1111 West Georgia Street, Vancouver, British Columbia, V6E 4M4

(hereinafter called "Terasen Gas")

OF THE FIRST PART

AND:

HUSKY ENERGY MARKETING INC. (Inc. No. 208018507), a company incorporated under the laws of Alberta having an office in the City of Calgary, Alberta

(hereinafter called "Husky")

OF THE SECOND PART

RECITALS

WHEREAS:

- A. Husky, or an affiliate of Husky, operates a plant ("**Plant**") in the city of Prince George, British Columbia, and requires Gas for its operations;
- B. Terasen Gas owns and operates a Gas transmission pipeline which is connected to the Plant in Prince George, British Columbia;
- C. Husky requires a Gas delivery pressure of 425 PSIG;
- D. Terasen Gas has determined that adequate capacity exists on the Terasen Gas transmission pipeline to provide the DTQ hereinafter defined;
- E. Husky and Terasen Gas agree to enter into a bypass agreement which allows Husky to receive firm transportation service at the required delivery pressure from Terasen Gas at rates reasonably equivalent to the costs that would have been incurred by Husky had it constructed the Bypass Pipeline hereinafter defined; and
- F. The British Columbia Utilities Commission has endorsed the concept of negotiated rates that are competitive with the Bypass Pipeline alternative.

Order No.: G-82-05

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

NOW THEREFORE THIS AGREEMENT WITNESSES THAT in consideration of the premises, the covenants and agreements contained herein, the parties agree as follows:

ARTICLE 1

DEFINITIONS

1.01 Except where the context expressly states another meaning, the following words will have the following meaning:

"**Bypass Pipeline**" means those facilities (4", 7.1 km pipeline) which, but for this Agreement, Husky would construct in order to allow Husky to bypass the Terasen Gas transmission and distribution Gas system and provide the DTQ by transporting Gas from the interconnect of Westcoast Energy Inc. to the Plant.

"DTQ" means, subject to Section 2.06, the quantity of Gas that Terasen Gas is obligated to transport on a firm basis to a shipper at the Delivery Point on any particular Day and is specified in a Transportation Agreement.

"Terms and Conditions of the Transportation Rate Schedule" means the terms of the applicable transportation Rate Schedule and Transportation Agreement thereunder, and the General Terms and Conditions of Terasen Gas, as accepted for filing or approved by the British Columbia Utilities Commission.

1.02 Except for those terms defined under Section 1.01, terms or expressions used in this Agreement will have the meanings described in the Terms and Conditions of the Transportation Rate Schedule.

ARTICLE 2

TRANSPORTATION SERVICE

- 2.01 Terasen Gas will provide firm transportation service to Husky for its Plant operations near Prince George, British Columbia and Husky will accept such transportation service in accordance with this Agreement, the then prevailing provisions of Terasen Gas Rate Schedule 22 and the accompanying Transportation Agreement for the term of this Agreement and such provisions are incorporated herein by reference.
- 2.02 Terasen Gas represents and warrants that adequate capacity exists on the Terasen Gas transmission system, and will continue to exist during the term of this Agreement, to provide the DTQ to Husky.
- 2.03 Notwithstanding the provisions of Section 2.01, where anything in the Terms and Conditions of the Transportation Rate Schedule conflicts or is inconsistent with the terms and conditions set out in this Agreement, this Agreement governs.

Order No.: G-82-05

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

- 2.04 Husky shall be entitled to take service under a transportation Rate Schedule other than Rate Schedule 22 subject to the terms and conditions of such Rate Schedule. In such case, Sections 2.01 and 2.03 will apply in the same way to the elected Rate Schedule.
- 2.05 If Husky in any Contract Year elects to take transportation service pursuant to a Rate Schedule under which monthly Gas balancing provisions result in benefits to Husky which would not have been available if the Bypass Pipeline had been constructed, the rates specified in Article 4 will be increased for that Contract Year by an amount equal to Husky's savings resulting from Terasen Gas' monthly Gas balancing on the tolls that would have been incurred on the Westcoast pipeline system had Husky constructed the Bypass Pipeline. The said amount will be determined by Terasen Gas and subject to approval by Husky, such approval not to be unreasonably withheld or delayed. Husky shall pay the said amount, upon approval, to Terasen Gas at the end of that Contract Year.
- 2.06 Husky shall be entitled to the capacity on Terasen Gas' pipeline system as set out in Schedule 3. The minimum DTQ is 3,670 GJs at an after meter delivery pressure of 425 PSIG.

TERM OF AGREEMENT

- 3.01 Subject to Article 9, the initial term ("**Term**") of this Agreement will be for a period of 10.75 Years, effective from the 1st Day of February 2006 up to the 1st Day of November 2016 ("**Termination Date**").
- 3.02 The Term of this Agreement will be extended by one full year beyond the initial Termination Date or any subsequent extension thereof upon the terms and conditions of this Agreement, including the rates and charges provided herein, unless Husky provides written notice to Terasen Gas of its desire to terminate this Agreement at least 12 months prior to the then current Termination Date. If it so desires, Husky may also extend the term of this Agreement by more than one year through written notice at least 12 months prior to the then current Termination Date, provided that such extension will be for increments of a full year in each case. Terasen Gas shall not unreasonably withhold agreement to such extension of the Term of this Agreement.

ARTICLE 4

RATES AND CHARGES

4.01 Other than as expressly provided in this Agreement, Husky will not be charged for, nor be liable for, any of the charges and costs set out in the Terms and Conditions of the Transportation Rate Schedule.

Order No.: G-82-05

Effective Date: February 1, 2006

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

- 4.02 Subject to the adjustments hereinafter provided, Husky will, each Month, pay to Terasen Gas for services provided hereunder the following rates:
 - (i) subject to Section 4.04, the monthly charge ("**Monthly Charge**") as set out in Schedule 1; and
 - (ii) all charges pursuant to Sections 4.03, 4.04, 4.05 and 4.06.
- 4.03 The rates will be adjusted by Terasen Gas to reflect any changes in costs which would have been incurred by Husky as a result of: increasing the capacity of the Bypass Pipeline; or modification or addition to facilities which may reasonably be required for any other reason, had Husky constructed and operated the Bypass Pipeline. Any adjustments will be subject to approval by Husky, not to be unreasonably withheld or delayed.
- 4.04 The Monthly Charge as specified in Schedule 1 will be adjusted effective November 1 of each Contract Year on the following basis to compensate Terasen Gas for the following changes in its costs:
 - the operating, maintenance and property tax expenses are included in the Monthly Charge and will be subject to adjustment to reflect the annual percentage change in the Consumer Price Index for the City of Vancouver for the Month of August. The adjusted operating, maintenance and property tax expenses will form the basis for the following Contract Year adjustment; and
 - (ii) any new or increased tax related to the operation of the Bypass Pipeline that would have been levied had it been constructed and operated by Husky will be calculated on a monthly basis (the annual cost divided by 12) and added to the Monthly Charge specified in Section 4.02 (i) and Schedule 1.
- 4.05 Unauthorized Overrun Gas, Backstopping Gas and Balancing Gas will be charged for and paid by Husky in accordance with the rate set out in Rate Schedule 22.
- 4.06 Husky acknowledges that Terasen Gas is obligated to pay Franchise Fees to certain municipalities in which Terasen Gas provides services to Customers. In the event that Terasen Gas is required by statute, agreement, regulation, an arbitrator, a court of competent jurisdiction, or the British Columbia Utilities Commission to pay Franchise Fees or similar or equivalent fees on revenues received from Husky, or in relation to services provided hereunder, in addition to those Franchise Fees calculated and paid by Terasen Gas at the commencement of this Agreement, Husky will pay an amount equal to such additional fees to Terasen Gas and the rates and charges payable by Husky as set out in Sections 4.02 to 4.05 above and the Schedules attached hereto will be adjusted to reflect such change.

Order No.: G-82-05

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

FACILITIES INSTALLED

- 5.01 The rates herein are based upon the costs associated with the construction of a hypothetical 4", 7.1 km Bypass Pipeline to provide transportation service to Husky including facilities deemed by Terasen Gas to maintain the integrity of supply to other Customers. Such facilities include pipeline, metering, regulating, control and communications equipment. The Bypass Pipeline is based on the assumption that Terasen Gas controlled the design and installation of its facilities while Husky provided the following facilities to Terasen Gas requirements and specifications:
 - 1. Foundation;
 - 2. Foundations and building for meter and regulating housing;
 - 3. Foundation and building for telemetry housing;
 - 4. Secured compound;
 - 5. Electrical service at 120/240 VAC (at delivery point), 70 A, 60 HZ, 1 phase;
 - 6. Lighting and outlets to meter and regulating housing installed to Class1, Division 1, Group D, Canadian Electrical Code; and
 - 7. Telephone Service.

ARTICLE 6

FORCE MAJEURE

6.01 Notwithstanding any of the provisions contained herein or in any transportation Rate Schedule under which Husky takes service. Husky will not be entitled to credits of any monthly charges from Terasen Gas as a result of Shipper being entitled to relief by reason of Force Majeure as defined in the Rate Schedule 22 Transportation Service Agreement.

ARTICLE 7

ARBITRATION

- 7.01 Any dispute between the parties arising from this Agreement will be resolved by a single arbitrator pursuant to the *Commercial Arbitration Act* of British Columbia or successor legislation, save as expressly provided herein.
- 7.02 Either party may commence arbitration proceedings by sending to the other party a demand for arbitration setting out the nature of the dispute.

Order No.: G-82-05

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

- 7.03 The parties will have 10 Days from receipt of the demand referred to in Section 7.02 of this Agreement to agree upon the arbitrator, failing which either party may apply to the Supreme Court of British Columbia to select the arbitrator. The arbitrator must be sufficiently qualified by education and training to decide the particular questions in dispute. Unless otherwise agreed, the arbitrator may not be a past or present employee, officer or director of any of the parties or their respective successors, permitted assigns or affiliates, or any customer or supplier of Husky or Terasen Gas.
- 7.04 The arbitrator will proceed immediately to hear and determine the matter in dispute and will render a written decision, signed by the arbitrator, within 45 Days after the appointment, subject to any reasonable delay due to unforeseen circumstances. Notwithstanding the foregoing, if the arbitrator fails to render a decision within 60 Days after the appointment then either party may elect to have a new arbitrator appointed in like manner as if none had previously been appointed.
- 7.05 The decision of the arbitrator will be final and binding upon the parties and the parties will abide by the decision and perform the terms and conditions thereof. Notwithstanding the outcome of the arbitration proceeding, each party will bear its own costs, including third party costs, related to the arbitration but will share equally in the costs of the arbitrator.

CONDITIONS PRECEDENT

8.01 All obligations of the parties to this Agreement are subject to their formal execution of this Agreement and acceptance for filing by the British Columbia Utilities Commission of the rates, terms and conditions set out herein.

ARTICLE 9

TERMINATION

- 9.01 In addition to any other rights either party may have to terminate this Agreement:
 - Notwithstanding Article 3, Terasen Gas may terminate this Agreement if at any time the British Columbia Utilities Commission disallows Terasen Gas the recovery from the other Customers and/or Shippers of Terasen Gas of any revenue shortfalls, resulting from the negotiated rates under this Agreement;
 - (ii) Notwithstanding Article 3, Husky may terminate this Agreement if at any time the British Columbia Utilities Commission sets rates for Husky at a level in excess of the negotiated rates set out or provided for herein, or if the operations of Husky, or an affiliate of Husky, at the Plant permanently cease or if Husky, or an affiliate of Husky, ceases to use Gas permanently at the Plant; and

Order No.: G-82-05

Effective Date: February 1, 2006

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

- (iii) Notwithstanding Section 9.01(ii), Husky may terminate this Agreement at any time after November 1, 2026 without any payment under this Article 9.
- 9.02 Any termination pursuant to Section 9.01 will take effect on the next November 1 following a notice period of not less than 12 Months.
- 9.03 Except as otherwise stated in this Agreement, termination of this Agreement by Husky will require immediate payment by Husky to Terasen Gas of an amount equal to the undepreciated value of the Bypass Pipeline as specified in Schedule 2 as of the effective date of the termination and any additional costs pursuant to the application of Section 4.03 above. Notwithstanding any such payment, all Terasen Gas owned facilities serving Husky shall remain the property of Terasen Gas.
- 9.04 If Husky fails or neglects at any time to perform the obligations imposed upon Husky under this Agreement, and does not, within thirty (30) days after Terasen Gas has given notice of such failure or neglect, commence to remedy with due diligence and thereafter continue to remedy the matter with respect to which it is in default, then Terasen Gas, in its sole discretion, may terminate this Agreement, or may discontinue transportation of Gas to Husky under this Agreement but such discontinuance by Terasen Gas will not relieve Husky from its obligations under this Agreement after such failure has been fully remedied.
- 9.05 If Husky becomes bankrupt or insolvent or commits or suffers an act of bankruptcy or of insolvency or if a receiver is appointed over the Husky operations pursuant to a statute or under a debt instrument or if Husky seeks protection from the demands of its creditors pursuant to any legislation enacted for such purpose, Terasen Gas will have the right, at its sole discretion, to terminate this Agreement by giving notice in writing to Husky and thereupon Terasen Gas may cease further transportation of Gas to Husky under this Agreement.
- 9.06 Termination of this Agreement by Terasen Gas pursuant to Section 9.04 or 9.05 will entitle Terasen Gas to require immediate payment by Husky to Terasen Gas of an amount equal to the undepreciated value of the Bypass Pipeline as specified in Schedule 2 as of the effective date of the termination and any additional costs pursuant to the application of Section 4.03 above. Notwithstanding any such payment, all Terasen Gas facilities serving the Husky operations shall remain the property of Terasen Gas.
- 9.07 Any termination payment made in full in accordance with Section 9.03 or 9.06 shall be subject to a discount as determined by Terasen Gas to reflect the earlier receipt of payments due under this Agreement. The reduced amount of the payment will be based upon discounting the future payments to the date payment is made at Terasen Gas' average cost of funds. Terasen Gas' average cost of funds will include its weighted average before tax cost of short and long term debt and after tax return on preferred and common equity.

Order No.: G-82-05

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

NOTICES

10.01 If in any Year an executed Transportation Agreement is not in place, then the notice provisions of the latest executed Transportation Agreement and Rate Schedule will apply to this Agreement.

ARTICLE 11

MISCELLANEOUS

- 11.01 Husky shall have the right to release all or part of its DTQ in effect under this Agreement to Terasen Gas, provided Husky has obtained the prior written approval of Terasen Gas to absorb the capacity to meet incremental requirements on the distribution path. If Terasen Gas, in its sole discretion, determines that the requested release can be accommodated and absorbed without detrimental impact to its system or other Customers and/or Shippers, then Terasen Gas will prepare the necessary revisions to this Agreement, to reduce the volume appropriately. Husky will thereby be relieved of its obligations under this Agreement in respect of the DTQ or part thereof so released with the approval of Terasen Gas.
- 11.02 Notwithstanding Section 11.01, Terasen Gas may assign in its sole and absolute discretion, without the consent of Husky, Terasen Gas' rights and obligations under this Agreement to a party which acquires all or substantially all of Terasen Gas' Gas utility operations.
- 11.03 Notwithstanding Section 11.01, Husky may assign in its sole and absolute discretion, without the consent of Terasen Gas, Husky's rights and obligations under this Agreement to any party which acquires all or substantially all of Husky's Plant operations served by Terasen Gas pursuant to this Agreement.
- 11.04 This Agreement will enure to the benefit of and be binding on the parties and their respective successors and permitted assigns including, without limitation successors by merger, amalgamation or consolidation.
- 11.05 This Agreement and all amendments, modifications, alterations or supplements hereto will be governed by the laws in force in the Province of British Columbia.
- 11.06 In this Agreement the words, phrases or expressions which are not defined herein or in the Terms and Conditions of the Transportation Rate Schedule or in the Rate Schedule 22 and which in the usage custom of the business of the production, transportation, distribution or sale of Gas has a generally accepted industry meaning will bear that meaning.

Order No.: G-82-05

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

11.07 A party shall have the right, at its own expense, upon reasonable notice and at reasonable times, to examine the books and records of the other party only to the extent reasonably necessary to verify the accuracy of any statement, charge, payment or computation made under this Agreement. This examination right shall not be available with respect to proprietary, confidential or personal information not directly relevant to this Agreement. All invoices and billings shall be conclusively presumed to be final and accurate unless objected to in writing, with adequate explanation and/or documentation, within two years of their issuance.

IN WITNESS WHEREOF the parties have executed this Agreement effective February 1, 2006.

AGREED TO AND ACCEPTED:	AGREED TO AND ACCEPTED:
This _ <u>14th_</u> day ofJuly, 2005.	This _ <u>11th_</u> day of <u>July</u> , 2005.
TERASEN GAS INC.	HUSKY ENERGY MARKETING INC.
BY: <u>Original signed by Rick Parnell</u>	BY: <u>Original signed by Donald R. Ingram</u>
RICK PARNELL (Name – Please Print)	DONALD R. INGRAM
Director, Large Commercial & Industrial Markets	Senior Vice President
	BY:

. (Signature)

(Name - Please Print)

(Title)

Order No.: G-82-05

Effective Date: February 1, 2006

BCUC Secretary: <u>Original signed by R.J. Pellatt</u>

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs
Schedule 1

Bypass Pipeline Monthly Charge

Monthly Facilities Charge	\$ 14,740
Monthly Operation and Maintenance Charge (subject to increase as per Section 4.04)	\$ 775
Monthly Property Taxes (subject to increase as per Section 4.04)	\$ 2,365
Monthly Charge	\$ 17,880

Order No.: G-82-05

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

> Tariff Supplements G-20 Original Page 10

Schedule 2

Undepreciated Value of the Bypass Pipeline

Capital Cost Recovery for Early Termination of Contract

As at February 1, 2006	\$ 1,179,679
As at November 1, 2006	\$ 1,139,507
As at November 1, 2007	\$ 1,085,944
As at November 1, 2008	\$ 1,032,381
As at November 1, 2009	\$ 978,818
As at November 1, 2010	\$ 925,255
As at November 1, 2011	\$ 871,693
As at November 1, 2012	\$ 818,130
As at November 1, 2013	\$ 764,567
As at November 1, 2014	\$ 711,004
As at November 1, 2015	\$ 657,441
As at November 1, 2016	\$ 603,878
As at November 1, 2017	\$ 550,316
As at November 1, 2018	\$ 496,753
As at November 1, 2019	\$ 443,190
As at November 1, 2020	\$ 389,627
As at November 1, 2021	\$ 336,064
As at November 1, 2022	\$ 282,501
As at November 1, 2023	\$ 228,939
As at November 1, 2024	\$ 175,376
As at November 1, 2025	\$ 121,813
As at November 1, 2026	\$ 0

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

Schedule 3

Bypass Pipeline Capacity Based upon 425 PSIG

Duke Inlet Pressure	Capacity (10 ³ M ³ /hr)	Capacity (GJs/hr)
500 - 549	4.1	153
550 - 599	7.297	273
600 - 649	9.661	361
650 - 699	11.712	418
>700	13.592	508

Note: Duke Inlet pressure is measured at Terasen Gas' Prince George #1 lateral.

Order No.: G-82-05

Effective Date: February 1, 2006

BCUC Secretary: Original signed by R.J. Pellatt

Issued By: Scott Thomson, Vice President Finance and Regulatory Affairs

> Tariff Supplements G-20 Original Page 12

Attachment 28.1

REFER TO LIVE SPREADSHEET MODEL

Provided in electronic format only

FILED CONFIDENTIALLY

(accessible by opening the Attachments Tab in Adobe)

Attachment 39.1

1 5.0 PROPOSED CHANGES TO SERVICE CHARGES

BC Gas proposes to revise a number of its customer related service charges to better reflect the real cost of providing such services. In particular, revising the Application for Service charges brings the charges more in line with the cost of providing service. A review of other utilities supports this revision.

9

2

Table A provides a review of the Application/Connection and 10 Reconnection fees charged by the other British Columbia 11 utilities. Table B summarizes the Application/Connection and 12 Reconnection fees charged by some other major Canadian gas 13 The activities associated with reconnection of a 14 utilities. service after discontinuance at a customer's request are 15 similar to the Application for Service, and the corresponding 16 fees from various utilities are provided for comparison. 17

18

BC Gas currently charges \$10 for each Application for Service, regardless of whether the application is simply to transfer an existing account from one person to another or to actually install a new service line and meter set. In many situations the \$10 collected does not match the BC Gas administrative costs associated with providing the service.

25

5.1 Cost Analyses

26 27

BC Gas devotes considerable resources and incurs significant 28 cost, to process requests to initiate or terminate gas 29 service. However, the nature of the work required to process 30 applications and terminations varies depending upon the 31 the three basic The following are circumstances. 32 application/termination scenarios that develop: 33

34

Account Transfer - Existing Installation, Service Active
 In this case, the premise is already served with gas and

apply for service. The complete transaction can usually 2 be dealt with over the telephone or at a branch office. 3 4 Account Transfer - Existing Installation, Service 2) 5 Inactive 6 7 In this case, the premise is aready serviced with gas but 8 the gas is not flowing. While the administrative process 9 can be completed over the telephone or at the branch 10 office, BC Gas Customer Service personnel must be 11 service, including reactivate qas to dispatched 12 relighting the gas appliances. 13 14 New Account - New Service Required 15 3) In those situations where natural gas service is not 16 currently available at the premises, a new gas service 17 line installation is required. The customer must sign up 18 for the new installation in person at a branch office. 19 20 In each scenario, some or all of the following activities 21 occur: 22 23 The Applicant's information is obtained either over the 24 a) telephone or at a branch office. 25 26 The Applicant's information is keyed into the Company's 27 b) computer system. 28 29 An evaluation of the Applicant's credit worthiness is 30 C) determined and, if any outstanding bills from other 31 accounts exist, collection activities are undertaken 32 before new service is provided. 33 34 Information regarding the terminating customer is sought, d) 35 in particular a forwarding address for the final bill is 36

the gas is flowing at the time a new occupant calls to

1

obtained. Billing adjustments are required in some
 cases.

- 4 e) In some cases Customer Service personnel are dispatched
 5 to obtain a meter reading on the transfer date; in most
 6 situations the final meter reading is estimated.
- 8 f) When there is an existing service, but it is inactive, 9 Customer Service personnel are dispatched to reactivate 10 the service and relight the gas appliances. In some 11 cases the gas meter must be reset.
- g) In certain cases the account may be temporarily
 transferred into the landlord's name, for a period of
 time between tenants.
- h) When a new service installation is required, additional
 information is required for permanent installation
 records and a site visit prior to installation is
 necessary for 20-25% of all installations. Following
 installation of the new service, site records are
 completed in greater detail.
- 24 The activities listed generate costs in the following areas:
- 25 clerical time
- 26 computer system utilization
- 27 customer service personnel and vehicles
- 28 mains and services representatives and vehicles
- 29 associated overhead.
- 30

3

7

12

16

23

31 Given the various activities that are required to complete 32 applications/terminations, the following cost estimates for 33 each of the three scenarios has been developed.

1	Acc	count Transfer - Existing Installation, Service Active	
2			
3	1.	Interior Costs	
4		Clerical Time:	
5		Customer Service Representative 3	
6		OTEU Collective Agreeement \$17.69/hr	
7		- Group 5 (3 yrs experience)	
8		Benefits @ 17.3% \$ 3.06	
9		Concessions @ 15.5% <u>\$ 2.74</u>	
10		Total \$23.49/hr	
11		Average total application processing time: .5 hrs = \$11.	14
12			
13		Computer System Utilization:	
14		On-screen average costs for interior billing system \$ <u>4.</u>	<u>71</u>
15		Total Interior = \$16.	45
16			
17	2.	Lower Mainland Costs	
18		1993 B.C. Hydro customer accounts, service and	
19		collection contracts average cost \$ 9.	15
20			
21	3.	Weighted Average	
22		Lower Mainland - B.C. Hydro billing system -	
23		72% of customers x \$9.15 \$ 6.	59
24		Interior billing system -	
25		28% of customers x \$16.45 \$ <u>4</u> .	<u>52</u>
26		\$ <u>11.</u>	<u>21</u>
27			
28			
29	AC	count Transfer - Existing Installation, Service Inactive	
30			
31	1.	Clerical time & computer system use, same as above \$11.	21
32	±•		
32	2	Site Vigit	
34	2.	IBEW collective agreement \$21.77/hr	
35		Benefits @ 20.9% $$4.55$	
36		Concessions @ 36%	
30		Total $334.16/hr$	
30			
20		Average corvige deactivation/reactivation time = $1.0 \text{ hr}^* = -534$.	16
10		Nobialo coste $0.66 \cdot 10/hr \times 1.0 hr = 6.6$	10
40			<u>-</u> Ξ Δ7
41 40		10tar - 5 <u>51.</u>	
42		t Real time on account has to be reactivated the cost	
45		of doortivating gas sorvice has already been incurred	
44		OI REACTIVATING day pervice way atteady peem tweatter.	

Tab 12 Page 11

1 New Account - New Service Required	
2	
3 1. <u>Clerical Time</u>	
4 Construction Planning Clerk	
5 OTEU Collective Agreement \$17.69/hr	
6 - Group 5 (3 yrs experience)	
7 Benefits @ 17.3% \$ 3.06	
8 Concessions @15.5% \$ 2.74	
9 Total \$23.49/hr	
10 Average total application processing time: 1.5 h	s = \$35.24
11	
12 2. Computer System Utilization	
13 On-screen average costs for interior billing system	m \$ 4.71
14	
15 3. <u>Site Visit/Confirmation</u> (25% of installations are	non-standard)
16 Mains & Service Representative	
17 OTEU Collective Agreement \$22.47/hr	
18 - Group 8 (3 yrs experience)	
19 Benefits @ 17.3% \$ 3.89	
20 Concessions @15.5% \$ <u>3.48</u>	
21 Total \$29.84/hr	
22	
23 Average total application processing time: 1.5 h	s = \$44.76
24 Vehicle cost @ \$6.10/hr X 1.5 hrs =	\$ <u>9.15</u>
25 Total:	\$ <u>93.86</u>
26	
27 BC Gas proposes to recover a greater portion	on of the above
28 costs through increased fees charged for A	pplications for
29 Service as follows:	
30	
31 Current Pr	posed
32 <u>Charge</u> C	arge Increase
34 1) Application for Service I - \$10	25 \$15
35 Existing Installation -	
36 Account Transfer	
3/	
37 38 2) Application for Service II - \$10	75 \$65

During the year ending December 31st, 1992, BC Gas processed 1 approximately 205,000 Applications for Service. Approximately 2 181,000 of these applications were for account transfers only 3 and 24,000 applications were for new service installations. 4 An increase of \$15 per Application for Service I would 5 generate an additional \$2.715 million/year assuming 180,000 or 6 more Applications for Service are received in future years. 7 An increase of \$65 in Application for Service II, where a new 8 service line and meter set is required, would generate an 9 additional \$1.3 million per year assuming 20,000 such 10 installations occurred each year. BC Gas has installed an 11 average of 23,100 new services per year for the years 1990, 12 13 1991 and 1992.

14

The increased revenues arising from these revised charges serve to mitigate the impact of annual bill increases caused by the proposed rate design. The use of these funds is discussed in the "Implementation and Phase-in of the Proposed New BC Gas Rates" section for Residential, Commercial, General Service, and Industrial Customers in Tabs 6, 7, 8 and 9, respectively. 49

The proposed General Terms and Conditions provoked relatively little discussion during the hearing. The large volume industrial customers objected to the unlimited right of BCGUL under Section 13.2, to curtail gas to any of its customers in the event of failure of BCGUL's gas supply for <u>any</u> reason, arguing that this should not include the case of failure to deliver gas by the Utility's own suppliers. However, the Commission rejects this argument and is satisfied that a utility must have the final decision on emergency curtailment but will use these broad powers responsibly.

The Commission believes the adoption of consolidated General Terms and Conditions is a logical accompaniment to consolidation for rate making purposes. Adoption of simplified and clearer Conditions should improve customer understanding and simplify contract administration by the Utility. The Commission therefore approves adoption of the proposed consolidated General Terms and Conditions.

11.3.1 Service Charges: Connection and Reconnection Fees

The proposed new General Terms and Conditions contain a supplemental schedule of standard fees and charges related to connection fees and disputed meter testing charges.

BCGUL proposes to set the fee for Account Transfers (whether service is active or inactive) at \$25 and proposes a fee of \$75 for new installations. In its application the Utility demonstrates that the former charge is close to the average cost of servicing active and inactive account transfers. The "new installation" charge moves much closer to full cost recovery than the formerly charged \$10 fee, but still falls short of full cost recovery.

The Commission accepts that the proposed change in service charges is directionally correct, is satisfied that the charges are reasonably consistent with comparable charges of other Canadian utilities, and approves the application as filed.

11.4 BCGUL Application for Hearing Cost Recovery

By an August 9, 1993 letter addressed to the Commission, BCGUL requested permission to recover Phase B Rate Design hearing costs in the amount of \$487,179, plus \$52,944 of capitalized FDC modelling costs. These costs excluded any consideration of Commission costs arising from the hearing or any consideration of participant funding by the Utility which might arise from a Commission award under Section 133 of the Utilities Commission Act ("the Act").

Attachment 54.1

REFER TO LIVE SPREADSHEET MODEL

Provided in electronic format only

(accessible by opening the Attachments Tab in Adobe)